Chordotonal organ mechanosensory pathways as potential targets for the control of mosquito borne diseases

Matthew Paul Topping

University College London

RRDCMPSING01 - Research Degree: COMPLEX

I, Matthew Paul Topping confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Word count: 73082

Mathees Topping

Signed:

Abstract

Mosquito-borne infectious diseases are the cause of millions of deaths every year, with disease transmission inseparably linked with mosquito physiology and biology. Mechanosensory signalling plays an essential role in multiple parts of the mosquito life cycle, making it a potential target for insecticides. In spite of this, and the existence of several 'mechanotoxins', these pathways have so far not been exploited. My thesis examined the various aspects of mechanosensation-dependent insect behaviour, as well as the effect the chordotonal organ-specific compound pymetrozine had on these behaviours.

Behavioural studies using *Drosophila melanogaster*, including competition assays and flight tests, were completed to investigate the impact of pymetrozine exposure. Exposure produced significant decreases in flight ability and male reproductive fitness. *Drosophila melanogaster* lines previously reported to have insecticidal resistance were exposed to pymetrozine to check for potential cross-resistance, with no pymetrozine resistance being found in any case tested.

Three mosquito species were then studied in depth - Aedes aegypti, Anopheles gambiae and Culex quinquefasciatus. Using an already existing Drosophila auditory model, electrophysiological experiments were conducted to compare between different species and sexes. A considerable level of sexual dimorphism was found throughout, especially in the auditory nonlinearities associated with transducer gating. Energy and displacement gains with regards to the mechanical auditory system were approximated and the nerve response to different stimuli was investigated. Male and female responses to compound injection were also calculated.

Pymetrozine exposure was found to lead to complete loss of ChO mechanosensory function in these three species, as well as two additional insecticidal resistant *Anopheles gambiae* lines. Mathematical modelling of pymetrozine was discussed, particularly with regards to control programmes. 'Mechanotoxins' have so far been vastly underutilised as both insecticides and as a tool to explore auditory systems. These compounds hold great potential as methods of insect control.

Acknowledgements

I would like to thank and acknowledge the large number of people who both offered and provided assistance throughout my PhD, including but not limited to: Shahida Begum from the London School of Hygiene and Tropical Medicine; Dr Gareth Lycett and Adriana Adolfi from the Liverpool School of Tropical Medicine; Dr Max Reuter and the rest of the Reuter lab at University College London; and Dr Andrew Straw, Etienne Campione and the rest of the Straw Lab at the Institute of Molecular Pathology in Vienna.

In addition to this I need to recognise the help of all the members of the Jörg Albert Mechanosensation Lab during my time there, especially Alessandra Casamento, Ross Harper, Ryan Kavlie, Jason Somers and Camille Tardieu.

Special thanks are required for Nicholas Boyd-Gibbons, as without his patient explanations and demonstrations of key concepts and experimental procedures the project would have been exponentially more challenging.

My sincerest gratitude is given to Professor Johnathan Ashmore for his role as a secondary supervisor, as well as Dr Martin Walker and Professor María-Gloria Basáñez for their guidance and for all of their involvement in the supervisory process. The greatest thanks and most sincere appreciation however must be reserved for my primary supervisor, Professor Jörg Albert. Without his constant understanding and unwavering support this project would have been impossible. It is not possible within the confines of this manuscript to express how indebted I am to him, so I can only repeat myself and thank him again.

Finally, I would like to acknowledge and thank Ms Chang Su for everything that she has done throughout my PhD.

Table of contents

1. Introduction	33
2. Literature review	35
2.1. Major vector species of mosquito borne diseases	35
2.1.1. Aedes aegypti	
2.1.2. Anopheles gambiae	
2.1.3. Culex quinquefasciatus	38
2.2. Mosquito-borne disease transmission	39
2.3. Current control programmes	41
2.3.1. Vector control	42
2.3.1.1. Insecticidal nets and spraying	
2.3.1.2. Larval management and oviposition traps	43
2.3.1.3. Attractive toxic sugar bait	44
2.3.1.4. Proposed control methods utilising fungal, bacterial and mechanisms	l genetic 45
2.3.2. Pathogen control	
2.3.3. Vaccination	47
2.4. Insecticidal and antibiotic resistance	48
2.5. Use of Drosophila melanogaster as a model organism	50
2.6. Sensory organs and mechanotransduction	51
2.7. Auditory systems in <i>Drosophila melanogaster</i> and mosqui	to
species	54
2.8. Auditory mechanotransduction and efferent feedback in m	osquito
species	
2.9. Pymetrozine	61

2.10. Key components of insect life cycles that pymetrozine could	
target	63
2.10.1. Flight	63
2.10.2. Courtship	65
2.10.3. Circadian rhythm regulation	70
2.11. Mathematical modelling	72
3. Project rationale, objectives and hypotheses	73
3.1. Project rationale	73
3.2. Project objectives	74
3.3. Hypotheses	76
4. Drosophila melanogaster behavioural assays	78
4.1. Introduction	78
4.2. Materials and methods	80
4.2.1. Drosophila melanogaster rearing	80
4.2.2. Compound preparation	80
4.2.3. Compound exposure – ingestion	81
4.2.4. Drosophila melanogaster fertility assay	82
4.2.5. Drosophila melanogaster fecundity assay	83
4.2.6. Drosophila melanogaster life span assay	84
4.2.7. Flight initiation assay	85
4.2.8. Drosophila melanogaster Flycube flight assay	86
4.2.9. Circadian rhythm temperature entrainment of pymetrozine expo	sed
and unexposed Drosophila melanogaster males	88
4.2.10. Effect of pymetrozine on Drosophila melanogaster courtship s	ongs
	89
4.2.11. Drosophila melanogaster competitive reproduction assay	92
4.3. Results	94

4.3.1. <i>Drosophila melanogaster</i> fertility and fecundity assays
4.3.2. Drosophila melanogaster life span assay
4.3.3. Flight initiation assay
4.3.4. Drosophila melanogaster Flycube flight assay
4.3.5. Circadian rhythm temperature entrainment of pymetrozine
exposed and unexposed Drosophila melanogaster males
4.3.6. Effect of pymetrozine on Drosophila melanogaster courtship songs
4.3.7. Drosophila melanogaster competitive reproduction assay
4.4. Discussion
4.4.1. The effect of pymetrozine on Drosophila melanogaster fecundity and
fertility
4.4.2. The effect of pymetrozine on Drosophila melanogaster lifespan 105
4.4.3. The effect of pymetrozine on Drosophila melanogaster flight
initiation and maintenance106
4.4.4. The effect of pymetrozine on Drosophila melanogaster circadian
rhythms
4.4.5. The effect of pymetrozine on Drosophila melanogaster courtship. 108
5. Mosquito behavioural assays and physiological measurements
5.1. Introduction
5.2. Materials and methods113
5.2.1. Mosquito rearing113
5.2.2 Compound preparation113
5.2.3. Compound exposure – ingestion
5.2.4. Larvicidal assay 114
5.2.5. Impact of pymetrozine on reproductive success in males and
females

5.2.6 Mosquito Elycube flight assay	116
5.2.7 Mosquito flagellum length measurements	117
5.2.7. Mosquito hagenum length measurements	110
5.2.6. Mosquito wing beat frequency measurements	
5.3. Results of mosquito behavioural assays and physiologica	140
measurements	119
5.3.1. Larvicidal assay	119
5.3.2. Impact of pymetrozine on reproductive success in male and	I female
mosquitoes	
5.3.3. Mosquito Flycube flight assay	122
5.3.4. Mosquito flagellum length measurements	123
5.3.5. Mosquito wing beat frequency measurements	
5.4. Discussion	125
5.4.1. Effect of pymetrozine on <i>An. gambiae</i> larvae	125
5.4.2. Effect of pymetrozine on An. gambiae female fertility and fe	cundity
	126
5.4.3. The potential of Flycubes for tracking mosquito flight patter	r ns 126
5.4.4. Mosquito flagellar length and surface area measurements	127
5.4.5. Mosquito wing beat frequency measurements	128
6. Drosophila melanogaster pymetrozine vibrometry and	
electrophysiology	129
6.1. Introduction	129
6.2 Materials and methods	131
C 2.4. Drecontile melonogostor receing	404
6.2.1. Drosopnila melanogaster rearing	
6.2.2. Compound preparation	
6.2.3. Insecticide resistant <i>Drosophila melanogaster</i> crosses	132
6.2.4. Compound exposure method - injection	133
6.2.5. Vibrometry preparation and procedure – free fluctuations	

6.2.6. Free fluctuation fitting procedure	135
6.2.7. Stimulated recordings – force steps	137
6.2.8. Stimulated recordings – force step analysis procedure	139
6.3. Results of Drosophila melanogaster pymetrozine injections	with
respect to vibrometry and electrophysiology	144
6.3.1. Effective stiffness and best frequency changes	144
6.3.2. Force step electrophysiology recordings	148
6.4. Discussion	155
6.4.1. The effect of pymetrozine on Drosophila melanogaster audito	ry
systems as measured by free fluctuations	155
6.4.2. The effect of pymetrozine on <i>Drosophila melanogaster</i> audito	ry 450
systems as measured by force step stimulation electrophysiology.	156
7. Mosquito vibrometry and electrophysiology	158
7.1. Introduction	158
7.2. Materials and methods	161
7.2.1. Mosquito rearing	161
7.2.2 Compound preparation	161
7.2.3. Compound exposure method – injection	162
7.2.4. Mosquito vibrometry preparation	163
7.2.5. CO ₂ sedation experiments – free fluctuation measurements	164
7.2.6. CO_2 sedation experiments – electrophysiological recordings	165
7.2.7. Free fluctuation fitting procedure	165
7.2.8. Energy gain calculations	166
7.2.9. Mosquito apparent antennal mass estimations	168
7.2.10. Step recording analysis	170
7.2.11. White noise stimulus experiments	171
7.2.12. Pure tone sine stimulus experiments	173

7.2.13. TTX and TeNT injection series
7.3. Results
7.3.1. Mosquito free fluctuations - CO ₂ sedation and energy gain
7.3.2. Mosquito free fluctuations - effective stiffness
7.3.3. Mosquito apparent antennal mass estimations
7.3.4. Step recordings
7.3.4.1. Mechanical and nerve responses to force steps
7.3.4.2. Two state model of a single transducer population in mosquitoes
7.3.4.3. Dynamical stiffness and CAP amplitude
7.3.4.4. Potential mechanisms to maintain constant frequency tuning:
compensatory stiffness antagonises the gating compliance?
7.3.4.5. Effects of hypothesised multiple ion channel populations 202
7.3.5. White noise stimulus displacement gain experiments
7.3.6. Pure tone stimulus displacement gain experiments
7.3.7. TTX and TeNT injection series
7.4. Discussion
7.4.1. Mosquito auditory systems as measured by free fluctuations 224
7.4.2. Mosquito apparent antennal mass estimations
7.4.3. Mosquito auditory systems as measured by force step stimulation
electrophysiology
7.4.4. Systems of compensatory stiffness maintenance in mosquito
species
7.4.5. Multiple transducer populations
7.4.6. Extent of sexual dimorphism in mosquito auditory systems with
regards to displacement gain estimates233

7.4.7. Extent of sexual dimorphism in mosquito auditory systems with
regards to compound injection designed to sever efferent feedback loops
8. Mosquito pymetrozine vibrometry and electrophysiology238
8.1. Introduction
8.2. Materials and methods
8.2.1. Mosquito rearing
8.2.2. Compound preparation
8.2.3. Compound exposure methods – feeding and injection
8.2.4. Analysis of force step stimulation electrophysiology experiments
following pymetrozine exposure via ingestion
8.2.5. Analysis of electrophysiological force step experiments before and
after pymetrozine exposure via injection for Ae. aegypti, An. gambiae
(Kisumu) and Cx. quinquefasciatus
8.2.6. Analysis of electrophysiological force step experiments before and
after pymetrozine exposure via injection for <i>An. gambiae</i> (Ngusso and
after pymetrozine exposure via injection for <i>An. gambia</i> e (Ngusso and Tiassale)
after pymetrozine exposure via injection for <i>An. gambiae</i> (Ngusso and Tiassale)
after pymetrozine exposure via injection for <i>An. gambiae</i> (Ngusso and Tiassale)
after pymetrozine exposure via injection for <i>An. gambiae</i> (Ngusso and Tiassale) 243 8.2.7. Analysis of pure tone stimulation experiments following pymetrozine exposure via injection 243 8.3. Results 244
after pymetrozine exposure via injection for <i>An. gambiae</i> (Ngusso and Tiassale) 243 8.2.7. Analysis of pure tone stimulation experiments following pymetrozine exposure via injection 243 8.3. Results 244 8.3.1. Step recordings – pymetrozine feeding 244
after pymetrozine exposure via injection for <i>An. gambiae</i> (Ngusso and Tiassale)
after pymetrozine exposure via injection for <i>An. gambiae</i> (Ngusso and Tiassale) 243 8.2.7. Analysis of pure tone stimulation experiments following pymetrozine exposure via injection 243 8.3. Results 244 8.3.1. Step recordings – pymetrozine feeding 244 8.3.2. Changes observed in antennal free fluctuation recordings following pymetrozine injection 244
after pymetrozine exposure via injection for An. gambiae (Ngusso and Tiassale) 243 8.2.7. Analysis of pure tone stimulation experiments following pymetrozine 243 exposure via injection 243 8.3. Results 244 8.3.1. Step recordings – pymetrozine feeding 244 8.3.2. Changes observed in antennal free fluctuation recordings following 244 8.3.3. Force step recordings – pymetrozine injection 245
after pymetrozine exposure via injection for <i>An. gambiae</i> (Ngusso and Tiassale) 243 8.2.7. Analysis of pure tone stimulation experiments following pymetrozine exposure via injection 243 8.3. Results 244 8.3.1. Step recordings – pymetrozine feeding 244 8.3.2. Changes observed in antennal free fluctuation recordings following pymetrozine injection 244 8.3.3. Force step recordings – pymetrozine injection 250 8.3.3.1. Mechanical and nerve responses to force steps 250
after pymetrozine exposure via injection for <i>An. gambiae</i> (Ngusso and Tiassale) 243 8.2.7. Analysis of pure tone stimulation experiments following pymetrozine exposure via injection 243 8.3. Results 244 8.3.1. Step recordings – pymetrozine feeding 244 8.3.2. Changes observed in antennal free fluctuation recordings following pymetrozine injection 244 8.3.3. Force step recordings – pymetrozine injection 250 8.3.3.1. Mechanical and nerve responses to force steps 250 8.3.3.2. Changes to dynamical stiffness properties following pymetrozine
after pymetrozine exposure via injection for An. gambiae (Ngusso and Tiassale) 243 8.2.7. Analysis of pure tone stimulation experiments following pymetrozine exposure via injection 243 8.3. Results 244 8.3.1. Step recordings – pymetrozine feeding 244 8.3.2. Changes observed in antennal free fluctuation recordings following pymetrozine injection 244 8.3.3. Force step recordings – pymetrozine injection 245 8.3.3.1. Mechanical and nerve responses to force steps 250 8.3.3.2. Changes to dynamical stiffness properties following pymetrozine injection 250 8.3.3.2. Changes to dynamical stiffness properties following pymetrozine injection 250
after pymetrozine exposure via injection for <i>An. gambiae</i> (Ngusso and Tiassale)

8.3.4.1. Force step electrophysiological recordings for insecticide
resistant mosquitoes before pymetrozine exposure
8.3.4.2. Force step electrophysiological recordings for insecticide
resistant mosquitoes following pymetrozine exposure
8.3.5. Mosquito flagellar sensitivity to pure tone stimulation following
pymetrozine injection
8.4. Discussion
8.4.1. The effect of pymetrozine ingestion on mosquito auditory systems
8.4.2. The effect of pymetrozine injection on mosquito auditory systems –
free fluctuations
8.4.3. The effect of pymetrozine injection on mosquito auditory systems –
force step stimulation and compensatory stiffness mechanisms
8.4.4. Auditory systems of insecticide resistant mosquito species and the
effect of pymetrozine injection on these systems
8.4.5. The effect of pymetrozine injection on displacement gain as
calculated in mosquito auditory systems via pure tone stimulation 271
9. Conclusions
9.1. Discussion
9.1.1. Conclusions of behavioural experiments
9.1.2. Conclusions of electrophysiological experiments
9.2. Critique
9.2.1. Experimental critique
9.2.2. Statistical and modelling critique
9.3. Future work
References

Appendix A. Selected free fluctuations of *Drosophila melanogaster* CantonS females used in the fertility assay (in section 4.3.1) that were Appendix B. Selected free fluctuations of Drosophila melanogaster CantonS males used in the temperature entrainment assay (in section Appendix C. Selected free fluctuations of Drosophila melanogaster CantonS females and males used in the courtship song assay (in Appendix D. Selected free fluctuations of Drosophila melanogaster males used in the competitive reproduction assay (in section 4.3.7) Appendix F. Comparison of pre- and post-ringer nerve and mechanical responses for Drosophila melanogaster and mosquito species325 Appendix G. Median values obtained from fitting the two-state model for two independent transducer populations to post-ringer injection data for all Drosophila melanogaster lines investigated in section 6.328 Appendix H. Median velocity fit values for all mosquito species and Appendix I. P-values for comparisons between mosquitoes of two Appendix J. Changes to K_{PEAK} and K_{STEADY} for sedated mosquitoes ... 340 Appendix K. Statistical comparisons between single population and Appendix L. CAP amplitude and dynamical stiffness changes following

Appendix M. Effective stiffness and best frequency shifts between	
control, sedated and pymetrozine injected states in mosquitoes, as	
well significance values for all relevant comparisons	345

List of tables

Table 3. Number of surviving An. gambiae larvae 24, 48 and 72 hours afterpymetrozine exposure for each of the five concentrations of pymetrozine tested(numbers in brackets are the percentage of the starting population remaining at thattime point, starred values are significantly different from the equivalent control groupvalues).119

Table 5. Median values for total length of the flagellum from the point of laser focus to the pedicel, the total length of the entire flagellum and the total surface area of the entire antenna (standard errors are given in brackets below the median values, significant differences between conspecific female and male mosquitoes are starred).

Table 6. Median first and second harmonic estimates for all species and sexes(number in brackets indicates standard error values) alongside ratios of male tofemale first harmonics, as well as previously reported first harmonic estimates: Ae.aegypti published data is from (352), An. gambiae published data is sourced from(353, 442), whilst Cx. guinguefasciatus published data is from (354).

Table 7. Median best frequency values for all Drosophila melanogaster lines testedcomparing post-ringer and post-pymetrozine states (standard errors given inbrackets) as well as P-values for statistical comparisons between the two states. .. 144

 Table 8. Median values of effective stiffness for all Drosophila melanogaster lines

 investigated comparing post-ringer and post-pymetrozine states (numbers in brackets)

Table 11. Median values of flagellar best frequency for male and female mosquitoesfrom each species investigated comparing post-ringer and post-sedation states(numbers in brackets are standard errors) as well as P-values for statisticalcomparisons between the two states.179

Table 12. Median velocity fit parameters and energy gain calculated for each mosquito species and sex tested in both the active and sedated states (standard errors are given in brackets; the number of mosquitoes used to calculate the median energy gain value for each group is equal to the size of the relevant passive state group). 181

Table 13. Median values of effective stiffness for male and female mosquitoes fromeach species investigated comparing post-ringer and sedated states (numbers inbrackets are standard errors) as well as P-values for statistical comparisons betweenthe two states.183

Table 14. Mean values for mosquito apparent antennal mass estimates (numbers inbrackets refer to standard errors). Significant differences between conspecific malesand females are starred.184

Table 16. Parameter estimates obtained fitting the two-state model of a singletransducer population to male and female Ae. aegypti, An. gambiae and Cx.quinquefasciatus mosquitoes after steady state stiffness has been subtracted fromthe system.199

Table 18. Median displacement gains estimated using white noise stimulation for allmosquito species and sexes investigated (standard errors are shown in brackets).Significance levels between different sexes within a single species are shown next tothe sex that was estimated as having the significantly greater level of energyinjection.210

Table 19. Median values of the best frequency of the flagellum and nerve responses to pure tone stimulation (numbers in brackets refer to standard errors). Female fundamental WBF and the best frequency estimates for the flagellum from the velocity fits are provided to act as reference values. Two of the most prominent distortion products are also shown, with the difference tone being equal to the difference between the male and female WBF and the cubic distortion product being equal to the difference. 213

Table 21. P-values for statistical comparisons between the different injection statesfor male and female mosquitoes from each species investigated.222

Table 22. Median values of flagellar best frequency for male and female Ae. aegypti,An. gambiae and Cx. quinquefasciatus mosquitoes (standard errors are given inbrackets, with sample sizes for after ringer and after pymetrozine groups providedrespectively below the species and sex type) as well as P-values for statisticalcomparisons between the two states246

Table 23. Median values of flagellar best frequency for male and female Ae. aegypti,An. gambiae and Cx. quinquefasciatus mosquitoes (standard errors are given inbrackets, with sample sizes for after ringer and after pymetrozine groups providedrespectively below the species and sex type).248

Table 25. Median values for two state model of a single transducer populationparameter estimates for both An. gambiae Ngusso and Tiassale strains, grouped bysex (values in brackets are standard errors).258

Table 28. Comparisons between before and after ringer injection states of the medianvalues for a Drosophila melanogaster line (BL1283) and Ae. aegypti femalemosquitoes using the two state model for either a single population of transducers ortwo independent populations (no significant differences identifiable between any ofthe model parameters when comparing before and after ringer states).327

Table 30. Median values obtained by fitting the two-state model for two independenttransducer populations to post-ringer injection data for the following Drosophilamelanogaster lines (numbers in brackets refer to standard errors): w¹¹¹⁸xGAL4, J287:1xGAL4 and J28 8:1xGAL4.329

Table 32. Median velocity fit parameters and energy gains for male and female Ae.aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes after ringer injection (withstandard errors provided in brackets).331

Table 33. Median velocity fit parameters and energy gains for male and female Ae.aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes after TTX injection (withstandard errors provided in brackets).332

Table 34. Median velocity fit parameters and energy gains for male and female Ae.aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes after TeNT injection (withstandard errors provided in brackets).333

Table 37. Median velocity fit parameters and energy gains for male and female Ae.aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes after pymetrozineinjection (with standard errors provided in brackets).336

Table 38. Median velocity fit parameters for CO2 sedated male and female Ae. aegypti,An. gambiae and Cx. quinquefasciatus mosquitoes after pymetrozine injection (withstandard errors provided in brackets).337

Table 39. P-values for statistical comparisons between male and female mosquitoesfrom the same species for parameters obtained from the two state model fits of asingle transducer population.338

Table 40. P-values for statistical comparisons between female mosquitoes fromdifferent species for parameters obtained from the two state model fits of a singletransducer population.339

Table 41. P-values for statistical comparisons between male mosquitoes fromdifferent species for parameters obtained from the two state model fits of a singletransducer population.339

Table 42. Number of individuals from each mosquito species and sex which are fit better assuming two independent transducer populations rather than a single population, as determined using the AICc (following previous reports (394)), as well as the AICc for each fit type for female and male mosquitoes from each species. ... 342

between the two states	. 346
(numbers in brackets are standard errors) as well as P-values for all comparisons	
male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes	
Table 44. Median values of effective stiffness for sedated or pymetrozine exposed	1

List of figures

Figure 1: Overview of the major sense organs within Drosophila melanogaster, with arrows linking behaviours and stimuli to the appropriate sensory organ (adapted from	
(25))	
Figure 2. A) Physiological outline of a standard ChO showing key components of the	
organ (adapted nom (272))	
<i>B) Relative locations in a ChO of several ion channels considered important for proper ChO function (adapted from (273)).</i>	
Figure 3. A) The antennal ear of Drosophila melanogaster (adapted from (281))55	
B) The internal distribution of scolopidia, as well as the effect of rotations of A3 on these neuronal populations (adapted from (282))	
Figure 4. A) The antennal shaft of a male Culex quinquefasciatus mosquito57	
B) Generalised male mosquito antennal system (adapted from (287)	
Figure 5. A) The reported courtship pattern identified in Drosophila melanogaster mating attempts (adapted from (339))	
B) Section of a courtship song produced by a Drosophila melanogaster male with important segments highlighted. 66	
Figure 6. A) Harmonic convergence of individual male and female mosquitoes as	
demonstrated by changes in wing beat frequency over time; red and blue lines show	
female and male wing beat frequencies respectively for different harmonics, with the	
bottom image showing a better resolution of changes to the third female and second male harmonic (adapted from (357))	
B) Distortion product formation from the interaction of two pure tone stimuli, f_1 and f_2 , with the potentially relevant 2 f_1 - f_2 distortion product highlighted (taken from (276)). 68	
Figure 7. Experimental outline of the flight initiation assay, with the landing areas	
which denote good or bad fliers being highlighted (adapted from (272))	
Figure 8. A) The Flycube setup, indicating the positions of several of the cameras as	
well the computer arrangement required (image taken from (401))	
B) Aggregated flight trajectories from multiple Drosophila melanogaster whilst	
exposed to the 'figure-8' stimulus paradigm (adapted from (402))	

Figure 9. Schematic of triplets used in the competition assay, with b referring to
Drosophila melanogaster with brown eyes from the LH _B –UCL line, m referring to
Drosophila melanogaster with red eyes from the LH _M –UCL line and exposed/
unexposed referring to whether or not the fly had been exposed to pymetrozine
(sample sizes are shown next to each triplet type)
Figure 10. A) Total egg count values per individual female for the two groups involved
in the fertility experiment (sample size in brackets, black dots correspond to the 5 th
and 95 th percentiles)
B) Larval count data per couple for all four groups tested for fecundity (sample size in
brackets, black dots correspond to the 5 th and 95 th percentiles)
Figure 11. A) Survival rate of each of the four groups over the entire course of the
experiment for flies that were kept individually (sample size in brackets)
B) Median number of flies surviving per vial for flies that were kept in groups of ten
(sample size given in brackets)95
Figure 12. Ratio of good to bad fliers for control flies or flies exposed to pymetrozine
at a concentration of either 100, 500 or 1000ppm (sample size in brackets and
significant differences between groups are starred)
Figure 13. A) Aggregated two dimensional trajectories for each of the three stimuli
patterns for control Drosophila melanogaster (n=60), with the number and total time
length of trajectories indicated above each stimulus
B) Two dimensional heat maps indicating the most commonly occupied positions by
control Drosophila melanogaster (n=60) being tracked during each stimulus type. The
colour gradient goes from blue to red (low to high average occupancy rate)
Figure 14. A) Aggregated individual two dimensional trajectories for each of the three
stimuli patterns for pymetrozine exposed (1000 ppm) Drosophila melanogaster (n=60),
with the number and total time length of trajectories indicated above each stimulus.98

Figure 15. Turning probability after onset of the 'figure 8' stimulus at relative time 0s for each of the pymetrozine exposure levels tested (n=60 for all groups, with dark

Figure 17. A) Intrapulse frequency values for each of the four groups (black dots correspond to the 5th and 95th percentiles). 102

B) Two dimensional heat maps indicating the most commonly occupied positions by female Cx. quinquefasciatus mosquitoes (n=12) being tracked during each stimulus. The colour gradient goes from blue to red (low to high average occupancy rate). ... 122

Figure 21. Free, unstimulated fluctuation data for an individual Drosophila melanogaster from the 91-S line, with a frequency range of between 51 – 3200Hz and

B) Flagellar displacements (dashed line) and CAP amplitudes (solid line) for pymetrozine unexposed (left, purple) or exposed (right, grey) Drosophila melanogaster showing the time delay between flagellar and nerve responses (stimulus onset is at start of flagellar displacement and force is constant throughout).

B) Dynamic stiffness calculations for K_{INFINITY} (red dashed line), K_{STEADY} (blue dashed line) and K_{PEAK} for pymetrozine unexposed flies (left) and exposed (right) flies (solid black lines represent two state model fits of two independent transducer populations).

Figure 25. A) Free, unstimulated fluctuation data for a 91-S fly (left) before and after pymetrozine, with the equivalent states in a 91-R fly shown to the right. The frequency range for both flies is 51 – 3200Hz, with individual data points (in purple and grey for before and after pymetrozine injection states respectively) representing Fourier-transformed velocity amplitudes at each frequency contained within the transform range. Solid black lines show the velocity amplitude function fits for each recording.

Figure 26. A) Effective stiffness values for all Drosophila melanogaster lines investigated comparing post-ringer and post-pymetrozine injection. 147
B) Best frequency values for different Drosophila melanogaster lines comparing post-ringer and post-pymetrozine injection. 147
For both A) and B), from left (also labelled in graph): 91-S, 91-R, BL1283, BL 1675, w¹¹¹⁸xGAL4, J28 7:1xGAL4 and J28 8:1xGAL4. Starred values indicated significant differences between groups (with resistance types being documented in section 6.2.3)

Figure 30. A) Flagellar displacement (top) and CAP amplitude (middle) in response to the corresponding pure tone stimuli (bottom) for Aedes aegypti females - the left pure tone stimulus frequency is set at 185Hz whilst the pure tone stimulus to the right has a frequency of 445Hz.

Figure 31. Free, unstimulated fluctuations both before and during sedation over a frequency range of 100 to 2000Hz for A) an Ae. aegypti female, B) an An. gambiae female and C) a Cx. quinquefasciatus male. Individual data points represent Fourier-

transformed velocity amplitudes whilst solid black lines show the velocity amplitude function fits for each recording. 180

Figure 33. Flagellar displacement (top) and CAP amplitude (middle) in response to the corresponding force steps (bottom) for A) Ae. aegypti females (in red) and B) Ae. aegypti males (in blue).

Figure 36. Changes in dynamical stiffness and CAP amplitude in response to changes in antennal displacement to a maximum displacement of ±2000nm for female and male mosquitoes from each species investigated: A) Ae. aegypti females, B) An. gambiae females, C) Cx. quinquefasciatus females, D) Ae. aegypti males, E), An. gambiae males and F) Cx. quinquefasciatus males. The bold lines represent two-state gating spring model (for a single transducer population) fits to the median data (285).

Figure 37. Dynamical stiffness changes in response to changes to displacement calculated using either values calculated at the peak displacement (K_{PEAK}) or at the steady state (K_{STEADY}) for A) Ae. aegypti females, B) Ae. aegypti males, C) An. gambiae females, D) An. gambiae males, E) Cx. quinquefasciatus females, F) Cx. quinquefasciatus males and G) a Drosophila melanogaster male (CantonS strain).. 196

Figure 39. Changes in dynamical stiffness calculated by subtracting K_{STEADY} values from K_{PEAK} values for A) Ae. aegypti females, B) An. gambiae females, C) Cx. quinquefasciatus females, D) Ae. aegypti males, E), An. gambiae males, F) Cx. quinquefasciatus males and G) Drosophila melanogaster males (CantonS line). Solid lines represent two-state gating spring model (for a single transducer population) fits. 201

Figure 41. Changes in dynamical stiffness in response to changes in antennal displacement to a maximum displacement of ±2000nm plotted using two-state gating spring model (for either one or two independent transducer populations) fits to the median data for: A) Ae. aegypti females, B) An. gambiae females, C) Cx. quinquefasciatus females, D) Ae. aegypti males, E), An. gambiae males and F) Cx. 207

C) Changes in the median best frequency of the flagellum relative to the unstimulated best frequency as the stimulus force increases for An. gambiae and Cx. quinquefasciatus females (dashed lines represent median best frequency of the passive systems).

D) Changes in the median best frequency of the flagellum relative to the unstimulated best frequency as the stimulus force increases for male Cx. quinquefasciatus mosquitoes (dashed lines represent median best frequency of the passive systems). 209

Figure 43. Displacement gain values estimated using white noise stimulation for each species and sex (sample sizes are given in brackets, with significant differences

Figure 44. Sensitivity (dashed line) and nerve response (solid line) changes in response to changes in stimulus frequency for A) Ae. aegypti females, B) Ae. aegypti males, C) An. gambiae females, D) An. gambiae males, E) Cx. quinquefasciatus females and F) Cx. quinquefasciatus males. Fundamental WBFs are indicated as well as the difference tone and cubic distortion product (labelled 'diff tone' and 'cubic DF').

Figure 48. A and B) Time series of representative antennal velocities before and after ringer injection for An. gambiae males compared to TTX (A) or TeNT (B) injection. 220

C and *D*) Time series of representative antennal velocities before and after ringer injection for Cx. quinquefasciatus males compared to TTX (C) or TeNT (D) injection.

Figure 49. Energy gain values calculated for different injection states for A) Ae. aegypti females, B) Ae. aegypti males, C) An. gambiae females, D) An. gambiae males, E) Cx. quinquefasciatus females and F) Cx. quinquefasciatus males (significant differences are starred, with black dots corresponding to the 5th and 95th percentiles).

Figure 50. Changes in dynamical stiffness (A) and CAP amplitude (B) in response to changes in displacement for Cx. quinquefasciatus female mosquitoes exposed to

Figure 53. Best frequency and effective stiffness values after ringer injection, sedation or pymetrozine injection for male and female mosquitoes from A) Ae. aegypti, B) An. gambiae and C) Cx. quinquefasciatus, with significant differences between different states within a sex starred (black dots correspond to the 5th and 95th percentiles)... 249

Figure 54. Flagellar displacement (top) and CAP amplitude (middle) in response to the corresponding force steps (bottom) after pymetrozine exposure for A) Ae. aegypti females and B) Ae. aegypti males – N.B. an artifact (as the result of crosstalk between the stimulus and electrodes) is present in the nerve response for the larger force step.

Figure 58. Changes in dynamical stiffness and CAP amplitude in response to changes in antennal displacement to a maximum displacement of ±2000nm for each of the An. gambiae strains and sexes investigated after ringer and after pymetrozine: A) Ngusso females, B) Ngusso males, C) Tiassale females and D) Tiassale males. The bold lines represent two-state model of a single transducer population fits to the median data. Standard error bars (represented by black vertical lines) are included for Tiassale but not Ngusso because for that strain only one injection experiment was recorded for both sexes.

Figure 59. A) Changes in mechanical sensitivity in response to pure tone stimulation both post-ringer and post-pymetrozine injection for a male Cx. quinquefasciatus...263

Figure 61. Free, unstimulated fluctuations of female Drosophila melanogaster involved in the fertility assay described in section 4.3.1 for either a pymetrozine unexposed female CantonS (A) or a pymetrozine exposed female (B). The examples shown here are representative of all flies measured from each experimental group.319

Figure 63. Free, unstimulated fluctuations of a pymetrozine unexposed female CantonS (A), a pymetrozine exposed female CantonS (B), a pymetrozine unexposed

List of abbreviations

- CAP = Compound Action Potential
- ChO = chordotonal organ
- DD = Dark/Dark
- IPF = Intrapulse frequency
- IPI = Interpulse interval
- JO = Johnston's organ
- LDV = Laser Doppler Vibrometry
- LD = Light/Dark
- Nan Iav = Nanchung/Inactive
- ppm = parts per million
- TTX = Tetrodotoxin
- TeNT = Tetanus neurotoxin
- WBF = Wing Beat Frequency
- WHO = World Health Organisation

1. Introduction

Significant decreases in mortality and morbidity rates associated with some mosquito borne diseases have been seen worldwide over the past decade, particularly with regards to malaria and lymphatic filariasis (1, 2). These reductions have been possible as a result of control programmes involving indoor residual spraying of insecticides, the distribution of insecticidal mosquito nets and advances in therapeutic and preventative treatment of the diseases (for example, using artemisinin combination therapy to combat malaria) (3-6).

Increasingly, however, these significant advances are being placed under pressure through a combination of insecticidal and antibiotic resistance from the vector and parasite respectively (7-9). These dangers are further enhanced by the relatively limited number of insecticides that are available for distribution, some of which share similar mechanisms of action thus making resistance development to a single insecticide class highly problematic (10-12). Insecticidal resistance has not yet become sufficiently prevalent as to render continued usage of currently available insecticides ineffective, with insecticide treated bed nets still being reported to be more effective than untreated nets regardless of resistance levels for example (13). However, the serious problems that emerged during previous disease control attempts which relied too heavily on too few control measures (such as DDT) can serve as a reminder of the importance of using multiple methods of control (14, 15).

Alongside these growing concerns of resistance, the increasing spread of some mosquito species to previously uninhabited sections of the world could lead to increased disease transmission – *Aedes aegypti* and *Aedes albopictus*, which act as vectors of a number of different diseases such as dengue, have already been identified in Southern Europe and are predicted to increase in range considerably over the coming decades (16, 17). These changes in distribution are driven partly by climate change (18, 19) but also by the unprecedented scale of urbanisation and globalisation taking place worldwide (20, 21).

These twin sets of pressures place increasing demands on mosquito control programmes – insecticides and antibiotics are becoming more and more essential at a time when their effectiveness is decreasing constantly. It is clear therefore that new insecticides, with preferably novel mechanisms of action, are required to reduce the burden placed on the compounds that have currently been approved for use. Given the substantial delays that can occur between the identification of a promising compound and the actual introduction of that substance to the field, insecticides that have already been demonstrated as efficacious as well as safe for use alongside humans would be ideal for this purpose.

Slow acting feeding inhibitors (referred to here as mechanotoxins) for example are currently used as insecticides for crop pests but are also a potentially promising class of compounds for adaptation to combatting mosquito populations. One member of this class is pymetrozine (Syngenta, Switzerland), which has been shown to eliminate chordotonal organ (ChO) function in several insect species (22, 23). ChOs are ciliated stretch receptors which act as external sensory organs in many insect, but not mammalian, species; in *Drosophila melanogaster* ChOs are involved in both proprioceptive and auditory sensing (24, 25). Abolishing ChO function in plant-sucking insects has been reported to cause starvation (26).

The ablation of ChO function that follows pymetrozine exposure occurs specifically in these mechanoreceptors because the compound targets the Nanchung/ Inactive (Nan-Iav) ion channel heterodimeric complex, which appears to be solely confined to these units (27), and as such at low concentrations pymetrozine has been declared safe for use as a generic pesticide. This mechanism of action renders the drug distinct from other insecticides already in widespread use and makes it an interesting candidate for investigation.

Proper insect mechanosensory function is reliant on ChOs (28). Mechanosensation and mechanosensory feedback has been identified as being involved in flight (29), courtship rituals (30, 31), host identification by female mosquitoes for blood feeding attempts (32) and circadian rhythm regulation (33). Mosquito courtship in particular has been studied in depth because of the unusual phenomenon of frequency matching within couples at higher order harmonics of the fundamental flight tone (34). Given that this process necessitates both flight and auditory function before successful copulation can occur, it is a promising target for compounds which could theoretically deprive mosquitoes of mechanosensory capabilities.

Whilst much more is known of mechanosensation in *Drosophila melanogaster* than in mosquito species, there are still gaps in the general knowledge of the mechanosensory system. Pymetrozine therefore offers two different topics of interest; first as a proven pesticidal compound against crop pests that has the potential to be effective against mosquitoes and could therefore relieve some of the pressure felt by the currently used toxins; second as a tool to pharmacologically ablate mechanosensory function in insects and thus allow for experimentation and investigation into a relatively unexplored area of biological importance. That a compound with this mechanism of action has not yet been tested in mosquitoes increases its potential relevance as both a toxin and a laboratory tool.

This project therefore seeks to integrate these two possible usages by attempting to assess the potential of pymetrozine (and thus also other mechanotoxins that act in a similar manner) for use in mosquito control programmes whilst also utilising the drug to explore the relative importance of mechanosensation at different stages of the insect life cycle.

2. Literature review

2.1. Major vector species of mosquito borne diseases

There has been a significant shift in the relative contributions of communicable and non-communicable diseases towards the global burden of disease over the past century, with non-communicable diseases now being responsible for approximately 60% of the total burden (35-37). This burden is unequally distributed however, with many low- and middle-income countries tending to suffer inordinately more from communicable diseases than high-income countries in 2012 but was not present in any other income-bracket list (38). Whilst malaria is responsible for the majority of the morbidity and mortality caused by mosquito-borne diseases worldwide, other pathogens (in particular, lymphatic filariasis (39) and dengue (20)) also significantly contribute to this global problem; mosquito transmitted diseases clearly therefore remain a serious issue in certain areas of the world (35, 40).

Whilst there are currently over 3500 different identified mosquito species worldwide (41), the vast majority of these are not necessarily relevant when considering disease transmission – only 70 of the 460 formally recognised *Anopheles* species have been proven to be able to transmit malarial pathogens for example (42). Further than this several major mosquito species play a disproportionally large role in the transmission process of many of the most epidemiologically important mosquito borne diseases. These include *Anopheles gambiae, Ae. aegypti* and *Culex quinquefasciatus,* which together can be considered as the major vectors of malaria, dengue, Chikungunya, yellow fever and lymphatic filariasis (43-46).

As such, whilst the role other mosquito species play in the transmission of various diseases should not be neglected (such as *Anopheles arabiensis* and *Ae. albopictus* with regards to malaria and dengue respectively), these three species remain the major targets of many vector control programmes; there are significant differences between these species however that prevent all-encompassing control measures from being applied (47-49).

Differences in the biology of the various mosquito-borne pathogens can also result in enforced changes to disease control programmes. For example whilst the incubation periods of many mosquito borne diseases are temperature dependent they can also differ significantly between pathogens - the average incubation period for dengue has been approximated between 4.7 and 6.5 days whilst for lymphatic filariasis (when considering the *Wuncheria bancrofti* pathogen) this period can be between 10 and 20 days (50-53).

2.1.1. Aedes aegypti

The global expansion of *Ae. aegypti* has proceeded at a significant rate since the 1980's as urbanisation and climate change has increased this mosquito species' survival range to include even southern Europe (16, 54). This expansion has fuelled a similarly significant increase in the number of cases of dengue and the Chikungunya virus, with dengue in particular being the most prevalent human arbovirus worldwide (20).

Transmission of these two diseases occurs predominantly in urban environments as a result of the strong preference of *Ae. aegypti* mosquitoes for such living habitats, with egglaying and thus development of larval and pupal stages tending to occur in small man-made containers around human habitats (17, 44, 53). With the aforementioned increased rate of urbanisation that has occurred over the past 50 years (which shows no signs of halting) this has placed almost half of the world's population at risk for *Ae. aegypti* borne diseases (55). Females deposit single, unfloated eggs from which larvae, which rely on a siphon for respiration and lie at an angle to the water surface, emerge (56).

Female *Ae. aegypti* are highly anthropophilic (with reports of over 90% of blood meals coming from humans) and will generally only feed from other food sources in the absence of humans (which is uncommon in densely populated urban environments) (47, 57). Females also often take multiple blood meals within a single gonotrophic cycle (which lasts typically 3 to 4 days), greatly increasing the risk of disease transmission should a female become infectious (58-60).

In addition to this, dengue transmission tends to occur during the daytime and outdoors, thus placing the mosquito outside the range of the insecticidal nets and indoor insecticidal spraying that have proven highly effective against *An. gambiae* in some locations (61). As such control of *Ae. aegypti* borne diseases has proven particularly challenging and requires a variety of different control mechanisms, which tend to target egg laying sites as well as the immature stages of adult development (62).
2.1.2. Anopheles gambiae

In contrast to the preference shown by *Ae. aegypti* and *Ae. albopictus* for urban environments *An. gambiae* has overall demonstrated a strong inclination towards more rural areas of sub-Saharan Africa, where it remains the dominant malaria vector (43). As malaria was responsible for approximately 429,000 mortalities in 2015, with 92% of these deaths in the African region, malaria control programmes have heavily targeted this species in order to reduce the burden of disease (1, 63).

Female *An. gambiae* act as highly effective vectors of malaria because of their strong anthropophilic tendencies – a reported 90% of blood meals are taken from humans and females from this species are able to identify and target individual humans within a crowd of other animals (64). Females are capable of taking multiple blood meals every gonotrophic cycle (which lasts approximately 2 days) but this is highly dependent on the size of the mosquito, with smaller females requiring more blood feeds (65, 66). Although female *An. gambiae* feed from sources of sugar throughout their life-cycles this does not seem to significantly alter their blood feeding habits with regards to humans (67, 68).

An. gambiae females have a strong propensity for indoor biting at night and such vector control programmes have found significant success by distributing mosquito nets sprayed with insecticides as well as indoor spraying of houses with insecticides (1, 69, 70). Females from this species also tend to lay their eggs on the surface of calm, clean bodies of water which can then be targeted using larval source management; these eggs are laid singly on floats, with the emerging larvae lacking a siphon and lying parallel to the water surface (56, 71). Whilst extensive control programmes have produced significant reductions in mortality and morbidity, malaria control has not yet been completely successful and future elimination attempts require even greater efforts to be made.

2.1.3. Culex quinquefasciatus

Similar to Ae. aegypti, Cx. quinquefasciatus mosquitoes tend to prefer urban environments in tropical and sub-tropical regions - Cx. quinquefasciatus larvae are able to survive in heavily polluted water, such as sewage tanks, however whilst Ae aegypti larvae can only survive in relatively cleaner environments (56, 72). Cx. quinquefasciatus is the major vector species for West Nile Virus, Saint Louis encephalitis and the nematode *Wuncheria bancrofti* (73).

Although *Cx. quinquefasciatus* eggs are deposited on the surface of the water as large rafts of more than 100 eggs, the larvae and pupae which emerge and develop from these eggs act in a similar manner to *Ae. aegypti* of a comparative developmental stage – adult mosquitoes from these two species also rest parallel to the ground in an identical manner (56).

Cx. quinquefasciatus females are highly opportunistic and will acquire blood meals from multiple sources depending on nearby availability, meaning that estimates of anthropophilic behaviour can vary significantly depending on the environment of the region studied (74, 75). There is evident plasticity in female feeding habits as well, with an overall tendency to attempt to acquire blood meals indoors during the night but also a significant amount of biting attempts outdoors in some locations - this high level of variation extends to the percentage of females which take multiple blood meals per gonotrophic cycle, which lasts for around 3 days on average (72, 76, 77).

As a result of the flexibility demonstrated in *Cx. quinquefasciatus* biting habits, control programmes have been forced to utilise multiple methods of vector control, ranging from attractive toxic sugar bait stations in key locations for egg laying to the controlled introduction of bacterium targeting mosquito larvae (78, 79).

2.2. Mosquito-borne disease transmission

As is clear from section 2.1, there are significant differences between the major mosquito species associated with disease transmission. There are some similarities that are maintained across the three species however, with frequency matching of males and females flight tones at higher order harmonics of the fundamental wing beat tone being reported as an essential part of courtship in all of these species - females from all three species must also consume blood meals in order to produce viable offspring, with this process directly linked to disease transmission (34, 56).

Transmission of a mosquito-borne disease can occur when a female mosquito that is infected with a disease takes a blood meal from a human and transfers a pathogen in the process. Beyond this fundamental step in the process, tremendous variation is observable even in mosquito biting habits (47, 80, 81), pathogen development (82-84) and tendency towards anthropophilic behaviour (64, 85, 86) (as noted in section 2.1). These differences are seen not only between species but also between members of the same species in different areas (87). This has made the biting process itself a key target for insecticidal control measures because of its ubiquity amongst all mosquito species of interest.

Species specific preferences for human blood meals (as opposed to blood from other animals) can play an important role in disease transmission (88, 89). Mosquito attraction to humans is mediated by a number of different factors including mosquito genetic predisposition as a result of odorant receptor evolutionary pathways (90) and human genetics (91). There are potentially many physical and chemical cues that females use to identify suitable hosts (92), such CO₂, temperature and body odour (93-96) – female mosquitoes may then integrate these separate sensory cues during host seeking (97).

Olfactory regulation in females, particularly with respect to odorant receptors in *An. gambiae*, has therefore become a potential target for control measures that seek to prevent biting events from occurring (98-100). The increasing capability to produce transgenic lines in laboratory settings (such as those created to study the odorant receptor co-receptor gene *Orco*) has also provided greater levels of insight into olfactory centre organisation (101).

Pathogen infection can create significant changes to insect blood-feeding behaviour; for example, *Anopheles gambiae* mosquitoes infected with the *Plasmodium* parasite are far more attracted to human odorants and are more likely to attempt to obtain blood meals than their uninfected counterparts (102-104).

In addition to this, mice infected with *Plasmodium chabaudii* (the rodent malarial parasite) display enhanced attractiveness to mosquitoes during key periods of the pathogen cycle in which the mammals are highly infectious (105).

The blood feeding process itself follows a reasonably standard pattern; after a female mosquito has located a potential source of blood and landed upon it, she will skim her labellar tip over the surface of the skin in order to identify promising insertion points. Once she found a satisfactory position she will separate her forelegs symmetrically around her entire proboscis so as to generate as much force as possible whilst her fascicle penetrates the skin in a series of rapid thrusts. The labium and maxilla will then follow through the puncture (106). Both the labium and fascicle have been suggested to be essential to the blood feeding process, with females lacking a labium unable to penetrate human skin (107) whilst those without a fascicle have drastically reduced blood sensory capability (108).

The flexibility of the fascicle tip allows it explore fully the freshly created perforation in search of a blood vessel; once one has been located and entered, the maxillary palps vibrate in phase with the entire mouthparts of the female to stimulate blood flow and begin the feeding process (109). There are several steps therefore throughout the feeding procedure in which mechanosensation could play an important role, with the reported presence of ChOs in the female mosquito labellar lobes also suggestive of a mechanosensory component of the biting procedure (110, 111).

2.3. Current control programmes

The significant variation observable between mosquito species in terms of biting habits, breeding locations and global distributions (as discussed in section 2.1) necessitates targeted control programmes that change focus according to the vector and disease in question. Certain situations require specific control tactics; for example indoor residual spraying may be effective against *An. gambiae* females that prefer to bite indoors but could be relatively ineffective against other *Anopheles* species that tend to blood feed outside the confines of the home (112, 113).

The tendency for *Ae. aegypti* and *Ae. albopictus* females to thrive in urban environments whilst *An. gambiae* prefer more rural areas is also indicative of the unsuitability of generic control programmes as different control measures are required for the different environments (114, 115). Vectors of the same disease, such as lymphatic filariasis, may even require different insecticidal strategies depending on the continent that they are found upon (116).

At their most general therefore, most current mosquito control programmes rely on a mixture of targeting the disease vector via some manner of insecticide application (117) whilst also combatting the pathogen with either prophylactic mass drug administration (118) or rapid treatment of individuals presenting signs of illness with antibiotics (or similar medical compounds), which not only help to reduce morbidity associated with the disease but also decrease the risk of transmission events occurring (119). Some prophylactics can also have an effect on the mortality rate of biting mosquitoes, further enhancing their usefulness (120). These generic programmes are then tailored to specific situations in order to be as effective as possible.

Further than this, vaccines for certain diseases are already available, such as yellow fever (121) and dengue (122). Although a malaria vaccine will hopefully become available for inclusion in control programmes within the foreseeable future the precise timelines for its potential deployment are unclear at this time (123).

2.3.1. Vector control

2.3.1.1. Insecticidal nets and spraying

There are four major categories of insecticide classified as suitable for use in order to combat mosquitoes, all of which have distinct mechanisms of action and can be used in various combinations; these are the pyrethroids, the organophosphates, the organochlorines and the carbamates (12). These can then be further classified as targeting either voltage-gated sodium channel proteins (with both the pyrethroids and the organochlorides preventing the closure of these channels (124, 125)) or the enzyme acetylcholinesterase, which is essential for the termination of synaptic transmission - the carbamates reversibly inactivate this enzyme whilst the organophosphates permanently inactivate it (126, 127).

The two most common methods of introducing one of these insecticide types to the mosquito vector are indoor residual spraying of pesticides and long lasting insecticidal bed nets (128). Both have proven highly effective in reducing disease prevalence in a variety of circumstances by decreasing the number of disease transmission events – both by eliminating female mosquitoes that are about to take a blood meal and by removing females within the mosquito population that are able to successfully produce offspring (129-131).

These two control mechanisms can be used in conjunction with one another in order to maximise coverage, though the effectiveness of such a control programme is dependent on many different variables including location, spraying frequency and bed net coverage rates (132-134). Both of these methods of control are far less effective against mosquito species that tend to take blood meals outdoors or are less anthropophilic than other vector species and so require supplementation as part of an integrated programme (135, 136).

Preventing biting events from occurring in regions where biting and disease transmission tend to occur outdoors (and as such outside the effective range of indoor pesticide spraying) requires different strategies. The effectiveness of topical insecticidal repellents to combat mosquitoes attempting to bite in such places is uncertain, with large variation between the results of different studies (137). Low compliance rates, inappropriate usage of repellents and suboptimal coverage compound coverage can all play a significant role in this however rather than there necessarily being fundamental issues with the use of repellents (138, 139). Insecticide treated clothing has shown more effective results in terms of biting protection but requires repeated treatment of insecticides to counteract the removal of compounds when the clothing is washed (140).

2.3.1.2. Larval management and oviposition traps

Another method that allows for targeting of mosquitoes species that do not bite indoors (and as such could be considered beyond the reach of bed nets and indoor spraying), as well as for the utilisation of different pesticides than those used to target adult insects, is the targeting of the larval stages of the mosquito developmental cycle (141). This has shown some positive results in certain situations (142-144) and has thus led to the trial of many different types of control compounds (145, 146) as well as unique mechanisms of compound dissemination (147).

Identification of areas containing high larval densities can be difficult for mosquito species that tend to prefer urban environments (such as *Ae.* aegypti) and can result in significant costs associated with door-to-door searching efforts (62, 148). Targeting specific locations, such as junkyards, which are likely to contain suitable locations for mosquito egg-laying to occur can help to make larvicidal applications more time and cost efficient (149, 150). Area-wide low volume insecticide dispersal using commercially available sprayers has also been trialled and found to be effective in reducing larval emergence, though a corresponding decrease in adult population size was unfortunately not observed (151, 152).

This seeming disparity between significant reductions in apparent larval populations and a seemingly steady population size for adults has been reported for several studies that targeted the larval stages of mosquitoes using a variety of methods and could be due to inadequate distribution of the larvicidal compound or potential effects of larval density on adult survival rates (150, 153, 154). Regardless of the precise cause, the lack of a decrease in the adult mosquito population size raises some doubts over the effectiveness of larval management programmes (62).

One method that could be therefore be used as a supplementary measure alongside larvicidal applications is the introduction of oviposition traps; small pots of water containing both oviposition stimulants and an insecticide so that when a female mosquito attempts to lay eggs she is exposed to the compound (62). These traps are particularly suitable for targeting *Aedes* species in urban environments and have shown positive results in field trials, though they are likely to be less effective if used to target *Anopheles* species, whose egg laying habits are significantly different (155, 156).

2.3.1.3. Attractive toxic sugar bait

Another control method that can be utilised against mosquito species that tend to bite outdoors is the usage of attractive toxic sugar baits, which involve doping a sugar-based source of mosquito food with a pesticide and have proven successful as well in specific circumstances, provided the mosquito species that is being targeted tends to feed regularly from specific sugar sources (157, 158).

Utilising insecticidal ingestion as the method of exposure can circumvent issues involved with some cuticular resistance mechanisms and can lead to increased compound efficacy but does still require the compounds involved to be safe for use around humans (159).

There are two separate versions of the attractive toxic sugar bait measure, with one entailing spraying the toxic solution onto vegetation near ponds and other water sources known to contain mosquito larvae whilst the other involves setting up toxic sugar bait stations inside residences – field trials utilising both methods have noted reductions in estimated adult population sizes of greater than 80%, with the female population who had completed multiple gonotrophic cycles being particularly effected (160, 161).

This specific population is of great interest for control programmes as it is these females which are the most likely to be infected with a pathogen (given the necessary disease incubation period) and as such reduction in the size of group can lead to a disproportionate decline in disease incidence. Treating livestock with similar insecticides, alongside other methods utilising zooprophylaxis, has also shown some potential (162, 163).

2.3.1.4. Proposed control methods utilising fungal, bacterial and genetic mechanisms

Trials of fungal and bacterial (principally *Wolbachia*) pathogens have demonstrated significant potential as alternative methods for targeting mosquito species (164-166). Fungal pathogens can be sprayed indoors in a comparable manner to that of traditional insecticides in indoor residual spraying programmes and have the benefit that mosquito mortality schedules can be pre-determined; this in theory should decrease the possibility of resistance developing as well as reducing the number of infectious bites that occur (167-169).

Wolbachia pathogens have been utilised in a completely different manner however; Wolbachia infected Ae. aegypti that have been released in semi-field trials have been reported to invade existing mosquito populations and block dengue transmission because of pathogen-mediated changes to mosquito host biology (such as increased feminisation of males and cytoplasmic incompatibility) (170, 171). Whilst this technique appears highly promising and could be extended to the control of many mosquito borne diseases, there are some concerns over the impact that *Wolbachia* could have on localised biodiversity (172).

A variety of genetic methodologies also seek to disrupt disease transmission via the controlled release of laboratory reared mosquitoes into the field – these include the sterile insect technique and the release of mosquitoes carrying a dominant lethality (173, 174). The first technique involves the release of artificially sterilised males who are thus unable to produce offspring despite mating with females with (theoretically) the same likelihood as non-sterilised wild males, which should result in a decrease in population size (although this has not always been the case in field tests) (175, 176). Male mosquitoes released utilising the second technique are mutants carrying dominant lethal mutations which result in their female offspring suffering from a drastic reduction in competitive fitness (177, 178). Genetic techniques have a significant level of potential with regards to mosquito control but have yet to become fully integrated in vector control programmes.

There are therefore a large variety of potential vector control tools that can be used – unfortunately however several of these measures (such as the genetic techniques) have not been widely implemented whilst others (such as larval source management) require extremely high levels of coverage and expenditure to have any observable effect. As such, insecticidal nets and indoor residual spraying continue to provide the backbone of vector control efforts. Whilst they have so far done so successfully, this could cause problems in the future if insecticidal resistance reaches a critical threshold; they are also unable to substantially help control efforts targeting female mosquitoes that bite outdoors.

2.3.2. Pathogen control

Preventative treatment reduces both disease incidence and transmission by stopping disease progression should an protected individual be bitten by an infected mosquito, with mass drug administration of compounds such as ivermectin having been reported to lead to significant decreases in the prevalence of diseases like lymphatic filariasis (179, 180). Chemoprevention is only possible for diseases which have appropriate antibiotic treatments available however and so is not suitable to combat certain diseases spread by *Ae. aegypti* for example (181, 182).

The successes observed for such programmes targeting lymphatic filariasis and onchocerciasis have resulted in such control measures being suggested for expansion to malaria control programmes, whilst recognising certain caveats regarding the possibility of increased antibiotic resistance prevalence (183-185). Seasonal malaria chemoprevention using a combination of amodiaquine and sulfadoxine-pyrimethamine has already been recommended in some regions of sub-Saharan Africa and has the potential for significant decreases in malaria-associated morbidity (186, 187).

In addition to preventative treatment programmes, immediately treating patients who present any signs of illness can also help to decrease disease transmission as well as reduce key morbidity and mortality factors (188, 189). Unfortunately whilst this is possible for malaria because of the existence of artemisinin combination therapy, which has rapidly become the gold-standard antibiotic course for malaria in spite of increasing reports of antibiotic resistance (189, 190), there are no specific remedies available for treating infection with dengue or the chikungunya virus. This means that only treatment of the disease symptoms is possible, as well as attempting to reduce the possibility of further mosquito biting events (by transferring infectious individuals to hospitals for treatment and continually keeping them under insecticidal nets for example) (191).

2.3.3. Vaccination

Vaccination has repeatedly been demonstrated as among the most effective methods of combatting infectious diseases, with pathogens such as smallpox, polio and measles either being greatly reduced in incidence or completely eradicated due to sustained vaccination programmes (192). Unfortunately vaccine development can be a drawn-out and arduous process with no guarantee of success – the development of an effective vaccine to combat malaria for example has been particularly challenging (193-195).

As a result of this, much has been made of, and many hopes attached to, the RTS,S/AS01 malaria vaccine. Trials so far conducted have produced mixed results – there is published evidence of infection prevention over a four year period (196) but this has been found to strongly wane over time, which can cause significant issues as the average age of infection increases (197). In addition to this, the original goal of 50% efficacy has not been met during any of the later stage trials and the vaccine is not effective against *Plasmodium vivax* strains (198). Whilst other vaccines targeting malaria are being researched, they remain well behind the RTS,S/AS01 vaccine in terms of the developmental pipeline, meaning that for the foreseeable future at least vaccines will play at best a supporting (rather than a major) role in malaria control initiatives (195, 199).

Vaccines are available for certain mosquito-borne diseases however, with the yellow fever vaccine in particular proving to be safe, long-lasting and highly efficacious (200). A dengue vaccine, CYD-TDV, has been licensed for use in some countries and although it has encountered some issues regarding serotype susceptibility it could be become an effective tool in regions suffering from a high dengue burden (122, 201). Early stage trials for potential Zika virus vaccines have been initiated as well, though given the experiences of such trials for malaria vaccines early stage results should be treated with extreme caution (202, 203).

Whilst suitable vaccines are not available to combat every infectious disease, where they are obtainable they form a key part of control strategies. Nevertheless it must be stressed that even the most effective vaccines acting alone will not lead to the elimination of many mosquito borne diseases and as such should be utilised as part of a wider ranging programme of control measures that also feature different mechanisms of disease control.

The use of vaccines as a control measure does hold a significant advantage over the other two methods of control however in that it alone is not subject to the growing concerns regarding insecticidal and antibiotic resistance that could cause severe problems for disease control initiatives in the future (7, 190, 204).

2.4. Insecticidal and antibiotic resistance

Evolutionary pressures have led to the development of high levels of insecticidal resistance in many mosquito species (205-207). The extent of the selective pressures faced by mosquitoes can be judged by the rapidity in which susceptible populations develop resistance – for example the mosquito mortality rate in some locations after insecticide exposure (at currently advised levels) has fallen from 100%, when the toxin was first introduced, to less than 50% after several years of constant use (208, 209). In particular, within three years of spraying in Mexico (210) and after less than a decade of interventions in Tanzania mosquito populations with almost no discernible resistance levels became highly resistant to pesticides (211).

Resistance to the pyrethroid class of insecticides is of particularly significant importance because insecticidal bed nets can only be treated with members of this category of compounds and a significant proportion of toxins used for indoor spraying are pyrethroid based (12, 212). This has made reports of the development and globalisation of pyrethroid resistance all the more concerning (213-215) especially when taken in conjunction with the evolution of resistance to other insecticides classes (as well as resistance to multiple insecticides, which is often linked to cytochrome P450 enzyme mutations) (216-219).

Resistance can be classified into one of four types; cuticular (absorption of the compound is reduced due to cuticular modification (220)), metabolic (enzyme activity increases prevent toxin effectivity (221)), target-site (change in compound binding-site structure reduces probability of successful insecticide binding (222)) or behavioural (behavioural adaptations allow for toxin avoidance (223)). Behavioural resistance is the least well categorised and understood of the four but has been suggested as having a sizeable impact on vector control (224, 225).

Resistance mechanisms can occasionally have indirect consequences on important aspects of mosquito behaviour. $ace-1^R$ mutants for example display resistance to both carbamate and organophosphate based toxins but also demonstrate altered biting frequency and rhythmicity as result of changes to their salivary protein expression patterns (226). The $ace-1^R$ mutation confers a fitness cost on mutants (227), which is not uncommon for mutant alleles. This reduction in competence can then lead to a reduced response to pathogen infection in the mosquito, causing a relative increase in infectivity potential in these resistant individuals and thus potentially increasing the likelihood of infectious bites occurring as population resistance grows (228, 229).

Insecticidal resistance does not however necessarily have to mean total resistance to all concentrations of a specific compound, meaning that insecticide spraying and bed nets can retain some usefulness even in resistant areas (230). In addition to this, testing the youngest mosquito age bracket could artificially inflate the apparent level of population resistance as insecticide resistance has been suggested to decrease as mosquitoes grow older (231, 232) - no complete causal link has been definitively identified yet however and the confounding influence of other factors (such as natural selective pressures) may also play a significant role. The general overall increases observed in resistance levels however help to force the case that new insecticides are required for use in the field.

The increasing reports of insecticidal resistance have been matched by a similarly growing number of accounts of antibiotic and drug resistance within various pathogens. For example, artemisinin resistance has been reported for almost a decade (8) and has become substantially more prevalent since then (7, 233), whilst ivermectin resistance has been noted more and more as the compound is distributed to wider populations (234). This pattern of isolated outbreaks of resistant pathogens becoming widespread has been seen previously, with previous frontline antibiotics such as chloroquine becoming useless within the span of a decade (235).

Resistance can emerge as a result of non-completion of antibiotic treatment or due to inadequate doses of the treatment being administered to patients because of the distribution of counterfeit, substandard drugs, with the considerable selective pressures placed on pathogens by mass drug administration being particularly important as well (236, 237).

There are few novel antibiotics sufficiently advanced along the developmental pipeline to have the potential for a significant impact in the next few years (238), with neglected tropical diseases in particular suffering from a lack of research funding in this area (239). Most new treatments proposed in the near future involve repurposing compounds already in use against one disease, such as using the anti-filarial drug ivermectin against malaria (240, 241) – this is a common occurrence within the sphere of helminth infections for example (242). These approaches can yield some successes and help to maximise the effectiveness of the repurposed compounds by targeting multiple diseases, but can also increase the likelihood of resistance developing.

The development of resistance has repeatedly prevented global eradication of several diseases over the past century (243-246), though this has not been helped by the overreliance of control efforts on a limited number of vector control mechanisms (247). The expansion of the available tool set can only help the disease control effort.

2.5. Use of Drosophila melanogaster as a model organism

Drosophila melanogaster is widely used as a model organism because of a number of major advantages associated with the insect; they are relatively cheap to maintain, reproduce quickly and efficiently and their genomes are well mapped and open to manipulation (248). Drosophila genetics have been at the forefront of the overall field of genetics for decades (249, 250), with the recent development of the CRISPR-Cas9 system in Drosophila melanogaster opening even more pathways for exploration (251, 252). The CRISPR-Cas9 system allows for targeted deletions of specific coding sequences of the genomes, enabling efficient investigation of gene knockouts – perhaps most interestingly with regards to this project it also enables the introduction of mutations that lead to insecticidal resistance, the consequences of which can then be explored (253, 254).

Most pertinently for this thesis, *Drosophila melanogaster* has been used not only as a model of hearing loss (255, 256) but also of both insecticidal resistance (257, 258) and mosquitoes – in terms of their genomes (259, 260), parasite development within a host (261) and disease transmission (262). Both types of insect also rely on ChOs for proprioception and auditory function (28, 263) and as such mechanotoxins which are able to successfully target *Drosophila melanogaster* could potentially be included in anti-mosquito programmes.

Whilst there remain significant differences between the two fly types, certain key sections of the *Drosophila melanogaster* life cycle can be considered as simplified versions of the corresponding mosquito behaviour – *Drosophila melanogaster* auditory communication during courtship for example does not require the complex harmonic interaction and exploitation of distortion products that is necessary in mosquitoes species (34) but both courtship processes involve sensory location of sexual partners and the active hearing of wing beats (264, 265).

It seems plausible therefore that if a mechanotoxin was able to significantly impair *Drosophila melanogaster* habits, especially copulatory functions, then the corresponding effect on the relatively more complex and delicate mosquito systems would be even greater. Taken in conjunction with the increased simplicity of arranging *Drosophila* species based experiments, as well as the available depth of literature and knowledge regarding *Drosophila melanogaster* mechanosensation as compared to mosquitoes, this provides a powerful argument in favour of using *Drosophila melanogaster* as a basis for behavioural tests which can then be extended to mosquito species when desired.

2.6. Sensory organs and mechanotransduction

There are two major classes of mechanosensory organ found within *Drosophila melanogaster*, with Type I sense organs being ciliated whilst Type II organs are non-ciliated – Type I organs can be further distinguished into bristles, campaniform sensillae and ChOs (25, 266, 267). There is some overlap between the general roles played by each of the receptor classes – this is shown in figure 1 below (which illustrates in detail the stimuli each sensory organ is receptive to) with regards to proprioception in particular, with several different receptor types being involved in this important sensory activity. There also appears to be some overlap in the mechanotransduction channels themselves (for example the ion channel NompC is present throughout the Type I organs and appears to be linked to the mechanotransduction process in each receptor (268-270)). There still exists a great deal of specificity in organ function however, with this thesis being specifically focussed on the role of ChOs with regards to auditory and proprioceptive mechanotransduction.



Figure 1: Overview of the major sense organs within Drosophila melanogaster, with arrows linking behaviours and stimuli to the appropriate sensory organ (adapted from (25)).

ChOs are the primary substrate sites for graviception and auditory mechanosensation and mechanotransduction in insect species (25, 271). These sensory organs are ciliated stretch receptors present throughout the insect body which transduce stimulated displacements into electrical signals, and are comprised of repeated sub-units known as scolopidia (267). Each scolopidia can contain between one and four sensory neurons - figure 2 contains both a physiological outline of the structure of the organ and the relative locations of some of the most important ion channels found within ChOs (24). NompC and the heterodimeric ion channel formed by Nanchung and Inactive are of particular interest, with both being involved in the mechanotransduction process and the Nan-lav heterodimer being reported as the site of action for pymetrozine (all of which are discussed in sections 2.7, 2.8 and 2.9).



Figure 2. A) Physiological outline of a standard ChO showing key components of the organ (adapted from (272))
B) Relative locations in a ChO of several ion channels considered important for proper ChO function (adapted from (273)).

Whilst the site of transduction within ChOs has been widely agreed to be the distal cilium, the molecular origin has yet to be determined. There are two competing models currently proposed; a NompC model in which the ion channel NompC (=TRPN1) plays the role of transduction mediator whilst the Nan-Iav heterodimeric ion channel complex acts as amplifier and a Nan-Iav model in which the places are essentially reversed and NOMPC now acts as a pre-amplifier (274). In both models NompC is not involved in wind/gravity sensation, which is instead exclusively associated with Nan-Iav (274).

Whilst there is no overwhelming evidence in support of either model, NompC's confirmation as a certified transduction channel subunit directly gated by mechanical stimuli lends that theory extra credence (269), though there remains the possibility that another channel, such as Piezo (275), is in fact the site of auditory mechanotransduction – it could also be the case that the basic, single mechanotransducer channel model is not sufficient complex to capture the full process. Irrespective of the precise transduction machinery however, ChO function is essential for (ChO mediated) proprioception, audition and at least some aspects of temperature sensation (28).

2.7. Auditory systems in *Drosophila melanogaster* and mosquito species

Both mosquito species and *Drosophila melanogaster* have antennal ears that are utilised as auditory receivers, with transduction taking place within the Johnston's organ (JO) of these insect species (see figures 3 and 4 for organ illustrations) (76, 276). The JO is housed in the pedicel or second antennal segment of the antennal receiver and is the largest ChO within the insect body; indeed the male *Ae. aegypti* JO is the largest ChO in the insect world as it is comprised of approximately 7300 scolopidia and 16000 sensory neurons (277). In contrast to this, female *Ae. aegypti* have only half this number (278) and the *Drosophila melanogaster* JO encloses merely 500 neurons (279).

The *Drosophila melanogaster* antennal ear is comprised of three separate parts. These consist of a feathery arista which is directly attached to the third antennal segment (known as A3 or the funiculus), with movement of A3 resulting in the stretching and compression of the two opposing JO neuronal populations housed within the second antennal segment (referred to as A2 or the pedicellus) – figure 3 contains a schematic of how the rotation of A3 causes this stretching of mechanoreceptors. A3 is capable of one dimensional rotation only in response to stimulation (274).

Of the aforementioned 500 *Drosophila melanogaster* JO neurons there are five recognized neuronal subgroups, denoted types A - E (271). These subgroups were classified using UAS-GAL4 lines which expressed GFP in JO neurons with images of the JO from these strains then being used for classification purposes, alongside electrophysiological experiments, calcium imaging and gravitaxis behavioural assays - classes A and B have been thus identified as involved in auditory transduction whilst C and E are essential for wind/ gravity sensation (with type D's exact function currently unclear) (24, 280).

This apparent specification of transduction, with NompC only expressed in type A and B neurons whilst Nan-Iav is expressed across almost all neurons, further complicates the proposed transduction models discussed previously in section 2.6 and has led to the classification of two transducer populations; an auditory 'sensitive' population (comprising classes A and B) and an auditory 'insensitive' population (comprised of the other classes).



Figure 3. A) The antennal ear of Drosophila melanogaster (adapted from (281)) B) The internal distribution of scolopidia, as well as the effect of rotations of A3 on these neuronal populations (adapted from (282)).

As A3 has only a single mechanical degree of freedom with regards to its rotational movement, the *Drosophila melanogaster* antennal system can be modelled as a damped harmonic oscillator – Laser Doppler Vibrometry (LDV) allows for extremely sensitive, unstimulated free fluctuation measurements of the antennal ear to be recorded, from which assessments of the level of auditory function of the individual can be made (283, 284). The antennal frequency tuning of different *Drosophila species* for both males and females is set to match the spectral composition of the species-specific male courtship songs (274, 284).

The transducer machinery within A2 has been described by a gating spring model of transduction, in which elastic elements open ion channel gates in response to stimulus forcing - the major hallmarks of transducer gating and active hearing have been well documented and demonstrated in this species (and are discussed in section 6.1) and include energy injection into the auditory system, self-sustained optoacoustic emissions, frequency selectivity and sensitivity and a compressive non-linearity that results in an inverse relationship between sound intensity and auditory sensitivity (282, 283, 285).

Drosophila melanogaster courtship requires only the male to produce wing beats, which the female is capable of detecting at the fundamental harmonic (279). Mosquito auditory systems by comparison contain a significantly greater level of complexity because of the higher order of intricacy involved in acoustic communication for many mosquito species; whilst *Toxorhynchites brevipalpis* are able to communicate during courtship using first harmonic wing beats (264), species such as *Ae. aegypti* are forced to harmonically converge on higher order wing beat harmonics for courtship to occur (34).

This frequency range is typically above 1000Hz, although neither male nor female appear to show interest in frequencies greater than 400Hz (34). That the two sexes can still communicate is due to the manipulation of auditory distortion products generated by the direct interaction of the wing beats produced during regular flight in both males and females (276, 286).

This type of communication process requires complicated auditory machinery, including a flagellum capable of multi-dimensional movement (with the mosquito flagellum being modelled as an inverted pendulum with two degrees of mechanical freedom) as well as a considerably larger JO than in *Drosophila* species. Significant progress has still been made in comprehending the system however, with investigations into frequency tuning mechanisms and the presence of amplificatory machinery being conducted in addition to reports of mathematical modelling of the system as a force-damped oscillator (287-290).

In both male and female mosquitoes the auditory system is comprised of a flagellum (composed of 13 subunits known as flagellomeres) which is attached to two basal components, the pedicel and the scape (76) – figure 4 contains images of both the flagellum and the pedicel for reference. In contrast to *Drosophila melanogaster* however, there is a significant level of sexual dimorphism in the auditory system of many mosquito species (276); in those species where there is a low level of dimorphism such as *Opifex fuscus* copulation is completely different from that observed otherwise (291, 292).

This sexual dimorphism is evident not only in the dense fibrillae which extend from the male flagellomeres (which are more structurally variable than in females) but also in the size and composition of the JO (293-295). The male auditory system in particular is incredibly sensitive and finely tuned, although the female is also in possession of a far more sensitive system than most other insect species (288). Auditory sensitivity across the different species may vary greatly but in general female auditory systems tend to be more sensitive than males at lower frequency values (below 200Hz) and individual components of the female auditory system have been reported to be tuned to different frequencies (296).



Figure 4. A) The antennal shaft of a male Culex quinquefasciatus mosquito B) Generalised male mosquito antennal system (adapted from (287).

As compared to the five subsets of scolopidia identified in the *Drosophila melanogaster* JO, four different scolopidia types (labelled A - D) have been reported in the equivalent organ in mosquito species (277, 278). The vast majority of these JO scolopidia are type A, which comprise over 90% of the total neuronal population in both males and females and are inserted radially along the inner surface of the pedicel's prongs (76).

Whilst males and female *Aedes aegypti* have approximately equal numbers of both Type B and C scolopidia as well as similar numbers of prongs, only the male JO contains a Type D scolopidium – prongs found in the male *Ae. aegypti* pedicel are also far longer and thinner than those of the female (76). In contrast to this however the pedicel of female *Anopheles stephensi* contain a single Type D scolopidium, though females of this species have only 47 prongs as compared to 78 for males (297, 298).

Antennal fibrillae extension does not occur in male *Ae. aegypti* and *Cx. quinquefasciatus*, with fibrillae remaining constantly erect in both of these species (299-301). Male *An. gambiae* do demonstrate fibrillae erection however during time periods associated with copulatory activity (302, 303). The mechanism by which this erection is modulated is not fully understood but is thought to be related to the release of specific flagellar neurotransmitters which then result in secondary messenger systems being released (304).

Such complex auditory machinery is essential because, unlike in *Drosophila* species where both male and female antennal systems are broadly tuned to the same stimulus, male and female mosquitoes have separate flagellar and JO frequency tuning (which is discussed in more detail in sections 2.10.2 and 7) (34). Male auditory function has tended to be the primary focus of investigations because of the significantly larger JO present in males. This emphasis has for example allowed for the detection and confirmation of nonlinearities in their auditory systems (305), the source of which is likely to be active transducer modules within the auditory neurons (276).

This large JO, in addition to dense antennal fibrillae to increase flagellar surface area, is necessary for males because they are required to locate females for courtship to be possible (305). Male mosquitoes not only seem to place significantly greater energy resources into locating females but also appear to exert more effort in order to induce copulation, with modulation of the fundamental WBF of male *Ae. aegypti* for example being greater than for females of that species (306).

2.8. Auditory mechanotransduction and efferent feedback in mosquito species

Auditory mechanotransduction per se has not yet been directly studied for mosquitoes as it has for *Drosophila* species and its role in mosquito auditory function is not fully clear. This is partly because of evidence concerning spontaneous oscillations, a phenomenon unique to male mosquitoes in natural situations which results in relatively large flagellar oscillations (on the order of 1 μ m) in the absence of stimulation.

Spontaneous oscillations have only been elicited in female mosquitoes (as well as in *Drosophila melanogaster*) via DMSO injection (283, 289) – such injections also result in oscillations in male mosquitoes. This is likely the result of decreases in the stiffness of the system following injection, possibly as the result of DMSO's potential effect on membrane stiffness (307). Within the gating-spring model of mechanotransduction (as previously described for *Drosophila melanogaster* for example (285)) this could imply that the steady state stiffness has decreased and as a result the gating spring stiffness has been reduced to a negative value (i.e. the system has become so compliant that active forcing is no longer necessary for self-sustained oscillations to occur).

In male mosquitoes these spontaneous oscillations have been hypothesised as arising in the absence of compound injection as a result of dynein-tubulin motors present in ciliated sensillae, a process that should thus not involve mechanotransduction (308). The reasoning behind this exclusion is based on the results of colchicine injection, an alkaloid which should disrupt microtubules in dendritic sensillae, into the JO of a male mosquito whose flagellum was spontaneously oscillating. The compound caused the cessation of the oscillations whilst theoretically leaving mechanotransduction intact (though the significant reduction in JO nerve response following injection suggests this assumption may not be reasonable).

This does not necessarily prove the independence of the two systems however, as that would require spontaneous oscillations to occur in a system that no longer retained mechanotransducer functionality, whereas here colchicine may act to prevent oscillations whilst leaving the basic mechanotransduction apparatus intact. In either case, the role of mechanotransduction in mosquito auditory systems has not yet been completely defined. The existence and possible importance of efferent modulation of auditory function within several insect species (including mosquitoes) is also presently undecided. Whilst the auditory systems of *Drosophila* species display many of the hallmarks of active hearing and share remarkable similarities with vertebrate hair cells and ears, previous experiments in *Drosophila melanogaster* found no evidence of efferent feedback within the auditory system.

This was determined by injection with tetrodotoxin (TTX), a powerful toxin that targets voltage gated sodium channels and prevents stimulus conduction along neurons effected (309). TTX should therefore be able to sever any efferent feedback loops present within the *Drosophila melanogaster* auditory system. Whilst TTX injections resulted in an almost total loss of action potential production, there was no observable effect on the mechanics of the auditory system (laboratory data, unpublished).

Another test for the existence of efferent feedback in *Drosophila melanogaster* utilised the conditional expression of tetanus neurotoxin (TeNT) in a pan-neuronal manner (310). TeNT acts at neuromuscular junctions by blocking presynaptic membranes, thus preventing the release of neurotransmitters in this region (311, 312). Obstructing synaptic transmission across all neurons did not prevent mechanical feedback gain in *Drosophila melanogaster* and as such this compound, like TTX, was judged to have no effect on the auditory system. This result was interpreted as meaning that no efferent control system exists within the *Drosophila melanogaster* auditory system, a conclusion which is in agreement with previously published literature (313). It is important to stress however that thus far only evidence of the absence of an efferent system has been presented rather than proof that such a system does not exist.

Efferent modulation of audition had long been suspected for mosquitoes without any real confirmation of such systems being reported (290). Male *Cx. quinquefasciatus* mosquitoes however have recently been found to contain such an efferent system within their auditory framework, possibly to allow for modulation of sensory neurons (314). The presence of efferent systems within male mosquitoes could indicate a greater complexity necessary for proper auditory function within mosquito species. Males from other mosquito species seem highly likely to also contain efferent auditory networks as a result of this and could help to further explain some of the systems of auditory control within male mosquitoes.

2.9. Pymetrozine

Pymetrozine is a slow acting feeding inhibitor currently distributed as a crop spraying insecticide for use against use against pests such as aphids which feed by inserting their stylus into crops and imbibing phloem sap – the compound is sprayed onto crops that require protection from such pests, which then ingest pymetrozine if they attempt to feed (26). Pymetrozine exposure does not directly result in insect mortality as mechanosensory function is not a directly essential requirement for insect survival. It does however cause starvation in insects that can no longer feed naturally and are not provided with a replacement source of nutrition; it is also possible that in the field a lack of proprioceptive and auditory capabilities leads to a decrease in survival probability as predators become more difficult to avoid (23).

Pymetrozine has been confirmed as safe for use in its current application method and has shown high levels of efficacy at concentrations as low as 10⁻⁷ M during injection tests (315). These concentration estimates are misleading however as the usual mechanism of exposure is ingestion, which therefore requires greater concentrations to be effective especially when applying the toxin in the field.

Previous studies using locusts identified ChOs are the target of pymetrozine, though the mechanism of action and ion channel target site was not reported at that time (22). Since then the major target of pymetrozine in *Drosophila melanogaster* has been confirmed as the Nan-Iav heterodimeric ion channel complex, which is involved in the conduction of calcium (316, 317). All other channels tested requiring at least a hundred fold increase in concentration for any effect to be seen (27).

Further than this, the compound only affects these transient receptor ion channels when they form a single heterodimer and otherwise leaves them undisturbed. As it is only in ChOs that this heterodimer is found within insects, pymetrozine can be considered to be a ChO specific mechanotoxin. Pymetrozine acts as an agonist to this channel complex and compound exposure results in ablation of ChO mechanosensory function as mediated by this complex by promoting an increased cellular calcium influx that functionally deteriorates ChO neurons (27).

TRPV homologs of Nanchung and Inactive have been identified in *Ae. aegypti* male ChOs (318). This provides evidence for pymetrozine being able to affect mosquitoes in a similar manner as for other species – this is also supported by reports of strong effects of pymetrozine on a variety of species (26, 319).

There are no published reports of the impact of pymetrozine exposure on mosquito species and as such nothing is currently reported as to how long the effect of pymetrozine exposure would last for mosquitoes or whether such exposure would affect feeding from sugar sources, searching for hosts or even fecundity.

These important topics have begun to be investigated in other insect species - for example there are reports of a decrease in aphid fertility after pymetrozine consumption and pymetrozine exposed *Drosophila melanogaster* are able to successfully feed from sugar sources in laboratory conditions (although the *Drosophila melanogaster* mechanism of feeding is distinctly different from that of insects that require stylus insertion to feed) (27, 320). Pymetrozine injection experiments in *Drosophila melanogaster* have demonstrated a reduction in aristal sensitivity following exposure and an increase in overall aristal effective stiffness (27). This is also the case if the insect has ingested the compound either as an adult or during the larval stage; this reduction insensitivity has been identified as well if *Drosophila melanogaster* are exposed to pymetrozine solely during the larval stage and are not exposed as adults (unpublished data, Jörg Albert lab).

As would unfortunately be expected in a widely used insecticide, resistance has been documented in several insect species, though the basis of this resistance has not yet been reported (321-323). This resistance has developed over prolonged exposure to the toxin however and could potentially be mitigated by using combinations of insecticides together.

Pymetrozine exposure could have a negative impact in various ways on mosquito species because any behaviour mediated by ChOs could be affected by compound exposure. For example, pymetrozine could alter blood feeding behaviour, either directly by worsening flight ability or indirectly by altering circadian rhythm regulation. It could also decrease copulation success by damaging auditory capabilities (as well as the previously mentioned potential impact on flight) or it could simply decrease mosquito fitness to the extent that the mosquito is outcompeted by a competitor or consumed by a predator.

2.10. Key components of insect life cycles that pymetrozine could target

2.10.1. Flight

Flight initiation and regulation in *Drosophila melanogaster* is the product of multiple sensory feedback mechanisms including mechanosensation (324). Mechanosensory neurons within the JO itself play a key role in flight maintenance by controlling wing motor reflexes (325) and also by maintaining constant flight speeds (326). A specific sub-class of neurons (denoted as class D) has been proposed as the site of this maintenance activity and has been identified as wind sensitive (327), supported by the significant antennal deflections required to produce activation of these neurons (274).

Aristaless *Drosophila melanogaster* demonstrate significant increases in groundspeed variability compared to *Drosophila melanogaster* which have functional aristae (29), which is most likely due to sensorimotor delays associated with flight maintenance solely mediated via visual feedback (328). The usual regulatory feedback loops become increasingly unstable as instabilities gradually build up throughout the entire system, eventually preventing proper flight control. Proprioceptive feedback loops in particular have been described as essential to regular flight pattern establishment (329) and mutants for mechanosensory genes expressed in the JO are demonstrably worse at flight initiation than wildtype controls (272).

Whilst there is a high level of variability in the level of flight ability necessary for survival (with female urban *Ae. aegypti* being reported to require much shorter flights between blood meal sources and oviposition sites than rural *An. gambiae* females, which are forced to cover longer distances (330, 331)), successful flight initiation and maintenance is in general essential across all species and both sexes, particularly with regards to predator avoidance and the localization of sources of glucose (56, 76).

The processes which underlie successful flight are regulated by a variety of mechanisms, but of particular interest to this project is the reported role that the antennal systems of mosquito species seem to perform in flight maintenance – graviception has also been suggested to be localised to the JO in mosquito species and air speed modulation is presumed to be a result of drag on the antenna (76, 332, 333).

Even partial removal of the flagellum results in significant if slight changes to wing beat amplitude, though interestingly removing equal amounts of both flagella minimises these changes (76). This suggests that the halteres (modified hindwings involved with flight control and mainly populated with campaniform sensillae (334, 335)) are more central to flight maintenance for mosquitoes than the antennal system (332).

The importance of the maintenance of proper flight mechanisms in mosquitoes with regards to not only copulation but also for females to find suitable blood meals should not be underestimated and bears repeating (336, 337). Assuming that mosquito flight regulation is mediated in a similar manner to *Drosophila melanogaster*, it seems possible that ablation of ChO mechanosensory function using pymetrozine would lead to significant decreases in overall flight capability.

2.10.2. Courtship

Many insect species display intricate courtship rituals designed to communicate mating suitability and availability (amongst other factors), with courtship events in both *Drosophila* and mosquito species tending to follow specific, identifiable processes (34, 338).

Drosophila melanogaster courtship rituals follow a well-recognised pattern (339), depicted in Figure 5 below; the male first uses sensory stimuli to locate a suitable female before using a foreleg to tap the female's abdomen. Provided the female has remained receptive up until this point, the male will extend a wing and begin to produce two distinct types of song, denoted as either pulse or sine songs. Pulse songs are comprised of rhythmic downward strokes of the wing at an interval and frequency specific to different *Drosophila* species (and which is closely correlated with the frequency tuning of the antennal system (284)) whilst sine songs consist of the constant production of sine waves (with figure 2 containing examples of each of these song types). The dynamic frequency range of both types of song is relatively small, which taken in conjunction with the one-dimensionality of the *Drosophila melanogaster* sound receiver requires the male to get as close to the female as possible during song production (340).

If the female judges the song to be of acceptable quality she will allow the male to proceed with genital licking and finally copulation. The entire process is dependent on the female's willingness to proceed, without which the male will be unable to successfully copulate (341). After impregnation, the female generally remains unreceptive to further courtship attempts for at least three days (342).

Whilst pulse songs have been identified as crucial to successful copulation attempts due to its' proposed role in species identification, sine songs have been considered as inessential (343) and are thought to be mainly present to enhance female receptivity to mating (344). The distribution of song type seems to be non-random and the result of the males' judgement of the courtship stage as well as the state of the female (345).

Two key components in determining the quality of a pulse song are the time interval between separate pulses within a pulse train (commonly referred to as the interpulse interval (IPI)) and the frequency of a pulse itself (known as the intrapulse frequency (IPF)) (346) - a diagrammatic representation of both the IPI and the IPF is included in Figure 5 part B. For *Drosophila melanogaster* typical IPI values of about 35ms have been reported, alongside IPF estimates of between 150 and 250Hz and sine song frequency values of approximately 150Hz (281, 284).





B) Section of a courtship song produced by a Drosophila melanogaster male with important segments highlighted.

The roles that auditory feedback and proprioception play with regard to song production should theoretically be significant – ChOs are distributed throughout the bodies of *Drosophila* species, including their wings, and could therefore be involved with both the males' auditory perception of the songs he is creating and with the proprioceptive sensation that his wings are moving. These feedback loops could then allow the male to constantly control and optimise the intrinsic characteristics of his wing beats and thus improve his chances of reproductive success (340, 347).

In spite of this the majority of reports have not found compelling evidence of any major auditory requirements for the production of quality songs. *Drosophila melanogaster* male mutants without aristae or antennae (*aristaless* and *antennaless*) were able to create songs statistically indistinguishable from control flies (329, 348, 349) whilst physical removal of the entire antenna did not prevent copulation from occurring (350). On the other hand, other *atonal* mutants have been found to produce significantly different songs from wildtype males –this could however be the result of developmental pleiotropic effects that were not properly controlled for within the experimental design (340).

Previous work within the Albert lab found that flies with ablated antennal function had a significantly greater IPF than control flies, though the IPI and sine song frequency remained constant throughout (unpublished data, Jörg Albert lab). This ablation of function was brought about by gluing either one or both of the antennae to the head, with both cases producing similar results. As such it remains unclear whether auditory and proprioceptive feedback mechanisms (and therefore ChO mechanosensory function) are important for male *Drosophila melanogaster* during courtship attempts.

None of the previous attempts at investigating this topic have used a pharmacological method of eliminating antennal function. Given this, and the current lack of clarity on the topic, pymetrozine's possible effect on courtship behaviour overall (and song production specifically) is worthy of assessment.

In contrast to *Drosophila melanogaster*, female mosquitoes can play a much greater role in mosquito courtship behaviour – this is more similar in some ways to the duetting seen in *Drosophila virilis*, which has been found to rely on auditory sensation to succeed (351). Harmonic convergence of male and female flight tones on shared higher order harmonics (typically the third female and second male harmonic) has been reported for *Ae. aegypti, An. gambiae* and *Cx. quinquefasciatus*, with an example of such convergence shown in figure 6 part A (352-354).

This frequency convergence has been suggested as conveying information about the genetic quality (i.e. greatest fitness benefits associated with mating) of the male to the female in a similar manner as the pulse song in *Drosophila* species (355, 356). Whilst it seems highly likely that harmonic flight tone convergence is an active process as both male and female mosquitoes change their flight tones in response to one another (with deafened mosquitoes being reported as unable to harmonically converge with mosquitoes of the opposite sex (352)), this frequency matching may be less important than the auditory distortion products created by this convergence (357, 358).

For example, whilst the mechanical tuning of the male flagellum is theorised to be focussed on the female fundamental wing beat frequency (WBF), the female flagellum is tuned to the cubic distortion product of both her own and her prospective partner's wing beats i.e. the difference of the male WBF and twice the female WBF (as highlighted in figure 6 part B) (358).

The JO of male mosquitoes on the other hand is tuned to the difference tone (the difference between the male and female WBFs); the tuning of the female JO is not presently completely understood and could be more related to identifying predators or blood meal sources than courtship, with female *Culex territans mosquitoes* for example displaying a high degree of phonotaxis towards amphibian auditory cues (358-360). If the female JO is indeed tuned to specific frequencies for this reason, it provides an interesting point of comparison for the relative importance each sex places on copulation in contrast to other activities.



Figure 6. A) Harmonic convergence of individual male and female mosquitoes as demonstrated by changes in wing beat frequency over time; red and blue lines show female and male wing beat frequencies respectively for different harmonics, with the bottom image showing a better resolution of changes to the third female and second male harmonic (adapted from (357)).

B) Distortion product formation from the interaction of two pure tone stimuli, f_1 and f_2 , with the potentially relevant 2 f_1 - f_2 distortion product highlighted (taken from (276)).

The male auditory system may require a significant level of sensitivity because of the behavioural mating patterns of mosquitoes – swarms of males form normally around dusk and these males are required to identify and locate female wing beats from the overwhelming noise surrounding them (which includes the sound generated by their own wing beat) (361, 362). The WBFs of different mosquito genera are identifiable and measurable using velocity-sensitive pressure gradient microphones to record flight attempts by individual mosquitoes which can then be later analysed (353, 363).

In the field, copulatory events for *Aedes*, *Anopheles* and *Culex* species tend to occur mid-flight, though copulation for *Ae. aegypti* can occur occasionally on the ground (56, 76). Males have a short period of time in which to transfer his sperm or adjust his position once the mosquitoes are attached to each other so that both partners can sustain constant flight (303, 364). Within caged laboratory conditions, mating during flight is still possible but the later stages of copulation (particularly insemination) are more commonly observed for couples attached to the sides of the cage (76, 365).

Female mosquitoes generally mate once per lifetime but there is evidence across multiple species that a small yet significant proportion of the female population that undergoes multiple insemination events (366, 367). This polyandry appears to depend on the post mating interval between copulation attempts by separate males and could have a significant impact on control programmes that utilise sterilised or modified male mosquitoes in order to reduce mosquito population sizes (by decreasing the number of females that can produce viable offspring), as females would then become more likely to mate with wildtype, non-sterilised males (368).

2.10.3. Circadian rhythm regulation

The maintenance of an internal circadian clock bestows numerous benefits on any organism that develops such a system, with circadian rhythms of activity having been reported in a wide variety of different species (369, 370). Understanding the processes behind the maintenance of these rhythms, as well as the effect such rhythms can have on behaviour, is of particular interest for anthropophilic mosquito species because of the strong rhythmicity of female biting habits and male copulatory events (76, 371).

Circadian rhythms in *Drosophila melanogaster* have been well documented using a variety of systems, with robust entrainment being shown using a variety of different stimuli (or Zeitgeber) such as light, temperature and vibration (372). Typically such circadian activity is examined using monitoring systems which quantify the time spent in different activity phases during and after periods of entrainment to stimulation, either using simple activity counts or more complex video analyses (373, 374).

In mosquito species, circadian activity can be quantified using either laboratory experiments or biting catches in the field (in which female mosquitoes are captured during biting attempts and a count is kept of which times captures most commonly occur). Laboratory experiments were previously conducted by keeping individual mosquitoes in glass containers alongside auditory detection equipment so that flight activity could be detected. These mosquitoes were then scored for each minute during each hour spent flying (and thus received a score of between a minimum of 0 and a maximum of 60) which could be aggregated across groups of interest (76). Recently this system has in general been superseded by techniques more similar to those seen for *Drosophila melanogaster* (375).

An. gambiae and Cx. quinquefasciatus mosquitoes both display crepuscular peaks of activity whilst Ae. aegypti is considered to be a diurnal species (76). Mosquito circadian rhythm activity has been described in detail previously (376, 377), most notably when investigating changes to the circadian patterns of females once they have successfully obtained a blood meal (378). This has been reported to be the result of a downregulation of clock genes once the female has procured a blood meal (375) - indeed there is a strong circadian component with regards to gene expression throughout the mosquito body, particularly in the antenna (379). As noted previously, the antennal fibrillae of male An. gambiae become erect during specific periods of the day that tend to be correlated with an increased likelihood of copulatory events (303).

Male and female mosquito activity patterns are clearly distinct, with males tending to show peaks in activity earlier in the day than females – both are more active during nocturnal periods however (380, 381).

Interestingly *Plasmodium* strains of the malaria parasite also exhibit strong circadian activity regulatory mechanisms in order to stay in synchrony with the circadian clock of their host (382, 383). If they become de-synchronised from this rhythm they suffer dramatic decreases in fitness as compared to those that are able to maintain constant time-matching.

Light is generally considered to be the most powerful Zeitgeber for entrainment of the circadian clock, though as mentioned above behavioural events also play a role (384). Entrainment to temperature cycles has also been shown to be possible in mosquito species (385). In *Drosophila melanogaster* ChOs have been proven to be involved in circadian rhythm regulation in terms of both temperature and vibrational entrainment and as such it seems plausible that they could also have a role in the same regulation in mosquito species (33, 386). Given the target of pymetrozine, exposure to the compound could lead to a change in circadian rhythm regulation within the affected mosquito which could then result in sub-optimal feeding or mating behaviour.

2.11. Mathematical modelling

Given the complexity involved in modern disease control programmes, it is essential that as many eventualities as possible are accounted for and the tools involved in these programmes are utilised as effectively as possible. Extensive mathematical modelling of many aspects of mosquito borne diseases has thus been conducted – these range in complexity from the original compartmental Ross model of malaria transmission (and its later development into the Ross-MacDonald model) to Bayesian models of *Ae. aegypti* abundance (188, 387-389).

Given pymetrozine's indirectly fatal nature and the necessity of compound ingestion by an insect for effective exposure rather than cuticular uptake, modelling strategies previously used for fungicides could be appropriate starting points for model development for this toxin (168). Feeding cycle models in general could also be useful as reference points as they provide sufficient complexity to capture the unusual exposure method involved with regards to pymetrozine as well as allowing for investigation into the possible long-term impacts of insecticide spraying (390, 391).

Even these relatively progressive models do not properly account for male actions within the overall population however. Indeed male mosquitoes in general remain unaccounted for and, in the case of pymetrozine in particular, could be significant (392). Spraying of insecticides onto vegetation has not been investigated in depth either, although some modelling with regards to maximising the effectiveness of attractive toxic sugar bait solutions has been attempted (393).

Models of insect mechanosensation have also been reported, with the aforementioned gating spring model of auditory mechanotransduction in *Drosophila* species forming the basis for future investigations described in this thesis into the auditory system of both *Drosophila melanogaster* and several mosquito species (as discussed in sections 6 and 7) (285, 394).

Overall, estimating pymetrozine's potential impact in the field would require a mathematical model that could account for both mosquito sexes whilst including all the potential targets for spraying with the compound. This would necessitate some type of feeding cycle model which allowed for multiple generations to be followed.
3. Project rationale, objectives and hypotheses

3.1. Project rationale

From the literature reports presented in section 2, it is clear that insecticides with different mechanisms of action to the currently existing options are increasingly required for mosquito control programmes. Pymetrozine is a promising candidate for usage based off the compounds' current deployment as a crop spray – its effectiveness against mosquitoes however remains to be tested. This project therefore attempts to ascertain the efficacy of pymetrozine (and by extension the potential of other mechanotoxins) at disrupting copulatory, flight and auditory systems in both *Drosophila* and mosquito species with a view to the inclusion of pymetrozine in future mosquito control measures.

In addition to these experiments, using pymetrozine as a powerful pharmacological tool to ablate mechanosensation will allow for the exploration of insect ChO-dependent mechanosensory systems. Assessing the role of mechanosensation in courtship not only provides concrete support or dismissal of pymetrozine as an pesticide but also enables investigation of what is and what is not necessary for insects to function competitively.

Fundamental aspects of auditory function in mosquitoes are still - to some extent open for interpretation; pymetrozine can contribute to a greater understanding of the overall system by allowing for comparisons between different states. Whilst the final conclusion of this work will be on the suitability of pymetrozine for mosquito control initiatives, the results demonstrated to support this recommendation should also provide the basis for a deeper comprehension of insect mechanosensation.

Three different mosquito species have been chosen for experimentation – *Ae. aegypti, An. gambiae* and *Cx. quinquefasciatus.* All three of these mosquito species are major vectors of disease worldwide and as such are targets of disease control programmes. Each species has also been reported to rely on harmonic convergence for copulation to occur. Significant variation in feeding, oviposition and copulatory behaviour has been observed between these mosquito species – by examining all three therefore it is hoped that the study can be as broad and inclusive as possible.

Careful comparisons of auditory systems can also be made across species, as well as relative assessments of the extent of sexual dimorphism observed; by using three different strains of *An. gambiae* (Kisumu, Ngusso and Tiassale) for example it should be possible to study more closely the auditory systems present within this species.

3.2. Project objectives

The major objectives of this thesis are:

1) To conduct behavioural investigations into the effect of pymetrozine on several aspects of the mechanosensory mediated behaviour of *Drosophila melanogaster*. This includes the potential impact of pymetrozine on fertility, fecundity, life expectancy, flight ability (both with regards to initiation and maintenance), courtship song production and competitive fitness, and is discussed in section 4.

2) To conduct behavioural investigations into the effect of pymetrozine on several aspects of the mechanosensory mediated behaviour of mosquito species; this includes the potential impact of pymetrozine on mosquito larvae as well as fertility and fecundity in adult male and female mosquitoes; this is discussed in section 5.

3) To conduct electrophysiological experiments in conjunction with LDV measurements in order to investigate the mechanical signatures of transducer gating in mosquito species using similar principles as previous reports and pre-existing models developed in *Drosophila melanogaster*, this is discussed in section 7.

4) To conduct electrophysiological experiments to investigate the effect of pymetrozine exposure on these mechanical signatures in both *Drosophila melanogaster* and mosquito species; this includes *Drosophila melanogaster* and mosquito lines that have been reported as expressing insecticidal resistance and is discussed in sections 6 and 8 respectively.

5) To calculate energy gain estimates for male and female mosquitoes following exposure to compounds designed to sever existing efferent feedback loops, as well as following exposure to pymetrozine, with comparisons between the different injection states enabling investigation into the amplification machinery present in both sexes; this is discussed in section 7.

6) To measure the WBF of male and female mosquitoes for all three species, which will then be used to calculate distortion products for comparison with the results of pure tone stimulation experiments designed to identify the peak mechanical sensitivity and nerve response frequencies for all mosquito groups tested; this is discussed in section 7.

7) To calculate displacement gain estimates (using either pure tone or white noise stimulation) for male and female mosquitoes from all species tested in order to investigate sex-specific differences in both the extent of gain observed and the best frequency of mechanical responses; this is discussed in section 7.

3.3. Hypotheses

The major hypotheses of this thesis are as follows:

1) Mechanical signatures of mechanotransducer gating have been identified for *Drosophila melanogaster* and gating spring models of transduction have been published (285). The first hypothesis is thus that the same basic models as used for *Drosophila* species can be used to investigate similar signatures in mosquito species and can allow for quantification of mechanotransducer gating in the JO of mosquitoes; this includes amplification, adaptation and auditory non-linearities.

2) Pymetrozine has been shown to target the Nan-lav heterodimeric ion channel complex in *Drosophila melanogaster* and acts as an irreversible agonist (27). Given the reported conservation of this channel in mosquito species (318), the second hypothesis is that pymetrozine should have a comparable effect on mosquitoes as it does on *Drosophila melanogaster* in terms of the ablation of ChO mechanosensory function; this includes the effect of pymetrozine on mechanical signatures of mechanotransducer gating.

3) Given the role that the Nan-Iav heterodimeric ion channel complex plays in ChO mechanotransduction (274), the third hypothesis is that exposure to pymetrozine should lead to a disruption in those behaviours (in both *Drosophila melanogaster* and mosquitoes) which are mediated by mechanosensation; this includes flight ability and courtship ritual performance.

4) Efferent synapses have been identified in male *Cx. quinquefasciatus* mosquitoes (314). The fourth hypothesis is that if amplification is transducer based in mosquito species then exposure to pymetrozine should abolish this amplification. In addition to this, if efferent synapses are involved in regulating this amplification then severing these synapses pharmacologically should lead to a change in amplification i.e. synapses act to downregulate/upregulate amplification then severing synaptic function should increase/decrease amplification.

5) Amplification and flagellar frequency tuning are closely correlated with the sexually dimorphic communication behaviour of male and female mosquitoes, a major component of which is harmonic convergence (34, 276). The fifth hypothesis is that the specific frequency tuning of both the flagellum and the maximum nerve response is directly correlated with the tuning of the WBF and the resultant distortion products created by interactions between conspecific male and female mosquitoes.

In total therefore this thesis aims to not only make suggestions as to the efficacy and potential of pymetrozine as an insecticide but by analysing the data collected using pymetrozine as a pharmacological tool also offers future proposals to enhance the general understanding of ChO-mediated mechanosensation in mosquito species.

4. Drosophila melanogaster behavioural assays

4.1. Introduction

To date there are no published reports of the effect of pymetrozine (or any other mechanotoxin) on any mosquito species. The little that has been reported on pymetrozine's efficacy and mechanism of action has utilised other insect species, such as *Drosophila melanogaster*, and has so far not fully explored the potential impact of ablating ChO mechanosensory function on the lifecycle of an insect species (22, 23, 27).

Thus by testing various experimental procedures using *Drosophila melanogaster* as a model organism, it seems possible to not only produce data that can be analysed to form the basis of future, more complex experiments in mosquito species but also to investigate the importance of mechanosensation with regards to essential activities in this species; these include flight for example, or courtship behaviour (as detailed in section 2.10).

As discussed in section 2.5, *Drosophila melanogaster* serve as a highly effective model organism for many reasons – among the most important being not only the genetic mutability of the species but also the enormous array of tools and assays which have been reported and published (395-397). This diverse literature enables many specific aspects of various important parts of the *Drosophila melanogaster* lifecycle to be investigated. For example, experiments investigating the fertility and fecundity of various mutant and control flies are possible because of the short generation time of this species (398, 399). Conceptually similar assays can thus be used to investigate whether pymetrozine exposure causes reductions in fertility and/ or fecundity (as had been reported for aphids) (320).

In a similar manner to this flight assays, such as those devised to test either flight initiation or flight maintenance, can also be utilised (400, 401). Whilst flight initiation attempts can be relatively straightforward to quantify, with basic comparisons being drawn between individuals who attempted flights when dropped into containers and those who did not, flight maintenance can be explored in a variety of ways: the multi-camera tracking systems referred to as Flycubes for example, which have been reported to accurately track the flight patterns of *Drosophila melanogaster* in response to visual stimulation, provide an excellent paradigm for such investigations because of the reproducibility of stimulation patterns and the accuracy of the recording equipment (401). Flycubes consist of a free-flight environment containing multiple LCD screens which can present synchronised visual stimuli, alongside high resolution cameras capable of tracking insect flight responses to stimulation (402).

Flycubes take advantage of *Drosophila melanogaster*'s optomotor response to widefield visual motion by presenting rotating stimuli which cause individual *Drosophila* to rotate in-line with the stimulus (402, 403). By presenting stimuli with different rotational velocities for example, Flycubes are thus able to provide information on the number of flights attempted, how long each flight lasted, the flight pattern produced and the flight response following stimulation (401). This data can then be compared between different *Drosophila* groups to look for significant differences in flight behaviour and maintenance.

Circadian rhythm regulation fulfils a vital role in the control and regulation of numerous behaviours for *Drosophila* and mosquito species (76, 372, 404). *Drosophila melanogaster* has served as an ideal model for circadian rhythm studies in many previous publications and has also previously been reported to require ChO function for proper entrainment to temperature stimuli (369, 386). Given that mosquitoes also entrain to temperature cycles and pymetrozine allows the precise ablation of ChO function, investigating the circadian activity patterns of *Drosophila melanogaster* that have been exposed to pymetrozine (in comparison to controls) will allow for a greater understanding of the role of ChOs in circadian rhythm regulation as well as the formation of hypotheses regarding the potential effect of pymetrozine on the circadian rhythms of mosquitoes (385).

Section 2.10.2 described the courtship routine of *Drosophila melanogaster*, including the production of pulse and sine songs from the male fly. Although the previous literature does not support the idea that auditory feedback is necessary for the production of songs of sufficient quality, the ability of pymetrozine to pharmacologically ablate ChO function (both with regards to audition and proprioception) may result in significant differences being observed (329, 348). The results of such experiments could thus deepen the understanding of the potential role that feedback mechanisms fill with regards to song production.

Finally, the previous experiments described above all investigated the effect of pymetrozine on individuals in non-competitive environments; in the proposed fecundity assay for example male *Drosophila melanogaster* are not provided with competition for females. Providing competition in such environments can enable greater insights into the true extent of pymetrozine's impact which may otherwise remain hidden because of the optimisation of the environment i.e. ample food and no competitors for courtship.

Numerous competitive fitness assays have been described for *Drosophila melanogaster*, which frequently involve two males competing with each other to mate with a single female (405, 406). Thus by exposing one male to pymetrozine and allowing the other to remain unexposed it should be possible to analyse the potential impact that the compound could have on fitness due to the ablation of ChO mechanosensory function.

4.2. Materials and methods

4.2.1. Drosophila melanogaster rearing

All *Drosophila melanogaster* stocks, unless otherwise stated, were reared in incubators at 25°C and 60% relative humidity and were provided with food prepared according to the Chippendale recipe. The incubators used a 12 hr: 12hr Light: Dark (LD) cycle which the lines were kept at during all developmental stages.

All wildtype flies used in section 4, with the exceptions of experiments described in sections 4.2.8 and 4.2.11, were the Jörg Albert labs' own wildtype CantonS strain and had no known resistance to any insecticide. Control flies used in experiments described in section 4.2.8 were from the lab of Dr Andrew Straw (Institute of Molecular Pathology, Vienna) and were also from the CantonS line – these flies had no reported insecticidal resistance to any compound. The LH_M–UCL and LH_B–UCL fly lines used in the experiment described in section 4.2.11 were made available by the Dr Max Reuter lab (University College London, London) and had no reported insecticidal resistance (407).

4.2.2. Compound preparation

Pymetrozine: No significant differences had been seen in previous experiments between using DMSO or using milli-Q water as a solvent for pymetrozine, provided the solution was vortexed thoroughly for several minutes (laboratory data, unpublished). Therefore milli-Q water was utilised in all relevant experiments to avoid any possible issues arising in using DMSO as a control substance. A stock solution of concentration 1000 parts per million (ppm) was created by mixing 50mg of pymetrozine with 50ml of milli-Q water. This stock was then shaken and vortexed until the compound had fully dissolved and was diluted to necessary concentrations as required.

4.2.3. Compound exposure – ingestion

Male and female *Drosophila melanogaster* were deprived of food for 24 hours in empty plastic vials containing only damp cotton for humidity control before being transferred to new vials containing 2ml of a mixture composed of 7.5% glucose and 1.5% agar. Either 20 µl of an appropriate concentration of pymetrozine or 20 µl of milli-Q water (for control purposes) had been directly spread on the surface of the food in each individual vial. The flies were kept on the pymetrozine exposed/ unexposed food for a varying length of time depending on the experiment and were then returned to regular stock food. Only 10 *Drosophila melanogaster* were kept in each vial, including control vials, to prevent increases in insect mortality that had been seen previously for pymetrozine exposed *Drosophila melanogaster* if the density of flies per vial was too high.

The length of time during which *Drosophila melanogaster* were allowed to feed from doped food was chosen to minimise the possibility of multiple feedings whilst maximising the number of flies that would feed at all. Pymetrozine exposed *Drosophila melanogaster* have a clear phenotype because of their lack of mechanosensory capabilities – unlike control flies, exposed *Drosophila melanogaster* are slow to move about and reluctant to climb up the walls of a vial (27). The conclusive testing for pymetrozine exposure, which was performed wherever possible, was done using LDV (as described in section 6); this allowed for an unambiguous identification of pymetrozine exposed flies. As this testing was destructive however it could only be done upon the completion of an experiment.

4.2.4. Drosophila melanogaster fertility assay

Male and female CantonS virgin *Drosophila melanogaster* were selected upon eclosion using CO_2 sedation and transferred in groups of 10 to vials containing standard fly medium for 24 hours, with male and female flies being kept separately. After this, half of the females were transferred to food doped with pymetrozine at a concentration of1000ppm whilst the remaining half were placed on control food (i.e. food that not been doped with pymetrozine). All males were transferred to control food (with both sexes kept separate).

After two days on this food 40 pairs (each consisting of one male and one female) were placed in individual courtship containers via aspiration. Twenty of these pairs contained a female that had not been exposed to pymetrozine with the other twenty females having been exposed to the compound (no male had been exposed to pymetrozine). Each pair was given 24 hours in which to engage in courtship before the males were isolated and the females transferred to vials containing fresh food. After a further 24 hours on this food, the females were transferred again to another set of vials containing fresh food, where they remained for a final 24 hours before being removed for confirmation of pymetrozine exposure/non-exposure.

All egg counts were done immediately after removing the females from each vial and were then aggregated over each experimental day. In total 40 couples were included in the final analysis; 20 of which contained a female *Drosophila melanogaster* that had been exposed to pymetrozine, whilst the other 20 contained a female that had not been exposed to pymetrozine.

Statistical tests for differences between groups were calculated using Mann-Whitney rank sum tests with a significance level of 0.05 using the statistics software package Sigmaplot (Systat Software Inc., London). Post-hoc calculations of statistical power for the sample sizes and significance level used and effect size observed gave an estimate of 49% power, which is considerably lower than would be usually advised and would therefore lead to an increased likelihood of type II errors.

4.2.5. *Drosophila melanogaster* fecundity assay

Male and female CantonS virgin *Drosophila melanogaster* were selected upon eclosion via CO_2 sedation, after which they were transferred in groups of 10 to vials containing standard fly food medium for 24 hours (with females and males being kept separately). They were then transferred onto either food doped with pymetrozine at a concentration of 1000ppm or control food for two days.

Pairs consisting of one male and one female were then placed together on transparent food that allowed for larval counting. In total there were four different couple phenotypes –both male and female unexposed to pymetrozine (which served as a control), both male and female exposed to pymetrozine and either the male or the female exposed to pymetrozine whilst the other sex remained unexposed. 30 couples from each group were prepared at the beginning of the experiment.

Each couple was allowed to copulate for five days before being transferred to a new vial where they stayed for another five days before being tested for pymetrozine exposure. Larval counting was done immediately upon the couple's removal from the vial. Couples in which either fly died at any point during the experiment were removed from the final analysis – this meant the final analysis included 23 couples from the control group, 22 couples in which only the female was exposed to pymetrozine, 28 couples in which only the male was exposed and 22 couples in which both male and female were exposed.

Statistical tests for differences between groups were calculated using ANOVA on ranks tests with a significance level of 0.05 using Sigmaplot. Post-hoc calculations of statistical power for the sample sizes used and effect sizes observed gave estimates of less than 50% power for each group investigated when compared to the control group, and less than 40% statistical power estimates were calculated for comparisons between groups which contained at least one fly that had been exposed to pymetrozine. This level of statistical power is considerably lower than would be usually advised and would therefore lead to an increased likelihood of type II errors.

4.2.6. *Drosophila melanogaster* life span assay

Male and female CantonS virgin *Drosophila melanogaster* were collected after eclosion via CO₂ sedation and were transferred in groups of 10 to vials containing standard fly food medium for 24 hours (with females and males being kept separately). *Drosophila melanogaster* were then transferred in groups of 10 to either pymetrozine doped food (at a concentration of1000ppm) or control food for two days, with males and females still kept apart. Following this all flies were then transferred to vials containing standard fly food medium, with male and female flies still being maintained separately.

Drosophila melanogaster were either kept alone in individual food vials or in single sex groups of 10. In total there were 48 pymetrozine exposed males and 44 pymetrozine exposed females kept individually, as well as 67 unexposed males and 44 unexposed females; 8 group vials of 10 exposed males and 4 group vials of 10 exposed females were maintained as well, alongside 7 group vials of unexposed males and 4 group vials of unexposed females. All *Drosophila melanogaster*, whether alone or in groups, were transferred into vials containing fresh fly food every 48 hours until all flies involved in the experiment had died. After each transfer fly mortality levels were recorded by counting the number of dead flies in each newly empty vial.

As this study was designed to be purely descriptive and to provide an estimate for fly mortality, no formal calculations for statistical power were made. This means that the results of this study should be treated with caution and should not be considered statistically significant.

4.2.7. Flight initiation assay

Drosophila melanogaster from the CantonS line were sedated using CO₂ upon eclosion and females were inspected under a microscope for apparent wing defects. Those females without apparent wing defects were then transferred into vials containing standard fly food for 24 hours. After this, the flies were transferred in groups of 10 onto fly food doped with pymetrozine at a concentration of either 100ppm, 500ppm or 1000ppm (or a control food that was not doped with pymetrozine). After being kept for two days on their respective food types all flies were transferred onto standard fly food for a further 24 hours.

Paraffin oil was applied to the interior of a 500ml measuring cylinder and a funnel was securely attached to the top of the cylinder (with figure 7 including an outline of the experimental setup). Female flies were then emptied from their vials (each of which contained 10 flies) into the cylinder, whereupon they attempted to initiate flight and adhered to the paraffin oil coating the interior of the cylinder or fell to the bottom. After each vial the position at which each fly landed was noted and the flies were removed from the inside of the cylinder. These results were then aggregated and split into two phenotypes dependant on whether the flies landed in the top ('good fliers') or bottom ('bad fliers') half of the measuring cylinder, and as such were judged to initiate, or not initiate, flight.

The data was then analysed using Fisher's exact test (with an applied Bonferroni correction to account for multiple testing) using Sigmaplot, such that the significance level was set at P = 0.0083. No formal statistical power calculations were made for sample size estimates, with sample sizes being chosen based on a previous report which used a maximum sample size of 84 – all groups investigated had a sample size greater than this except for the controls (with 66 control flies, 89 flies exposed to 100ppm concentration pymetrozine, 149 flies exposed to 500ppm concentration pymetrozine and 159 flies exposed to 1000ppm concentration pymetrozine included in the final analysis) (272).



Figure 7. Experimental outline of the flight initiation assay, with the landing areas which denote good or bad fliers being highlighted (adapted from (272)).

4.2.8. Drosophila melanogaster Flycube flight assay

The mechanisms behind the Flycube setup have been described in detail previously (401). The Straw lab CantonS line was used for all *Drosophila melanogaster* experiments. 3 day old flies were deprived of food overnight by being placed in a vial containing nothing but water-soaked cotton (for humidity control). The following day the flies were then sedated on ice and females were transferred in groups of 10 to vials containing food doped with pymetrozine at a concentration of either 100, 500 or 1000ppm, or control, unexposed food.

After remaining on the food for 4 hours, the flies were then sedated on ice again and females were individually selected after being checked under a microscope for apparent wing defects. 10 females were placed overnight into each Flycube, which contained soaked cotton wool to provide sufficient humidity but no food source. In total, 6 repeats were made each of the three different concentrations of pymetrozine that were tested, as well as the control group, meaning that 60 flies were tested for each level of exposure and were included in the final analysis.

The Flycube room was kept in constant darkness and at a constant temperature of 25°C in order to counteract any potential electronic heating difficulties. Each Flycube consisted of 5 video screens, as shown in figure 8 part A, which were able to display visual stimuli once an insect had been judged to successfully begin flying (but otherwise remained white) and a glass slide that sealed the top of the cube and prevented escape attempts.

The stimulation type and length was adjustable in accordance with the animal's flight pattern. In this experiment three different types of stimulation were used – a 'figure 8' stimulus which forced the insect to fly in a double loop and two control stimulus patterns (one of which presented stimuli designed to encourage the insect to fly at a constant height whilst the other presented an entirely neutral stimulus) for comparative purposes (401, 402).

As described in section 4.1, *Drosophila melanogaster* turn in accordance with a rotational stimulus in order to negate the effect of the rotational motion. Thus the 'figure 8' stimulus was able to force individual *Drosophila melanogaster* to complete two 180^o rotations in the shape of a figure 8 (in a similar style to the trajectories included in figure 8 part B) by providing stimuli with the appropriate rotational velocities; in this case a checkerboard stimulus pattern was used, with control stimuli providing altered versions of the same pattern which were not designed to stimulate rotational movements (401).



Figure 8. A) The Flycube setup, indicating the positions of several of the cameras as well the computer arrangement required (image taken from (401)). B) Aggregated flight trajectories from multiple Drosophila melanogaster whilst exposed to the 'figure-8' stimulus paradigm (adapted from (402)).

The Flycube setup involved 5 cameras as depicted in Figure 8 part A, which were able to track flying *Drosophila melanogaster* with high levels of accuracy. Once a fly initiated flight, it was followed until a certain length of time had passed (set here to be 1 second) and the initiation attempt had been judged to be successful. At this point the stimuli were provided in a set pattern – first the standard control stimulus, then the 'figure 8' stimulus and finally a z-control stimulus, with the stimulus pattern being looped if the flight continued for long enough. Only one fly at a time could be monitored by the camera system and so even if multiple flies attempted to follow the stimulation pattern only the flight of the original insect was recorded. Only stimulated flights of at least 1 second in length were saved for analysis.

Once an experiment had been terminated, all data was automatically uploaded to a shared cloud space and basic automated analysis was completed. Flight trajectories recorded during figure 8 stimulus playback were compared to expected flight patterns and the number of complete loops made by individual flies was calculated. This analysis was then aggregated with previous data set analyses to provide a database of the total number of tracked flights per stimulus type, the time length of each flight recorded (which was then aggregated across all flights for each stimulus type to produce an estimate of total flight time per stimulus), 'heat' maps of the most likely fly position during each stimulation type and turning probability in response to 'figure 8' stimulus onset. The turning probability data is of particular interest because it is able to show the median response time of a group of *Drosophila melanogaster* to the 'figure 8' stimulus as well as the likelihood of any response being elicited.

4.2.9. Circadian rhythm temperature entrainment of pymetrozine exposed and unexposed *Drosophila melanogaster* males

CantonS *Drosophila melanogaster* were sedated upon eclosion using CO₂ and virgin males were transferred in groups of 10 to vials containing standard food medium. The next day half of the virgin males were again transferred in groups of 10 to vials containing either food doped with pymetrozine at a concentration of 1000ppm or control food. After spending two days on this food, the males were again sedated using CO₂ and transferred individually to activity tubes containing standard food on one side only. In total 96 exposed males and 96 unexposed males were transferred. The activity tubes were then inserted in groups of 32 into 6 DAM2 monitors (Trikinetics, MA), which were themselves kept in a temperature, humidity and light controlled incubator. An environmental monitor was kept in the incubator in order to ensure that the correct temperature and light programmes were followed consistently.

DAM2 monitors enable an estimation of circadian activity to be calculated by shining a single infrared beam down the centre of each of the 32 vials in the monitor. Each time a fly crossed the centre of a vial (which is designed to reduce as much as possible the fly's vertical movement) the beam was broken and a count was recorded. These counts were them summed together in 15 minutes bins across the length of the experiment.

The following entrainment pattern was used for all flies; first a four day 12 hr: 12hr LD entrainment at a constant temperature of 25°C was used to ensure entrainment to a light stimulus. Then a Dark: Dark (DD) free run, also at 25°C, was completed between 00:00 relative circadian time on day five of the experiment and 16:00 on day eight. After this a four day temperature entrainment cycle was initiated in constant darkness, with temperature shifting between 16 and 25°C every 12 hours. This meant there had been an 8 hour advance in stimulus onset to minimise any potential effects from the first entrainment. Finally, flies entered another DD free run at 25°C which lasted for 3 days.

A collection of scripts called the fly toolbox was used to analyse the summated count data utilising the Matlab software package (MathWorks, Cambridge) (408). Actograms (which display the number of beam breaks recorded for individuals over every 15 minute period in the experiment) of each fly were investigated, with flies that demonstrated no activity in the last 24 hours of the final free run being excluded from any further analysis. In total therefore after applying this selection process 81 pymetrozine exposed males and 88 unexposed males were included in the analysis. Sensitivity analysis calculations using these sample sizes estimated that when using Mann-Whitney rank sum tests at a significance level of 0.05 an effect size of 0.46 could be identified with 90% statistical power.

4.2.10. Effect of pymetrozine on *Drosophila melanogaster* courtship songs

Male and female virgin CantonS *Drosophila melanogaster* were collected upon eclosion using CO₂ sedation and were transferred in groups of 10 to vials containing standard food medium for 24 hours (with both sexes being kept separately). Half of the male and female populations were then transferred to vials containing food doped with pymetrozine at a concentration of1000ppm, with the rest of the males and females being transferred to vials containing control food. After 2 days in these vials all *Drosophila melanogaster* were then transferred to vials containing standard fly food medium, with males and females still being kept apart.

2 days after being transferred to standard fly food medium, males and females were individually sedated on ice and screened for apparent wing defects. Following this one male and one female were chosen from the pymetrozine exposed/ unexposed groups and placed into plastic courtship chambers. These chambers consisted mainly of a small compartment approximately 1cm in diameter and 0.5cm deep. A metal slide in the middle of the compartment allowed for the chamber to be split in order to separate the male and female.

A microphone (Emkay NR 3158 miniature pressure-gradient microphones, Knowles Electronics Inc., Illinois) was positioned directly parallel to the bottom of the chamber for recording purposes. The microphone output was initially run through an amplifier, before being put through a CED 1401 data acquisition board. It was then digitized at a 10 kHz rate using the Spike2 software (Cambridge Electronic Design Ltd., Cambridge). All recordings were made between one and four hours after the incubator lights had turned on (meaning that the flies should have been within the morning peak of activity), with room temperature and relative humidity maintained at 22°C and approximately 55% respectively.

The couple was allowed 10 minutes to recover from the sedation, during which time the slide prevented contact between the sexes. Once the recording started, the slide was removed and the couple was allowed 5 minutes together before the recording was halted. Only songs produced during the 5 minute period immediately following the couple being able to physically interact were analysed. After the recording had finished both male and female *Drosophila melanogaster* were collected so that the effect of pymetrozine exposure/nonexposure could be confirmed using LDV. Four different couple phenotypes were recorded – both male and female unexposed to pymetrozine (control), both male and female exposed to pymetrozine, only the male exposed to pymetrozine or only the female exposed to pymetrozine. Seven couples from each group were recorded from and included in the final analysis. Pulse and sine songs were identified in the Spike2 recording manually and exported for use in Matlab using criteria based on existing literature reports and laboratory procedures (284, 356, 409).

For a section of the recording to be labelled as a pulse song the following criteria had to be satisfied:

1) Within a time of 0.5s at least 5 pulses must be identifiable

2) The apparent pulse train must have an amplitude of more than double the background noise level; this background level was assessed visually by manually placing horizontal cursors on the upper and lower lowers limits of the noise recorded by the microphones before the slide had been removed

3) If the time window between adjacent pulses was more than twice that of the gap recorded between previous adjacent pulses than a new potential pulse train was judged to have begun

Similarly, for a section of the recording to be labelled as a sine song the following criteria had to be satisfied:

1) The sine train must last for at least 0.5s, during which time the frequency of the train must not change by more than 10%

2) The apparent sine train amplitude must be more than double the background noise level (judged in a similar manner as for the pulse song criteria)

3) A new sine song was determined to have started provided the gap between adjacent songs was longer than 0.1s and the previous criteria were also satisfied

Identified pulse and sine songs were then analysed in Matlab using a bespoke software package (as previously detailed (284)), which provided frequency estimates for each individual sine song and IPI and IPF values for each individual pulse song. The sine song frequency estimates were calculated by applying fast Fourier transforms to the extracted sine songs after first applying a high-pass filter (>70Hz). Applying the previous criteria for sine song identification allowed for a frequency resolution of less than 3Hz. Peak frequencies could then be identified for individual males.

It was not possible to extract accurate estimates of the IPF by directly applying fast Fourier transforms to the extracted pulse data because the frequency resolution of individual pulses is low whilst applying such transforms to the entire pulse train leads to high levels of noise. To circumvent these issues a Matlab program was used that detected each pulse created within a manually defined region of the extracted recording by comparing the width and power of amplitude peaks above a pre-determined threshold - this detection procedure was then manually screened to ensure that no pulses had been misidentified or missed.

All pulses were centred on their largest amplitude peak, which was then normalised across the train such that all pulses had the same maximum amplitude (with negative pulses being inverted so that their peaks became positive in order to maintain phase coherence). A Hamming window function was applied to this normalised data to minimise noise. The IPI could then be estimated by extracting the time intervals between the maximum amplitude peaks and calculating the median of these intervals.

The individual pulses were then rearranged such that the distance between neighbouring peaks was constant. The distance value was chosen by the program so that the fast Frequency transform of the reconstructed pulse train had a frequency resolution of 1Hz, from which the IPF could be extracted by identification of the peak frequency.

Statistical comparisons between the IPI, IPF, sine song frequency and number of pulse trains produced from males from each of the four groups were then calculated using ANOVA tests in Sigmaplot. Post-hoc calculations of statistical power indicated that whilst calculations for significant differences in the IPI and sine song frequency between groups in which males had not been exposed to pymetrozine and those in which they had been exposed had a high level of power (over 95% in all cases tested) comparisons made between groups in which neither male had been exposed (or both had been exposed) were highly underpowered (<50% in all cases tested).

This was also true for statistical tests involving the IPF and the number of pulse trains produced: none of the statistical tests performed involving these parameters had a power above 50% and as such had a lower statistical power than would normally be advisable. This means that there is an increased likelihood of type II errors and that tests which do not find significant differences between groups should be treated with caution.

4.2.11. Drosophila melanogaster competitive reproduction assay

The LH_M–UCL line has a homozygous dominant allele which results in a red eye colour phenotype whilst the LH_B–UCL line has been bred to maintain a homozygous recessive allele, resulting in a brown-eye phenotype. The first generation offspring of a cross between these two lines should therefore solely have a red-eyed phenotype, with only the first generation offspring of two brown-eyed parents having brown eye colour. Adults from these lines, maintained in the Reuter lab (407), were kept for 24 hours in vials containing standard fly food medium (with both lines being kept separately). Both male and female adults were then removed and isolated. Four days after this removal, larvae that emerged from eggs that had been laid were picked and transferred into fresh standard food vials. Each vial contained 50 larvae as a density control measure.

Virgin males from both lines and virgin females from the LH_B –UCL line were selected upon eclosion via ice sedation and were transferred to vials containing standard fly food for two days – male and female flies were kept apart, as were males from each line. Males from both lines were then distributed equally in groups of 10 between vials containing either pymetrozine doped food (1000ppm) or control, unexposed food. All females were placed separately on the control food as well in groups of 10. All flies were kept on this food for 2 days to ensure saturating exposure to the compound.

Following this, pymetrozine exposed and unexposed LH_M –UCL males, pymetrozine exposed and unexposed LH_B –UCL males and LH_B –UCL females were placed individually in separate food vials (following sedation on ice). Pairs of males were then introduced to the vials containing individual females via aspiration. These pairs always consisted of one LH_M –UCL male and one LH_B –UCL male and were of one of three types – both males unexposed to pymetrozine (as a control), only the LH_M –UCL male exposed or only the LH_B –UCL male exposed. The triplets were then left together for 90 minutes before both males were removed from the vial via aspiration and kept separately so that they could be later tested using LDV in order to confirm their pymetrozine exposure phenotype.

Females were then allowed 4 days to lay eggs before also being removed from the vials. The eye colour of the offspring of females which had successfully mated was then checked upon eclosion, with red eyes indicating a LH_M –UCL father and brown eyes indicating a LH_B –UCL father. The final data analysis included 51 control triples, 47 triplets in which the LH_B –UCL male had been exposed to pymetrozine and 34 triplets in which the LH_M –UCL had been exposed (with figure 9 providing a schematic of the experiment).

Previous experiments in the Reuter lab (laboratory data, unpublished) had suggested a small competitive advantage conferred to the LH_M –UCL as compared to the LH_B –UCL line, meaning a slight bias may have been introduced to the control results. A generalised linear model of the results, using eye colour as a variable, was fitted using the statistics software programme R, with the control triplet utilised as a reference (410).



Figure 9. Schematic of triplets used in the competition assay, with b referring to Drosophila melanogaster with brown eyes from the LH_B –UCL line, m referring to Drosophila melanogaster with red eyes from the LH_M –UCL line and exposed/ unexposed referring to whether or not the fly had been exposed to pymetrozine (sample sizes are shown next to each triplet type).

4.3. Results

4.3.1. Drosophila melanogaster fertility and fecundity assays

No significant differences were seen in the number of eggs laid by females either exposed to pymetrozine or those that were only exposed to a control (P >0.05). There were also no significant differences seen between any group in the fecundity assay over any five day period or over the entire 15 day length of the experiment (P> 0.05 in all cases tested). Figure 10 contains distributions of egg and larval counts observed for each group tested.

Flies from both experiments had their auditory capabilities tested using LDV (free fluctuations, as defined in section 6.2.6, of representative flies from the fertility assay are shown in appendix A).



Figure 10. A) Total egg count values per individual female for the two groups involved in the fertility experiment (sample size in brackets, black dots correspond to the 5th and 95th percentiles).

B) Larval count data per couple for all four groups tested for fecundity (sample size in brackets, black dots correspond to the 5th and 95th percentiles).

4.3.2. *Drosophila melanogaster* life span assay

Mortality rates of pymetrozine exposed *Drosophila melanogaster* males and females kept individually were much greater during the first 25 days of the assay, after which a decrease in mortality rate for the exposed flies saw all four groups reach approximately the same percentage of survivors after 90 days. Some exposed flies were capable of living far longer than this however, with ages of over 120 days being reached, as is illustrated in figure 11 part A.

Mortality rates of pymetrozine exposed males and females kept in groups of ten were greater than their unexposed counterparts whilst the median number of flies per vial was greater than 1; once this stopped being the case and vial density was thus reduced then mortality rates stabilised and population numbers became equal throughout the four groups (as seen in figure 11 part B).



Figure 11. A) Survival rate of each of the four groups over the entire course of the experiment for flies that were kept individually (sample size in brackets).B) Median number of flies surviving per vial for flies that were kept in groups of ten (sample size given in brackets).

4.3.3. Flight initiation assay

Exposure to pymetrozine at any concentration tested led to a significant decrease in the ability to successfully initiate flight, as can be seen in figure 12 (P<0.001 for all comparisons made). The experimental group in which flies which had been exposed to pymetrozine at a concentration of 100ppm had a significantly greater ratio of good to bad fliers than those groups where the exposure concentration of pymetrozine was 500 or 1000ppm (P<0.005 for both comparisons made). No statistically significant differences were seen between the group in which exposure level was 500ppm and that in which the level was 1000ppm (P>0.05).



Pymetrozine concentration (ppm)

Figure 12. Ratio of good to bad fliers for control flies or flies exposed to pymetrozine at a concentration of either 100, 500 or 1000ppm (sample size in brackets and significant differences between groups are starred).

4.3.4. Drosophila melanogaster Flycube flight assay

Control *Drosophila melanogaster* that were not exposed to pymetrozine were able to routinely respond to stimuli in the expected manner. There were 1116 recognised flights during the figure 8 stimulus display resulting in a total flight time of 3266.4s, as compared to 626 and 538 flights for the control and Z-control stimuli which totalled 1061.5s and 932.1s respectively. Average flight times for the three stimuli types were thus approximately 2.93 seconds for the 'figure 8' stimulus, 1.70 seconds for the control stimulus and 1.73 seconds for the Z-control stimulus. The heat maps shown in figure 13 part B demonstrate the ability of the Flycube system to force *Drosophila melanogaster* to fly in the desired 'figure 8' pattern.





B) Two dimensional heat maps indicating the most commonly occupied positions by control Drosophila melanogaster (n=60) being tracked during each stimulus type. The colour gradient goes from blue to red (low to high average occupancy rate).

After pymetrozine exposure (at 1000ppm concentration), the number of *Drosophila melanogaster* flights successfully maintained long enough for the system to begin tracking decreased substantially for every stimulus type. There were only 56 recognised flights during the 'figure 8' stimulus display resulting in a total flight time of 112.1s, as compared to 78 and 90 flights for the control and Z-control stimuli (which totalled 134.0s and 147.3s respectively).

Average flight time for the two control stimuli thus stayed approximately the same as for the control flies (at 1.72 and 1.64 seconds for the control and Z-control stimuli respectively) but was reduced for the 'figure 8' display stimulation to 2.00s. No successful 'figure 8' attempts were recorded by the system, with the heat maps provided in figure 14 part B indicating this clearly, especially when compared to the heat maps of control *Drosophila melanogaster* shown in figure 13 part B.





This decrease in successful flight maintenance and the number of trajectories tracked compared to the control is also visible in the results from the other exposure concentrations (to a progressively smaller extent as the concentration level is correspondingly reduced). A dose dependent effect is also visible in the turning probabilities in response to stimulation (demonstrated in figure 15), with increasing concentrations of pymetrozine leading to decreases in turning probability following stimulation onset. This reduction in turning probability is reflected in the elimination of successful 'figure 8' attempts in pymetrozine exposed flies shown in figure 14 part A.



Figure 15. Turning probability after onset of the 'figure 8' stimulus at relative time 0s for each of the pymetrozine exposure levels tested (n=60 for all groups, with dark lines representing median probabilities whilst the shaded areas represent 95% confidence intervals).

4.3.5. Circadian rhythm temperature entrainment of pymetrozine exposed and unexposed *Drosophila melanogaster* males



Figure 16. A) Median activity counts during the free run following light entrainment of both pymetrozine exposed and unexposed Drosophila melanogaster, with the white/ dark background segments represent the period during light entrainment in which the incubator lights were turned on/ off respectively (standard errors are represented by vertical lines at each data point).

B) Median activity counts for pymetrozine exposed and unexposed Drosophila melanogaster during the second, post temperature entrainment free run. The white/ red background segments represent the period during temperature entrainment when the temperature was kept at 16/25°C respectively (standard errors are represented by vertical lines at each data point).

There were no significant differences in median activity count observed between the pymetrozine exposed and unexposed *Drosophila melanogaster* groups at any time point during either the free run after light entrainment or the free run after temperature entrainment. Two activity peaks are observable for both sets of flies in the free run that followed the light entrainment section of the experiment, whilst only one is observable in the temperature entrainment free run (as displayed in figure 16). Selected representative free fluctuations from flies included in the experiment are included in Appendix B.

4.3.6. Effect of pymetrozine on Drosophila melanogaster courtship songs

Significant differences in the IPI were seen between the two groups that contained a male *Drosophila melanogaster* unexposed to pymetrozine and the two groups that featured an exposed male (Control vs only male exposed = P < 0.006, Control vs both exposed = P < 0.006, only female exposed vs only male exposed = P < 0.04, only female exposed vs both exposed = P < 0.05). Significant differences were also seen between the same groups for the sine song frequency (Control vs only male exposed = P < 0.004, Control vs both exposed = P < 0.0003, only female exposed vs only male exposed = P < 0.004, Control vs both exposed = P < 0.0003, only female exposed vs only male exposed = P < 0.001, only female exposed vs both exposed = P < 0.0003, only female exposed vs only male exposed = P < 0.001, only female exposed vs both exposed = P < 0.0001. No other significant differences were calculated for these parameters. Figure 17 below displays these significant/ non-significant differences.

No significant differences were observed between any groups for either the IPF or the number of pulse trains produced (P>0.05 in all cases tested). Table 1 contains median values for all parameters of interest and appendix C contains representative examples of free fluctuations for male and female *Drosophila melanogaster* involved in the experiment.

Table 1. Median IPI, IPF, sine song frequency and number of pulse trains produced for each couple type (standard errors are given in brackets below each median value, with significant differences when compared to the control couples starred).

Couple type	Couple type		Sine frequency	Number of
Couple type	(ms)	(Hz)	(Hz)	pulse trains
Neither	34.20	180.90	135.94	59.0
exposed (n = 7)	(0.35)	(4.08)	(1.37)	(9.04)
Only male exposed (n = 7)	*36.28 (0.38)	182.33 (5.30)	*126.17 (2.96)	42.0 (8.67)
Only female exposed (n = 7)	33.12 (0.42)	185.41 (5.71)	132.24 (1.73)	36.0 (6.39)
Both exposed (n = 7)	*36.61 (0.28)	174.80 (6.65)	*127.49 (1.78)	36.0 (6.22)



Figure 17. A) Intrapulse frequency values for each of the four groups (black dots correspond to the 5th and 95th percentiles).

B) Interpulse interval values for each of the four groups (significant differences are starred, black dots correspond to the 5th and 95th percentiles).

C) Sine song frequency values for each of the four groups (significant differences are starred, black dots correspond to the 5th and 95th percentiles).

4.3.7. Drosophila melanogaster competitive reproduction assay

As shown in table 2, exposure to pymetrozine significantly decreased the likelihood of a male *Drosophila melanogaster* from either line successfully out-competing an unexposed rival male and copulating with a female, with exposure to pymetrozine leading to a significant reduction in the percentage of offspring having the eye colour of the male that was exposed. Appendix D contains examples of free fluctuations for males involved in the experiment

Table 2. Percentage of females from all triplet types whose offspring had brown eyes and the associated significance values (-/+ refers to pymetrozine unexposed and exposed males respectively, starred values represent significant differences when compared to the reference couple).

Triplet type	Number of	Percentage of triplets with	Z value
(males)	triplets	brown eyed offspring	(Pr(>z))
Lh _B -UCL -	51	13 10/	Ref
Lh _M -UCL -	51	43.1%	(-)
Lh _B -UCL -	24	95 20/	*3.628
Lh _M -UCL+		00.076	(0.000286)
Lh _B -UCL +	17	23 /0/	*-2.04
Lh _M -UCL -	47	23.4 %	(0.041306)

4.4. Discussion

4.4.1. The effect of pymetrozine on *Drosophila melanogaster* fecundity and fertility

There was no evidence found in the data presented in section 4.3.1 to support the idea that reductions in insect fecundity previously reported for aphids after pymetrozine exposure could extend to *Drosophila melanogaster*, with no significant differences being calculated between control and pymetrozine exposed females with respect to either egg laying or production of larvae (meaning that fertilised, viable eggs had been produced) (320).

Females which had copulated with pymetrozine exposed males also produced viable offspring in statistically similar numbers as those who had mated with control males, suggesting that pymetrozine does not affect male *Drosophila melanogaster* fertility.

These results agree with previous reports that female mechanosensory mutants (for example *hemingway* mutants, which have abolished auditory mechanical amplification) have wildtype levels of fertility – the equivalent male mutants, as well as male *tilB* mutants, are sterile because of the roles played by these genes with regards to ciliary motility which pymetrozine is unlikely to replicate given it's mechanism of action (411, 412). Although these findings do not necessarily by themselves rule out possible effects on mosquito fertility or fecundity, they do reduce the probability that such an effect exists.

Both tests were underpowered however and as such the lack of a significant difference between the control groups and groups exposed to pymetrozine may be the result of a type II error – previous studies investigating differences in the number of offspring produced by female flies required comparison groups of at least 50 females in order to cope with the high degree of variability observed in offspring counts between individuals (413, 414). As such it would require experiments including much larger sample populations to reach concrete conclusions on the possible effect of pymetrozine on fertility and fecundity in *Drosophila melanogaster*.

4.4.2. The effect of pymetrozine on *Drosophila melanogaster* lifespan

Taken together, the two experiments within the lifespan assay shown in section 4.3.2 provide some support for the idea that density plays a significant role in survival probability after pymetrozine exposure, with significantly higher mortality rates seen for vials containing multiple pymetrozine exposed flies (both male and female) than for control vials, with these rates reducing once fly density had been reduced. This concurs both with anecdotal evidence that *tilB* mutants demonstrate similar mortality rates when kept in comparatively dense vials (laboratory data, unpublished) and with published reports that increased adult density leads to decreases in *Drosophila* life span (415, 416).

Mutations to olfactory, gustatory and other sensory peripheral neurons have been reported to affect *Drosophila melanogaster* lifespan and mortality rates, and as such mechanosensory deficits have the potential to induce similar reductions in survival probabilities in relatively challenging environments (417, 418). Retaining mechanosensory capability in crowded environments could therefore provide a competitive advantage.

However the results of the test where all flies were housed individually, in which both male and female flies exposed to pymetrozine had greater mortality rates than unexposed individuals, suggest that fly density may not be the most significant factor determining fly lifespan. Indeed, the results of this assay seem to imply that pymetrozine itself may have an impact on short-term mortality, with those exposed flies that died early potentially having consumed more of the compound than those that survived for longer.

There could also be compensatory mechanisms involved, or some degree of potential reversibility (although no such indications have so far been identified in laboratory tests), such that after a certain length of time the fly is able to adapt to pymetrozine exposure and is fully capable of obtaining lifespans on the scale as unexposed *Drosophila melanogaster*. Residual levels of geotaxis have been identified in *Nan* and *lav* null mutants which are entirely abolished from pymetrozine exposed flies, suggestive of compensatory mechanisms which attempt to fill the role that should be occupied by ChOs (27).

These lifespan assays were designed to be exploratory and test the upper potential limits of the lifespan of pymetrozine exposed *Drosophila melanogaster*, and as such do not carry substantial statistical weight. Potential significant differences between groups (or the lack thereof) do not carry statistical significance and further, larger assays using group sizes in line with the published reports would be necessary to fully explore the impact that pymetrozine could have on *Drosophila melanogaster* mortality rates (419, 420).

4.4.3. The effect of pymetrozine on *Drosophila melanogaster* flight initiation and maintenance

Given that *Drosophila melanogaster* use at least three different sensory-motor reflexes to maintain only the vertical aspects of flight, ablation of ChO-mediated sensation via pymetrozine could be expected to significantly reduce flight ability – this was indeed the major conclusion of sections 4.3.3 and 4.3.4, with pymetrozine exposure leading to reductions in both flight initiation and maintenance (421).

Beyond this, there also appears to be a dose-dependent effect of the compound with further significant differences calculated between different exposure concentrations as concentration increased. This may be the result of inefficient exposure to the compound at lower concentrations, with toxin levels within the insect being insufficient to affect every ChO present. The retention of some level of ChO function could thus be enough to maintain some (reduced) flight capability. As the exposure concentration is increased so is the likelihood of complete ablation of all ChOs in the individual fly and therefore further deterioration of flight ability. Such dose dependent effects on flight (and climbing) have been identified for CO_2 exposure for example (422).

Whilst there was no statistically significant difference found between *Drosophila melanogaster* exposed to pymetrozine at either 500 or 1000ppm in terms of flight initiation capability, there was a further reduction in flight turning probability at the higher exposure level – this could be the result of different internal mechanisms which control flight initiation and turning in *Drosophila* species (421, 423, 424). This suggests that, based upon these results, the highest concentration of pymetrozine possible should be tested for use in future trials, though this should be adjusted according to safety and tolerability concerns.

Whilst the Flycube system is highly efficient at tracking the flight patterns of *Drosophila melanogaster* and can provide a wealth of information about the resulting flight behaviour, it is unable to provide data regarding groundspeed variability and cannot force individual flies to initiate flight behaviour (which can lead to relatively low numbers of flight attempts) (29, 328). Further tests which utilise tethered flying *Drosophila melanogaster* to investigate changes in groundspeed in conjunction with JO neuron activity could provide further detail on the role of ChO neurons in flight maintenance, and the resulting effect of pymetrozine exposure on these processes (325).

4.4.4. The effect of pymetrozine on Drosophila melanogaster circadian rhythms

The ability of pymetrozine exposed *Drosophila melanogaster* to entrain to temperature changes, as evidenced by the lack of significant differences observable in section 4.3.5 between the control and pymetrozine exposed groups in terms of their second free run activity pattern, was surprising given the prior literature on the requirement of functioning ChOs for *Drosophila melanogaster* to entrain to temperature (386). The impact of pymetrozine was assessed after the experiment using LDV (with examples shown in appendix B) and every fly that had been exposed to pymetrozine showed a clear and expected phenotype, suggesting that the lack of a significant difference between the two groups is not the result of non-exposure to pymetrozine in the test group.

Previous studies investigating the importance of ChOs for temperature entrainment used mutants that completely lacked all ChO function (386). It could therefore be the case that pymetrozine exposed *Drosophila melanogaster* are still able to entrain to a temperature Zeitgeber because there are potentially multiple sensory input pathways to ChOs and only one of these is affected by the compound. For example, *Drosophila* ionotropic receptor 25a has been shown to form part of a pathway to the *Drosophila melanogaster* circadian clock that is able to detect changes in temperature in the absence of all currently reported temperature sensors located in the antenna (373). The ability of pymetrozine exposed *Drosophila melanogaster* to entrain to vibrational stimulation could provide greater insight into the various potential sensory input pathways to ChOs and the circadian clock (33).

As a result of the *Drosophila melanogaster* experiment, it seems unlikely that pymetrozine would have an effect on temperature entrainment of mosquito circadian rhythms and so may not affect the clock-related activity patterns observed for both males and females, particularly with regards to biting frequency and habits (379, 425).

4.4.5. The effect of pymetrozine on Drosophila melanogaster courtship

Male *Drosophila melanogaster* are the producers of both pulse and sine songs within a couple. That the differences in IPI and sine song frequency described in section 4.3.6 are seen between both groups which contain pymetrozine unexposed males and both those that contain exposed males, in addition to the sizable extent of the differences seen, suggests pymetrozine has a significant effect on both pulse and sine song production in males.

Both the median IPI and IPF values for each group lay within the ranges previously reported for *Drosophila melanogaster* (approximately 35ms and between 150 to 250Hz respectively); however there was an observed reduction in median sine song frequency when compared to the expected value of 150Hz in all groups tested. This could be the result of the documented changes in major song component characteristics in response to temperature shifts (426, 427).

Given that previous reports did not find auditory feedback to be required for successful production of sufficient quality songs (as discussed in section 2.10.2), and that unpublished laboratory data suggested that ablating single antennal function led to increases in IPF but not IPI, it is possible that the pharmacological effect of pymetrozine in some way led to the demonstrated increase in IPI and decrease in sine song frequency. The statistical power of tests for differences in IPF between different groups was too low however to conclusively determine that the lack of significant differences is real and not the result of a type II error.

This pharmacological effect for example is not limited to just ChOs within the JO and should extend to all ChOs throughout the fly body. This could then lead to wing beat control problems, which would lend further credence to the previously presented evidence in sections 4.3.3 and 4.3.4 of a reduction in flight capability following pymetrozine exposure (as well as the aforementioned literature described in section 2.10.1 reporting the necessity of a functional JO for proper flight control) (29, 325).

There is also the possibility that pymetrozine has a secondary target (as although a very high concentration of the compound has been reported to be necessary to affect other ion channel complexes, only a few such channels have been tested (27)) and it is the impact of the toxin on this target that results in an altered song phenotype – for example *Drosophila* species with mutations in the *slowpoke* locus have recently been found to have relatively lower frequency sine songs than controls (428).
These significant changes to important pulse and song characteristics are likely to be related to the decrease in competitive fitness detailed in section 4.3.7 for pymetrozine exposed male *Drosophila melanogaster*. Given sufficient time and a low fly density, male *Drosophila melanogaster* were still able to copulate with females after pymetrozine exposure and as such this change in song structure does not complete eliminate reproductive viability. It may however severely decrease the likelihood of outcompeting rivals during copulation attempts (especially when taken in tandem with the other effects that pymetrozine has on insect physiology), which is not only interesting from a theoretical stand-point but also from the point of view of control programme design.

As the IPI has been reported to be an essential characteristic of a pulse song (which is itself deemed crucial to successful courtship events) whilst the sine song is considered to some extent unnecessary in *Drosophila melanogaster*, it could be that this parameter specifically is responsible for the reduction in fitness observed for males following pymetrozine exposure (343, 344).

The possibility that it is specifically a reduction in comparative song quality that led to the apparent reduction in fitness demonstrated in section 4.3.7 could be further investigated by repeating the experiment with the same male phenotypes but this time using only females that had been exposed to pymetrozine; if pymetrozine unexposed males no longer hold a competitive advantage this would suggest that song quality is the major determining factor in mate selection (346). If pymetrozine exposure still results in a reduction in male fitness then this could instead be due to other effects of pymetrozine on the males, with compound exposure leading to a loss of proprioception and thus a reduction in physical fitness for example (429). It seems reasonable to hypothesise however that it is combination of all of the effects of pymetrozine exposure (loss of proprioception, change to song characteristics etc.) that results in the calculated decrease in competitive fitness.

Regardless of the exact cause, using a directly competitive environment shows the extent of the difficulties facing pymetrozine exposed *Drosophila melanogaster* males during courtship. If pymetrozine is able to reduce competitive fitness in male mosquitoes in addition to impacting courtship behaviour in both sexes then the compound could become an attractive proposition for mosquito control.

5. Mosquito behavioural assays and physiological measurements

5.1. Introduction

The results described in section 4 showed the impact that pymetrozine exposure can have on *Drosophila melanogaster*, with significant reductions in flight ability and competitive fitness noted. Whilst *Drosophila melanogaster* may be able to serve as a basic model organism, in order to investigate the potential effects that pymetrozine may have on mosquito species investigations utilising mosquitoes must be performed.

Modern mosquito control programmes employ multiple intervention methods in order to maximise the impact on mosquito populations (as discussed throughout section 2.3) (430, 431). This includes larval source management, with identification of larval sources proving vital to control programmes targeting *Aedes* and *Anopheles* species for example (62, 144).

Whilst a variety of different larvicides are currently used (with many more proposed for use), the unique mechanism of action demonstrated by pymetrozine would make any potential larvicidal effect of particular interest for future use as an intervention (432-434). Testing the potential impact of pymetrozine exposure on *An. gambiae* larvae would therefore provide information regarding the compound's efficacy with regards to larval stages of the mosquito lifecycle.

Although pymetrozine's potential as a larvicide should not be neglected, given the compound's effect on auditory function in *Drosophila melanogaster* and the results of the competition assay presented in section 4.3.7, it also seems applicable to investigate the potential impact on mosquito fertility and fecundity (27). Previous reports have assessed the impact of interventions or mutations on fertility and fecundity, and thus have provided the basis for investigations using pymetrozine (435, 436).

Repeating such tests using *An. gambiae* mosquitoes (which are highly relevant given the role of this species as a major vector of disease) would therefore be able to provide evidence as to whether pymetrozine exposed male mosquitoes are able to mate with unexposed females or, alternatively, whether pymetrozine exposed female mosquitoes are able to copulate with unexposed males. Experimental results shown in section 4 suggested that whilst pymetrozine led to a significant reduction in the competitive fitness of male *Drosophila melanogaster*, in the absence of competition exposed males were still able to copulate with females and no significant differences were found in the fertility and fecundity of pymetrozine exposed female *Drosophila melanogaster* (though those studies were found to be underpowered). Given the differences in courtship routines however between mosquito and *Drosophila* species it may be the case that pymetrozine has a more potent effect on mosquito courtship; investigations into this topic using *An. gambiae* mosquitoes will help to determine the pymetrozine's potential for future use for targeting mosquito species.

The results of *Drosophila melanogaster* flight initiation and maintenance assays which compared different levels of pymetrozine exposure (as described in sections 4.3.3 and 4.3.4 respectively) seemed to show not only an effect of pymetrozine on flight ability but also a dose-dependent relationship between increases in compound concentration and decreases in flight ability. Given the importance of flight to mosquito species, it would be therefore of interest to investigate whether pymetrozine has a similar effect on mosquitoes as it does on *Drosophila melanogaster* (56, 76).

However, mosquitoes have a significantly different flying style than *Drosophila* species because of differences in wing placement on the body as well as body mass increases (76, 437). The visual system of mosquitoes is also less well-understood and analysed than the corresponding *Drosophila melanogaster* system (438-440). Thus before the experiments conducted in section 4.3.4 can be repeated for mosquitoes, an exploratory study is necessary in order to optimise parameter settings in the Flycube and ensure that mosquitoes not exposed to pymetrozine are able to respond to stimuli in a regulated and controllable manner.

The majority of the data so far published using LDV to investigate insect auditory function used displacement data that was non-angular i.e. did not provide the angular movement of the antenna but instead gave the direct displacement (285, 394). Providing angular displacement data however removes systematic biases associated with measuring force deflections at different lengths along the flagellum (288).

Calculation of the antennal length for both male and female mosquitoes from each species that will be investigated in section 7 will thus enable the future conversion of flagellar displacement data into angular flagellar deflections if necessary and will also enable future comparisons between energy and displacement gain estimates calculated for different species and sexes (as calculated in section 7) to be viewed in the context of the distance of the laser focus point from the pedicel.

Given that this thesis will focus on the three species detailed in section 2.1 (*Ae. aegypti, An. gambiae* and *Cx. quinquefasciatus*) because of their relevance as major disease vectors, these measurements will be taken for male and female mosquitoes from these species.

The tuning of male and female mosquito antennal systems (in terms of their mechanical sensitivity and JO tuning) has been reported to depended on the WBF of males and females in that species, particularly with regards to distortion products formed by the interaction of male and female flight tones (34, 276, 358). Although WBF estimates have been reported for all three species investigated in this thesis (as described in section 2), the WBF can change according to environmental factors such as temperature (34, 357).

As such it is vital that WBF estimates are calculated in the exact environmental conditions present during mosquito experiments performed throughout this thesis so that accurate estimates of both the fundamental flight tones and the resulting distortion products can be calculated. This will allow the results of future investigations into the frequency tuning of the mosquito flagellum detailed in section 7 to be compared and contrasted with this data.

5.2. Materials and methods

5.2.1. Mosquito rearing

All, *Ae. aegypti* and *An. gambiae* (Kisumu strain) and *Cx. quinquefasciatus* (Muheza strain) mosquitoes used for experiments were provided by Shahida Begum from the London School of Hygiene and Tropical Medicine. All mosquitoes were reared using a 12 hr: 12hr LD cycle at 26°C and 75% relative humidity and were fed using a 10% glucose mixture. Horse blood feeding, where appropriate, was completed by a trained research assistant using the Hemotek system (Discovery Workshops, Accrington).

5.2.2 Compound preparation

Pymetrozine: Preparation of the compound was identical to that described in section 4.2.2 for feeding experiments in *Drosophila melanogaster*.

5.2.3. Compound exposure – ingestion

Male and female experimental mosquitoes were deprived of food overnight before being given a mixture of 1000ppm pymetrozine stock solution and a 10% glucose solution which was created in such a manner as to give the required pymetrozine concentration – control mosquitoes were provided with a solution containing only 10% glucose. The mosquitoes were exposed to pymetrozine exposed/ unexposed food for 6 hours before it was replaced with non-doped food.

The length of time during which the mosquitoes were allowed to feed from doped food was chosen to minimise the possibility of multiple feedings whilst maximising the number that would feed at all. Confirmation of effective pymetrozine exposure was made using LDV wherever possible, though this could only be done after the conclusion of the experiment as this form of testing is destructive.

5.2.4. Larvicidal assay

Following World Health Organisation (WHO) guidelines (441), groups of 30 third instar *An. gambiae* larvae were added to test cups that already contained 100ml of water. All larvae were screened under a microscope after being transferred to confirm their general health. 10 ml of a solution containing the relevant concentration of pymetrozine was added to the appropriate cup to produce a final dilution of the compound – the final concentrations tested were 1, 10, 100 and 1000ppm. A control solution was also tested which was composed only of the solvent used to dissolve pymetrozine (in this case milli-Q water), and as such did not contain any pymetrozine.

Each concentration tested (as well as the control) included a replicate, meaning that 60 larvae in total was analysed for each dilution group included in the series. Sufficient food was added to each cup to prevent larval death from starvation.

The cups were kept between 25 to 27°C with constant 70% humidity and had a 12hr: 12hr LD photoperiod. Mortality was measured at 24, 48 and 72 hours after the beginning of the experiment. Statistical comparisons were made between the control group and groups that had been exposed to some concentration of pymetrozine using Fisher's exact test with the Bonferroni correction to account for multiple testing using Sigmaplot, such that the significance level was set at P = 0.0125.

WHO guidelines for larvicidal assays recommend that at least five replicates are made for each compound concentration tested in order to reach an accepted level of statistical power, meaning that this assay as conducted should be considered statistically underpowered and therefore has an increased likelihood of type II errors.

5.2.5. Impact of pymetrozine on reproductive success in males and females

An *An. gambiae* colony was blood fed in the laboratory of Dr Gareth Lycett at the Liverpool School of Tropical Medicine, and then allowed to lay eggs, which were reared until the pupal stage. The pupae were then sexed under a microscope with males and females being transferred to separate vials at a constant density of 10 pupae per container. After emergence the mosquitoes were given one meal of 10% glucose food before being deprived of food overnight.

All mosquitoes were then given 6 hours to feed from a glucose solution containing pymetrozine (at a concentration of either 50, 100 or 1000ppm) or from a control solution that did not contain pymetrozine. There were in total seven different experimental groups; apart from the control group in which neither sex had been exposed to pymetrozine, all groups contained one sex exposed to the compound (with a concentration of either 50, 100 or 1000ppm depending on the group) and the other entirely unexposed.

Between 50 and 55 female mosquitoes and 45 and 50 male mosquitoes were aspirated into different courtship containers, where they were provided with a constant source of glucose. The containers were stored at a constant humidity and temperature of 75% and 26°C respectively in a room that was kept on a 12hr: 12hr LD cycle with dawn and dusk photoperiods included. Blood meals were provided on both the second and third day that the mosquitoes were kept in the containers and blood feeding was confirmed visually for the majority of females.

After five continual days of containment female mosquitoes were aspirated into individual chambers containing damp cotton to allow them to lay eggs. The females were then monitored over the following 72 hours and any mosquito that laid any number of eggs was recorded. These eggs were then monitored to confirm whether hatching occurred, which would therefore indicate prior insemination of the female mosquito.

The total number of females included in the final analysis is included in table 4 below i.e. 39 females from the control group, 45 females exposed to pymetrozine at a concentration of 50ppm, 50 females exposed to pymetrozine at a concentration of 100ppm, 28 females exposed to pymetrozine at a concentration of 1000ppm, 12 pymetrozine unexposed females coupled with males exposed to pymetrozine at a concentration of 50ppm and 33 unexposed females coupled with males exposed to pymetrozine at a concentration of 1000ppm were included in the final analysis. Statistical testing was conducted using Fisher's exact test with the Bonferroni correction to account for multiple testing using Sigmaplot. Calculation of post-hoc tests of statistical power indicated that statistical power throughout all comparisons was <50%, which is much lower than would usually be advised and as such statistical test results should be treated cautiously because the study is highly statistically underpowered.

5.2.6. Mosquito Flycube flight assay

The experimental set up was similar to that described previously for *Drosophila melanogaster* in section 4.2.9 with 5 to 7 day old female *Cx. quinquefasciatus* mosquitoes being sedated on ice, checked for apparent wing defects and then placed overnight in the Flycubes in groups of three in order to judge the ability of the Flycube system to track mosquito flight patterns as well the mosquito response to stimulation. The stimulus types presented were identical to those presented to *Drosophila melanogaster* in section 4.2.9 and the experimental length, environment and analysis procedure were also duplicated.

In total 12 different mosquitoes were included in the final analysis, none of which had been exposed to pymetrozine. As this was an exploratory study, no formal statistical power calculations were made.

5.2.7. Mosquito flagellum length measurements

Male and female *Ae. aegypti, An. gambiae* and *Cx. quinquefasciatus* mosquitoes between 3 and 6 days old were first sedated on ice and then held by the thorax in a pair of inverted forceps. Using a pair of sharpened forceps under a microscope, the flagellum of each mosquito was removed from the pedicel and placed into a buffer solution. The flagella were then transferred to a microscope slide and a LSM 800 Zeiss confocal microscope (Zeiss, Cambridge) was used to examine each individual antenna. Images were taken of antennae which were judged to be intact. These images were then analysed using the LSM image browser software which enables measurement of each flagellar section.

Radial symmetry of the mosquito flagellum was assumed, meaning that it was assumed that the maximum length (as well as the width) of each flagellomere was identical in every 2-dimensional plane. The length and width of each section was recorded and then used to calculate the surface area of each flagella segment. The total flagellar surface area was then determined from these individual computations.

In total 8 *Ae. aegypti* female flagella, 10 *Ae. aegypti* male flagella, 9 *An. gambiae* female flagella, 22 *An. gambiae* male flagella, 23 *Cx. quinquefasciatus* female flagella and 16 *Cx. quinquefasciatus* male flagella were included in the analysis.

Statistical testing was done using Wilcoxon signed rank tests in Sigmaplot. Post-hoc statistical power tests found that whilst comparisons between the surface areas of male and female *Ae. aegypti* and *Cx. quinquefasciatus* had power estimates of over 95% in both cases, the same comparison made between male and female *An. gambiae* was significantly underpowered (<20%), and as such there was an increased likelihood of a type II error. All statistical comparisons of the antennal length were found to have a statistical power of over 90% in post-hoc tests.

5.2.8. Mosquito wing beat frequency measurements

3 to 6 day old male and female *Ae. aegypti, An. gambiae* and *Cx. quinquefasciatus* mosquitoes were sedated using ice and then attached to a wire filament using blue-light cured dental glue. The filament was glued to the dorsal side of the insects' thorax in such a way that the mosquito was still able to freely move its wings. The wire was then attached to a micromanipulator atop a vibration isolation table and the individual mosquito was given 5 minutes to recover from sedation.

The same microphone as used in the *Drosophila melanogaster* courtship song experiments described in section 4.2.11 (Emkay NR 3158 miniature pressure-gradient microphones) was placed perpendicular at a set distance of 1cm from the mosquito being measured, along with the same amplifier and spike2 hardware and software setup as previously used. The wing beat of the mosquito as it attempted to escape from its' tether was then captured using the microphone. Each recording lasted for 5 minutes before the mosquito was removed from the setup.

Individual spectrograms of 5s segments of wing beats were analysed in spike2, with a Hanning window being applied to the data. Both the first and second harmonics were then identified using a 4096 point FFT. In total 8 *Ae. aegypti* females, 10 *Ae. aegypti* males, 10 *An. gambiae* females, 11 *An. gambiae* males, 11 *Cx. quinquefasciatus* females and 11 *Cx. quinquefasciatus* males were included in the analysis.

Statistical analysis was conducted using ANOVA on ranks tests in Sigmaplot. Posthoc power calculations estimated that whilst the statistical power of comparisons between mosquitoes of the same species but a different sex were always above 95%, the statistical power of comparisons between same sex mosquitoes of different species was less than 40% in all cases tested (apart from the comparison made between male *Ae. aegypti* and *An. gambiae* mosquitoes, for which a statistical power of greater than 80% was calculated). This indicates that whilst comparisons between conspecific male and females were unlikely to have a high probability of type II errors occurring, the likelihood of these errors occurring for comparisons between species was much greater than would be usually advised.

5.3. Results of mosquito behavioural assays and physiological measurements

5.3.1. Larvicidal assay

Significant differences were calculated for comparisons between the control group and the *An. gambiae* larval group exposed to pymetrozine at concentrations of 1000ppm (P<0.01). For all other statistical comparisons no significant differences were identified between compared groups, in particular when comparing the control group with the groups exposed to pymetrozine at concentrations of 1 and 10ppm (as is evident in table 3).

Table 3. Number of surviving An. gambiae larvae 24, 48 and 72 hours after pymetrozine exposure for each of the five concentrations of pymetrozine tested (numbers in brackets are the percentage of the starting population remaining at that time point, starred values are significantly different from the equivalent control group values).

Pymetrozine	Larvae surviving	Larvae surviving	Larvae surviving
concentration (ppm)	at 24 hours	at 48 hours	at 72 hours
0 (control)	60	60	57
0 (control)	(100%)	(100%)	(95%)
	60	59	58
I	(100%)	(98.3%)	(96.7%)
10	60	56	54
10	(100%)	(93.3%)	(90%)
100	52	50	49
100	(86.7%)	60 (100%) 59 (98.3%) 56 (93.3%) 50 (83.3%) 15* (25%)	(81.7%)
1000	25*	15*	0*
1000	(41.7%)	(25%)	(0%)

5.3.2. Impact of pymetrozine on reproductive success in male and female mosquitoes

There was no data recorded for the group containing unexposed females and males exposed to a pymetrozine concentration of 100ppm due to damage sustained to the courtship container during the experiment. The group of unexposed females and males exposed to 50ppm concentration pymetrozine also suffered losses, but some females remained for transfer to individual vials (table 4 includes a full list of the number of females transferred and the percentage of those that were transferred that laid eggs, with figure 18 representing this data graphically).

Whilst it does appear that either male or female exposure to high concentrations (i.e. 100ppm or greater) of pymetrozine reduced the number of fertile females found to almost zero, with no fertile females found in the group that contained females exposed to pymetrozine at a concentration of 1000ppm, no statistically significant differences were identified between any of the groups.

Table 4. The number of egg laying and fertile females for each experimental group (values in brackets are percentages of the total population).

Treatment group (Females/ Males)	Number of females	Number that laid eggs	Number of fertile females
0ppm/ 0ppm	30	7	5
(control)		(17.9%)	(11%)
	45	16	11
Sobbut obbut	45	(35.6%)	(24.4%)
1000000/00000	50	4	1
	50	(8%)	(2%)
10000000/00000	28	0	0
	20	(0%)	(0%)
0ppm/ 50ppm	12	1	1
oppin/ soppin		(8.3%)	(8.3%)
0ppm/ 1000ppm	33	3	2
оррни тоооррни	55	(9.1%)	(6.1%)



Figure 18. Number of fertile females for each experimental group, with the numbers above each bar representing the relevant absolute number of fertile females as well as the total female population above each bar.

5.3.3. Mosquito Flycube flight assay

245 flights were recorded for mosquitoes whilst the 'figure 8' stimulus was being displayed, with a total flight time of 694.3s – this is in comparison to 332 and 698 flights for the control display and z-control display respectively (which resulted in total flight times of 963.9s and 1896.5s for each stimulus type). Flight durations for each of the stimulus types thus averaged 2.83 (8figure 8' stimulus), 2.90 (control stimulus) and 2.72 (Z-control stimulus) seconds, with the control flight times being much greater than previously seen for *Drosophila melanogaster* and so suggesting that the mosquito flight patterns were not significantly more affected by the test stimulation than by the control. The heat maps shown in figure 19 part B indicate that the flight response to the 'figure 8' and control stimuli was similar (though no statistical test was completed to confirm the potential statistical validity of this statement).



Figure 19. A) Aggregated individual two dimensional trajectories for each of the three stimuli patterns for Cx. quinquefasciatus female mosquitoes (n=12), with the number and time length of trajectories indicated above each stimulus.

B) Two dimensional heat maps indicating the most commonly occupied positions by female Cx. quinquefasciatus mosquitoes (n=12) being tracked during each stimulus. The colour gradient goes from blue to red (low to high average occupancy rate).

5.3.4. Mosquito flagellum length measurements

There is a statistically significant reduction in overall length and surface area of the flagellum for *Ae. aegypti* and *Cx. quinquefasciatus* male mosquitoes compared to females from the same two species (P <0.001 for all comparisons). Male *An. gambiae* have a significantly longer flagellum than female mosquitoes from the same species but there was no significant difference identified between the two sexes from this species in the terms of the total flagellar surface area (as seen in table 5). Appendix E contains a complete table which includes median values for each flagellomere individually.

Table 5. Median values for total length of the flagellum from the point of laser focus to the pedicel, the total length of the entire flagellum and the total surface area of the entire antenna (standard errors are given in brackets below the median values, significant differences between conspecific female and male mosquitoes are starred).

Species/ sev	Length from focus	Total length	Total surface
Species/ sex	point to base (µm)	(µm)	area (µm²)
Ae. aegypti females	1282.30	*1795.83	*154550.5
(n= 8)	(18.20)	(20.09)	(2897.9)
Ae. aegypti males	1244.11	1630.86	118863.5
(n= 10)	(17.74)	(21.52)	(2266.7)
An. gambiae females	1119.26	*1548.03	140408.8
(n= 9)	(11.72)	(12.08)	(3250.7)
An. gambiae males	1551.68	1736.68	139051.0
(n= 22)	(18.06)	(10.32)	(1288.1)
Cx. quinquefasciatus females	1447.17 (14.03)	*1863.21 (24.24)	*159103.1 (5515.7)
cx. quinquerasciatus males (n= 16)	1066.57 (7.96)	1400.99 (9.65)	99313.3 (1224.0)

5.3.5. Mosquito wing beat frequency measurements

Whilst significant differences were identified between the different sexes of each species (P <0.001 in all cases), no statistically significant differences were seen between females of different species (P>0.05 in all cases). *Ae. aegypti* males had a significantly higher WBF than *An. gambiae* males (P <0.05), but no other male comparison was found to be statistically significant. The ratio (shown in table 6) of male to female WBF of about 1.5 in all three species suggests that frequency matching during copulation attempts occurs at the second male, and third female, harmonic (in line with published reports).

Table 6. Median first and second harmonic estimates for all species and sexes (number in brackets indicates standard error values) alongside ratios of male to female first harmonics, as well as previously reported first harmonic estimates: Ae. aegypti published data is from (352), An. gambiae published data is sourced from (353, 442), whilst Cx. quinquefasciatus published data is from (354).

	Median first	Median	Ratio of male to	Previously
Species/ sex	harmonic/	second	female first	reported first
	Hz	harmonic/ Hz	harmonic	harmonic/ Hz
Ae. aegypti females	405.0	810.0		430.6
(n= 8)	(22.0)	(44.9)		(10.8)
Ae. aegypti males	624.5	1249.0	1.54	636.7
(n= 10)	(21.6)	(43.8)	1.54	(15.1)
An. gambiae females	384.0	768.0		492.6
(n= 10)	(16.7)	(42.2)		(2.4)
An. gambiae males	558.0	1116.0	1 45	769.0
(n= 11)	(10.2)	(20.4)	1.45	(3.9)
Cx. quinquefasciatus	390.0	780.0		428.3
females	(10.5)	(20.8)		(9.6)
(n= 11)	(1000)	()		()
Cx. quinquefasciatus	583.0	1166.0		542.4
males	(15.8)	(32.0)	1.49	(18.2)
(n= 11)	(10.0)	(02.0)		(10.2)

5.4. Discussion

5.4.1. Effect of pymetrozine on *An. gambiae* larvae

Very high concentrations of pymetrozine were required in order to lead to the significant levels of direct larval mortality reported for other compounds, as can be seen in section 5.3.1. This suggests that pymetrozine may not be a suitable for use as an larvicide given that the major challenges associated with larvicidal interventions tend to be associated more with regards to the most effective methods to locate and destroy larvae rather than which compounds should be used to do this, especially as efficacious larvicides already exist (62, 443, 444).

However, the aforementioned unpublished laboratory evidence that *Drosophila melanogaster* larvae exposed to pymetrozine doped food before eclosion which are then transferred to food not exposed to pymetrozine demonstrate severe reductions in auditory function (similar to the level observed for adults who are exposed to pymetrozine) when compared to controls may indicate that direct larval mortality may not be necessary for pymetrozine to be effective. If this effect holds for mosquito species then adult mosquitoes which emerge from the pupal stage with ablated ChO function may be sufficiently competitively disadvantaged that they are unable to control strategies for certain species (such as *Aedes* species) and as such new methods which increase the range of options available could still be useful (62).

This hypothesis would require testing however by exposing mosquito larvae to pymetrozine, allowing adults to emerge in the absence of pymetrozine and then using LDV techniques (described in detail in section 6 and discussed with regards to mosquitoes in sections 7 and 8) to measure the antennal state of the adult mosquito. This would immediately provide information as to whether pymetrozine exposure in the larval stage can result in altered adult auditory function.

Using WHO recommendations as a guideline, insufficient replicates were completed for each concentration of pymetrozine tested (441). This means that the results of the statistical tests should be treated with caution as there is an increased likelihood of statistical errors. A greater number of replicates would be necessary before conclusions can be drawn as to whether pymetrozine exposure directly leads to significant increases in larval mortality at low concentration levels.

5.4.2. Effect of pymetrozine on An. gambiae female fertility and fecundity

It has been reported that *An. gambiae* colonies tend to reproduce poorly in cages, with a small proportion of the females performing most of the egg-laying (445, 446). This theory is supported by the low number of fertile females present in the control group (seen in section 5.3.2) and therefore it is difficult to draw clear conclusions from comparisons made between different groups.

As such, whilst the results presented do not necessarily contradict the theory that pymetrozine can be used to not only target female mosquitoes in order to prevent biting attempts and reduce fertility but could also be used against males with the aim of decreasing mating success – unfortunately they also do not provide strong evidence that pymetrozine can have a large impact on mosquito fertility because of the low sample sizes.

The low sample sizes for each group, with few females from any group laying eggs, make it difficult to draw conclusions from statistical comparisons between groups because the experiment was statistically underpowered. In order to determine with any sense of accuracy the effect of pymetrozine on male and female mosquito fertility and fecundity many more replicates are necessary in order to account for the small proportion of females in each colony who lay eggs. Alternatively, a different mosquito species could be used in which a greater proportion of the females lay eggs (such as an *Aedes* species) (447).

5.4.3. The potential of Flycubes for tracking mosquito flight patterns

Whilst the results presented in section 5.3.3 show the Flycube system can accurately track mosquitoes during flight, they also unfortunately demonstrate that the stimuli provided was not able to recreate the earlier results seen for *Drosophila melanogaster*.

Although there are currently existing systems able to track mosquito flight patterns in their natural settings, the ability of the recording set-up to recognise and record mosquito flight patterns as well as the vast array of stimulus types that are available as part of the Flycube setup means that future tuning of the system to mosquitoes could enable precise tracking of individuals in response to different sensory cues – for example auditory stimuli which match the WBF of the relevant mosquito species (448). More experiments based upon previous publications (both with regards to the Flycubes and the basic underlying mechanisms of flight maintenance and sensory control) are ultimately necessary to more accurately understand the necessary stimulation types for the mosquito species before experiments utilising pymetrozine exposed mosquitoes are conducted (437, 449, 450).

5.4.4. Mosquito flagellar length and surface area measurements

The median length and surface area measurements of the flagellum for each mosquito species and sex (as estimated in section 5.3.4) are in reasonable agreement to prior reports available in the literature. A previous paper estimated the flagella surface area of *Anopheles gambiae s.s.* females to be in the region of 127000 μ m², as compared to the value given in section 5.3.4 of approximately 140000 μ m² for this species; the estimated flagellar length however was calculated to be 1136 μ m, which is far smaller than the estimate calculated in this thesis (451). In addition to this a separate report estimated the length of a male mosquito flagellum to be approximately 1300 μ m (452).

For *Ae. aegypti*, the length of the female flagellum has been reported as 2mm (±0.1mm), whilst the equivalent value for males was estimated as 1.6mm (±0.1mm) (363). Whilst the value calculated for female *Ae. aegypti* in section 5.3.4 is smaller than this reported value (at approximately 1.8mm), the estimate for male mosquitoes from this species is in agreement with the published report.

Whilst the distance from the pedicel to the focus point of the laser is statistically indistinguishable for male and female *Ae. aegypti*, there are significant differences between the same values for male and female *An. gambiae* and *Cx. quinquefasciatus* (with male *An. gambiae* having significantly longer flagella than females whilst male *Cx. quinquefasciatus* have smaller flagella than females from the same species). These differences in length are also seen between separate species.

The reduction in overall surface area for *Ae. aegypti* and *Cx. quinquefasciatus* male mosquitoes compared to females from the same two species is most likely the results of the measurement process itself, with causes the destruction of the fibrillae which form a visually distinct part of the male auditory system and are far less evident for females (76, 363). The absence of fibrillae drastically reduces the total available surface area of the male flagellum whilst the females are left comparatively unaffected. This is also true for *An. gambiae* males, but the greatly increased length of the male flagellum in this species compared to the females potentially resulted in no significant differences in surface area being calculated between the two sexes.

This antennal length data can now serve as a reference for future data conversions into flagellar angular displacements and as such could help to improve the comparability between different mosquito species of force-displacement data (as discussed in section 7).

5.4.5. Mosquito wing beat frequency measurements

The WBF estimates for *Cx. quinquefasciatus* and, in particular, *Ae. aegypti* mosquitoes from both sexes (as described in section 5.3.5 and included in table 6) match closely with previous reports, with the ratio of male and female fundamental WBFs in both species being approximately 1.5 (1.49 and 1.54 for *Cx. quinquefasciatus* and *Ae. aegypti* respectively) (352, 354).

Both male and female *Anopheles gambiae* mosquitoes however had far lower WBF values than the values published in the literature (353). This could be the result of many possible environmental and genetic variables, including changes in the rearing room temperature, differences in the strain background or the number of generations spent as a laboratory colony (306, 355, 442). Regardless of the cause, this decrease in WBF was found in both males and females and so the ratio of fundamental WBFs between the sexes was kept reasonably constant (1.45 compared to the 1.56 ratio previously calculated) (353).

This difference in estimated WBF could also indicate deeper issues with regards to the design of WBF measurement experiment; applying glue to the thorax of a mosquito may change the mechanics and resonance properties of the mosquito's thoracic box, with an increased stiffening of the system leading to potentially greater than expected WBF estimations (453). The gluing process could also potentially change the flight system in some other manner that would then lead inaccurate estimations of the natural WBF of the mosquitoes as compared to measurements made whilst the mosquito was free-flying – future estimates could therefore be made using a flight arena in which mosquitoes which are not tethered can be recorded from (358).

Whilst the differences in WBF between male and female mosquitoes were sufficiently large enough that significant differences were found (with every species having a difference between male and female median values larger than 150Hz), the sample sizes used for each group did not provide sufficient statistical power to allow for reasonably powered statistical comparison tests to be calculated for within-sex judgments across different species to be made. In order to produce more precise measurements of mosquito WBFs, which are essential for accurate estimation of relevant distortion products (as discussed in sections 2.10.2 and 7.3.6), both the inclusion of larger sample sizes and the utilisation of free-flying mosquitoes are required.

6. *Drosophila melanogaster* pymetrozine vibrometry and electrophysiology

6.1. Introduction

Experiments utilising LDV have allowed for intricate examinations of the sensory systems of many different types of insects, including mosquitoes, *Drosophila* species, bees and crickets (284, 289, 454, 455). Experiments utilising *Drosophila melanogaster* are particularly relevant for this thesis, with extensive investigation of signatures associated with mechanotransduction in the fly's antennal ear having been conducted and reported (274, 282). *Drosophila melanogaster* serves as an ideal tool for investigation into these signatures because of the accessibility of the arista and the genetic malleability associated with this model organism (276).

These signatures (as described for *Drosophila melanogaster* and discussed also in section 2.7) form the basic definitions for the active hearing process and include:

1) Power gain by motile mechanosensory cells i.e. active amplification of the antennal ear's mechanical stimulation (283)

2) self-sustained otoacoustic emissions that are the result of uncontrolled amplification (282)

3) increased antennal sensitivity in response to decreases in stimulus intensity as a result of a compressive nonlinearity associated with transducer gating (456)

4) increased frequency selectivity due to continuous cycle-by-cycle amplification (282, 290)

Given that auditory function in *Drosophila melanogaster* is reliant on ChOs, and pymetrozine has been demonstrated to ablate mechanosensory functionality in this organ, it is of interest to investigate the impact that the compound could have on these signatures not only with a view to inclusion in control programmes but also with regards to the fundamental machinery behind these processes (27).

There are two possible exposure methods that could be utilised in order to investigate the potential effect of pymetrozine on *Drosophila melanogaster* auditory function; oral pymetrozine ingestion or direct injection of the compound.

Ingestion experiments have the clear benefit that, compared to injection protocols, they more accurately mimic the method of exposure to pymetrozine that insects would face in the wild (26). Injection experiments are also limited in that the individual fly can become damaged as a result of the experimental procedure.

On the other hand, injection experiments also have some clear advantages – they allow for clear comparison between pre- and post-pymetrozine states within a single insect, removing the possibility that changes to the auditory system were due to a cause other than pymetrozine, and the full impact of the compound on the auditory system can be clearly observed as the entire body is soaked in pymetrozine. It also removes the possibility that the absence of a nerve response to stimulation is the result of incorrect placement of a recording electrode.

In addition to this the only previously published work regarding pymetrozine's effect on *Drosophila melanogaster* auditory functions utilised compound injection as the preferred exposure method (27). Thus throughout section 6 injection experiments were preferred for all detailed electrophysiological analyses to ensure that any changes in auditory function can be ascribed to pymetrozine exposure.

As described in sections 2.5 and 4.1, one of the major benefits of utilising *Drosophila melanogaster* as a model organism is the extensive library of genetic tools available; this has enabled the creation and maintenance of numerous *Drosophila melanogaster* lines which demonstrate resistance to insecticides used in mosquito control programmes through a variety of mechanisms (including direct expression of mosquito resistance genes) (257, 457, 458). The existence of *Drosophila melanogaster* lines which demonstrate resistance to insecticide classes allows for experiments to investigate the potential presence of cross-resistance mechanisms between pymetrozine and pyrethroids for example. It also enables investigations into the potential impact of resistance expression on auditory function.

Comparisons between before and after pymetrozine injection in terms of both the free fluctuations of the individual fly (as shown before in appendices A – D in terms of feeding experiments) will allow for any changes in antennal function as measured by laser vibrometry to be identified using previously published methods of mathematical analysis (described in full in section 6.2.6) (283). Analysis of force step stimulation electrophysiology, in line with previous reported methodologies, will also enable estimation of the compound effect on not only the CAP response to stimulation but also key parameters which form an essential part of gating spring models of mechanotransduction in ChO (as described in sections 2.7 and 6.2.8) (285, 394).

6.2. Materials and methods

6.2.1. Drosophila melanogaster rearing

All *Drosophila melanogaster* stocks, unless otherwise stated, were reared identically to those investigated in section 4; all lines were kept in incubators at 25°C and 60% relative humidity and were provided with food prepared according to the Chippendale recipe. The incubators used a 12 hr: 12hr LD cycle which all lines were kept at during all developmental stages.

6.2.2. Compound preparation

Ringer: A saline solution was produced for control injection purposes following a previously published external saline solution guideline (459); this consisted of 5mM KCL, 135mM NaCl, 0.5mM CaCl₂, 2Mg Cl₂, 5mM N-Tris[hydroxymethyl]methyl-2-aminoethanesulfonic acid (TES) and 36mM sucrose, with an overall solution pH of 7.4.

Pymetrozine: Preparation of the compound was identical to that described in section 4.2.2 for feeding experiments in *Drosophila melanogaster*.

6.2.3. Insecticide resistant Drosophila melanogaster crosses

Pyrethroid resistance: yw/Y; UAS-AaegCYP9J28 7:1 males were crossed with w¹¹¹⁸;; HR-GAL4 females to produce w¹¹¹⁸/Y; UAS-AaegCYP9J28 7:1;HR-GAL4 male offspring (hereon referred to as J28 7:1xGAL4), which were used for testing. Similarly, w¹¹¹⁸/Y; UAS-AaegCYP9J28 8:1;HR-GAL4 male offspring (hereon referred to as J28 8:1xGAL4) were produced by crossing yw/Y; UAS-AaegCYP9J28 8:1 males with w¹¹¹⁸;; HR-GAL4 females. Finally, yw/Y males were crossed with w¹¹¹⁸;; HR-GAL4 females, which resulted in w¹¹¹⁸/Y; HR-GAL4 (hereon referred to as w¹¹¹⁸xGAL4) males. These males were used as a control for both previously mentioned lines. Insecticidal resistance to pyrethroids is mediated in these lines via production of a specific enzyme capable of rapidly metabolising the insecticide (457).

Cyclodiene resistance: Bloomington line number 1675 (hereon referred to as BL1675) has the genotype Rdl^{MD-RR}, meaning that all males tested should demonstrate resistance to all cyclodiene-based compounds. This resistance is the result of a single amino acid mutation within the second membrane-spanning domain of the *Drosophila melanogaster* GABA receptor that renders the receptor insensitive to cyclodiene-based insecticides (458).

Carbamate resistance: Bloomington line number 1283 (hereon referred to as BL1283) has the genotype T(3;4)86D, kar¹ ry⁴¹ Ace¹/MRS, with the balancer being required to avoid lethality. All males should be resistant to carbamate-based compounds due to a single nucleotide mutation in the *Drosophila* acetylcholinesterase gene which results in the production of a resistance-conferring enzyme (460).

DDT resistance: Resistant and susceptible lines (hereon referred to as 91-R and 91-S respectively) came pre-selected and were maintained in the lab. The 91-S line was used as a control for not only the 91-R line but also for both the BL1283 and BL 1675 lines. All resistant males demonstrate DDT resistance because of the way that the lines were originally selected; this resistance is considered to be polygenic in nature and influenced by multiple chromosomes (461, 462).

6.2.4. Compound exposure method - injection

Drosophila melanogaster injection experiments followed an identical protocol regardless of the specific line under investigation: first micro-capillaries (Harvard Apparatus, Cambridge) were pulled in a micropipette puller (Sutter Instruments, CA). These sharpened capillaries were then filled with either a control ringer solution (consisting of deionised water that had been diluted in the ringer solution described in section 6.2.3. to the desired concentration) or a ringer solution containing pymetrozine at a concentration of 1000ppm. When required, the relevant solution was injected into the thorax of the insect in such a way as to flood the body and so increase the likelihood that the compound was able to circulate to the head and as such reach the JO.

All experiments were divided into three stages – before injection, after ringer and after pymetrozine. This allowed for investigation of potential significant differences between pre- and post-ringer injection states. Differences between these states are discussed in greater detail in appendix F - there was a small but significant decrease in CAP amplitude of between 15 and 25% for all *Drosophila melanogaster* and mosquito species tested following ringer injection

No individuals from any *Drosophila melanogaster* line examined appeared to show any significant mechanical differences when comparing before and after ringer injection states and the best frequency of the antenna did not change significantly for any species experimented on. For the purposes of the major body of this thesis, comparisons drawn during injection experiments were done by contrasting the post-ringer state only with the post-compound of interest state in order to simplify procedures.

6.2.5. Vibrometry preparation and procedure – free fluctuations

All Drosophila melanogaster antennal free fluctuation recordings and electrophysiology experiments were conducted in conjunction with a PSV-400 Laser Doppler Vibrometer with an OFV-70 close-up unit (70 mm focal length) and a DD-500 displacement decoder, for which the mounting procedure has been described in detail previously (284). Male and female Drosophila melanogaster were first glued, ventrum-down, using blue-light cured dental glue to the top of a Teflon rod. Glue was then applied sparingly to the entirety of the fly (except for the antennae and wherever possible the spiracles, particularly those placed at the thorax-leg transition points) in order to minimise disturbances caused by movement of the insect. After this the fly's left arista was completely glued to the head whilst the second antennal segment of the right antenna was stuck down to prevent unwanted movements - as such only the third antennal segment and arista of the fly's right antenna should be able to move freely. The mounted fly was then placed in a micro-manipulator atop a vibration isolation table. The fly's right arista was arranged such that it was perpendicular to the Laser Doppler Vibrometer system.

Using the aristal tip as a reference/ focus point for the laser, aristal displacements were measured using the Laser Doppler Vibrometer system (as shown in figure 20). All measurements were taken without the arista being stimulated, meaning that the antennal movement should be compromised solely of a passive motion due to thermal bombardment and an active component because of mechanical feedback from mechanosensory neurons. A CED 1401 A/D converter was utilised to ascertain displacement output at a sampling rate of 100 kHz in the Spike 2 software programme. All experiments involved 3-6 day-old flies, with room temperature maintained between 20 and 22°C.



Figure 20. The Laser Doppler Vibrometer experimental setup for measuring unstimulated, free fluctuations of the antennal ear, with the laser point focussed on the tip of the Drosophila melanogaster's arista (figure provided by Nicholas Boyd-Gibbins, Jörg Albert lab).

6.2.6. Free fluctuation fitting procedure

After unstimulated, free fluctuations of the *Drosophila melanogaster* antennal receiver were recorded a fast Fourier transform of the antennal velocity amplitudes was calculated for frequencies between 1 and 3200 Hz for all *Drosophila melanogaster* lines investigated. Recording measurements within the frequency range below 51Hz contained a significant level of noise and were excluded from these analyses.

These velocity amplitudes were then fitted using a modified function based upon a previously described forced damped oscillator function (283) – whilst the original function was used for squared displacement amplitudes ($X^{2}_{(\omega)}$) this modified version was instead used for velocity amplitudes ($X_{(\omega)}$) by converting between the function domains:

$$X^{2}(\omega) = \frac{(F_{O}/m)^{2}}{(\omega_{0}^{2} - \omega^{2})^{2} + (\omega \cdot \frac{\omega_{0}}{O})^{2}}; X^{2}(\omega) = (X(\omega) \cdot \omega)^{2}$$

where F_0 is the external force strength, *m* is the antennal apparent mass, ω is the angular frequency, ω_0 is the natural angular frequency and *Q* is the quality factor (which is defined as $Q = m\omega_0/\gamma$, where γ is the damping constant).

By taking advantage of these two equivalence relationships the following velocity amplitude fit function can be created:

•

$$X(\omega) = \frac{F_0 / m}{\sqrt{\omega^2 . ((\omega_0^2 - \omega^2)^2 + (\omega . \frac{\omega_0}{Q})^2)}}$$

This function provides estimates for F_0/m , ω_0 and Q, which could then be further utilised to calculate other parameters; of particular relevance were the best frequency and the overall stiffness of the antennal receiver. This data was then aggregated across individuals from groups of interest, which allowed for overall population estimates to be made. The velocity fit function well described the antennal velocity data obtained during free fluctuation recordings, with typical R² values of over 0.98 (with figure 21 containing an example of such a fit).



Figure 21. Free, unstimulated fluctuation data for an individual Drosophila melanogaster from the 91-S line, with a frequency range of between 51 – 3200Hz and individual data points (in purple) representing Fourier-transformed velocity amplitudes at each frequency contained within the transform range. The solid black line shows the velocity amplitude function fits for the recording.

The effective stiffness of the antenna was calculated by utilising the assumption that the system could be modelled as a forced-damped oscillator in order to take advantage of the Equipartition theorem:

$$\frac{1}{2}K\langle x^2\rangle = \frac{1}{2}K_BT$$

where *K* is the effective stiffness of the oscillator, $\langle x^2 \rangle$ is the sum of the squared Fourier displacement amplitudes, K_B is the Boltzmann constant (1.38 × 10⁻²³ J/K) and T is the absolute temperature (estimated at approximately 295 K). Rearranging this equation gives:

$$K = \frac{K_B T}{\left\langle x^2 \right\rangle} = \frac{K_B T}{\int X^2(\omega) d\omega}$$

In total, 7 *Drosophila melanogaster* males from each of the 91-S, 91-R, BL1283 and BL 1675 lines were analysed, as well as 8 males from the J28 7:1 xGAL4 line and 10 males from each of the w¹¹¹⁸xGAL4and J28 8:1 xGAL4 lines. Statistical comparisons of the antennal best frequency and stiffness were made between before and after pymetrozine states using paired before and after t-tests (with a significance level of 0.05) using Sigmaplot. Post-hoc calculations of statistical power consistently showed statistical power estimates of over 95% for all comparisons made.

6.2.7. Stimulated recordings – force steps

All *Drosophila melanogaster* investigated during stimulated antennal recordings were first mounted in the same manner as described for free fluctuation recordings in section 6.2.5. A charging electrode was then inserted into the insect in order to raise its electrostatic potential, compared to ground, to -20V. Two electrostatic actuators were brought into position roughly symmetrically around the arista. These actuators allowed for electrostatic stimulation of the arista once the insect's electrostatic potential reached the aforementioned level (394).

A recording electrode was inserted into the base of the right antenna in order to measure mechanically-evoked compound action potentials (CAPs) - the charging electrode provided a reference point for these CAP recordings. Figure 22 provides a diagrammatic layout of the experimental set-up used during stimulated recordings. The type of stimulus provided to the arista changed depending on the particular experiment being conducted and is described in more detail in the following sections.

Unstimulated fluctuation measurements of the arista were recorded at the start and end of each injection stage of the experiment in order to judge the retention of auditory function throughout the experiment; flies whose free fluctuations showed a shift in best frequency of more than 10% prior to pymetrozine injection were not included in the final analysis (this percentage shift was adopted as a control measure based on the results of previous experiments and mainly served as an estimate of declining auditory function as a result of experimentation).

All force step recordings were conducted in line with previous reports (282, 285). This required the production of a symmetric step stimulus using the electrostatic actuators which lasted for 50ms and was calibrated to push the antenna to a maximum displacement of \pm 8000nm. This force step was then exponentially attenuated such that the final step stimulus resulted in an antennal displacement of \pm 40-50nm. At each step the antennal displacement and CAP response were recorded, as well as the stimulus itself – figure 23 part A contains examples of each of these response types. This data set allowed for investigation of numerous aspects of the auditory system, such as changes in antennal dynamical stiffness in response to changes in stimulus intensity (as discussed in section 6.2.8).



Figure 22. The Laser Doppler Vibrometer experimental setup for measuring stimulated fluctuations of the antennal ear as well as compound action potentials which propagate as a result of stimulation, with the laser point is focussed on the tip of the Drosophila melanogaster's arista (figure provided by Nicholas Boyd-Gibbins, Jörg Albert lab).

6.2.8. Stimulated recordings – force step analysis procedure

After the onset of a force step stimulus, the *Drosophila melanogaster* antennal receiver displays an initial displacement overshoot in the direction of the stimulus before rapidly recoiling before reaching a steady state displacement. These three steps (overshoot, recoil and then excursion) are reflective of the direct mechanical gating and adaptation of transducer channels in both vertebrate hair cells and the transducer machinery for hearing in *Drosophila* species (285, 463, 464). CAP responses are produced with a sub-millisecond delay following the aristal displacement as a result of the stimulation. This process is illustrated in figure 23 for both control and pymetrozine exposed *Drosophila melanogaster*.



Figure 23. A) Flagellar displacements (top) and CAP amplitudes (middle) in response to the corresponding force steps (bottom) for Drosophila melanogaster before pymetrozine exposure (in purple) or after exposure (in grey) – X_{PEAK} and X_{STEADY} are highlighted as part of the flagellar displacement plot.

B) Flagellar displacements (dashed line) and CAP amplitudes (solid line) for pymetrozine unexposed (left, purple) or exposed (right, grey) Drosophila melanogaster showing the time delay between flagellar and nerve responses (stimulus onset is at start of flagellar displacement and force is constant throughout). Following exposure to pymetrozine, *Drosophila melanogaster* auditory systems no longer produce CAP responses to stimuli and the previously mentioned signatures of mechanotransduction are also greatly reduced. Instead the aristal displacement to equivalent force step sizes is more than halved, with the majority of this decrease coming as the result of a reduction in the size of the initial overshoot. This initial overshoot displacement (hereon referred to as X_{PEAK}) is highlighted in red in figure 23 part A, with the steady state displacement (hereon referred to as X_{STEADY}) being shown in blue.

Whilst it has been reported that X_{STEADY} largely scales linearly with increasing force (as demonstrated in figure 24 below), X_{PEAK} shows a nonlinear force-displacement relationship in the auditory systems of fully functional *Drosophila* species (285). The corresponding dynamic stiffness equations, assuming that *m* is the apparent antennal mass, *F* is the external force and \tilde{X}_{PEAK} is the acceleration calculated at the initial displacement peak, are:

$$K_{PEAK} = \frac{\delta(F - mX_{PEAK})}{\delta X_{PEAK}}; \quad K_{STEADY} = \frac{\delta F}{\delta X_{STEADY}}$$

This nonlinear behaviour can be modelled by using a two-state gating spring transducer model (285) and is no longer present after pymetrozine exposure. The decreases in K_{PEAK} for small force amplitudes are thus greatly reduced, though they not abolished.

Assuming that X_{STEADY} is linearly proportional to force, K_{STEADY} should remain constant across all force step displacement magnitudes – this assumption does not seem to hold completely true however, as is discussed in detail in section 7.3.3.4. In any case, K_{STEADY} provides important information on the steady state elasticity of the antennal joint. Taken in conjunction with $K_{INFINITY}$, which is defined as the asymptotic stiffness of the antennal receiver in response to large force stimuli, the gating spring stiffness (K_{GS}) can be calculated. K_{GS} is defined as the difference between $K_{INFINITY}$ and K_{STEADY} (i.e. $K_{INFINITY} - K_{STEADY}$) and provides an estimate of transducer mechanical integrity (394).





As mentioned above, a two state gating spring model of a single transducer population has previously been reported to describe the ion channel populations of the *Drosophila melanogaster* auditory system in JO (285), with force being defined as:

$$F = K_{INFINITY} X - p_O(X)Nz + F_0 - m X_{PEAK}$$

where *F* is the external force, *X* is the antennal displacement, $p_0(X)$ is the ion channel open probability at displacement *X*, *N* is the number of ion channels, *z* is the single channel gating force and F_0 is a constant offset term, with the other terms as defined above. Electrophysiology experiments utilising force steps allow for fitting of this model to force-displacement data for the antennal receiver, which then provides estimates of *N*, *z*, $K_{INFINITY}$ and X_0 (the displacement at which the ion channel open probability is equal to 0.5).

The ion channel open probability for a single transducer population is defined as:

$$p_O(X) = \frac{1}{1 + e^{-\frac{z(X - X_0)}{K_B T}}}$$

where $p_O(X)$ is the ion channel open probability at displacement X, K_B is the Boltzmann constant and *T* is the absolute temperature, with the other terms as defined above. For all *Drosophila melanogaster* analyses X_0 was determined to be 0nm i.e. the resting position.

A model of two independent transducer populations was further developed from this starting point as a result of the identification of different functional roles for the different populations of scolopidia contained in the JO (327). This adapted model follows the same fundamental principles as the single transducer population model but allows for two separate ion channel populations to exist – these are denoted as the sensitive and insensitive populations to reflect their sensitivity, or lack of sensitivity, to auditory stimulation (as discussed in section 2.7) (394). These developments first require a reformulation of the force equation given above so that the displacement-dependent antennal stiffness can be calculated:

$$K(X) = K_{INFINITY} - (\frac{Nz^2}{K_B T}) p_o(X)(1 - p_o(X))$$

Introducing a second, independent transducer population requires supplementing this stiffness equation with another term to include the effects of this additional ion channel population:

$$K(X) = K_{INFINITY} - \left(\frac{N_{s} z_{s}^{2}}{K_{B} T}\right) p_{OS}(X)\left(1 - p_{OS}(X)\right) - \left(\frac{N_{i} z_{i}^{2}}{K_{B} T}\right) p_{Oi}(X)\left(1 - p_{Oi}(X)\right)$$

where N_S , z_S and $p_{OS}(X)$ are the number of ion channels, the single channel gating force and the ion channel open probability at displacement X for the sensitive population (with N_i , z_i and $p_{Oi}(X)$ being the equivalent parameters for the insensitive population). The ion channel open probability equation also requires modification in order to account for the newly introduced transducer population:

$$\frac{p_{OS}(X) + p_{Oi}(X)}{2} = \frac{1}{2} \left(\frac{1}{1 + e^{-\frac{z_S(X - X_0)}{K_B T}}} + \frac{1}{1 + e^{-\frac{z_i(X - X_0)}{K_B T}}} \right)$$

where all terms are as defined above.

For all *Drosophila melanogaster* gating spring model analyses included throughout this thesis, this two state model of two independent transducer populations was utilised for displacements between \pm 5000nm. This provided information on the two apparent types of ion channel populations in the system, including the estimated population sizes and single channel gating forces, as well as $K_{INFINITY}$. The previously reported standard apparent antennal mass of 5ng was used for all *Drosophila melanogaster* analysis (394).

In total, 7 male flies from each *Drosophila melanogaster* line investigated (i.e. 91-S, 91-R, BL1283, BL 1675, w¹¹¹⁸xGAL4, J28 7:1xGAL4 and J28 8:1xGAL4) were included in the final force-step analysis. Statistical comparisons of K_{INFINITY}, K_{STEADY} and K_{GS} were made between before and after pymetrozine states using paired before and after t-tests (with a significance level of 0.05) in Sigmaplot. Post-hoc calculations of statistical power consistently showed statistical power estimates of over 90% for all comparisons made, except for the statistical comparison made between before and after pymetrozine states using paired before in this case was calculated to be less than 30%, which therefore indicates a higher likelihood of a type II error than would usually be recommended.

6.3. Results of *Drosophila melanogaster* pymetrozine injections with respect to vibrometry and electrophysiology

6.3.1. Effective stiffness and best frequency changes

Table 7 contains the statistically significant increases in best frequency seen for all lines tested, with the minimum increase observed being 200 Hz (with a maximum of approximately 400Hz). The free fluctuation plots for male *Drosophila melanogaster* shown in figure 25 are demonstrative of changes seen after pymetrozine exposure throughout all experiments – an increase in best frequency from between 100-300Hz to around 500Hz in association with a decrease in maximum antennal velocity amplitude in line with a statistically significant increase in effective stiffness (as shown in table 8 and figure 26).

Table 7. Median best frequency values for all Drosophila melanogaster lines tested comparing post-ringer and post-pymetrozine states (standard errors given in brackets) as well as P-values for statistical comparisons between the two states.

Drosophila	Best frequency median	median Best frequency median		
<i>melanogaster</i> line	after ringer (Hz)	after pymetrozine (Hz)	F-value	
91-S	210.48	603.94	P = 0.00002	
(n = 7)	(10.4)	(31.0)	F = 0.00002	
91-R	249.20	443.25	P = 0.000425	
(n = 7)	(12.3)	(26.5)	F = 0.000435	
BL1283	281.88	494.17	P = 0.000006	
(n = 7)	(18.3)	(15.3)	F = 0.000006	
BL 1675	239.33	544.45	P - 0.00001	
(n = 7)	(9.28)	(28.5)	F = 0.00001	
w ¹¹¹⁸ xGAL4	144.08	389.59	P = 0.000001	
(n = 10)	(4.46)	(19.7)	r = 0.000001	
J28 7:1 xGAL4	178.09	385.27	P = 0.000000	
(n = 8)	(5.94)	(12.75)	1 - 0.000009	
J28 8:1 xGAL4	164.42	370.62	P = 0.000034	
(n = 10)	(9.75)	(27.8)	r = 0.000034	


Figure 25. A) Free, unstimulated fluctuation data for a 91-S fly (left) before and after pymetrozine, with the equivalent states in a 91-R fly shown to the right. The frequency range for both flies is 51 – 3200Hz, with individual data points (in purple and grey for before and after pymetrozine injection states respectively) representing Fourier-transformed velocity amplitudes at each frequency contained within the transform range. Solid black lines show the velocity amplitude function fits for each recording.

B) Velocity amplitude function fits produced using median population values for each of the fit parameters both before (in purple) and after (in grey) pymetrozine for 91-S (left) and 91-R (right) lines.

Significant increases in the calculated effective stiffness were also seen for all *Drosophila melanogaster* lines (as is demonstrated in figure 26) – table 8 contains all effective stiffness estimates in both post ringer and post pymetrozine states.

Table 8. Median values of effective stiffness for all Drosophila melanogaster lines investigated comparing post-ringer and post-pymetrozine states (numbers in brackets are standard errors) as well as P-values for statistical comparisons between the two states

Drosophila	Effective stiffness median	ffective stiffness median after ringer (μN/m)Effective stiffness median after pymetrozine (μN/m)P-value		
<i>melanogaster</i> line	after ringer (µN/m)			
91-S	1.9	58.9	P =	
(n = 7)	(0.32)	(6.56)	0.000064	
91-R	1.4	19.6	P = 0.016	
(n = 7)	(0.13)	(6.87)		
BL1283	2.5	63.1	P =	
(n = 7)	(0.40)	(5.84)	0.000042	
BL 1675	3.5	48.5	P =	
(n = 7)	(0.83)	(3.57)	0.000035	
w ¹¹¹⁸ xGAL4	0.3	20.1	P = 0.002	
(n = 10)	(0.03)	(16.22)		
J28 7:1 xGAL4	0.7	15.5	P = 0.008	
(n = 8)	(0.07)	(3.19)		
J28 8:1 xGAL4	0.4	18.3	P = 0.016	
(n = 10)	(0.09)	(3.63)	1 - 0.010	



Figure 26. A) Effective stiffness values for all Drosophila melanogaster lines investigated comparing post-ringer and post-pymetrozine injection.

B) Best frequency values for different Drosophila melanogaster lines comparing postringer and post-pymetrozine injection.

For both A) and B), from left (also labelled in graph): 91-S, 91-R, BL1283, BL 1675, $w^{1118}xGAL4$, J28 7:1xGAL4 and J28 8:1xGAL4. Starred values indicated significant differences between groups (with resistance types being documented in section 6.2.3) and black dots correspond to the 5th and 95th percentiles.

6.3.2. Force step electrophysiology recordings

Full lists of median parameters obtained by fitting the two state model of two independent transducer populations to the force displacement data obtained after ringer injection (but prior to pymetrozine injection) are available in tables 29 and 30 in appendix G. Following pymetrozine injection, there was a severe ablation of auditory transduction in all *Drosophila melanogaster* lines tested (as is evident in figure 27). CAP responses to stimuli were no longer produced and changes in dynamic stiffness around the resting position became less pronounced, although not entirely eliminated for some lines.







Figure 27. Changes in dynamical stiffness and CAP amplitude in response to changes in antennal displacement to a maximum displacement of \pm 5000nm comparing postringer (in purple) and post-pymetrozine (in grey) states for each of the control and resistant Drosophila melanogaster lines investigated: A) 91-S, B) 91-R, C) BL1283, D) BL 1675, E), w¹¹¹⁸xGAL4, F) J28 7:1xGAL4 and G) J28 8:1xGAL4. Individual data points represent median values calculated at that displacement, with vertical black bars representing standard errors.

Significant changes to $K_{INFINITY}$ and K_{STEADY} , and thus by extension K_{GS} , were also identified, with significant decreases being calculated for all three of these parameters across all *Drosophila melanogaster* lines investigated (as shown in tables 9 and 10 as well as figure 28) – with the exception of the 91–S line, for which K_{GS} was not found to be statistically significantly different when comparing between the post ringer and post pymetrozine states.

Table 9. Median values for dynamic stiffness parameters after ringer injection and after pymetrozine injection for the following Drosophila melanogaster lines (standard errors given in brackets): 91-S, 91-R, BL 1283 and BL 1675. P-values for all statistical comparisons are also included.

	91 – S	91 – R	BL 1283	BL 1675
	(n = 7)	(n = 7)	(n = 7)	(n = 7)
KINFINITY after ringer	47	60	75	63
(µN/m)	(3.0)	(1.2)	(2.0)	(1.8)
K _{INFINITY} after pymetrozine (μN/m)	38 (4.4)	40 (1.2)	69 (3.1)	41 (1.8)
Significance value	P = 0.013	P = 0.00006	P = 0.002	P = 0.0001
K _{STEADY} after ringer	31	37	48	39
(µN/m)	(2.6)	(1.0)	(1.5)	(1.2)
K _{sτεady} after pymetrozine (μN/m)	21 (2.7)	26 (0.7)	46 (2.1)	26 (1.5)
Significance value	P = 0.001	P = 0.016	P = 0.013	P = 0.016
K _{GS} after ringer	16	24	25	23
(µN/m)	(0.7)	(0.5)	(0.6)	(0.7)
K _{GS} after pymetrozine (µN/m)	17 (1.9)	15 (0.6)	21 (1.3)	14 (0.7)
Significance value	P = 0.992	P = 0.0002	P = 0.004	P = 0.0001

Table 10. Median values for dynamic stiffness parameters after ringer injection and after pymetrozine injection for the following Drosophila melanogaster lines (standard errors given in brackets): w¹¹¹⁸xGAL4, J28 7:1xGAL4 and J28 8:1xGAL4. P-values for all statistical comparisons are also included.

	w ¹¹¹⁸ xGAL4	J28 7:1xGAL4	J28 8:1xGAL4
	(n = 7)	(n = 7)	(n = 7)
KINFINITY after ringer	58	61	59
(µN/m)	(0.9)	(1.5)	(1.6)
KINFINITY after	49	39	47
pymetrozine	(0.7)	(0,6)	(3.1)
(µN/m)	(0.7)	(0.0)	(0.1)
Significance value	P = 0.0001	P = 0.002	P = 0.000458
K _{STEADY} after ringer	38	40	40
(µN/m)	(0.6)	(1.2)	(0.7)
K _{STEADY} after	35	26	34
pymetrozine	(0.5)	(0.5)	(2.4)
(µN/m)	(010)	(0.0)	(=)
Significance value	P = 0.0001	P = 0.002	P = 0.016
K_{GS} after ringer	20	21	21
(µN/m)	(0.3)	(0.4)	(1.0)
K _{GS} after	14	12	12
pymetrozine	(0.5)	(0.3)	(1.8)
(µN/m)	(0.0)	(0.3)	(1.0)
Significance value	P = 0.00016	P = 0.002	P = 0.008





Figure 28. Estimated stiffness parameter values (including K_{INFINITY}, K_{STEADY} and K_{GS}) in both post-ringer and post-pymetrozine states for all Drosophila lines investigated (significant differences between states are starred, black dots correspond to the 5th and 95th percentiles): A) 91-S, B) 91-R, C) BL1283, D) BL 1675, E), w¹¹¹⁸xGAL4, F) J28 7:1xGAL4 and G) J28 8:1xGAL4.

6.4. Discussion

6.4.1. The effect of pymetrozine on *Drosophila melanogaster* auditory systems as measured by free fluctuations

Best frequency and effective stiffness estimates calculated from the free fluctuation fits for all *Drosophila melanogaster* lines tested in section 6 prior to pymetrozine exposure were in broad agreement with previous published values (274, 284, 310). In addition to this, all *Drosophila melanogaster* lines tested in section 6.3.1 showed significant increases in both best frequency and effective stiffness after pymetrozine injection compared to after ringer injection. The frequency values reached after pymetrozine injection were similar to those previously found in section 4 for susceptible *Drosophila melanogaster* from the wildtype CantonS line exposed to pymetrozine via oral ingestion (as shown in the free fluctuations provided as evidence of pymetrozine's effect on auditory function in appendices A - D); in particular, these CantonS results were similar to estimates of the 91-S and w¹¹¹⁸xGAL4 lines, which lacked insecticidal resistance and were thus used as controls for other lines which demonstrated some form of resistance.

The changes in best frequency and effective stiffness in pymetrozine injected insects are comparable to the differences previously reported when comparing control flies with auditory mutants (283); for example, $nompA^2$ and btv^{5Pl} and mutants had reported median best frequencies of approximately 406 and 528Hz, and median effective stiffnesses of approximately 35 and 60 µN/m, respectively. As such, pymetrozine injected *Drosophila melanogaster* from the 91-S, 91-R, BL1283 and BL1675 lines measured in this section match closely with btv^{5Pl} mutants, whilst the pymetrozine injected three UAS-based lines were to some extent similar to $nompA^2$ mutants.

Another ChO mutant, *tilb*², was reported to have a much greater best frequency and effective stiffness (approximately 750 Hz and 120 μ N/m respectively) than any of the *Drosophila melanogaster* lines tested here, and as such had an antennal receiver much more similar to a dead fly than an alive, pymetrozine exposed fly (283).

The three UAS-based lines (including the line used as a control) all had lower best frequency and effective stiffness values than the other insecticidal resistant lines tested before pymetrozine injection (though still with the expected range of values for *Drosophila melanogaster* (283, 284)), suggesting that this is the result of the UAS construct rather than any insecticidal resistance mechanism; after pymetrozine injection the three lines still demonstrated significant increases in both best frequency and effective stiffness.

6.4.2. The effect of pymetrozine on *Drosophila melanogaster* auditory systems as measured by force step stimulation electrophysiology

For all *Drosophila melanogaster* lines tested in section 6.3.2, regardless of insecticidal resistance status, pymetrozine exposure caused an elimination of CAP production and a reduction in the dynamical stiffness decreases seen in response to relatively small stimuli that were present after ringer injection (with the lack of CAP response to stimulation in agreement with that previously reported for pymetrozine exposed *Drosophila melanogaster* (27) as well as for ChO mutants such as *tilB*(412)).

The decrease in dynamical stiffness around the resting position seen in *Drosophila melanogaster* with fully functional ChO and mechanotransduction machinery appears to be somewhat maintained in several lines after pymetrozine injection. In all cases however this decrease is greatly reduced, indicating that the auditory nonlinearity (a hallmark of the active hearing process) has been mostly abolished by pymetrozine (282). It could also be possible that the remaining nonlinearity is not an auditory nonlinearity but is instead related to other processes, with non-auditory nonlinearities being previously reported for *NompC* null *Drosophila melanogaster* mutants for example (394).

The partial retention of some level of compliance is particularly interesting when comparing the 91–R DDT resistant line to its' control line, 91-S. Whilst susceptible 91–S flies appear to lose all compliance increases after exposure to pymetrozine, 91–R flies still show an apparent decrease in dynamical stiffness around the resting position (in figure 27). This could potentially indicate some level of cross-resistance to pymetrozine, but the significant reduction in dynamic stiffness changes (in addition to the complete ablation of CAP responses to stimulation) suggest that if any resistance is present in this line it is not sufficient to maintain wildtype levels of auditory function after pymetrozine exposure.

For all lines tested there were clear and significant reductions in K_{STEADY} and $K_{INFINITY}$ which led to a significant decrease in K_{GS} in all lines except the 91-S line (as demonstrated in table 9). Similar reductions in K_{STEADY} and $K_{INFINITY}$ have been reported for NompC null *Drosophila melanogaster* mutants as well as *Drosophila melanogaster* whose auditory neurons were ablated (394). As this decrease in K_{GS} was also significant for the 91–R resistant line (for which the 91–S line acts as a control), this could potentially indicate that there exists some difference in the molecular substrates of these two lines. The 91–S line does appear however to be completely susceptible to pymetrozine as there is a complete elimination of CAP response and compliance changes around the resting position following pymetrozine injection.

Whilst there was a clear and complete abolition of ChO mechanosensory function (as measured by LDV) in all *Drosophila melanogaster* lines tested using injection protocols, no lines were tested using oral ingestion exposure methods. Given that ingestion is the regular method of exposure, and that direct compound injection could be more efficacious than ingestion exposure procedures, future experiments investigating the impact of pymetrozine could utilise ingestion procedures for comparative purposes (26).

Changes in dynamic stiffness around the resting position for several *Drosophila melanogaster* lines (for example 91-S flies) were also to some extent noisy compared to other lines. Whilst the sample sizes for each group were sufficient to identify significant differences between stiffness parameters of interest, increasing the sample sizes of future investigations could help to reduce this noise level. Other insecticidal resistant *Drosophila melanogaster* lines are available for testing (including lines resistant to spinosad for example), and increasing the number of resistant mechanisms studied would not only provide information on potential occurrences of cross-resistance with pymetrozine but also offer potential insights into how insecticide resistance could affect auditory systems (465).

7. Mosquito vibrometry and electrophysiology

7.1. Introduction

For male mosquitoes (particularly those from each of the three species described in section 2.1, which are the focus of this thesis) to increase the probability of successfully identifying and locating conspecific females their auditory systems must remain tuned within specific frequency ranges (34). This process is made more complicated by the high level of background noise produced by mosquito swarm (76, 290). Female mosquitoes are also poor sound radiators and produce relatively faint emissions during flight (305).

As such the male mosquito antennal receiver must not only be selective in distinguishing what could potentially be a female of the same species that is available for courtship but must also be sensitive enough to differentiate the female in the first place. This requires an auditory system that is able to both inject large amounts of energy (should that be required) and maintain a constant frequency tuning (290).

Though these auditory requirements may be more stringent for male mosquitoes, a large burden is also placed on females from some species as well because of the nature of mosquito courtship rituals (363). In species where harmonic convergence forms an essential part of the courtship ritual (for example, *Ae. aegypti, An. gambiae* and *Cx. quinquefasciatus*), both male and female mosquitoes require significant levels of auditory function in order to match their WBFs (34, 352).

These three mosquito species are particularly relevant for disease control programs because of their status as major vectors of multiple diseases (as discussed in sections 2.1 and 2.3) and will be the focus of all experiments conducted in section 7.

The mosquito antennal system is thus the result of trade-offs between these multiple requirements. Previous studies have used laser vibrometry to investigate the frequency selectivity and tuning that underlies these systems, particularly with regards to copulatory activities (289, 354, 358). These reports have not however necessarily examined potential mechanical signatures that arise from the direct gating of mechanotransducers, with mechanotransduction in mosquito auditory systems not fully explored (as discussed in section 2.8, although some aspects of these signatures, such as auditory nonlinearities, have been studied) (276, 287).

Such signatures of transducer gating (as described in section 6.1) have been extensively examined for *Drosophila melanogaster* however, with the application of gating spring models (as discussed in section 6.2.8) having been reported to be sufficient to describe this simple, one dimensional system (285).

These models have enabled the estimation of transducer populations and gating forces, alongside antennal stiffness projections (282). Applying these models therefore to mosquito species could therefore provide a wealth of information about mosquito auditory transduction processes, though the inherent additional complexity of the mosquito antennal system (for both males and females) means that this application needs to be done carefully and any conclusions that are made require caution.

Previous reports have also estimated energy gain in *Drosophila melanogaster* lines, which allowed for comparisons of energy injection between controls and ChO mutants (283). Applying these principles to male and female mosquitoes from different species (by comparing the free fluctuations of the flagellum of individual mosquitoes in active and passive states) would enable estimation of the extent of auditory sexual dimorphisms in these species.

Displacement gain calculations could provide insight into these sexual dimorphisms as well, with previous calculations being made in mosquito and *Drosophila* species using either white noise or pure tone stimulation (34, 272, 284, 289). Pure tone stimulation at specific frequencies tends to be used for displacement gain calculations because pure tones are potentially more biologically relevant than white noise stimulation (which covers a far larger range of frequencies but at a reduced intensity), with pure tones having the potential to model mosquito wing beats.

White noise stimulation however ensures as broad a stimulus as possible however such that a large range of frequencies can be investigated. Therefore given the nature of the two stimuli, comparing the results of pure tone stimulation with white noise stimulation could help to explore the extent of frequency selectivity present in mosquito auditory systems.

Whilst these investigations would provide quantifiable information regarding the differences between the auditory systems of male and female mosquitoes, using the already established vibrometry and electrophysiology protocols (detailed in section 6.2.8) could also allow for hypotheses to be made about the systems of frequency selectivity within the mosquito antenna.

For example, one possible method mosquitoes could use to maintain a constant frequency tuning is to minimise dynamic stiffness reductions that result from direct mechanotransducer gating - this would then decrease the extent of nonlinearity within the auditory system and reduce changes to the flagellar best frequency associated with transducer gating (466). This can be investigated by comparing the changes in the K_{PEAK} and K_{STEADY} (as defined in section 6.2.8) in response to decreasing antennal displacements and estimating the dynamic stiffness of the system in the absences of the steady state stiffness.

In addition to this, utilising extensions to the gating spring model and the ion channel open probability function (also as described in section 6.2.8) would allow for exploration of the conceptual existence of multiple transducer populations (which have been identified for *Drosophila melanogaster* (394)) in mosquito species.

Finally, spontaneous oscillations of the flagellum remain unique to males in the absence of compound injection - these oscillations, which can have displacement amplitudes on the order of hundreds of micrometres, may therefore play a male-specific role during mosquito courtship rituals but the mechanisms which control these oscillations are not fully understood (289, 308). The recent identification of efferent feedback loops within the auditory system of male *Cx. quinquefasciatus* mosquitoes could suggest the potential involvement of such loops in regulating these oscillations (314).

Injecting compounds which have been reported to sever such efferent feedback networks in other organisms into mosquitoes should therefore reveal the underlying mosquito auditory system (310, 467). This would enable the potential presence of such systems in both male and female mosquitoes from each of the different species investigated to be examined, as well as exploring the necessary requirements for spontaneous oscillations to occur. Sexual dimorphisms between the responses of male and female mosquitoes can also be analysed so that differences between the sexes in terms of auditory function can be described.

7.2. Materials and methods

7.2.1. Mosquito rearing

Unless otherwise noted, all mosquito species were reared in an identical manner as described in section 5: all *Ae. aegypti, An. gambiae* (Kisumu) and *Cx. quinquefasciatus* (Muheza) mosquitoes used for experiments were provided by Shahida Begum from the London School of Hygiene and Tropical Medicine. All mosquitoes were reared using a 12 hr: 12hr LD cycle at 26°C and 75% relative humidity and were fed using a 10% glucose mixture.

7.2.2 Compound preparation

Ringer: Preparation of the ringer solution was identical to that described in section 6.2.2 for injection experiments in *Drosophila melanogaster*.

Pymetrozine: Preparation of pymetrozine was identical to that described in section 4.2.2 for feeding experiments in *Drosophila melanogaster*.

TTX: A 100 μ M stock TTX solution was obtained from Professor Jonathan Ashmore (University College London, London). This was diluted to 5 μ M in the ringer solution previously described, with this concentration being used in all injection experiments.

TeNT: 166µl of the aforementioned ringer solution was injected into 25µg of a powdered version of the compound to obtain a 1 µM solution that could be used as a stock. 20 µl of this solution was then further mixed with 480 µl of ringer solution to create a 20nM solution that was used for all injection experiments.

7.2.3. Compound exposure method – injection

Injection experiments in mosquitoes entailed a similar protocol as described in section 6.2.3 for *Drosophila melanogaster* injection experiments: first micro-capillaries (Harvard Apparatus, Cambridge) were pulled in a micropipette puller (Sutter Instruments, CA). These sharpened capillaries were then filled with either a control ringer solution (consisting of deionised water that had been diluted in ringer to the desired concentration to match the compound of interest) or a ringer solution containing the compound of interest (pymetrozine, TTX or TeNT). Upon injection of either control or compound, the solution was injected into the mosquito in such a way as to flood the body and so increase the likelihood that the compound was able to circulate to the head and as such reach the JO.

All experiments were again divided into three stages – before injection, after ringer and after compound of interest. This allowed for investigation of potential significant differences between both pre- and post-ringer injection states as well as post-ringer and post-compound of interest injection states. Similar to *Drosophila melanogaster*, reductions in maximum CAP responses were observed after ringer solution injection but in general the mechanics of the system did not significantly change (appendix F contains representative comparisons for *Ae. aegypti* females and more discussion on this topic). As such, all comparisons made for injection experiments compared post-ringer and post-compound of interest states to simplify procedures.

7.2.4. Mosquito vibrometry preparation

Whilst the major principles of mounting were the same for both mosquitoes and *Drosophila melanogaster*, the increased size (with regards to both the entire body and more specifically the antennal system) of individual mosquitoes from each of the mosquito species tested resulted in minor modifications being made to the mounting procedure to ensure that whilst the flagellum remained free to move the rest of the mosquito body was firmly held in place. This meant that mosquitoes were glued dorsal-side down and as such glue was applied to each leg individually to prevent interference with the measurements.

Once the mosquito was mounted, the experimental setup was identical to that described in section 6.2.7 for *Drosophila melanogaster*, with the exception that the laser point was no longer focused on the antennal tip – instead the bottom of the third flagellomere from the tip was used as the focal point for female mosquitoes from all three species, whilst the corresponding point for males was the bottom of their second flagellomere from the tip.

This change in laser focus point was introduced to minimise disturbances in free fluctuations noted when focussing on the antennal tip, whilst also attempting to maximise stimulated displacements and providing as stable a focal point as possible. The frequency range of fluctuation measurements was also increased to between 1 and 10000Hz (with a 1.5625Hz resolution), in order to investigate the possibility of high frequency harmonics becoming a factor in any recording.

Male mosquitoes that were considered to be spontaneously oscillating at the beginning of an experiment were not included in any of the analyses – this was judged on the basis of the maximum flagellar displacements identified during free fluctuation measurements, with males demonstrating displacements greater than 1µm being considered to be spontaneously oscillating. No female mosquitoes were judged to demonstrate spontaneous oscillations in the absence of solution injection.

Electrophysiological experiments utilising force step stimuli were completed in the same manner as for *Drosophila melanogaster* and as such are as described in section 6.2.8, up to and including the type and duration of stimuli.

7.2.5. CO₂ sedation experiments – free fluctuation measurements

For CO₂ sedation experiments the mounting procedure was identical to regular experiments but the mounted mosquito was instead put inside a rectangular steel chamber (6 x 6 x 2.5cm), which was held in place opposite the vibrometer by a micromanipulator. A small glass window in the one of the sides of the chamber was positioned perpendicular to the vibrometer and the mosquito was placed inside the chamber so as to still allow for free fluctuation measurements to be taken.

A cavity under the chamber allowed for CO_2 to enter the chamber and sedate the mosquito, which was kept as close to the cavity as possible. A flow regulator (Flowbuddy, Flystuff) enabled a constant flow rate of 3 l/min to be maintained. A plastic case (3.5 x 2.5 x 2.5cm) was placed surrounding the mosquito in order to prevent rapid losses of CO_2 once the gas flow had been halted.

After an initial unstimulated fluctuation recording had been taken to assess the health of the antennal system pre-sedation, CO_2 was allowed to flow into the chamber for one minute, with fluctuation measurements being taken in a constant loop to judge the state of the auditory system. At that point, the flow was stopped and a free fluctuation of the passive state of the flagellum was recorded. After 5 minutes of recovery time had elapsed, another free fluctuation was recorded so that the level of recovery could be analysed.

Mosquitoes which did not exhibit signs of recovery of auditory function were excluded from the final analysis; this recovery was judged by assessing the maximum antennal velocity amplitude reached after cessation of CO_2 flow as well as the best frequency of the flagellum, with shifts of more than 10% in either of these values compared to the presedation state resulting in the exclusion of the mosquito from the data set.

7.2.6. CO₂ sedation experiments – electrophysiological recordings

Force step electrophysiological experiments (as described in section 6.2.8) conducted whilst a mosquito was sedated by CO_2 required the insertion of charging and recording electrodes into the mosquito and thus prevented the use of the plastic case. They also required a constant flow of CO_2 within the chamber to ensure that the mosquito did not awaken whilst the recordings were being taken. As the flow could cause disruptions to laser measurements of the smallest stimuli this required a significantly greater than normal amount of force steps to be recorded for averaging purposes, with a balance being maintained between a completely sedated state and allowing the mosquito time to recover from the effects of sedation before the auditory system became permanently damaged.

In practice this entailed regularly halting the gas flow within the chamber to allow the mosquito to recover, confirming the apparent health of the auditory system by taking a free fluctuation recording so that comparisons could be made between the current state and the original baseline and confirming the CAP amplitude response to stimulation in the active state before sedating the mosquito again and proceeding with the step recordings. A complete set of force steps was recorded in the active state prior to any sedation attempts to provide a control for each mosquito recorded from.

7.2.7. Free fluctuation fitting procedure

Unstimulated, free fluctuations of the mosquito flagellum were analysed using a similar procedure to that described for *Drosophila melanogaster* in section 6.2.6: flagellar free fluctuations were recorded and a fast Fourier transform of the antennal velocity amplitudes was calculated for frequencies between 1 to 2000Hz for all mosquito species. Recording measurements within the frequency range below 101Hz contain a significant level of noise and were excluded from these analyses. These velocity amplitudes were then fitted using the same velocity amplitude fit function as described for *Drosophila melanogaster*, from which estimates of the parameters F_0/m , ω_0 and Q for mosquito species were obtained.

52 Ae. aegypti females, 39 Ae. aegypti males, 42 An. gambiae females, 35 An. gambiae males, 37 Cx. quinquefasciatus females and 43 Cx. quinquefasciatus males were included in the analysis of non-sedated mosquitoes; 35 Ae. aegypti females, 29 Ae. aegypti males, 33 An. gambiae females, 24 An. gambiae males, 28 Cx. quinquefasciatus females and 31 Cx. quinquefasciatus males were included in the analysis of sedated mosquitoes.

7.2.8. Energy gain calculations

In order to estimate the level of energy gain in the auditory system, ratios of the total fluctuation power in the active and passive auditory systems of mosquitoes were calculated. The procedure builds upon a previously published approach (282, 283) and thus follows the following rationale: the passive energy of the system can be assumed to be equal to the sum of the squared Fourier displacement amplitudes in the passive state, $\langle x_p^2 \rangle$, multiplied by the spring constant k_p as well as a proportionality constant α :

$$E_p = \alpha k_p \left\langle x_p^2 \right\rangle$$

The passive spring constant itself is equal to the product of the antennal apparent mass, *m*, and the square of the natural frequency of the system, ω_n^2 :

$$k_p = m\omega_p^2$$
,

Therefore it can be assumed that:

$$E_{p} = \alpha k_{p} \left\langle x_{p}^{2} \right\rangle = \alpha m \omega_{p}^{2} \left\langle x_{p}^{2} \right\rangle$$

Applying the same procedure to the active state provides an equivalent equation for that state:

$$E_{a} = \alpha k_{a} \left\langle x_{a}^{2} \right\rangle = \alpha m \omega_{a}^{2} \left\langle x_{a}^{2} \right\rangle$$

Energy gain is defined here as

$$Energy gain = \frac{E_a - E_p}{E_p}$$

Thus:

Energy gain =
$$\frac{\alpha m \omega_a^2 \langle x_a^2 \rangle - \alpha m \omega_p^2 \langle x_p^2 \rangle}{\alpha m \omega_p^2 \langle x_p^2 \rangle}.$$

Assuming that the proportionality constant α is the same in both active and passive states, this can be finally simplified to:

Energy gain =
$$\frac{\omega_a^2 \langle x_a^2 \rangle}{\omega_p^2 \langle x_p^2 \rangle} - 1$$

In order to calculate energy gain therefore it is necessary to obtain the best frequency estimates, as well as the sum of the squared Fourier displacement amplitudes, in both the active and passive states - passive states of the auditory system are estimated by sedating the insect using CO₂. The best frequency estimates are obtained by fitting the velocity amplitude fit function described in section 6.2.6 to the fast Fourier transforms of the antennal velocity amplitudes in the active and passive states whilst the sum of the squared Fourier displacement amplitudes is estimated by integrating the displacement squared fit function also described in section 6.2.6 over all angular frequency values, i.e.

$$\left\langle x_{i}^{2}\right\rangle = \int_{0}^{\infty} x_{i}^{2}(\omega) d\omega$$

In total, 35 *Ae. aegypti* females, 29 *Ae. aegypti* males, 33 *An. gambiae* females, 24 *An. gambiae* males, 28 *Cx. quinquefasciatus* females and 31 *Cx. quinquefasciatus* males were included in the final analysis (complete lists of sample sizes used in every energy gain calculation made for all mosquito species tested in section 7 are given in the tables included in appendix H).

Tests for significant differences between male and female mosquitoes were made using Wilcoxon signed rank tests with a significance level of 0.05 in Sigmaplot. Post-hoc statistical power calculations gave estimates of below 50% power for comparisons made between male and female mosquitoes for both *Ae. aegypti* and *An. gambiae* species. This indicates that the comparisons were underpowered, with an increased likelihood therefore of a type II error occurring.

7.2.9. Mosquito apparent antennal mass estimations

All *Drosophila melanogaster* step recordings analysed in section 6 used an apparent antennal mass estimated from CO_2 sedation experiment results (283). No such apparent antennal mass has previously been reported for mosquito species however and as such it was necessary to calculate new values, using the same method as for *Drosophila melanogaster*. Thus individual mosquitoes were sedated, with free, unstimulated fluctuations of the mosquito antenna being recorded whilst the mosquito was sedated. A harmonic oscillator model was then fitted to the resulting curve, which enabled extraction of the apparent antennal mass.

This follows from the procedure previously reported in the literature (283): assuming that the mosquito auditory system is in a passive state (i.e. there is no active injection of energy into the auditory system) then the fluctuations of the mosquito flagellum should obey the Equipartition theorem (as in section 6.2.6):

$$\frac{1}{2}K\langle x^2\rangle = \frac{1}{2}K_BT$$

where *K* is the effective stiffness of the oscillator, $\langle x^2 \rangle$ is the sum of the squared Fourier displacement amplitudes, *K*_B is the Boltzmann constant (1.38 × 10⁻²³ J/K) and *T* is the absolute temperature (estimated at approximately 295 K).

Assuming that *K* is equal to the spring constant of the oscillator system, K_s (i.e. $K = K_s$), whilst the mosquito is sedated then the relationship between the spring constant, apparent antennal mass *m* and the natural frequency of the system ω_0 can be modified accordingly:

$$K_{S} = m\omega_{0}^{2}$$
$$\Rightarrow m = \frac{K_{B}T}{\omega_{0}^{2} \langle x^{2} \rangle}$$

Therefore, in order to obtain apparent mosquito antennal mass estimates the natural frequency of the oscillator system and the sum of the squared Fourier displacement amplitudes are required.

Both of these values can be estimated using methods described in section 7.2.8 which detailed the energy gain calculation procedure, with the natural frequency of the system being approximated from the velocity amplitude fit function and $\langle x^2 \rangle$ following from

$$\left\langle x_{i}^{2}\right\rangle = \int_{0}^{\infty} x_{i}^{2}(\omega)d\omega$$

It was assumed that the apparent antennal mass remained constant regardless of whether the mosquito auditory system was in an active or a passive state (an assumption reported to be reasonable for *Drosophila melanogaster* (283)).

For many mosquitoes two passive state fluctuations were taken; one before and one after pymetrozine. No significant differences in the apparent antennal mass estimates were found between the two states using paired before and after t-tests with a significance level of 0.05 in Sigmaplot, and tests for normality found the data to be normally distributed. In order to maximise the size of the dataset therefore, a two-state mixed effects model was fitted in R using the lme4 package to account for the fact that not all measurements were independent of each other - this allowed for estimation of means and standard errors (468).

In total, 56 measurements from *Ae. aegypti* females (35 before pymetrozine exposure, 21 after), 45 measurements from *Ae. aegypti* males (30 before pymetrozine exposure, 15 after), 50 measurements from *An. gambiae* females (33 before pymetrozine, 17 after), 31 measurements from *An. gambiae* males (22 before pymetrozine exposure, 9 after), 50 measurements from *Cx. quinquefasciatus* females (29 before pymetrozine exposure, 21 after) and 54 measurements from *Cx. quinquefasciatus* males (33 before pymetrozine exposure, 21 after) and 54 measurements from *Cx. quinquefasciatus* males (33 before pymetrozine pymetrozine exposure, 21 after) were included in the final analysis.

Tests for statistically significant differences in the apparent antennal mass of conspecific male and female mosquitoes were made using two-tailed t-tests with a significance level of 0.05 in Sigmaplot. Post-hoc calculations of statistical power gave estimates of over 99% power for comparisons made for *An. gambiae* and *Cx. quinquefasciatus* mosquitoes, but below 50% for *Ae. aegypti*. This is a lower statistical power than would usually be recommended and would lead to an increased likelihood of type II errors.

7.2.10. Step recording analysis

As this was the first thorough examination of mechanotransduction in the mosquito auditory system, unless otherwise stated the original two state model of a single transducer population (as detailed in section 6.2.8) was utilised throughout. This fit provided estimates of the number of ion channels, the channel gating force, K_{INFINITY}, K_{STEADY} and K_{GS}.

The fitting of mosquito force-displacement data to the two state model of two independent populations (as discussed in section 7.3.3.4) was conducted as described for *Drosophila melanogaster* in section 6.2.8. Statistical comparisons between different model fits were made by calculating the AICc by fitting both models for either individual mosquitoes using Sigmaplot and selecting the fit with the lesser value or to the median dynamical stiffness values for different mosquito groups and selecting the lesser value.

Maximum displacements of ±2000nm were selected both because CAP saturation points were reached at these values for both sexes and to avoid the increased level of noise in the dynamic stiffness of the system observed at greater displacements, which is potentially associated with non-transduction related nonlinearities. The analytical process was identical to that of *Drosophila melanogaster* lines tested in section 6, except for an extra round of run averaging applied to the effective force values for male mosquitoes to counteract the increased noise observed in their force-displacement data sets.

The extent of nonlinearity has been reported to be a key measure of transduction introduced nonlinearities within the *Drosophila melanogaster* auditory system, and is significantly involved in modifying the best frequency of the antennal receiver in response to changes in stimulus intensity (456). An estimate of the extent of auditory nonlinearity was calculated for all mosquitoes tested using a previously reported formula (from (285)):

Extent of nonlinearity =
$$\frac{Nz^2}{(4K_BT * K_{INFINITY})}$$

where N, z, K_B , T and $K_{INFINITY}$ are as defined in section 6.2.8.

In total, 21 *Ae. aegypti* females, 18 *Ae. aegypti* males, 18 *An. gambiae* females, 15 *An. gambiae* males, 17 *Cx. quinquefasciatus* females and 15 *Cx. quinquefasciatus* males were included in the final analysis. Statistical comparisons between different mosquito species were made using ANOVA on ranks tests in Sigmaplot. Statistical tests to determine whether dynamical stiffness estimates were non-negative utilised one sided t-tests with a significance level of 0.05 in Sigmaplot.

7.2.11. White noise stimulus experiments

Male and female mosquitoes between 3 and 7 days old were mounted as previously described in section 7.2.4 and then, via charging electrode insertion, had their electrostatic potential raised to -20V above ground. The force step stimulation programme described previously was then utilised to provide an estimate of the mosquito's auditory function and to calibrate the maximum displacement of the flagellum to approximately ±25000nm. The programme also enabled the calculation of a proportionality coefficient for conversion between different units of force (in this case, volts and Newtons).

A white noise stimulus, programmed in PSV 9.1 (Polytec software, Polytec), was then provided between 1 and 10000Hz, with attenuation to this stimulus provided by an external attenuation system (Electronics workshop, University of Cologne). At first a maximum attenuation of 80dB was applied, which was then lifted in 5db steps until 0dB was reached. At each step free fluctuations of the flagellar response to stimulation were taken, with a final, unstimulated fluctuation being taken at the end of the experiment to help assess whether the maximum force stimulus had damaged the system and caused a loss of antennal function.

The white noise stimulus itself was also recorded at each step, which allowed for calculation of the ratio of the antennal velocity amplitude and stimulus intensity at each frequency and the fitting of a harmonic oscillator model to the resulting data (with no assumptions made during this fitting procedure as to whether the flagellum itself is best modelled as a harmonic oscillator) (283); this enabled calculation of the mechanical sensitivity at each level of stimulus attenuation.

These mechanical sensitivity values for each stimulus attenuation were then fitted using a three parameter-sigmoidal function in the software package Sigmaplot, with all fits that were accepted having R^2 values greater than 0.9. This enabled the estimation of the displacement gain by comparing the values for maximum and minimum attenuations. Figure 29 shows a typical displacement gain plot as well as the sigmoid function fit; for reference the female mosquito used for the figure had a calculated displacement gain of 3.10.

In total, 7 *Ae. aegypti* females, 7 *Ae. aegypti* males, 9 *An. gambiae* females, 7 *An. gambiae* males, 13 *Cx. quinquefasciatus* females and 13 *Cx. quinquefasciatus* males were included in the final analysis. Statistical comparisons were made using Wilcoxon signed rank tests with a significance level of 0.05 in Sigmaplot.



Figure 29. Changes in mechanical sensitivity in response to increases in white noise stimulus intensity for an individual Cx. quinquefasciatus female. Each point represents the mechanical sensitivity at an individual attenuation of the stimulus and the black line represents the best fit of a sigmoid function.

7.2.12. Pure tone sine stimulus experiments

Male and female mosquitoes between 4 and 8 days old were prepared in an identical manner as described for the white noise and force step stimulation experiments (in section 7.2.4), including utilising the force step stimulation protocol to estimate the relevant proportionality coefficient for conversion between different units of force. A recording electrode was also inserted into the mosquito in order to record CAP responses. After this, pure tone sine wave stimuli were used to stimulate the antenna. The stimuli ranged between 15 and 695Hz, in 10Hz intervals. This range was chosen to completely encompass all significant mechanical and nerve responses seen in test run-throughs of the set-up.

Mechanical and nerve responses at higher fundamentals of WBF estimates were found to be negligible when compared to the responses within the above frequency ranges and so were not included in the analysis. At every frequency included, the stimulus lasted continuously for 2.5 seconds before stopping for a further 2.5 seconds; this pattern was repeated 5 times for each frequency tested. After the conclusion of the pure tone stimulation process, a free fluctuation recording was taken to judge the final auditory capabilities of the flagellum.

Figure 30 part A displays the mosquito flagellar displacement in response to pure tone stimuli – there is a short period of constantly increasing displacements before a maximum displacement amplitude is reached. By fitting a sine wave function to a steady segment of the displacement response (after having first applied a DC remove to the data in order to centre the response around the resting position) an estimate of the peak flagellar displacement at each stimulus frequency can be obtained. Applying the same process to the stimulus itself at each frequency tested enabled a ratio of flagellar displacement to stimulus force to be calculated for all frequency values.

Plotting this estimated mechanical sensitivity data over frequency produces figures like that shown in Figure 30 part B. By fitting a Gaussian function to this data set an estimate of the maximum and minimum values can be obtained, with the point of maximum mechanical sensitivity in each mosquito being defined as the best frequency of the flagellum for that individual. Calculating the ratio of the maximum and minimum values obtained from the fit enables an estimation of the displacement gain (using pure tone stimulation) to be made in individual mosquitoes, which can then be aggregated together into groups of interest. The Gaussian function itself is not assumed to model the mosquito flagellum and is only used for estimation purposes.

Crosstalk between the stimulus and the recording electrode set up can lead to reflections of the stimulus amplitude in the nerve response, creating artifacts. These artifacts can cause problems during data analysis by artificially distorting the apparent nerve responses, though fortunately they cannot by themselves provoke nerve responses (285). Crosstalk can be immediately identified in nerve recordings because there is no time delay associated with the phenomenon and as such artifacts become immediately visible in both the stimulus and nerve channels.

One way of counteracting this issue is by taking advantage of the known phase of the artifact, as well as its' distinguishability from the real nerve response because of the lack of a time delay. Assuming that the crosstalk present in the system results in an overlaying of a copy of stimulus into the nerve response then by subtracting this artifact (after accounting for a change of scale) the real data should be revealed.

In order to achieve this, stimulus and nerve data were extracted from spike2 data files and exported to Matlab with only the steady state of the nerve being used for this analysis (i.e. after the initial onset had reached a constant state). After applying a DC remove to both sets of data, the stimulus at each frequency was subtracted from the corresponding nerve response - in order to account for a change in scale between the two channels the stimulus data was first multiplied by a constant which was selected such that the area under the curve of the remaining nerve response would be minimised.

In practice this meant allowing the constant to increase between 0 and 2 in steps of 0.01, subtracting the product of the constant and the stimulus from the nerve data and then calculating the resulting area under the curve. The nerve response at each frequency in the absence of crosstalk could then be estimated from the power spectrum of the residual nerve response, with the median best frequency of the nerve being defined as the frequency at which the nerve response magnitude was greatest.

In total, 8 *Ae. aegypti* females, 10 *Ae. aegypti* males, 7 *An. gambiae* females, 7 *An. gambiae* males, 8 *Cx. quinquefasciatus* females and 8 *Cx. quinquefasciatus* males were included in the final analysis. All statistical comparisons were made using Wilcoxon signed rank tests with a significance level of 0.05 in Sigmaplot.



Figure 30. A) Flagellar displacement (top) and CAP amplitude (middle) in response to the corresponding pure tone stimuli (bottom) for Aedes aegypti females - the left pure tone stimulus frequency is set at 185Hz whilst the pure tone stimulus to the right has a frequency of 445Hz.

B) A Gaussian fit (solid black line) to individual sensitivity data points, also for an Ae. aegypti female.

7.2.13. TTX and TeNT injection series

Male and female mosquitoes between 3 and 8 days old were mounted as described in section 7.2.4 before a reference electrode was inserted in order to ground the insect and a free fluctuation of their baseline auditory state was recorded. A ringer control solution was then injected into the thorax of the insect so as to displace the original biological fluids contained within the mosquito. Free fluctuations of the antenna were then taken for 10 minutes at 1-2 minutes intervals in order to assess any changes in auditory function.

Mosquitoes that were judged to be in a suitable condition after this time had elapsed were then injected with either TTX (5µM concentration) or TeNT (20nM concentration). Free fluctuations were then taken every 1-2 minutes over either the next 10 or 25 minutes (for TTX and TeNT injections respectively). More time was allowed for TeNT injections as preliminary investigations suggested that the mechanism of action of the compound, along with the lower concentration used, necessitated a longer wait before potential changes to antennal auditory function could be seen.

After the specified period of time had elapsed, the mosquito was sedated so that a measurement of its passive system could be taken. The mosquito was then allowed to recover, with recovery being assessed by free fluctuation recordings – this sedation process followed the protocol given for CO₂ sedation experiments in section 7.2.5. After the mosquito had been given five minutes to recover from sedation, pymetrozine (10nM) was injected into the thorax. After this final injection was completed, free fluctuations were once again taken at 2 minute intervals for the following 10 minutes. The mosquito was sedated again using CO₂ and a free fluctuation of the passive system was taken. The mosquito was allowed time to recover before a last fluctuation was recorded. After the conclusion of each TeNT experiment, each mosquito was disposed of according to the departmental protocols for toxic waste disposal.

Energy gain calculations were made utilising the same procedure as described in section 7.2.8, with the last free fluctuation recorded for each injection state serving as a baseline for the most stable system state recorded for each injection state. Pymetrozine energy gains were estimated using the second, post-pymetrozine sedated free fluctuation data whilst all other injection states used the first, pre-pymetrozine sedated free fluctuation data as a baseline.

The final analysis included 10 TTX and 11 TeNT injections for *Ae. aegypti* females, 10 TTX and 11 TeNT injections for *Ae. aegypti* males, 12 TTX and 12 TeNT injections for *An. gambiae* females, 7 TTX and 8 TeNT injections for *An. gambiae* males, 14 TTX and 15 TeNT injections for *Cx. quinquefasciatus* females and 15 TTX and 14 TeNT injections for *Cx. quinquefasciatus* males. As all injection experiments (regardless of whether TTX or TeNT was injected) included both ringer and pymetrozine injection, this meant that in total 21 *Ae. aegypti* females, 21 *Ae. aegypti* males, 24 *An. gambiae* females, 15 *An. gambiae* males, 29 *Cx. quinquefasciatus* females and 29 *Cx. quinquefasciatus* males were included in the final analysis of energy gain estimates following exposure to either ringer or pymetrozine.

Tests for significant differences between injection states within a specific mosquito group were done using paired before and after t-tests with a significance level of 0.05 in Sigmaplot. Statistical tests to determine whether energy gain estimates following pymetrozine injection were greater than zero utilised one sided t-tests with a significance level of 0.05. Post-hoc statistical power calculations found that all comparisons made between different injection states for *Ae. aegypti* and *An. gambiae* females had a power <50%, meaning that these statistical comparisons should be considered underpowered.

7.3. Results

7.3.1. Mosquito free fluctuations - CO₂ sedation and energy gain

Table 11 contains median estimates of the flagellar best frequency before and after sedation for male and female *Ae. aegypti*, *An. gambiae* and *Cx. quinquefasciatus*, as well as significance levels for all statistical comparisons made. Sedating female and male *An. gambiae* as well as *Cx. quinquefasciatus* males caused the best frequency of the system to shift; in the females this led to an increase of over 100Hz, whilst for both groups of males there was a significant decrease by approximately the same amount. Male and female *Ae. aegypti* however, along with *Cx. quinquefasciatus* females, did not show shifts in best frequency on this level (with the maximum shift identified for these mosquitoes being 15Hz for female *Cx. quinquefasciatus*).

Examples of these best frequency changes are given in figure 31; whilst the *Ae. aegypti* female shown in part A maintains a constant best frequency estimate across both active and passive states, the sedated *An. gambiae* female in part B show an increase in best frequency of approximately 100Hz when compared to the active state. It also seems that as a result of this shift in frequency the passive system is partially recognisable in the free fluctuations of the active system. The *Cx. quinquefasciatus* male example shown in part C also demonstrates a shift in best frequency of approximately 100Hz, though in this case this is a decrease rather than an increase. The passive system of this male is also observable whilst the mosquito is not sedated.

Table 11. Median values of flagellar best frequency for male and female mosquitoes from each species investigated comparing post-ringer and post-sedation states (numbers in brackets are standard errors) as well as P-values for statistical comparisons between the two states.

Mosquito species/ sex	Best frequency after ringer (Hz)	Best frequency after sedation (Hz)	P-value	
Ae. aegypti females	198.5	207.5	P <0.01	
(n = 36/ 35/ 36)	(1.42)	(3.99)	F \$0.01	
Ae. aegypti males	290.0	297.99	D = 0.706	
(n = 25/ 30/ 25)	(11.91)	(11.90)	P = 0.706	
<i>An. gambiae</i> females (n = 29/ 33/ 26)	215.0 (3.95)	325 (7.44)	P <0.001	
<i>An. gambiae</i> males (n = 20/ 22/ 18)	352.0 (13.01)	283.7 (5.70)	P <0.001	
Cx. quinquefasciatus females (n = 29/ 29/ 27)	221.6 (5.16)	206.5 (3.26)	P <0.01	
Cx. quinquefasciatus males (n = 28/ 33/ 35)	365.6 (6.88)	274.1 (10.81)	P <0.001	



Figure 31. Free, unstimulated fluctuations both before and during sedation over a frequency range of 100 to 2000Hz for A) an Ae. aegypti female, B) an An. gambiae female and C) a Cx. quinquefasciatus male. Individual data points represent Fourier-transformed velocity amplitudes whilst solid black lines show the velocity amplitude function fits for each recording.

Table 12 contains median values for the three parameters obtained from the velocity amplitude fits for free fluctuations (as detailed in section 7.2.7) as well as median estimates for the level of energy gain in all three mosquito species: significant differences in terms of the energy gain estimates were calculated between male and female *Cx. quinquefasciatus* (p<0.005) but not for either *Ae. aegypti* or *An. gambiae* (p>0.05 in both cases). Energy gain values for all three species are also shown in figure 32.
Table 12. Median velocity fit parameters and energy gain calculated for each mosquito species and sex tested in both the active and sedated states (standard errors are given in brackets; the number of mosquitoes used to calculate the median energy gain value for each group is equal to the size of the relevant passive state group).

	<i>Ae.</i> <i>aegypti</i> females	<i>Ae.</i> <i>aegypti</i> males	<i>An.</i> gambiae females	<i>An.</i> gambiae males	Cx. quinque- fasciatus females	Cx. quinque- fasciatus males
ACTIVE						
STATE						
Sample size	52	39	42	35	37	43
E0/m	6.11 x10 ⁻⁴	6.87 x10⁻⁴	7.71 x10 ⁻⁴	5.50 x10⁻⁴	7.12 x10 ⁻⁴	5.70 x10 ⁻⁴
FO/III	(1.8 x10 ⁻⁵)	(3.7 x10 ⁻⁵)	(3.2 x10 ⁻⁵)	(2.6 x10 ⁻⁵)	(3.8 x10 ⁻⁵)	(2.9 x10 ⁻⁵)
Best	203.06	293.83	219.70	336.46	212.96	387.89
frequency/Hz	(2.22)	(11.72)	(3.55)	(8.58)	(2.41)	(6.60)
0	3.32	1.59	1.19	2.87	5.76	7.70
Q	(0.21)	(0.47)	(0.24)	(1.32)	(2.74)	(4.34)
PASSIVE						
STATE						
Sample size	35	29	33	24	28	31
E0/m	5.16 x10⁻⁴	5.78 x10⁻⁴	6.62 x10 ⁻⁴	5.68 x10⁻⁴	5.41 x10 ⁻⁴	6.11 x10 ⁻⁴
10/111	(1.9 x10 ⁻⁵)	(4.3 x10 ⁻⁵)	(3.2 x10 ⁻⁵)	(4.2 x10 ⁻⁵)	(2.4 x10 ⁻⁵)	(3.0 x10 ⁻⁵)
Best	207.47	301.43	325.00	283.55	206.45	263.20
frequency/Hz	(3.99)	(12.32)	(7.44)	(5.97)	(3.32)	(8.70)
Q	1.04	0.94	0.67	0.91	1.11	1.00
	(0.04)	(0.06)	(0.03)	(0.07)	(0.04)	(0.05)
Energy gain	3.06	2.81	1.93	2.05	6.26	1.85
(kBT)	(0.62)	(1.69)	(0.25)	(0.58)	(2.05)	(2.40)



Figure 32. Energy gain estimates calculated for male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes (sample sizes given in brackets, significant differences between conspecific male and female mosquitoes are starred).

7.3.2. Mosquito free fluctuations - effective stiffness

All male mosquitoes have a higher best frequency (as shown in table 10 above) and effective stiffness than females of the same species in the active state after ringer injection, with the effective stiffness of all mosquito species and sexes significantly increasing whilst in the sedated state (as demonstrated in table 13).

Table 13. Median values of effective stiffness for male and female mosquitoes from each species investigated comparing post-ringer and sedated states (numbers in brackets are standard errors) as well as P-values for statistical comparisons between the two states.

Mosquito species/ sex	Effective stiffness after ringer (µN/m)	Effective stiffness (sedate state) (µN/m)	P-value
Ae. aegypti females	13.8	74.0	P <0.001
(n = 36/ 35/ 36)	(0.84)	(7.34)	1 <0.001
Ae. aegypti males	54.6	148.1	P <0.001
(n = 25/ 30/ 25)	(5.05)	(24.4)	P <0.001
An. gambiae	22.6	157.0	
females (n = 29/ 33/ 26)	(17.8)	(21.2)	P <0.001
An. gambiae males	85.6	162.5	D -0.001
(n = 20/ 22/ 18)	(8.51)	(12.4)	P <0.001
Cx. quinquefasciatus	3.84	58.7	P <0.001
(n = 29/ 29/ 27)	(0.5)	(2.35)	F <0.001
Cx. quinquefasciatus	18.3	107.9	R =0.001
maies (n = 28/ 33/ 35)	(6.06)	(21.6)	P <0.001

7.3.3. Mosquito apparent antennal mass estimations

Mean values of the mosquito apparent antennal mass are given in table 14 below: male and female *An. gambiae* and *Cx. quinquefasciatus* were found to have significantly different apparent masses (P <0.05 for *An. gambiae* and P <0.01 for *Cx. quinquefasciatus*), whilst no such significant differences were seen between male and female *Ae. aegypti* (P>0.05).

Table 14. Mean values for mosquito apparent antennal mass estimates (numbers in brackets refer to standard errors). Significant differences between conspecific males and females are starred.

Species/ sex	Sample size (before pymetrozine)	Sample size (after pymetrozine)	Mean apparent antennal mass (kg)
An angunti famala	35	21	4.054 x10 ⁻¹¹
Ae. aegypti leitiale		21	(2.592 x10 ⁻¹²)
Ac coguntimala	20	15	4.393 x10 ⁻¹¹
Ae. aegypti maie	30	15	(2.995 x10 ⁻¹²)
*An combiac fomale	33	17	4.535 x10 ⁻¹¹
An. gamblae lemale		17	(2.638 x10 ⁻¹²)
*An combios molo	22	0	5.414 x10 ⁻¹¹
An. gambiae male	22	9	(3.460 x10 ⁻¹²)
*Cx. quinquefasciatus	20	21	3.232 x10 ⁻¹¹
female	29	21	(1.435 x10 ⁻¹²)
*Cx. quinquefasciatus	22	21	3.918 x10 ⁻¹¹
male	33	21	(2.201 x10 ⁻¹²)

7.3.4. Step recordings

7.3.4.1. Mechanical and nerve responses to force steps

The response of both male and female mosquitoes to force step stimulation is similar to that seen for *Drosophila melanogaster* both in section 6.2.8 and in previous reports (285). This includes initial overshooting of the flagellum in response to the stimulus onset, negative recoil following this overshoot (with further flagellar overshoots and recoils taking place after this, particularly if small force stimuli are used as in figure 33 part B) and adaptation of the system to the stimulus such that a steady state displacement is reached. This response to stimulation, especially with regards to the extent of adaptation noted, is a key mechanical signature of the direct gating of transducer channels (and has been noted as such for *Drosophila melanogaster* and vertebrate hair cells for example) (285, 469-471).

Nerve responses to the largest force steps are also produced with a delay on the order of a millisecond (as seen in figures 33 and 34). The recoil action that occurs after stimulus onset for mosquito species takes longer to reach equilibrium than for *Drosophila melanogaster* however, particularly with regards to female mosquitoes, and which is therefore potentially indicative of an overall change in flagellar stiffness.



Figure 33. Flagellar displacement (top) and CAP amplitude (middle) in response to the corresponding force steps (bottom) for A) Ae. aegypti females (in red) and B) Ae. aegypti males (in blue).

In both mosquito sexes there is a reproducible delay between receiver displacement and nerve response of approximately 800-900 µs for large force steps (as seen in figure 34 below). The order of magnitude for this time interval is strongly suggestive of direct gating of transducer channels; for example the equivalent time delay for *Drosophila melanogaster* has been estimated at closer to 500 µs for analogous stimuli (which is comparable given the increased length of the mosquito flagellum) and is considered to be within the acceptable time range to indicate direct gating of mechanotransducer channels (285). That the nerve response returns to the baseline established prior to stimulus onset despite the stimulus continuing is also demonstrative of the adaptation of transducer channels (285).

For small force steps, the time delay between the receiver and nerve responses increases considerably in both sexes, which is also in agreement with *Drosophila melanogaster*. The male mosquito nerve response however now takes more than 1 ms longer than the female to appear, as well as being seemingly split into a number of parts (as shown in figure 34, which could be related to potential male-specific neuron populations).

Both male and female *Ae. aegypti* share similarities with regards to maximum CAP response but it is when comparing the smallest steps that the major differences can be identified – the male nerve response is an order of magnitude greater than the female response, potentially indicating the presence of a neuronal subgroup which demonstrates an increased sensitivity to minute displacements. The male nerve response to large stimuli also seems to contain a second spike after the major component has diminished, which may be the result of neuronal firing from this same sensitive population.



Figure 34. Flagellar displacements (dashed line) and CAP amplitudes (solid line) showing the time delay between flagellar and nerve responses for large (left) and small (right) stimuli for A) Ae. aegypti females and B) Ae. aegypti males (stimulus onset is at start of flagellar displacement and force is constant throughout).

7.3.4.2. Two state model of a single transducer population in mosquitoes

Table 15 gives median values of the parameters estimated by fitting the two state model of a single transducer population to the force-displacement data collected for each mosquito species. No significant differences were calculated for any parameter when comparing female *Ae. aegypti* and *Cx. quinquefasciatus* mosquitoes (p>0.05 for all tests), whilst significant differences were found between *An. gambiae* females with females from the other two species when comparing every parameter apart from the estimated number of ion channels (p<0.01 for all tests apart from the estimated number of ion channels, in which case p>0.05).

For male mosquitoes the differences between species appear less clear. In terms of the estimated number of ion channels, male *Cx. quinquefasciatus* had a significantly smaller population than estimated for either male *Ae. aegypti* or *An. gambiae* (p<0.05), who were not found to be significantly different (p>0.05). However male *Ae. aegypti* had significantly smaller estimates for the ion channel gating force, $K_{INFINITY}$ and K_{STEADY} than males from the other two species (p<0.05 for all tests), who were statistically not significantly different from each other (p>0.05 for all tests).

Further than this, K_{GS} was found to be significantly larger in male *An. gambiae* than for *Ae. aegypti* and *Cx. quinquefasciatus* males (p<0.05), with males from these two species being statistically indistinguishable (p>0.05). Finally, no significant differences were calculated between males from any species in terms of the estimated extent of nonlinearity (p>0.05). Complete lists of all P-values are available in appendix I. Table 15. Median values for two state model of a single transducer population parameter estimates for each species and sex, grouped by sex (values in brackets are standard errors). Starred values within a group indicate significant differences between this value and values for the same sex in the other species (all P-values are given in appendix I).

	Number of ion channels	Ion channel gating force (fN)	K _{INFINITY} (µN/m)	K _{STEADY} (µN/m)	K _{GS} (µN/m)	Extent of non- linearity
Ae. aegypti female (n = 21)	943.9 (178.1)	16 (2.9)	74 (1.8)	64 (1.3)	9 (0.7)	0.178 (8.7 x10 ⁻³)
<i>An.</i> gambiae female (n = 18)	982.1 (210.2)	*28 (1.9)	*134 (6.4)	*94 (4.0)	*38 (2.7)	*0.370 (0.018)
Cx. quinquefas- ciatus female (n = 17)	916.7 (205.0)	14 (1.4)	71 (2.7)	59 (1.7)	12 (1.2)	0.189 (0.010)
Ae. aegypti male (n = 18)	298.1 (130.6)	*27 (3.5)	*145 (8.4)	*106 (6.6)	41 (2.6)	0.097 (0.012)
An. gambiae male (n = 15)	189.6 (89.0)	41 (6.0)	181 (9.6)	135 (7.1)	*53 (3.6)	0.121 (0.024)
<i>Cx.</i> <i>quinquefas-</i> <i>ciatus</i> male (n = 15)	*77.2 (15.9)	58 (7.8)	176 (5.8)	145 (4.6)	41 (2.2)	0.097 (0.018)

The extent of nonlinearity in the auditory system (described in section 7.2.10) is an important measure of the nonlinearities introduced into the auditory system via transduction. That it is significantly greater in female *An. gambiae* mosquitoes (as demonstrated in figure 35) as compared to female mosquitoes from the other species investigated is in strong agreement with the aforementioned frequency shifts between active and passive states noted for females from this species, with relationships between nonlinear gating compliance and antennal best frequency having been reported previously (472).



Figure 35. Comparison of estimates for the extent of nonlinearity in the auditory systems of the different mosquito species investigated (sample sizes are given in brackets, with significant differences within a sex are starred and black dots correspond to the 5th and 95th percentiles).

7.3.4.3. Dynamical stiffness and CAP amplitude

As is evident from figure 36, there is considerable variability between not only different sexes but also different mosquito species in terms of both the calculated changes in dynamical stiffness and the CAP responses to stimulation. Broadly however the dynamic stiffness values for all mosquito groups shown demonstrate a decrease in dynamic stiffness around the resting position and the CAP response shows a saturating increase in magnitude as displacement increases.

This increase in compliance is characteristic of mechanotransduction activity (285). In general, female mosquitoes tended to show these decreases once stimuli had become smaller than 1500nm whilst males seemed to only display such drops for very small stimuli (approximately 250-400nm). These contrasting sensitivity levels are indicative of strong differences in the auditory system between the sexes.

Sexually dimorphic responses are shown not just in the dynamical stiffness changes but also in the nerve response; for all males more than half of the total maximum CAP amplitude was produced after displacements of only a few hundred nanometres. The maximum CAP amplitudes produced by male *Ae. aegypti* and *Cx. quinquefasciatus* were slightly larger than for females (though it took much larger displacements for the females to produce similarly sized CAP amplitudes whilst the males demonstrated maximum CAP amplitude saturation much more quickly, suggestive of a shift in working ranges).

The CAP responses of *An. gambiae* males were not only much greater than those produced by males from the other two species but also stood in sharp contrast to the low CAP amplitudes recorded for the females of that species – these CAP amplitude findings proved highly reproducible and are unlikely to be the result of sub-optimal recording electrode insertions.





Figure 36. Changes in dynamical stiffness and CAP amplitude in response to changes in antennal displacement to a maximum displacement of ± 2000 nm for female and male mosquitoes from each species investigated: A) Ae. aegypti females, B) An. gambiae females, C) Cx. quinquefasciatus females, D) Ae. aegypti males, E), An. gambiae males and F) Cx. quinquefasciatus males. The bold lines represent two-state gating spring model (for a single transducer population) fits to the median data (285).

The CAP amplitude plots also display potential saturation points at which different populations of ion channels contribute towards the total CAP response produced. As the biology of the mosquito JO is not fully understood it is difficult to assess exactly at what working ranges these different populations may be working or even how many different independent transducer populations there may be.

The male *An. gambiae* for example appears to have perhaps as many as three saturation points even within 2000nm deflections, at approximately 150, 600 and 1500 nm respectively. These are potentially indicated in the dynamical stiffness plots show above as well, with changes in stiffness gradient noticeable within those displacement ranges. The presence of highly sensitive male-specific neurons, which are highly likely to be involved in locating females given the exclusive focus males place upon courtship, could explain these initial saturation points.

7.3.4.4. Potential mechanisms to maintain constant frequency tuning: compensatory stiffness antagonises the gating compliance?

The dynamic stiffness calculations described so far have all used stiffness values calculated from the peak displacement of the flagellum (X_{PEAK}) after the force stimulus onset (referred to in section 6.2.8 as K_{PEAK}). Using this method, an increase in compliance is seen around the resting position. This is not the only stiffness present in the system however, with the steady state stiffness (calculated using X_{STEADY} values and referred to in section 6.2.8 as K_{STEADY}) also potentially playing a role in stiffness changes.

 K_{STEADY} was originally assumed to be constant across all displacement magnitudes as X_{STEADY} scales linearly with force. The dynamical stiffness plots in figure 37 however show the full extent of the changes seen in K_{STEADY} across the whole displacement range for both mosquito species and *Drosophila melanogaster*.

For *Drosophila melanogaster*, calculating dynamical stiffness using values from either the peak or steady displacement data leads to similarly shaped plots – the increase in compliance around the resting position appears far less pronounced but still visible over the same range of displacements that produce the characteristic inner part of the nonlinearity predicted by the two state gating spring model. This also seems to be the case for female *An. gambiae*, with both K_{PEAK} and K_{STEADY} matching closely at displacements smaller than ±500nm. This continues to support the previous evidence (including nonlinearity estimates, two state model fitting and CAP amplitude saturation points) that these females share remarkable similarities with *Drosophila melanogaster* in terms of their antennal responses to displacement force steps.

For all other groups of mosquitoes investigated however the K_{STEADY} values show increases in dynamic stiffness as displacement size decreases. *Cx. quinquefasciatus* and *An. gambiae* males display this behaviour to some extent but for *Cx. quinquefasciatus* females and both *Ae. aegypti* sexes these increases are much more evident, with a high degree of anti-symmetry between the K_{PEAK} and K_{STEADY} values.





Figure 37. Dynamical stiffness changes in response to changes to displacement calculated using either values calculated at the peak displacement (K_{PEAK}) or at the steady state (K_{STEADY}) for A) Ae. aegypti females, B) Ae. aegypti males, C) An. gambiae females, D) An. gambiae males, E) Cx. quinquefasciatus females, F) Cx. quinquefasciatus males and G) a Drosophila melanogaster male (CantonS strain).

These increases in dynamic stiffness could be reflective of a compensatory mechanism within the auditory system that prevents the system from becoming too compliant and attempts to maintain the best frequency of the active and passive systems within the very narrow frequency ranges that are necessary for successful mosquito courtship. Both of these suggestions imply some other kind of mechanism, possibly active, that somehow provides stiffness as the flagellum becomes compliant.

If the system is indeed active then it should be physiologically vulnerable to CO₂ sedation, which removes the ability to breakdown ATP, and so should change during sedation – these changes are shown in figure 38 below for an *Ae. aegypti* female and for a male *Ae. aegypti* mosquito, as well as male and female *Cx. quinquefasciatus* in appendix J.

Whilst the mosquito is sedated, auditory CAP responses are completely abolished this is unsurprising given the removal of ATP breakdown from the system. The sharp increase in compliance seen in K_{PEAK} in the active state for displacements less than ±500nm disappears during sedation, meaning the characteristic nonlinearity that was clearly present in the mosquito shown in figure 38 whilst awake is no longer evident during sedation; the remaining peak and trough in the sedated state could be due to the gas flow in the chamber.



Figure 38. Changes in dynamical stiffness in response to changes to displacement, calculated for an Ae. aegypti female using either values estimated at the peak displacement (K_{PEAK}) or at the steady state (K_{STEADY}); A) K_{PEAK} values for both active and sedated states, B) K_{PEAK} and K_{STEADY} comparison for the active state and C) K_{PEAK} and K_{STEADY} for the sedated state.

D) Changes in CAP amplitude responses in response to changes in displacement calculated for an Ae. aegypti female in both active and sedated states.

Assuming that the overall dynamic stiffness of the antennal receiver system is composed of these two active, counteracting components (with a compensatory increase in steady state stiffness around the resting position to balance out the increase in compliance seen in K_{PEAK}), it could potentially be the case that the previously calculated dynamic stiffness plots did not represent the full extent of the gating compliance present in the mosquito auditory system.

In order to calculate a more precise estimate of the entire scope of the gating compliance therefore it may be necessary to subtract the steady state stiffness from the peak stiffness (i.e. $K_{PEAK} - K_{STEADY}$), thus removing any compensatory processes from the system. Attempting this calculation for *Drosophila melanogaster* (shown below in figure 39) leads to dynamical stiffness values below 0 μ N/m around the resting position, which should thus result in spontaneous oscillations of the antenna (as can also be seen in Nanchung and Inactive mutants (27)).

Applying this same principle to mosquito species reveals similar results (shown in figure 39 with a full list of parameter estimates contained in table 16); for all three mosquito species, female mosquitoes show negative dynamical stiffness values once the extra stiffness provided by K_{STEADY} has been removed from the system. Dynamic stiffness estimates calculated for male *Ae. aegypti* and *Cx. quinquefasciatus* mosquitoes around the resting position are also negative, though equivalent values for male *An. gambiae* remain significantly greater than zero across all displacement ranges investigated (p>0.05).

Following the subtraction of the steady state stiffness from the system, the extent of nonlinearity substantially increased across all species and sexes tested (as can be seen in table 16), indicating an increase in auditory transduction related nonlinearities in the system.

Table 16. Parameter estimates obtained fitting the two-state model of a single transducer population to male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes after steady state stiffness has been subtracted from the system.

	Number of ion	lon channel gating	K _{GS}	Extent of	
	channels	force (fN)	(µN/m)	nonlinearity	
<i>Ae. aegypti</i> females (n = 21)	692.1	27	11	2.963	
An. gambiae females	2512.6	18	44	1.170	
(n = 18)					
<i>Cx. quinquefasciatus</i> females (n = 17)	699.2	24	13	1.800	
Ae. aegypti males (n = 18)	975.2	27	44	0.987	
<i>An. gambiae</i> males (n = 15)	111.4	63	53	0.515	
<i>Cx. quinquefasciatus</i> males (n = 15)	346.3	35	35	0.753	





Figure 39. Changes in dynamical stiffness calculated by subtracting K_{STEADY} values from K_{PEAK} values for A) Ae. aegypti females, B) An. gambiae females, C) Cx. quinquefasciatus females, D) Ae. aegypti males, E), An. gambiae males, F) Cx. quinquefasciatus males and G) Drosophila melanogaster males (CantonS line). Solid lines represent two-state gating spring model (for a single transducer population) fits.

7.3.4.5. Effects of hypothesised multiple ion channel populations

The single population ion channel open probability fits (which have previously been reported on for use in this context (285, 394)) are produced from the assumption that at zero displacement the probability of an ion channel being open is equal to the probability of it being closed i.e. $P_0 = 0.5$, and that there is a single population of ion channels producing the entirety of the CAP response – once all channels have become open then the maximum CAP amplitude should be produced.

The ion channel open probability fit included in figure 40 part A for *Drosophila melanogaster* includes a larger displacement range than in parts B or C because the second ion channel population is considered more insensitive than the first and so requires larger aristal deflections to become noticeable (394). Using these fits as an example (as the *Drosophila melanogaster* system is known to have more than one transducer population present (394)), the single channel fit displayed in figure 40 part A underestimates the minimum displacement necessary for the maximum CAP response as it assumes all channels have the same open probability and effective displacement ranges. Using a two-sigmoidal model however produces a significantly better fit, with one saturation point emerging at around 2500nm and another at around 6000nm.

Similarly, the single ion channel population fits for mosquitoes drastically reduce the complexity of the system – for the *An. gambiae* males shown in figure 40 part C for example every ion channel is predicted to open at displacements greater than 200nm. A two population model however suggests that whilst there is a population of ion channels that is completely open at this level of force stimulus, there is another group that requires a much greater relative force to open. The first saturation points for *An. gambiae* females (shown in figure 40 part B) do not emerge until flagellar deflections reach almost 1000nm when using a two population model, closely matching the dynamic stiffness changes seen earlier in figure 36.



Figure 40. CAP amplitude (left axis) and ion channel open probability (right axis) as functions of displacement for A) a Drosophila melanogaster male, B) An. gambiae females and C) An. gambiae males. The points represent CAP response amplitude at the relevant force displacement (standard errors where possible are included as black lines) whilst the different fit types are labelled within each image.

There is therefore some evidence that models which account for multiple ion channel populations should be used in future analyses. Figure 41 shows the differences seen between using single population fits and two independent population fits as a starting point for increasing the model complexity – median model fit parameters are included in table 17, with all fits being calculated assuming that $X_0 = 0$.

The calculated AICc values provided in table 41 in appendix K for fits to the two models for male *Cx. quinquefasciatus* and *An. gambiae* mosquitoes suggest that increasing the complexity of the model did not significantly increase the quality of the model; for all other mosquito types however using a two state model for two independent transducer populations produced a better fit than using a single population model even after applying a cost to utilising extra parameters. Both *Ae. aegypti* and *An. gambiae* females in particular show an improved fit, with the full extent of the increase of compliance for the smallest force stimulation steps now being captured.

Table 17. Parameter estimates obtained by fitting the two-state model for two independent transducer populations to both sexes for Ae. aegypti, An. gambiae and Cx. quinquefasciatus (sample sizes are given in brackets).

	Ae.	An.	Cx. quinq-	Ae.	An.	Cx. quinq-
	aegypti	gambiae	uefasciatus	aegypti	gambiae	uefasciatus
	females	females	females	males	males	males
	(n = 21)	(n = 18)	(n = 17)	(n = 18)	(n = 15)	(n = 15)
Number of						
first	6.0	70.7	27.0	97 /		
population	0.0	19.1	57.0	07.4	-	-
ion channels						
Number of						
second	026.7	1504 1	1042 0	0007 7	190.6	77.0
population	930.7	1504.1	1243.0	2337.7	109.0	11.2
ion channels						
First						
population	100	74	26	40		
gating force	122	74	30	48	-	-
(fN)						
Second						
population	1.1	10	10	7 2	11	59
gating force	14	10	12	7.5	41	50
(fN)						
KINFINITY	74	125	70	150	101	176
(µN/m)	74	155	12	150	101	170
K _{STEADY}	64	Q٨	50	106	125	145
(µN/m)	04	34	55	100	100	140
K _{GS}	10	Δ1	13	44	53	Δ1
(µN/m)			10			





Figure 41. Changes in dynamical stiffness in response to changes in antennal displacement to a maximum displacement of ± 2000 nm plotted using two-state gating spring model (for either one or two independent transducer populations) fits to the median data for: A) Ae. aegypti females, B) An. gambiae females, C) Cx. quinquefasciatus females, D) Ae. aegypti males, E), An. gambiae males and F) Cx. quinquefasciatus males.

7.3.5. White noise stimulus displacement gain experiments

Decreased attenuation (i.e. increased stimulus force size) led to greater antennal velocities but decreased sensitivity for all mosquitoes investigated (demonstrated in figure 42 parts A and B). Interestingly however different species and sexes showed distinct changes in the best frequency of their flagella as the stimulus intensity increased (shown in figure 42 parts C and D) – males showed a clear decrease whilst females either showed an increase (in the case of *An. gambiae*) or demonstrated a small (~5%) decrease followed by a suitably compensatory increase. The shifts in best frequency did not occur until a threshold stimulus intensity was reached.



Figure 42. A) Changes in antennal velocity magnitude seen in response to changes in the relative stimulus force attenuation for an individual Cx. quinquefasciatus female (shown here in dB).

B) Corresponding changes in sensitivity at each attenuation for the same female (N.B. attenuation axis is reversed here compared to A)).

C) Changes in the median best frequency of the flagellum relative to the unstimulated best frequency as the stimulus force increases for An. gambiae and Cx. quinquefasciatus females (dashed lines represent median best frequency of the passive systems).

D) Changes in the median best frequency of the flagellum relative to the unstimulated best frequency as the stimulus force increases for male Cx. quinquefasciatus mosquitoes (dashed lines represent median best frequency of the passive systems).

Similar to the results of the absolute energy gain calculations shown earlier, *Cx. quinquefasciatus* males demonstrate a significantly smaller displacement gain in response to white noise stimulation than females (with displacement gain in this sense being defined in section 7.2.11). Now however this difference has also been extended to both *Ae. aegypti* and *An. gambiae* males when compared to females of their own species – table 18 contains median values of estimated white noise displacement gains as well as significance values for all statistical comparisons made, whilst figure 43 shows the range of calculated displacement gains for each mosquito species and sex.

Table 18. Median displacement gains estimated using white noise stimulation for all mosquito species and sexes investigated (standard errors are shown in brackets). Significance levels between different sexes within a single species are shown next to the sex that was estimated as having the significantly greater level of energy injection.

Species/ sex	Median displacement gain (white noise)	Significance level
Ae. aegypti female	2.28	D - 0.029
(n= 7)	(0.58)	F = 0.036
Ae. aegypti male	1.51	
(n= 7)	(0.31)	-
An. gambiae female	2.14	P = 0.026
(n= 9)	(0.13)	1 = 0.020
An. gambiae male	1.19	_
(n= 7)	(0.37)	-
Cx. quinquefasciatus female	3.67	P - 0.031
(n= 13)	(0.54)	1 = 0.001
Cx. quinquefasciatus male	2.22	_
(n= 13)	(0.80)	



Figure 43. Displacement gain values estimated using white noise stimulation for each species and sex (sample sizes are given in brackets, with significant differences between male and female mosquitoes from the same species starred; black dots correspond to the 5th and 95th percentiles).

7.3.6. Pure tone stimulus displacement gain experiments

Significant differences can be seen in the flagellar and JO best frequency estimates in table 18 between male and female mosquitoes from the same species. Significant differences can also be seen between males from different species; *Anopheles* males have significantly lower best frequency values for both peak mechanical and nerve responses than *Ae. aegypti* males (p<0.05 in both cases) – *Cx. quinquefasciatus* males also have a significantly lower peak nerve response best frequency than *Aedes* (p <0.01). The only such difference seen throughout the females was a significantly lower frequency for the peak nerve response in *Cx. quinquefasciatus* females than in *Ae. aegypti* (p <0.01).

The best frequency tuning of the female flagellum and the male JO has previously been noted to be equal to the two major distortion products included in table 19; the cubic distortion product and the difference tone respectively. Whilst the median best frequency of the male *Ae. aegypti* JO was approximately 20Hz greater than would expected from the difference tone estimation, for all other mosquito types there was close agreement between the values approximated from calculating the distortion products of WBF measurements and those that were directly found in the pure tone experimental results.

There was almost no difference between the values estimated for the two relevant distortion products in *Cx. quinquefasciatus* (197 Hz for the cubic distortion product as compared to 193Hz for the difference tone) meaning that it may not be clear for this species as to which distortion product is relevant for the exact best frequency tuning – this was not an issue in the other species however as there were far greater differences in frequency between the two distortion products.

Table 19. Median values of the best frequency of the flagellum and nerve responses to pure tone stimulation (numbers in brackets refer to standard errors). Female fundamental WBF and the best frequency estimates for the flagellum from the velocity fits are provided to act as reference values. Two of the most prominent distortion products are also shown, with the difference tone being equal to the difference between the male and female WBF and the cubic distortion product being equal to the difference between the male WBF and double the female WBF.

	Ae.	Ae.	An.	An.	Cx.	Cx.
	aegypti	aegypti	gambiae	gambiae	quinque-	quinque-
	female	male	female	male	fasciatus	fasciatus
	(n= 8)	(n= 10)	(n= 7)	(n= 7)	female	male
					(n= 8)	(n= 8)
Median flagellum						
best frequency	195.0	360.0	205.0	325.0	205.0	335.0
(from pure tone	(5.49)	(12.31)	(7.51)	(4.36)	(6.66)	(7.43)
stimulation)/ Hz						
Flagellar free	203.1	384.4	219.7	332.0	213.0	311.2
fluctuations best	(2.22)	(20.5)	(3.55)	(5.81)	(2.41)	(5.52)
frequency / Hz	()	(20.0)	(0.00)	(0.01)	(2.11)	(0.02)
Cubic distortion	185 5	_	210.0	_	197.0	_
product / Hz	100.0		210.0		107.0	
Female	405.0		384.0		390.0	
fundamental	(22.0)	-	(16.7)	-	(10.5)	-
WBF/ Hz	(22.0)		(10.7)		(10.0)	
Median best						
frequency of	175.0	245.0	155.0	185.0	125.0	185.0
nerve response/	(6.66)	(10.62)	(3.69)	(9.69)	(7.43)	(5.90)
Hz						
Difference tone/	-	219.5	-	174 0	-	193.0
Hz		210.0				100.0

Figure 44 provides a representation of the best frequency tuning of the flagellum and the JO in the three mosquito species studied with figure 45 allowing for species specific comparisons of flagellar sensitivity and maximum nerve response between conspecific males and females. One of the most notable sex specific differences shared between all species is that the magnitude of the male nerve response was significantly greater than the magnitude of the female nerve response in all three species, especially when considering *An. gambiae*.

Female *Ae. aegypti* and *Cx. quinquefasciatus* however demonstrated much greater maximum flagellar sensitivity estimates than males from these two species, which agreed with the earlier presented results in figures 40 and 44 with regards to the energy gain and the displacement gain estimated using white noise stimulation. In contrast to this however the flagellar sensitivity of *An. gambiae* males was much greater than that of females when a pure tone stimulus was utilised.

From both the nerve and mechanical data, *Ae. aegypti* and *Cx. quinquefasciatus* males appear very similar after accounting for some frequency shifts. Both the maximum nerve response and peak mechanical gain values are similar across species, with this also being true for the females from these two species.



Figure 44. Sensitivity (dashed line) and nerve response (solid line) changes in response to changes in stimulus frequency for A) Ae. aegypti females, B) Ae. aegypti males, C) An. gambiae females, D) An. gambiae males, E) Cx. quinquefasciatus females and F) Cx. quinquefasciatus males. Fundamental WBFs are indicated as well as the difference tone and cubic distortion product (labelled 'diff tone' and 'cubic DF').



Figure 45. Sensitivity (left) and nerve response (right) in response to pure tone stimulation at different frequencies for male and female A) Ae. aegypti B) An. gambiae and C) Cx. quinquefasciatus. All points represent median values of individual experimental values at that frequency and bars signify standard errors at that point.
Table 20 includes the median displacement gains for all mosquito groups tested (estimated using pure tone stimulation) and demonstrates the sex-specific significant differences seen across all three species. *Cx. quinquefasciatus* females demonstrate significantly larger displacement gain values than males, with *Ae. aegypti* females also having significantly greater displacement gain values than males from that species (with all P-values included in table 20). Male *An. gambiae* however have significantly greater displacement gains than conspecific females. Figure 46 illustrates the range of median displacement gains for each mosquito species and sex investigated.

Table 20. Median displacement gains estimated using pure tone stimulation for all male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes (standard errors are shown in brackets) as well as P-values for comparisons between male and female mosquitoes from the same species

	Median	P-value	
Species/ sex	displacement gain		
	(Pure tone)		
Ae. aegypti female	14.75		
(n= 8)	(0.55)	1 = 0.00003	
Ae. aegypti male	8.24		
(n= 10)	(0.79)		
An. gambiae female	6.76		
(n= 7)	(0.33)		
An. gambiae male	11.92	P - 0.00003	
(n= 7)	(0.64)	F = 0.00003	
Cx. quinquefasciatus	16 52		
female	(0.89)	P = 0.00215	
(n= 8)	(0.03)		
Cx. quinquefasciatus	10.08		
male	(1.06)		
(n= 8)	(1.00)		



Figure 46. Displacement gain values estimated using pure tone stimulation for each mosquito species and sex (significant differences between males and females from the same species are starred, with black dots corresponding to the 5th and 95th percentiles).

7.3.7. TTX and TeNT injection series

Male and female mosquitoes across all three species showed significantly different responses to injection with either TTX or TeNT: figure 47 contains examples of free fluctuations from both male and female *Cx. quinquefasciatus* mosquitoes comparing both and after TeNT injection states. Whilst TeNT injection resulted in a significant increase in energy gain for the female *Cx. quinquefasciatus* (which was not the case for female *Ae. aegypti* or *An. gambiae* mosquitoes), the extent of this gain was significantly smaller than that seen in the male following injection (which began spontaneously oscillating). Injection with TTX also produced these sexually dimorphic responses in all three species.



Figure 47. Free, unstimulated fluctuations both before and after TeNT injection over a frequency range between 100 to 2000Hz for A) a Cx. quinquefasciatus female (linear scale) and B) a Cx. quinquefasciatus male (logarithmic scale). Individual data points represent Fourier-transformed velocity amplitudes whilst solid black lines show the velocity amplitude function fits for each recording.

Whilst both TTX and TeNT injection into male mosquitoes resulted in spontaneous oscillations of the flagellum, the length of time necessary for these oscillations depended to some extent on the compound type. The injection time courses contained in figure 48 show the extended length of time required for TeNT and TTX to cause spontaneous oscillations in an *An. gambiae* male as compared with a *Cx. quinquefasciatus* male – both compounds had an effect in the *Cx. quinquefasciatus* male almost immediately, whilst TeNT took far longer to have an effect on the *An. gambiae* male than TTX (which was almost instantly effective).



Figure 48. A and B) Time series of representative antennal velocities before and after ringer injection for An. gambiae males compared to TTX (A) or TeNT (B) injection. C and D) Time series of representative antennal velocities before and after ringer injection for Cx. quinquefasciatus males compared to TTX (C) or TeNT (D) injection.

The maximum antennal velocity reached, whilst still considerable when compared to the quiescent state, was far smaller for the *Cx. quinquefasciatus* male than for the *An. gambiae* male – flagellar parameter values obtained by fitting the velocity amplitude function to the free fluctuation data (as described in section 6.2.6) are provided in appendix H. Both mosquito species were able to maintain the level of energy injection required for such large oscillations for extended periods of time. Examples of the nerve and mechanical responses to TTX and TeNT injection in male and female mosquitoes are contained in appendix L.

Increases in energy gain for male mosquitoes (which could signify the removal of inhibitory feedback mechanisms) are shown in Figure 49 - full list of median values are included in tables 33 and 34 in appendix H. *Ae. aegypti* and *An. gambiae* female mosquitoes showed no significant change in energy gain following either TTX or TeNT injection when compared to the post ringer injection state (p>0.05 for all tests). The only significant change in energy gain for female mosquitoes from these two species was following pymetrozine injection, which resulted in the calculation of a significantly lower energy gain than for any other state (p<0.001 for all tests).

Female *Cx. quinquefasciatus* mosquitoes however were calculated to have significantly increased energy gains following injection by either TTX or TeNT (p<0.01 in both cases) – a significant decrease in energy gain was also calculated following pymetrozine injection (p<0.001 for all tests).

Simililarly, males from all three mosquito species had significantly increased energy gains following injection by either TTX or TeNT as well significantly decreased energy gains after pymetrozine injection (p<0.01 in all cases tested). The amount of energy gain however was much greater for all male mosquitoes regardless of species than for *Cx. quinquefasciatus* females. No female mosquito from any species was determined as having a spontaneously oscillating flagellum whilst all males that were injected with either TTX or TeNT were identified as having spontaneously oscillating flagellum. Significance levels for all comparisons made are included in table 21.

	Ae.	An.	Cx. quinq-	Ae.	An.	Cx. quinq-
	aegypti	gambiae	uefasciatus	aegypti	gambiae	uefasciatus
	females	females	females	males	males	males
Before and	P =	P = 0.610	P <0.01	P =	P =	P <0.001
after ringer	0.065	F = 0.019	P <0.01	0.011	0.001	P <0.001
After ringer	P =	P - 0 30/	P <0 001	P <0.001	P <0.001	P <0.001
and after TTX	0.232	F – 0.394	P <0.001	F <0.001	F <0.001	F <0.001
After ringer	P =	P - 0.266	P <0.001	P <0.001	P <0.001	P <0.001
and after TeNT	0.966	F = 0.200	F <0.001	F <0.001	F <0.001	F <0.001
After ringer						
and after	P <0.001	P <0.001	P <0.001	P <0.001	P <0.001	P <0.001
pymetrozine						
After TTX and						
after	P <0.001	P <0.001	P <0.001	P <0.001	P <0.001	P <0.001
pymetrozine						
After TeNT						
and after	P <0.001	P <0.001	P <0.001	P <0.001	P <0.001	P <0.001
pymetrozine						

Table 21. P-values for statistical comparisons between the different injection statesfor male and female mosquitoes from each species investigated.



Figure 49. Energy gain values calculated for different injection states for A) Ae. aegypti females, B) Ae. aegypti males, C) An. gambiae females, D) An. gambiae males, E) Cx. quinquefasciatus females and F) Cx. quinquefasciatus males (significant differences are starred, with black dots corresponding to the 5th and 95th percentiles).

7.4. Discussion

7.4.1. Mosquito auditory systems as measured by free fluctuations

The passive antennal receiver of *Drosophila melanogaster* has a best frequency of approximately 800Hz, whilst the best frequency of the active system is between 150 – 250Hz (283). The difference in best frequency between the two states is indicative of the active mechanotransduction processes taking place within the insects' auditory system (472). These active processes are important in order for *Drosophila melanogaster* to tune their antennal receiver to courtship songs but are linked to the development of nonlinearities.

In contrast to this, one of the most notable results shown in tables 12 and 13 is that the difference between active and passive states in terms of the best frequency of the flagellum of female *Cx. quinquefasciatus* mosquitoes is less than 15 Hz – this is also true for *Ae. aegypti* females. Such shifts are statistically significant but are unlikely to be biologically relevant because of the small absolute frequency change (473). *An. gambiae* females were alone amongst females in shifting from 219 to 325Hz in a sedated state. In addition to this, male *Ae. aegypti* did not show any significant changes in their best frequency when comparing active and passive states, although males from the other two species tested demonstrated a decrease in best frequency following sedation.

That these shifts were all either relatively small or non-existent suggests that frequency specificity is highly important for the mosquitoes. All female mosquitoes have similar best frequencies in the active state. This could imply that species specific differences may reside instead in the male auditory system, which seems to contain a greater level of complexity given the data presented in section 7.3.3.2 and previous reports (34, 289).

The flagellar best frequency estimates for female and male *Ae. aegypti* have previously been reported as approximately 230 and 380Hz respectively, both of which are higher than the equivalent values calculated in this thesis of approximately 200 and 280 Hz (363). These differences could be the result of changes in environmental conditions or variance between strains (in a similar manner to the differences noted in section 5.4.5 for differences between the reported and calculated values of mosquito WBF) (355, 442).

However, the best frequency of the male *Ae. aegypti* flagellum whilst it was spontaneously oscillating was estimated to be 380 Hz, exactly the same as the previous report (363). In addition to this, the best frequency of spontaneously oscillating male *Cx. quinquefasciatus* was estimated to be 311 Hz, which is similar to values reported previously

for this species – this similarity was not found for female *Cx. quinquefasciatus* however with best frequency estimates approximately 100Hz lower than previously published (308).

Male *An. gambiae* on the other hand were calculated to have a distinctly different best frequency in this thesis to that previously reported in either the fibrillae extended or collapsed state, although the flagellar tuning of females from this species was similar to the reported value (approximately 220 Hz as opposed to the reported estimate of 209 Hz) (353).

Flagellar effective stiffness increases following sedation (or pymetrozine) are in line with the results of section 6, as well as previous reports for *Drosophila melanogaster* using equivalent experiments, and are likely the result of a cessation of active processes which contribute to determining the stiffness of the flagellum due to CO_2 exposure (283).

In the non-stimulated, quiescent states used for the energy gain estimates, male *Ae. aegypti* and *An. gambiae* mosquitoes were not statistically significantly different from their female counterparts (which may be the result of an underpowered statistical comparison and so would require a larger sample size to confirm), whilst female *Cx. quinquefasciatus* had a significantly greater level of energy gain than males. These results suggest that in the absence of an biologically relevant stimulus, mosquitoes of both sexes inject comparatively low levels of energy into their auditory systems; wildtype *Drosophila melanogaster*, whose JO is considerably smaller than that of both mosquito sexes, typically exhibit energy gains of approximately 4.6 for example (283).

That the energy gain calculated in section 7.3.1 for mosquito species should be on the same scale as *Drosophila melanogaster* despite their JOs containing at the very least 10 times as many neurons may be the result of the configuration of the mosquito antennal receiver system (271, 277, 278). Whilst the one-dimensional auditory system in *Drosophila melanogaster* means that half the total number of ion channels can be opened by sufficiently large stimuli in a single direction (with the other half requiring stimulation in the opposite direction), the male mosquito flagellum is attached to between 60 to 70 radial prongs - this corresponds to approximately 6^o rotational precision (287).

The LDV measurements used to estimate the energy gain values are only taken in the plane of one pair of prongs and so do not account for other planes, which could have a significant effect on the energy gain calculation. For example, assuming there are 60 radial prongs and an approximately equal distribution of about 15000 neurons, this would mean there are approximately 250 neurons per direction (76). This is very similar to the estimated number of neurons in the *Drosophila melanogaster* JO, which are equally distributed into two separate groups to cover both possible displacement directions (271).

7.4.2. Mosquito apparent antennal mass estimations

There are no prior reports of mosquito apparent antennal mass values and as such there are no comparisons available to assess the accuracy of the estimates given in table 14. Whilst estimations of 276 ng have been reported for the effective mass of *Cx. quinquefasciatus* males (i.e. approximately 7 times greater than the reported apparent mass calculated in section 7.3.3), it is unfortunately difficult to compare effective and apparent mass estimates; not only are apparent mass estimates dependent on the measurement point chosen for recordings but also not all of the contents of the antennal system contribute towards auditory function as measured by the LDV protocol (308).

Of the three mosquito species investigated in section 7, significant differences were calculated between male and female mosquitoes for *An. gambiae* and *Cx. quinquefasciatus*, but not for *Ae. aegypti*. Given the plumose nature of the male mosquito's flagellum, it could be assumed that males from all mosquito species tested should have greater apparent antennal mass values than females. In *Drosophila melanogaster* however, apparent antennal mass and total body size are thought to be strongly correlated (laboratory data, unpublished). Whilst mosquito antennal systems show a greater level of sexual dimorphism than *Drosophila melanogaster*, it could be the case that body mass also plays a significant role in determining the apparent mass of the antenna for mosquito species as well.

Ae. aegypti males have been previously reported to be significantly smaller than females, with Ae. albopictus males also found to be significantly smaller than females from that species (335, 474). This could in turn help to explain why no significant differences are seen between the apparent antennal mass estimates for Ae. aegypti males and females; if females mosquitoes from that species are considerably larger than the males, this could compensate for the increased density of fibrillae attached to the male flagellum, meaning that the mass estimates for both sexes become approximately equal.

On the other hand, given the low value of the post hoc statistical power calculation given in section 7.2.9 for the comparison between male and female *Ae. aegypti* it is reasonable to believe that this comparative test is insufficiently powered to detect a significant difference. Greater sample size numbers would be necessary in order to sufficiently power the experiment and check the underlying cause of this lack of significance. Regardless of the significant differences (or lack of) seen, the estimation of these apparent mass values allows for the use of the previously reported gating spring model in order to analyse mosquito force-displacement data obtained from force step electrophysiology (285).

7.4.3. Mosquito auditory systems as measured by force step stimulation electrophysiology

Interpretation of the model results shown in table 15 must be done cautiously as it is unclear what the estimates for the number of ion channels and relevant gating forces mean for mosquito species – for example, assuming a standard 60-fold symmetrical arrangement of radial prongs linked to the male mosquito flagellar shaft with only two of these prongs (within the same plane) being affected by force-displacement steps, it could be suggested that the estimated number of channels in males should be multiplied by a constant to obtain a better estimate of the total number of ion channels within the male auditory system (76).

Significant differences identified between female *An. gambiae* and either *Ae. aegypti* or *Cx. quinquefasciatus* female mosquitoes could be indicative of the relative level of genetic similarity between *Ae. aegypti* and *Cx. quinquefasciatus* mosquitoes as compared to *An. gambiae* – such a level of similarity was not observed for male mosquitoes from these species however. Statistical comparisons between mosquitoes of the same sex but of a different species were also made more difficult to interpret because of a lack of statistical power for some comparisons; post-hoc power calculations indicated that these comparisons (such as those made for the number of ion channels in female mosquitoes) tended to have below 50% statistical power and as such should be considered underpowered. Larger sample sizes would be needed to categorically determine whether the lack of statistically significant differences is real or the result of these relatively underpowered comparisons.

The estimated values provided in table 15 can still provide some interesting results with regards to parameters that are more readily comprehensible across species. It is immediately clear that $K_{INFINITY}$, K_{STEADY} and K_{GS} are significantly greater within each mosquito species in the males as compared to the females for example, indicating that the male antennal system is significantly less compliant than the equivalent female system.

This maintenance of stiffness (with only narrow displacement windows of small increases in compliance) for male mosquitoes is likely due to the extreme importance of maintaining frequency selectivity, with the nonlinearities connected with gating compliances being reported to result in changes in the antennal best frequency of *Drosophila melanogaster* (and as previously discussed in section 6.1) (456).

The high levels of stiffness of the mosquito flagellum can be well demonstrated by comparing the proportional increases in compliance typically noted. For healthy *Drosophila melanogaster*, dynamic stiffness tends to decrease from a maximum of 70 to around 15 μ N/m for large and small displacements respectively, which equates to a greater than 4.5 fold difference in stiffness (as shown in figure 27 in section 6.3.2 and as has been reported previously) (285, 394). This change in stiffness is directly related to the differences in best frequency observed between passive and active states as stiffness and best frequency are proportional (as shown in section 7.2.9) (472).

By comparison, *An. gambiae* females (as shown in figure 36) regularly demonstrated maximal decreases in dynamic stiffness from 130 to 80 μ N/m, a reduction of less than 50%. Females from the other two mosquito species showed proportional decreases of just over 20% whilst estimates for males from all species were even smaller than this, although the baseline stiffness of the male mosquito flagellum is considerably higher in mosquitoes than for *Drosophila melanogaster*.

Despite this, the presence of a mechanical nonlinearity was clear for all mosquito sexes and species tested (as judged using previously reported methods and with data presented in figure 35), which is highly suggestive of the presence of the direct gating of mechanotransduction channels (285).

Only CAP recordings propagated by the axonal projections of neurons within the auditory system were able to be recorded via insertion of a recording electrode into the mosquito's antennal nerve because it was not possible to access any compound currents within the mosquito's auditory neurons. This means that whilst it was still possible to obtain some assessment of the auditory nerve response to stimulation, there are unfortunately time delays associated with spike generation and action potential propagation that are unavoidable using this recording method; these problems can be further compounded by differences associated with potential independent neuronal populations (285, 394).

Further than this, it seems unlikely that multiple populations of ion channels would be present in *Drosophila melanogaster* (denoted as sensitive and insensitive for this species) but not in mosquito species and therefore a single population model should not be sufficient to describe the additional layers of system complexity that are present (271, 394). This two state model of a single transducer population is used here however as a first step to quantify mechanotransduction with the mosquito auditory system (and specifically avoids including data that may lead to the analysis of non-auditory nonlinearities), and not as a definitive model of the mosquito mechanotransduction system; it can thus be further built upon in the future.

7.4.4. Systems of compensatory stiffness maintenance in mosquito species

The maintenance of an auditory system tuned within a specific frequency range involves active transduction-related processes which are associated with essential auditory nonlinearities (279). Nonlinear gating compliances and adaptation motors shift the best frequency of the antennal receiver in response to stimulation of different intensities (279, 456). In order to maintain such a flagellar tuning therefore the mosquito auditory system could seek to minimise these auditory mechanotransduction-based nonlinearities. Compensatory stiffness mechanisms could provide a method to reduce these non-linearities by ensuring that compliance changes are minimised. That such changes in K_{STEADY} are seen across all three mosquito species tested strongly supports this theory.

During sedation the sharp increase in K_{STEADY} seen around the resting position for the female *Ae. aegypti* mosquito included in figure 39 before sedation is no longer present. Instead both K_{PEAK} and K_{STEADY} appear identical apart from a constant background level increase in K_{PEAK} . This suggests that the additional stiffness provided in the steady state could have an active component. All sedation experiments across different species show similar conclusions – loss of nonlinearity, lack of auditory CAP response to stimulation and an absence of stiffness increases in K_{STEADY} around the resting position.

Given the low number of mosquitoes involved in the sedation experiments however, it is difficult to judge exactly how repeatable this loss of increases in K_{STEADY} is, in spite of the promising preliminary results. It may be the case that instead of an active process that leads to a change in stiffness occurring, viscoelastic elements within the auditory system provide compensatory mechanisms in a passive manner; such systems have previously been described in other situations (475, 476).

Within each individual recording taken whilst a mosquito was sedated, K_{PEAK} and K_{STEADY} seem almost identical aside from a constant vertical shift. This could be indicative of a systematic bias in the recording process or data analysis procedure, although the fact that K_{PEAK} and K_{STEADY} have distinctly different trends between different mosquito species and sexes suggests that this is perhaps not the case. The sharp peaks and troughs in dynamic stiffness around the resting position seen in both K_{PEAK} and K_{STEADY} are potentially the result of insufficient data averages being calculated for small displacements (for whom the effect of a constant flow of CO_2 is likely to be most important in terms of affecting displacement measurements); that these sudden changes in stiffness are observable for both K_{PEAK} and K_{STEADY} suggests that this phenomenon is truly the result of gas flow and could be removed by including more data into the averaging procedure.

Fitting the two state model for a single transducer population to the difference between K_{PEAK} and K_{STEADY} for male mosquitoes (as shown in section 7.3.4.4) provided a better fit in some respects than fitting to just K_{PEAK} ; it may be necessary therefore to remove the effect of K_{STEADY} from K_{PEAK} before conducting the force step stimulation analysis.

However the mosquito flagellar system is comprised of many interlocking systems designed to maintain a precise focus on narrow frequency ranges, whilst also exploiting distortion products in a manner possible only in a sharply tuned system (34). The removal of this potentially compensatory mechanism therefore could have substantial negative consequences on the entire auditory system; for example this counterbalancing stiffness might be necessary to prevent the system from constantly spontaneously oscillating.

One noticeable difference between male and female mosquitoes is the asymmetry evident in the dynamical stiffness estimates presented throughout section 7.3., with male (particularly *An. gambiae* and *Cx. quinquefasciatus*) mosquitoes displaying asymmetrical dynamic stiffness estimates around the resting position in response to positive and negative displacements of equal absolute magnitude. This may be due to differences between male and female mosquitoes in terms of the analysis procedure, with males undergoing a greater amount of data smoothing than females; this is unlikely to explain the significant differences seen however given that the smoothing should act to reduce outlying estimates.

The asymmetries could therefore be reflective of real differences in dynamical stiffness calculations for positive and negative displacements, which would thus depend on the orientation and positioning of the mosquito relative to the LDV experimental setup. Such asymmetries have not been noted for *Drosophila melanogaster* (285, 394).

These sexually dimorphic responses likely reveal the differing objectives of the male and female auditory systems; females may aim to prevent spontaneous oscillations whilst males, for whom spontaneous oscillations can naturally occur, focus on preventing increases in nonlinearity within the system (with male *An. gambiae* in section 7.3.4.4 for example having non-negative calculated dynamic stiffness estimates across all flagellar displacements tested) (289, 308).

Maintenance and minimisation of nonlinearities in the auditory system is essential for male mosquitoes. Whatever the mechanism by which steady state stiffness is maintained, removing this compensatory behaviour could result in a severe impairment on mosquito copulation attempts as the insects would no longer be able to maintain such a finely tuned system. The exact systems which maintain stiffness are therefore of great potential interest, and require further exploration.

7.4.5. Multiple transducer populations

The existence of only a single ion channel population in the auditory system of mosquitoes is, conceptually at least, unlikely given the complexity of the mosquito JO and the multiple populations reported for *Drosophila melanogaster* (271). There are unfortunately no available investigations reported as yet into these potential populations, which is why a two-state model for a single transducer population has been fitted throughout section 7. Investigating the changes in dynamical stiffness with increasing force steps (as shown in section 7.3.4.3) however clearly reveals a number of stiffness regimes present throughout, in a similar manner to that seen for *Drosophila melanogaster*. This is also visible from the CAP amplitude changes seen in figure 41, with several apparent changes in slope observable.

The improvement in ion channel open probability fit shown in section 7.3.4.5 for both male and female *An. gambiae* when a two transducer population was assumed was substantial but still requires modification. Given the low maximum CAP responses observed for female *An. gambiae* it may be more suitable to use another species, such as *Cx. quinquefasciatus*, for those fitting attempts as the maximum CAP response was more similar between males and females from this species.

Although the comparison of the relative quality of the single transducer population and two transducer population model fits using AICc values did suggest that the two population model described the force-displacement data more accurately for all female mosquitoes investigated, as well as male *Ae. aegypti*, no such improvement in quality was found for male *An. gambiae* and *Cx. quinquefasciatus* mosquitoes. This could indicate that only a single transducer population exists in these types of mosquitoes, though this would seem unlikely given the existence of multiple populations in *Drosophila melanogaster* and the indication of the existence in the other mosquito groups investigated (271).

The differences between the two fit types in terms of the AICc values were not particularly large even for those mosquito groups where the two ion channel population model was recommended, and as such this analysis does not provide solid evidence for the existence of a second transducer population. It therefore seems more likely that the lack of an improvement in fit quality results from a lack of understanding of the data and suggests that more research is necessary into the transducer populations before any conclusions can be drawn. This fuller understanding is vital as there are serious issues that can result from incorrectly using more complex models; for example additional populations would require the inclusion and investigation of additional resultant nonlinearities, which if not fully accounted for could result in misleading conclusions. These issues are also important when considering the ion channel open probability fits; whilst incorporating a second population into the fit function (as described in section 6.2.8) enables a seemingly more accurate description of the mosquito data because of an increase in the number of degrees of freedom associated with two populations, without a solid biological understanding of the mosquito system it could potentially lead to inaccurate assessments being made (394).

Thus it is important that the number of transducer populations in the mosquito system be properly estimated, potentially by the same mechanisms used to identify the auditory and wind/ gravity populations in *Drosophila melanogaster*, which will become possible hopefully in the foreseeable future as genetic manipulations of the mosquito genome become more widespread (271, 280).

Considering the extent of the sexual dimorphism between male and female mosquitoes (particularly with regards to auditory systems, such as the distribution of scolopidia populations) it may not be surprising if the two sexes also contained a different number of transducer populations (76). This is particularly relevant when taking into account the substantial CAP responses seen for even the smallest force steps for male mosquitoes from all three species, which are not present for female mosquitoes from any species tested.

7.4.6. Extent of sexual dimorphism in mosquito auditory systems with regards to displacement gain estimates

The best frequency of the peak mechanical sensitivity calculated from the pure tone stimulation data for each mosquito species and sex tested was similar to the estimates obtained by fitting the velocity amplitude fit function to free fluctuations of the flagellum (as shown in table 18 in section 7.3.6), The peak mechanical sensitivity of male mosquitoes from any species however did not well match the conspecific fundamental female WBF, in contrast to previous reports regarding tuning (358). There was a consistent difference of approximately 50Hz across all species, which could be the result of reported changes to the flagellar mechanical tuning due to temperature (308).

Both the peak mechanical sensitivity and nerve response to pure tone stimulation occurred at higher frequencies for male mosquitoes than for females across all three species tested, which is in agreement with prior reports (353, 358). Comparing these values with estimates of the difference tone and cubic distortion product for each species suggested various conclusions: whilst for *Ae. aegypti* both the peak male nerve response and the peak female mechanical sensitivity were approximately equal to the relevant calculated distortion products (as detailed in section 2.10.2), the equivalent tuning of responses for *An. gambiae* were too broad to reach any firm conclusions.

In addition to this, although *Cx. quinquefasciatus* males and females appeared to tune to the relevant calculated distortion product frequencies, the values calculated for the difference tone and the cubic distortion product were so similar that in practice it was not possible to distinguish which distortion product was relevant for which sex. This stands in contrast to previous reports for this species, in which the two distortion products could be individually distinguished (358).

This could be the result of imprecise estimates of the male and female WBF for each species (as discussed in section 5.4.5). Further experiments investigating the peak frequencies of mechanical sensitivity and nerve response to pure tone stimulation for these mosquito species would first require a more precise estimate of the WBF so that distortion products can also be calculated with greater accuracy. The sample size of the mosquito groups tested using pure tone stimulation should also be increased in order to investigate potential differences between female mosquitoes from different species in the best frequency of the peak mechanical sensitivity.

The mechanical tuning of the flagellum of male and female mosquitoes from all three species was discussed in section 7.4.1; the frequency of the peak nerve response differed significantly between the sexes as well as between *Ae. aegypti* males and males from the other two species. The frequency of the maximum nerve response of male *An. gambiae* was closely in line with that reported previously for this species whilst the tuning of male *Cx. quinquefasciatus* at a considerably lower value that than available in the literature (353, 358). Males from both these species had an estimated frequency tuning of 185Hz for their peak nerve response: given the resolution of the pure tone stimuli used in section 7.3.6 (which was presented in 10Hz steps) it may prove difficult even with a larger sample size to identify significant differences between the two species, should such differences exist.

Estimates of the displacement gain using white noise stimulation relied on fitting a sigmoidal model to the data obtained at each stimulus intensity and then calculating the ratio of the maximum and minimum values of mechanical sensitivity. The sharp decrease in sensitivity once the stimulus intensity had reached a threshold value however may be indicative of deeper underlying auditory mechanics. It may be the case therefore that modelling this sudden decrease in sensitivity as stimulus intensity increases requires the utilisation of a Hopf bifurcation model in order to account for the fact that mosquito auditory systems exist on the boundary of oscillatory instability (279, 290, 477).

The changes in best frequency seen for some types of mosquitoes during the white noise stimulation experiments (as shown in figure 42) strongly corresponded with the relative best frequencies of the active and passive systems for distinct mosquito sub-sets. Increased force caused the system to no longer require active gain mechanisms to function and as such the system tended towards the passive state and correspondingly tuned towards the best frequency of that state.

As such, females whose passive and active systems share identical frequency values showed almost no change in best frequency, whilst female *An. gambiae* demonstrated significant shifts which almost completely disappeared following pymetrozine injection. That the final frequency did not fully reach the estimated value for the completely passive system is likely the result of the force stimulus intensity not being powerful enough to reach a completely saturated state.

Previous reports on the mechanical sensitivity of the mosquito flagellum have generally indicated that male mosquitoes are more sensitive than females from the same species (289, 352, 354). This stands in sharp contrast to the displacement gain estimates calculated for both white noise and pure tone stimuli in sections 7.3.5 and 7.3.6 respectively, in which female mosquitoes were typically found to have higher gain estimates than conspecific males. These differences could be the result of different experimental methodologies (such as the specific focal point of the laser on the flagellum) or different mosquito lines used for analysis.

All sexes and species had significantly greater displacement gains when estimating the displacement gain using pure tone stimuli than with white noise stimuli. This could imply that pure tone stimulation is more biologically relevant for male and female mosquitoes from all species tested than broad-tone, lower intensity stimuli whose energy dissipates over a broad range of frequencies that are outside the frequency ranges that mosquito auditory systems are tuned to. This may therefore result in reduced white noise displacement gain estimates being calculated for both male and female mosquitoes due to interference with frequency-specific amplification mechanisms.

The evidence presented throughout section 7 has demonstrated repeatedly the considerable extent of sexual dimorphism in the auditory systems of mosquito species, both with regards to the changes in compliance and nerve responses to force step stimulation (shown in section 7.3.4.3) and also in the displacement and energy gain estimates.

Whilst the increased magnitude of CAP responses to stimulation (both in terms of force steps and pure tone) in male mosquitoes compared to females for all species tested could potentially simply be the result of the much larger JO in male mosquito, the finding that female mosquitoes in general have displacement gains that are either not significantly different to, or are significantly larger than, the corresponding estimates for males suggests that the female mosquito auditory system is potentially just as complex as the male. This is somewhat surprising given that female mosquitoes had previously been thought to possibly be deaf and it is only within the past few decades that female mosquito auditory function has been recognised as highly sensitive (363, 478, 479).

7.4.7. Extent of sexual dimorphism in mosquito auditory systems with regards to compound injection designed to sever efferent feedback loops

Injection of either TTX or TeNT into a mosquito should sever any efferant and afferant feedback mechanisms that exist within the auditory system and therefore potentially reveal the underlying system usually kept in check by constant damping feedback loops. Currently efferent feedback has only been reported in the auditory system of male *Cx. quinquefasciatus* mosquitoes, with some evidence also published alongside this suggesting that females from this species could potentially use efferent control loops to modulate auditory function (314). The results presented in section 7.3.7, with both TTX and TeNT injection leading to spontaneous oscillations in male mosquitoes, suggest that such systems exist in all male mosquitoes tested (including non-*Cx. quinquefasciatus* species) as well as potentially for female *Cx. quinquefasciatus*.

The impact on mosquitoes of injection with either TeNT or TTX is broadly sex specific, with only males exhibiting spontaneous oscillations in addition to corresponding extremely significant increases in energy gain after compound injection whilst females demonstrate much lower, or non-existent, levels – this is in line with previous literature indicating the sex-specificity of spontaneous oscillations (289, 308). This sex-specificity can be further picked apart: neither *An. gambiae* nor *Ae. aegypti* females showed any significant differences in energy gain between post-ringer and post TTX/TeNT states, though *Cx. quinquefasciatus* females showed a relatively small (compared to male mosquitoes) but significant increase when comparing these two states.

Of the three groups of female mosquitoes investigated, only *Cx. quinquefasciatus* females showed a significant difference. This could be suggestive of a potential dilution of neurotransmitters that occurs in the immediate aftermath of ringer injection, with increases in energy gain slowly being reversed until the system returns to a pre-ringer state after sufficient time has elapsed (though the time scale required is unclear given that energy gain estimates were calculated at least 10 minutes following compound injection). That such neurotransmitters could potentially exist only in *Cx. quinquefasciatus* females reflects the differences observed between female mosquitoes from different species.

On the other hand, the changes in male energy gain following TTX/ TeNT injection were much greater than those seen in females, with some *An. gambiae* males being calculated as having energy gains in the hundreds of thousands of kBT (contrasted with a post-ringer gain of less than 10).

Although all three male groups showed significant differences, the level of energy gain was considerably lower for *Ae. aegypti* males than for *Cx. quinquefasciatus* males, which was then in turn much smaller than that of *An. gambiae* males. This could be related to the extent of sexual dimorphism present in each species, with *An. gambiae* having being previously noted throughout sections 7.3 and 7.4 as exhibiting far greater differences between males and females in terms of auditory function than the other two species tested.

For all sexes and species tested, pymetrozine injection led to significant decreases in energy gain when compared to every other injection type. Whilst the energy gain remained above zero for almost all cases (and was tested and found to be significantly greater than zero), it resolutely remained at fractions of $1K_BT$, in agreement with displacement gain reports previously published (27).

The absence of spontaneous oscillations in female mosquitoes following TTX or TeNT injection clearly distinguishes the effects of TTX and TeNT on the antennal system from DMSO for example; DMSO injection may cause the antennal receivers of both male and female mosquitoes to spontaneously oscillate (289) but it also has this effect in *Drosophila melanogaster,* whose auditory systems do not contain efferent systems (310). Although TTX and TeNT have distinct mechanisms of action, the overall effect of compound injection should be to sever all afferent and efferent loops (should they be present) by preventing either voltage-gated sodium ion channels from firing or by blocking the release on neurotransmitters – that this causes spontaneous oscillations in male, but not female, *Cx. quinquefasciatus* suggests different functions for efferent networks in the two sexes.

The absence of energy gain in *Ae. aegypti* and *An. gambiae* females following TTX or TeNT injection could indicate that efferent feedback loops are not present in these two types of female mosquito or that these loops are present but serve a different purpose. It could also mean that the statistical power of the study is not great enough, although even if the power were to be increased and significant differences found, the scale of these differences would be minor compared to those calculated for male mosquitoes from any species or even *Cx. quinquefasciatus* females.

Identification of potential efferent control loops would require experiments similar to those reported for confirming the presences of such loops in males and as such all that can be drawn from the energy gain estimates is that female mosquitoes from all species tested do not contain efferent feedback control mechanisms within their auditory systems that operate in a similar manner to those seen in male mosquitoes (314). Given that only male mosquitoes demonstrate spontaneous oscillations in the absence of compound injection this could be indicative of the role that efferent feedback control plays in this phenomenon.

8. Mosquito pymetrozine vibrometry and electrophysiology

8.1. Introduction

Although section 5 included some behavioural experiments examining the impact of pymetrozine on mosquitoes, those assays were not as broad as the experiments conducted for *Drosophila melanogaster* in section 4. In addition to this, section 6 included detailed examinations of the auditory system of various *Drosophila melanogaster* lines both before and after pymetrozine injection (which were highly promising in terms of the ablation of ChO mechanosensory function) but did not include any mosquito species. Therefore it is important that the impact of pymetrozine on the mosquito auditory system is assessed in order to confirm whether the compound acts in a similar manner on both Dipteran species.

Previous data analysis presented in this thesis in section 6 utilised injection of pymetrozine in order to judge the impact of pymetrozine on the auditory system of *Drosophila melanogaster* (in line with previous reports (27)). The usual method of compound exposure in the field however is oral ingestion (as used throughout section 4, with free fluctuations of pymetrozine exposed *Drosophila melanogaster* provided in appendices A-D) (26). Comparisons between pymetrozine uptake via ingestion or via injection can enable analysis of potential significant differences between the exposure methods.

Whilst ensuring that pymetrozine ingestion is able to affect auditory function as measured by LDV is vital (as if pymetrozine is relatively ineffective after oral uptake it is not suitable for mosquito control programmes), changes from the expected auditory state observed in a mosquito after pymetrozine feeding exposure could not be unambiguously ascribed to the compound itself as there would exist no record of the state of auditory function prior to pymetrozine ingestion. Thus repeating the experiments completed in section 6 for *Drosophila melanogaster* will enable comparisons to be made within individual mosquitoes as well as between groups, in addition to enabling the determination of pymetrozine as the direct cause of any changes observed in auditory function.

This includes not only analyses of force step stimulation experiments but also free fluctuation recordings. The results of velocity amplitude fits to free fluctuation data were presented for mosquito species in section 7.3.1 and demonstrated that for changes in the best frequency of the mosquito flagellum following sedation did not correspond to changes seen for *Drosophila melanogaster* (both in section 6 and in previous reports (283)). Contrasting the changes observed following pymetrozine injection and those visible during sedation could allow for insight into the flagellar frequency tuning mechanisms.

238

The earlier investigations into the potential compensatory role that K_{STEADY} could fulfil within the auditory system (in section 7.3.4.4) suggested that this steady state stiffness could be the result of an active process because sedation of a mosquito prevented increases in dynamical stiffness observed in K_{STEADY} for small force stimuli from occurring. For all *Drosophila melanogaster* studied in section 6, pymetrozine was able to ablate ChO mechanosensory function and reduce increases in compliance related to K_{PEAK} around the resting position. Pymetrozine could thus also have a significant effect on relative changes in K_{STEADY} . This analysis can be completed in the same manner as in section 7.3.4 and could thus provide support to the theory of compensatory stiffness should K_{STEADY} change significantly following pymetrozine exposure.

Similar to the experiments conducted in section 6 using *Drosophila melanogaster* lines that demonstrated various forms of insecticidal resistance, the possible existence of cross-resistance between already existing insecticide resistance mechanisms in mosquitoes could reduce the potential of pymetrozine for future use as an insecticide.

Two *An. gambiae* strains, referred to as Ngusso and Tiassale, have been reported to exhibit significant levels of insecticide resistance, with the Ngusso line being reported as resistant to the organochloride class of insecticides and the Tiassale line resistant to compounds from all four major insecticide classes (described in section 2.3.1) (480, 481). Investigating whether pymetrozine is still able to ablate ChO mechanosensory function for these lines therefore could help provide information as to the potential usefulness of the compound for targeting mosquito populations in which there is a high prevalence of resistance to already existing insecticides.

The development of insecticidal resistance has been thought to be linked with relative reductions in competitive fitness (228, 482, 483). Before assessing the impact of pymetrozine on these resistant *An. gambiae* lines therefore it is important to judge the baseline, pre-pymetrozine auditory systems in order to compare the results with those seen for the susceptible *An. gambiae* Kisumu line tested in sections 5 and 7 and thus potentially identify any significant differences between the strains.

Finally, the energy gain analysis in section 7.3.7 provided further support for the existence of efferent feedback loops in male *Cx. quinquefasciatus*, which have been suggested to assist the male in locating suitable females for copulation (314). Male flagellar displacement modulation in response to pure tone stimuli (at frequencies which should mimic female flight tones) could be indicative of these feedback loops, and the potential impact of pymetrozine exposure on these loops could therefore be of interest in terms of identifying if pymetrozine can prevent male mosquitoes from identifying conspecific females.

8.2. Materials and methods

8.2.1. Mosquito rearing

Unless otherwise noted, all mosquito species were reared in an identical manner as described in section 5.2.1: all *Ae. aegypti, An. gambiae* (Kisumu) and *Cx. quinquefasciatus* (Muheza) mosquitoes used for experiments were provided by Shahida Begum from the London School of Hygiene and Tropical Medicine. *An. gambiae* (Ngusso and Tiassale strains) were provided by the lab of Dr Gareth Lycett from the Liverpool School of Tropical Medicine. All mosquitoes were reared using a 12 hr: 12hr LD cycle at 26°C and 75% relative humidity and were fed using a 10% glucose mixture.

8.2.2. Compound preparation

Ringer: Preparation of the ringer solution was identical to that described in section 6.2.2 for injection experiments in *Drosophila melanogaster*.

Pymetrozine: Preparation of pymetrozine was identical to that described in section 4.2.2 for feeding experiments in *Drosophila melanogaster*.

8.2.3. Compound exposure methods – feeding and injection

Feeding and injection experiments in mosquitoes entailed the same protocols as described in section 5.2.3 for pymetrozine ingestion and section 7.2.3 for pymetrozine injection. Pymetrozine exposure via ingestion was investigated in section 8.3.1 whilst all other sections included in the results utilised injection procedures for pymetrozine exposure.

8.2.4. Analysis of force step stimulation electrophysiology experiments following pymetrozine exposure via ingestion

Three *Cx. quinquefasciatus* female mosquitoes between 3 and 4 days old were transferred into individual vials and deprived of all food overnight. Following this they were provided with a glucose food source doped with pymetrozine (at a concentration of 1000ppm) for 24 hours. The female mosquitoes were then mounted and prepared for electrophysiological experiments in the same manner as described in section 7.2.4.

Electrophysiological force step recordings (as described in sections 6.2.8 and 7.2.10) were completed for all three female mosquitoes, with the data being analysed using the two state model for a single transducer population (previously described in section 6.2.8 and as used in section 7).

All three female *Cx. quinquefasciatus* mosquitoes were included in the final analysis. No formal power calculations were made for this experiment and as such it offers purely descriptive data as to CAP production and dynamical stiffness properties in response to force step stimulation in three individual *Cx. quinquefasciatus* female mosquitoes.

8.2.5. Analysis of electrophysiological force step experiments before and after pymetrozine exposure via injection for *Ae. aegypti*, *An. gambiae* (Kisumu) and *Cx. quinquefasciatus*

Male and female mosquitoes aged between 3 and 8 days old from each species investigated (*Ae. aegypti, An. gambiae* (Kisumu) and *Cx. quinquefasciatus*) were mounted and prepared for electrophysiological experiments in an identical manner as described in section 7.2.4. After the mounting procedure had been completed, free fluctuations of the flagellum were recorded and electrophysiological force step recordings (as described in sections 6.2.6 and 7.2.4) were completed in three separate states – before ringer injection, after ringer injection and after pymetrozine injection. Flagellar free fluctuation measurements were taken at the beginning and end of each of the three force step recordings taken per individual mosquito, as well as immediately after each compound injection. Compound injection was conducted in the same manner as described in section 7.2.3.

In total, 7 *Ae. aegypti* females, 6 *Ae. aegypti* males, 9 *An. gambiae* (Kisumu) females, 7 *An. gambiae* (Kisumu) males, 9 *Cx. quinquefasciatus* females and 9 *Cx. quinquefasciatus* males were included in the final analysis.

Analysis and model fitting of the free fluctuation data was completed as described in section 7.2.7 - this allowed for the fitting of the velocity amplitude function fit described in section 6.2.6 and thus the estimation of three key parameters described in that section (namely F_0/m , ω_0 and Q, with aggregated data for Ae. aegypti, An. gambiae (Kisumu) and Cx. quinquefasciatus mosquitoes being given in appendix H). Statistical comparisons between before and after pymetrozine states (as well as comparisons between sedated and post-pymetrozine injection states) were made using paired before and after t-tests with a significance level of 0.05 in Sigmaplot. Post-hoc calculations of statistical power found that all comparisons made had a power >80%, except for Ae. aegypti males for whom the power was below 50% (indicating an underpowered comparison).

Analysis of the force-displacement utilised the two state model for a single transducer population described in sections 6 and 7 (specifically section 7.2.10). Statistical comparisons between after ringer and after pymetrozine states for $K_{INFINITY}$, K_{STEADY} and K_{GS} for *Ae. aegypti, An. gambiae* (Kisumu) and *Cx. quinquefasciatus* were made using paired before and after t-tests with a significance level of 0.05 in Sigmaplot. Post-hoc calculations of statistical power found that these comparisons had a power of less than 50% (with the exception of comparisons made for *Ae. aegypti* males, where the power was calculated to be more than 90%) and such had should be considered statistically underpowered.

8.2.6. Analysis of electrophysiological force step experiments before and after pymetrozine exposure via injection for *An. gambiae* (Ngusso and Tiassale)

Male and female *An. gambiae* (Ngusso and Tiassale) mosquitoes aged between 3 and 10 days old were mounted and prepared for free fluctuation and electrophysiological experiments in an identical manner as described in section 7.2.4.

Once an individual mosquito had been mounted, a recording of the free fluctuations of the flagellum was taken to assess the health of the auditory system. Following this electrophysiological force step recordings were completed in three different states (before and after ringer injection, and after pymetrozine injection). Free fluctuations of the mosquito flagellum were recorded before and after each separate injection. Injection of both the ringer and pymetrozine solutions followed the protocol described in section 7.2.3.

In total, recordings from 2 *An. gambiae* (Ngusso) females, 3 *An. gambiae* (Ngusso) males, 6 *An. gambiae* (Tiassale) females and 5 *An. gambiae* (Tiassale) males were included in the before ringer injection analysis. The after ringer injection and after pymetrozine injection analyses each contained 1 *An. gambiae* (Ngusso) female, 1 *An. gambiae* (Ngusso) male, 3 *An. gambiae* (Tiassale) females and 3 *An. gambiae* (Tiassale) males.

Statistical comparisons between after ringer injection and after pymetrozine injection states used before and after Wilcoxon rank sum tests with a significance level of 0.05. Posthoc tests of statistical power found that whilst comparisons for *An. gambiae* (Ngusso) males had a power of over 80%, all other comparisons had statistical power estimates <50%, meaning that the comparisons should be considered underpowered.

8.2.7. Analysis of pure tone stimulation experiments following pymetrozine exposure via injection

Two male and two female *Cx. quinquefasciatus* mosquitoes between 5 and 6 days old were mounted and prepared for electrophysiological pure tone stimulation experiments in the same manner as described in section 7.2.4. These pure tone stimulation experiments (as described in section 7.2.12) were then repeated in each of the three injection states for all individuals studied (i.e. before ringer injection, after ringer injection and after pymetrozine injection). Analysis of this data was then completed in the same manner as described in section 7.2.12. These experiments were intended to be purely descriptive and as such no formal statistical power calculations were made.

8.3. Results

8.3.1. Step recordings – pymetrozine feeding

A complete elimination of CAP production in response to stimulation was observed for all three female *Cx. quinquefasciatus* female mosquitoes tested (as is evident in figure 50 below). Changes in dynamical stiffness for small stimuli (as seen in female *Cx. quinquefasciatus* not exposed to pymetrozine in section 7.3.4.3) are also no longer noticeable following pymetrozine ingestion. No comparisons to a before pymetrozine state were possible because of the compound exposure type used, which did not allow for prior electrophysiological recordings to be made.



Figure 50. Changes in dynamical stiffness (A) and CAP amplitude (B) in response to changes in displacement for Cx. quinquefasciatus female mosquitoes exposed to pymetrozine via ingestion of the compound (median values were taken from three separate individuals, with vertical black bars representing standard errors).

8.3.2. Changes observed in antennal free fluctuation recordings following pymetrozine injection

Pymetrozine injection did not lead to shifts in best frequency on the scale of those observed for *Drosophila melanogaster* tested in section 6 (as is demonstrated in figure 51). *Ae. aegypti* males showed no significant change in their flagellar best frequency and whilst *An. gambiae* and *Cx. quinquefasciatus* males demonstrated significant decreases in best frequency these shifts were far smaller than the minimum increase of 300Hz seen for *Drosophila melanogaster*. The significant differences observed for *Ae. aegypti* and *Cx. quinquefasciatus* females showed changes in best frequency of less than 13 Hz each.

Only *An. gambiae* females displayed increases in flagellar best frequency that could be considered comparable to those seen previously for *Drosophila melanogaster* lines – table 22 contains a full list of best frequency estimates for all mosquito groups both after ringer injection and after pymetrozine injection.



Figure 51. Free fluctuations for a Drosophila melanogaster male (A) and an Ae. aegypti female (B) showing both before and after pymetrozine states. The frequency ranges for A) and B) are 51 - 3200Hz and 100 - 2000Hz respectively. Individual data points represent Fourier-transformed velocity amplitudes at each frequency whilst solid black lines represent velocity amplitude function fits.

Table 22. Median values of flagellar best frequency for male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes (standard errors are given in brackets, with sample sizes for after ringer and after pymetrozine groups provided respectively below the species and sex type) as well as P-values for statistical comparisons between the two states

Mosquito species/ sex	Best frequency after ringer (Hz)	Best frequency after pymetrozine (Hz)	P-value	
Ae. aegypti females	198.5	210.2	P <0.01	
(n = 36/ 36)	(1.42)	(3.57)	1 <0.01	
Ae. aegypti males	290.0	292.7	P = 0.706	
(n = 25/ 25)	(11.91)	(11.49)	F = 0.700	
<i>An. gambiae</i> females (n = 29/ 26)	215.0 (3.95)	306.0 (7.07)	P <0.001	
<i>An. gambiae</i> males (n = 20/ 18)	352.0 (13.01)	319.1 (10.65)	P <0.01	
Cx. quinquefasciatus females (n = 29/ 27)	221.6 (5.16)	208.9 (1.90)	P <0.01	
<i>Cx. quinquefasciatus</i> males (n = 28/ 35)	365.6 (6.88)	258.7 (8.65)	P <0.001	

The best frequency estimates calculated following pymetrozine exposure are statistically indistinguishable from those calculated for mosquitoes whilst in the sedated state – table 44 in appendix M contains comparisons of these two states. Figure 52 shows comparisons between different states for both a female *An. gambiae* and a male *Cx. quinquefasciatus* mosquito in order to illustrate the high level of similarity between the sedated and pymetrozine exposed states in terms of the best frequency.





C and D) Free fluctuation recordings a Cx. quinquefasciatus male before and after pymetrozine (C) and before and during sedation (D).

The frequency range for all fits was between 100 and 2000Hz. Individual data points represent Fourier-transformed velocity amplitudes at each frequency whilst solid black lines show the velocity amplitude function fits for each recording.

Following pymetrozine exposure, the flagellar effective stiffness increases significantly for male and female mosquitoes from all three species investigated (as is evident in table 23). In a similar manner to the best frequency estimates following pymetrozine injection, these increases in effective stiffness are statistically indistinguishable from those calculated for mosquitoes in the sedated state (with table 44 in appendix M containing a full list of comparisons); the only exception is a significant difference in the effective stiffness calculated for *Ae. aegypti* females between sedated and pymetrozine injected states. Figure 53 shows the extent of this similarity, particularly in comparison to the post ringer state for both the best frequency and the effective stiffness estimates.

Table 23. Median values of flagellar best frequency for male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes (standard errors are given in brackets, with sample sizes for after ringer and after pymetrozine groups provided respectively below the species and sex type).

Mosquito species/ sex	Effective stiffness after ringer (µN/m)	Effective stiffness after pymetrozine (µN/m)	P-value	
Ae. aegypti females	13.8	40.5	P <0.001	
(n = 36/ 35/ 36)	(0.84)	(5.04)	1 <0.001	
Ae. aegypti males	54.6	91.7	P <0.001	
(n = 25/ 30/ 25)	(5.05)	(27.1)	F <0.001	
An. gambiae	22.6	125.9		
females (n = 29/ 33/ 26)	(17.8)	(13.5)	P <0.001	
An. gambiae males	85.6	182.6	P <0.001	
(n = 20/ 22/ 18)	(8.51)	(22.6)	F <0.001	
Cx. quinquefasciatus	3.84	42.3	5	
temales	(0.5)	(2.63)	P <0.001	
(n = 29/ 29/ 27)		. ,		
Cx. quinquefasciatus	18.3	86.6		
males	(6.06)	(8.36)	P <0.001	
(n = 28/ 33/ 35)	(/	()		



Figure 53. Best frequency and effective stiffness values after ringer injection, sedation or pymetrozine injection for male and female mosquitoes from A) Ae. aegypti, B) An. gambiae and C) Cx. quinquefasciatus, with significant differences between different states within a sex starred (black dots correspond to the 5th and 95th percentiles).

8.3.3. Force step recordings – pymetrozine injection

8.3.3.1. Mechanical and nerve responses to force steps

Male and female mosquito flagellar displacements and CAP production in response to force step stimulation in the absence of pymetrozine injection were previously displayed in figure 33 in section 7.3.4.1 (with equivalent responses for *Drosophila melanogaster* males both before and after pymetrozine injection shown in section 6.2.8 in figure 23). Figure 54 shows flagellar displacements and CAP responses to force step stimulation following pymetrozine injection for female and male *Ae. aegypti* mosquitoes.

In addition to the complete ablation of CAP response to force step stimulation of any size, the flagellar displacement response is also greatly changed – the initial overshoot is now far less prominent than for mosquitoes not exposed to pymetrozine and the recoil stage of the response to stimulation, though still present, undergoes far fewer corrections before reaching a steady state displacement.



Figure 54. Flagellar displacement (top) and CAP amplitude (middle) in response to the corresponding force steps (bottom) after pymetrozine exposure for A) Ae. aegypti females and B) Ae. aegypti males – N.B. an artifact (as the result of crosstalk between the stimulus and electrodes) is present in the nerve response for the larger force step.

Figure 55 includes data showing that mosquitoes of both sexes from all species investigated demonstrated a complete abolition of CAP production following pymetrozine injection. This occurred in tandem with a significant reduction in the decreases in dynamical stiffness observed around the resting position. As was the case for *Drosophila melanogaster* however, some of the mosquitoes tested (namely *Ae. aegypti* males and *Cx. quinquefasciatus* females, as evidenced in figure 55) seemed to retain an increased level of compliance for small displacements.

Almost all *Drosophila melanogaster* lines tested in section 6.3.2 showed reduced estimates for K_{STEADY} , $K_{INFINITY}$ and K_{GS} following pymetrozine injection. In contrast to this, only a few significant differences were calculated for either male of female mosquitoes (with table 24 providing a full list of median stiffness parameters and significance values): from the males, only *Ae. aegypti* showed a significant decrease in $K_{INFINITY}$ and K_{GS} following pymetrozine exposure, whilst females from both *An. gambiae* and *Cx. quinquefasciatus* species showed significant decreases in K_{STEADY} and significant increases in K_{GS} . No other significant differences were seen for any stiffness parameter.

Table 24. Median values for dynamic stiffness parameters before and after pymetrozine injection for male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus (numbers in brackets refer to standard errors and P-values below 0.05 are highlighted in bold).

	Ae.	An.	Cx. quinq-	Ae.	An.	Cx. quinq-
	aegypti	gambiae	uefasciatus	aegypti	gambiae	uefasciatus
	females	females	females	males	males	males
	(n = 7)	(n = 9)	(n = 9)	(n = 6)	(n = 7)	(n = 7)
KINFINITY after	75	135	76	136	181	175
ringer	(1.5)	(12 4)	(3.1)	(4.9)	(9.4)	(6.2)
(µN/m)	(1.5)	(12.4)	(3.1)	(4.3)	(3.4)	(0.2)
KINFINITY after	77	131	76	122	225	162
pymetrozine	(4.2)	(10.9)	(2.3)	(6.0)	(22.5)	(12.4)
(µN/m)	(1.2)	(10.0)	(2.0)	(0.0)	(22.0)	(12.1)
Significance	P>0.05	P\0.05	P\0.05	P-0.02	P>0.05	P\0.05
value	1 20.00	1 20.00	1 20.00	1 <0.02	1 20.00	1 20.00
K _{STEADY} after	66	103	63	98	137	139
ringer	(1.0)	(7.8)	(2 0)	(2.6)	(7.2)	(3.8)
(µN/m)	(1.0)	(1.0)	(2.0)	(2.0)	(7.2)	(0.0)
K_{STEADY} after	62	94	56	91	175	135
pymetrozine	(5.6)	(5.6)	(1.5)	(5.6)	(17.7)	(10.2)
(µN/m)	(0.0)	(0.0)	(1.0)	(0.0)	(17.7)	(10.2)
Significance	P>0.05	P<0.001	P<0.001	P>0.05	P>0.05	P>0.05
value	1 2 0.00			1 2 0.00	1 2 0.00	1 20.00
K _{GS} after ringer	9	37	15	37	48	32
(µN/m)	(0.7)	(4.8)	(1.4)	(2.9)	(4.2)	(2.7)
K _{GS} after	13	45	18	31	63	28
pymetrozine	(2.0)	(5.7)	(0.8)	(2 0)	(6.0)	(2.6)
(µN/m)	(2.0)	(0.7)	(0.0)	(2.0)	(0.0)	(2.0)
Significance	P>0.05	P<0 04	P <0 001	P<0.02	P>0.05	P>0.05
value	1 2 0.00	TAIAL		1 30.02	1 2 0.00	1 20.00




Figure 55. Changes in dynamical stiffness and CAP amplitude in response to changes in antennal displacement to a maximum displacement of ± 2000 nm for each of the mosquito species and sexes investigated both after ringer and after pymetrozine: A) Ae. aegypti females, B) An. gambiae females, C) Cx. quinquefasciatus females, D) Ae. aegypti males, E), An. gambiae males and F) Cx. quinquefasciatus males.

8.3.3.2. Changes to dynamical stiffness properties following pymetrozine injection

As can be seen directly from figure 56, pymetrozine injection significantly reduced dynamical stiffness changes in not only K_{PEAK} but also K_{STEADY} ; this is equivalent to the results obtained from the earlier sedation experiments discussed in section 7.3.4.4.

This reduction was different between male and female mosquitoes however – whilst K_{PEAK} and K_{STEADY} became almost completely linear following pymetrozine injection for female mosquitoes from all three species investigated, K_{STEADY} estimates for all three male mosquito types seemed to show a systematic increase in compliance around the resting position. This increase in compliance for K_{STEADY} was completely different however to the changes seen for this stiffness parameter before pymetrozine however, with figure 37 from section 7.3.4.4 showing that for all male mosquitoes tested there was a significant increase in dynamic stiffness for K_{STEADY} for small stimuli. Indeed, this increase in dynamic stiffness of stiffness to prevent changes in flagellar frequency tuning in section 7.4.4.





Figure 56. Changes in dynamical stiffness in response to flagellar displacement after pymetrozine injection plotted using either values calculated at the peak displacement (K_{PEAK}) or at the steady state (K_{STEADY}) for: A) Ae. aegypti females, B) An. gambiae females, C) Cx. quinquefasciatus females, D) Ae. aegypti males, E), An. gambiae males, F) Cx. quinquefasciatus males and G) a Drosophila melanogaster male.

8.3.4. Step recordings – pymetrozine injections with insecticide resistant lines

8.3.4.1. Force step electrophysiological recordings for insecticide resistant mosquitoes before pymetrozine exposure

The two insecticide-resistant strains of *An. gambiae* (Ngusso and Tiassale) have broadly similar nerve responses and flagellar dynamical stiffness changes as the susceptible Kisumu strain (figure 57 shows the full range of dynamics stiffness changes and CAP production whilst table 25 contains the median two-state model parameters for a single transducer system).

Male mosquitoes from both strains again demonstrated significantly greater CAP responses to stimulation and seemed much more sensitive (in terms of compliance increases) to small displacements than females. The dynamical stiffness changes in response to stimulated displacements for female *An. gambiae* mosquitoes from both strains fall much more in line with previous studies in *Drosophila melanogaster* than the male results, as was also the case for the Kisumu strain, with the nonlinearity estimates for females being almost three times greater than for males.

Table 25. Median values for two state model of a single transducer population parameter estimates for both An. gambiae Ngusso and Tiassale strains, grouped by sex (values in brackets are standard errors).

	Number of ion channels	Ion channel gating force (fN)	K _{inifnity} (μN/m)	K _{steady} (μN/m)	K _{GS} (µN/m)	Extent of non- linearity
Ngusso female (n = 2)	2320.5 (646.5)	24 (2.1)	166 (7.3)	116 (5.2)	51 (2.1)	0.468 (0.032)
Tiassale female (n = 6)	1489.6 (352.4)	25 (2.3)	149 (29)	109 (17)	41 (12)	0.361 (0.045)
Ngusso male (n = 3)	46.1 (16.4)	81 (38)	214 (30)	148 (30)	61 (1.7)	0.114 (0.030)
Tiassale male (n = 5)	119.8 (121.9)	57 (9.3)	212 (9.9)	148 (7.0)	59 (3.7)	0.103 (9.8 x10 ⁻³)

Α





Figure 57. Changes in dynamical stiffness and CAP amplitude in response to changes in antennal displacement to a maximum displacement of ±2000nm for each of the An. gambiae strains and sexes investigated: A) Ngusso females, B) Ngusso males, C) Tiassale females and D) Tiassale males. The bold lines represent two-state gating spring model fits (for a single transducer population) to median data points.

8.3.4.2. Force step electrophysiological recordings for insecticide resistant mosquitoes following pymetrozine exposure

Pymetrozine injection abolished CAP responses to stimulation in both sexes of both insecticide resistant *An. gambiae* strains (as shown in figure 58), although the low sample size used for all groups means that conclusions should be made cautiously. K_{STEADY} remained constant between post-ringer and post-pymetrozine states in females from both strains, whilst only the *An. gambiae* (Ngusso) males seemed to show an increase post-pymetrozine inject (with table 26 providing a full list of stiffness parameter values). Only one such male mosquito was injected with pymetrozine however so no firm conclusions can be drawn from this, except that pymetrozine seemed able to abolish auditory transduction in this specific individual male.





Figure 58. Changes in dynamical stiffness and CAP amplitude in response to changes in antennal displacement to a maximum displacement of ±2000nm for each of the An. gambiae strains and sexes investigated after ringer and after pymetrozine: A) Ngusso females, B) Ngusso males, C) Tiassale females and D) Tiassale males. The bold lines represent two-state model of a single transducer population fits to the median data. Standard error bars (represented by black vertical lines) are included for Tiassale but not Ngusso because for that strain only one injection experiment was recorded for both sexes.

Table 26. Median values for dynamic stiffness parameters before and after pymetrozine injection for An. gambiae (Ngusso and Tiassale) mosquitoes (standard errors are given in brackets were possible, with P-values for comparisons provided for male and female Tiassale mosquitoes).

	Ngusso	Tiassale	Ngusso	Tiassale
	female	female	male	male
	(n = 1)	(n = 3)	(n = 1)	(n = 3)
K _{INFINITY} after ringer (μN/m)	153	127 (24.2)	199	203 (22.6)
K _{INFINITY} after pymetrozine (μN/m)	149	121 (17.2)	317	195 (16.7)
Significance value	N/A	P>0.05	N/A	P>0.05
K _{STEADY} after ringer (μN/m)	113	94 (15.4)	121	146 (17.4)
K _{STEADY} after pymetrozine (µN/m)	104	86 (10.5)	224	146 (15.2)
Significance value	N/A	P>0.05	N/A	P>0.05
K _{GS} after ringer (μN/m)	40	33 (8.9)	78	68 (6.8)
K _{GS} after pymetrozine (μN/m)	45	35 (7.1)	93	57 (3.4)
Significance value	N/A	P>0.05	N/A	P>0.05

8.3.5. Mosquito flagellar sensitivity to pure tone stimulation following pymetrozine injection

Figure 59 demonstrates the effect that pymetrozine was able to have not only CAP production in response to pure tone stimulation but also on mosquito flagellar sensitivity; the male *Cx. quinquefasciatus* shown in the figure went from a relative gain of about 10 to approximately 3 following pymetrozine injection.



Figure 59. A) Changes in mechanical sensitivity in response to pure tone stimulation both post-ringer and post-pymetrozine injection for a male Cx. quinquefasciatus.
B) Nerve response to pure tone stimulation post-ringer and post-pymetrozine injection for a male Cx. quinquefasciatus mosquito.

Pymetrozine injection was also able to ablate the modulatory effect evident at the onset of the pure tone stimulus seen for *Cx. quinquefasciatus* males in figure 60 part A. This flagellar modulation of stimuli, which appears to be male specific, only occurred at frequencies around the peak flagellar frequency and showed the level of immediate increased gain that occurs before the system settled into a steady state. This effect was not seen in post-pymetrozine injected males (as shown in figure 60 part B), and was also absent essentially entirely from females in either a pre- or post-pymetrozine state.



Figure 60. A and B) Flagellar displacement following the onset at 0.1s of a pure tone stimulus (315 Hz) in the pre-pymetrozine (A) and post-pymetrozine (B) states in a Cx. quinquefasciatus male.

C and *D*) Flagellar displacement following the onset at 0.1s of a pure tone stimulus (215 Hz) in the pre-pymetrozine (*C*) and post-pymetrozine (*D*) states in a Cx. quinquefasciatus female.

8.4. Discussion

8.4.1. The effect of pymetrozine ingestion on mosquito auditory systems

Figure 50 demonstrates that all females investigated showed a complete loss of CAP responses to stimulation as well as an abolition of the expected decrease in dynamic stiffness around the resting position shown for *Cx. quinquefasciatus* female mosquitoes in section 7.3.4.3 disappeared. This suggests that pymetrozine has successfully ablated ChO mechanosensory function (as measured by LDV), and therefore implies that ingestion of pymetrozine is sufficient to induce this phenotype.

As discussed in section 6 with regards to *Drosophila melanogaster* injection experiments, experiments utilising pymetrozine ingestion as the exposure method are in some ways preferable to injection experiments because the exposure method used in the field is compound ingestion. Injection experiments therefore could give misleading information as to the extent of pymetrozine's effect on an individual mosquito if insecticide resistance mechanisms only affect compound ingestion (with pymetrozine resistance following ingestion having been documented in several species for example) (321, 322). That similar results were seen for both exposure methods, as observable when comparing the results of sections 8.3.1 and 8.3.3, the *Drosophila melanogaster* results presented in section 6 and previous reports, provides validity to the injection experimental results (27).

Only three female mosquitoes were tested in this section and all were from the same species. In order to confirm that pymetrozine ingestion reliably produces this phenotype across all mosquito species and sexes tested, more replicates must be done involving not only *Ae. aegypti* and *An. gambiae* mosquitoes but also *Cx. quinquefasciatus* males.

The concentration of pymetrozine used in these tests was also at an extremely high, saturating level – exposing mosquitoes to lower concentrations of the compound could therefore provide useful estimates with regards to the minimum exposure concentration necessary for an effect to be observable when the mosquito was tested using the LDV protocol. However, since the compound ingestion protocol does not allow for comparison with a pre-pymetrozine state it would be difficult to estimate an individual baseline for comparative purposes. Proper insertion of the recording electrode would also become more challenging as it could theoretically be possible that the absence of CAP responses is the result of not finding the antennal nerve rather than the effect of pymetrozine.

8.4.2. The effect of pymetrozine injection on mosquito auditory systems – free fluctuations

For all *Drosophila melanogaster* males tested in sections 4 and 6.3.1 (with an example provided in figure 51), pymetrozine exposure resulted in a significant increase in best frequency, from around 200 to 500Hz, in addition to decreases in the maximum antennal velocity amplitude and *Q*. Although decreases in these two parameters were seen for all mosquitoes tested as well, not all mosquitoes showed a significant change in best frequency after exposure – for example, there were no significant changes in best frequency calculated for male *Ae. aegypti* and the significant differences observed for female *Ae. aegypti* and the significant differences observed for female *Ae. aegypti* and *Cx. quinquefasciatus* involved frequency changes less than 15Hz and as such may only be statistically (rather than biologically) significant (484, 485).

For *Cx. quinquefasciatus* males and both sexes of *An. gambiae*, changes in flagellar best frequency were identical to those seen for the passive, sedated state. This stands in contrast to *Drosophila melanogaster*, for whom pymetrozine exposure results in a significantly different antennal state than that of sedation (as seen when comparing the pymetrozine injected *Drosophila melanogaster* presented in this thesis to previous reports of sedated flies (283, 284)). This strong frequency selectivity and maintenance is indicative of the differences between mosquito and *Drosophila melanogaster* auditory systems.

The difference in velocity amplitude fit function parameters between pymetrozine exposed and sedated states in *Drosophila melanogaster* allows for immediate validation that whilst the antennal system has been affected by pymetrozine, it is in a distinctly different state to that of sedation. For mosquito species however that comparison is much more difficult to make as there were few, if any, significant differences calculated between the sedated and pymetrozine exposed state.

Free fluctuation recordings for mosquitoes exposed to pymetrozine via the ingestion protocol (in section 8.3.1) showed similar peak antennal velocities and best frequency estimates to those injected with the compound, though pre-pymetrozine states could not be measured; thus the possibility remains that changes to the auditory system were the result of some factor other than insecticide ingestion.

Overall therefore, whilst recording free fluctuation measurements of the antennal ear appears to be a reasonable method of confirming the exposure of a mosquito to pymetrozine, comparisons between different auditory states require a greater level of caution for mosquito species than *Drosophila melanogaster*.

8.4.3. The effect of pymetrozine injection on mosquito auditory systems – force step stimulation and compensatory stiffness mechanisms

Compared to the male and female *Ae. aegypti* flagellar responses shown in section 7.3.4.1 (which contains the same force step stimuli for mosquitoes unexposed to pymetrozine) and to the previously reported responses for control *Drosophila melanogaster*, the active processes involved in the recoil of the flagellum after the initial overshoot are far less evident in section 8.3.3 with significantly less swinging of the flagellum taking place following stimulus onset (285). Whereas in a pre-pymetrozine state the receiver displacements in response to the smallest stimuli showed a high level of initial amplification when compared to the largest force steps, this is no longer evident after pymetrozine exposure, as each step response appears identical when adjusted for relative displacement.

The ablation of auditory CAP response to stimulation is evident for all stimuli, in agreement with the pymetrozine injection experiments using insecticide resistant *Drosophila melanogaster* lines in section 6; this indicates a loss of ChO mechanosensory function.

In contrast to the results calculated for *Drosophila melanogaster* in section 6 only one mosquito type studied (*Ae. aegypti* males) showed a statistically significant change in $K_{INFINITY}$ following pymetrozine injection, although female *An. gambiae* and *Cx. quinquefasciatus* mosquitoes did show significant changes to their estimates of K_{STEADY} and K_{GS} following pymetrozine exposure. This could be the result of the systems of compensatory stiffness that were earlier investigated in section 7.3.4, which means that following compound injection changes to stiffness parameters are more difficult to identify.

However, these compensatory systems were also significantly changed following pymetrozine exposure – rather than being almost anti-symmetrical to the values calculated for K_{PEAK} (with the notable exception of *An. gambiae* females), K_{STEADY} now seems almost identical to K_{PEAK} apart from a constant vertical shift. Indeed, calculating K_{PEAK} - K_{STEADY} (as was done in section 7.3.4 for control mosquitoes) now produces an essentially constant estimate of stiffness across all displacements values for male and female mosquitoes from all species tested (with differences in maximum and minimum values of less than 10μ N/m), which is indicative of the ablation of ChO mechanosensory function by pymetrozine.

The results shown in figure 56 are highly similar to those presented in section 7.3.4.4 for K_{STEADY} whilst the mosquito was sedated, providing further evidence that K_{STEADY} is the result of some active process (with estimates of K_{PEAK} - K_{STEADY} for sedated mosquitoes are almost identical to those of pymetrozine exposed mosquitoes).

Significant increases in the effective stiffness of the antenna following both sedation and pymetrozine injection were calculated in sections 7.3.2 and 8.3.2 for males and females from all three mosquito species investigated, which is suggestive of the key distinctions between effective stiffness (which includes all stiffnesses present in the entire antennal system) and dynamic stiffnesses (which provide specific information on the changes in stiffness for specific parameters over changes in flagellar displacement).

The significant increases in effective stiffness calculated following either sedation or pymetrozine exposure (as compared to the active, pre-pymetrozine state) were statistically indistinguishable for male and female mosquitoes from all three species tested. The only exception to this was the comparison made for *Ae. aegypti* females, for whom a significantly greater effective stiffness was calculated whilst mosquitoes from this group were in the sedated state than if they had been exposed to pymetrozine.

Given the lack of significant differences observed for other mosquito groups, as well as the absence of significant differences between sedated and pymetrozine exposed states identified in terms of the best frequency (as shown in appendix M), it seems that whilst this difference may be statistically significant it may not be biologically relevant.

8.4.4. Auditory systems of insecticide resistant mosquito species and the effect of pymetrozine injection on these systems

In general the auditory systems (as quantified by the LDV paradigm used throughout this thesis) of male and female *An. gambiae* from the Ngusso and Tiassale lines showed marked similarities to those of the *An. gambiae* Kisumu mosquitoes tested in section 7.3. This suggests that the development of different forms of insecticidal resistance has not led to changes in the auditory system that can be detected using this measurement system – given the low number of mosquitoes tested from the *An. gambiae* Ngusso and Tiassale lines however these comparisons must be made cautiously.

The extremely low CAP amplitudes produced by *An. gambiae* Kisumu females, as shown in section 7.3.4.3, could potentially have been the product of adaptation to cage conditions (although all female *Ae. aegypti, An. gambiae* (Kisumu) and *Cx. quinquefasciatus* mosquitoes tested in section 7 and section 8 had been kept in laboratory conditions for approximately the same length of time) – these adaptations can lead to alterations in mosquito physiology as a result of changes in selective pressure as well as the bottleneck formed by only a handful of female mosquitoes producing the majority of offspring (486). Such adaptations to laboratory conditions have been identified as having an impact on flight performance, as well as associated activities, in both *Drosophila melanogaster* and mosquito species (487, 488).

Both Tiassale and Ngusso strains utilised for experiments however also demonstrated relatively small CAP responses to stimulation in section 8.3.3 despite only been kept in laboratory cages for three generations, making it less likely these lines had fully adapted to cage conditions (for example, these colonies had yet to adapt to blood feeding via the Hemotek system and were still provided blood meals from human volunteers). It also seems implausible given the extremely large number of electrophysiology experiments completed using *An. gambiae* females from all lines throughout this thesis that this is the result of repeated incorrect insertions of the recording electrode.

It is therefore possible that this low level of nerve response to stimulation relative to females from other species is a species-specific feature of *An. gambiae* females, especially given the increased genetic differences between *An. gambiae* and the other two species investigated (as compared to the same differences between *Ae. aegypti* and *Cx. quinquefasciatus*). For example, a decreased focus on audition in female mosquitoes could have resulted in males injecting a greater amount of compensatory focus into their own auditory systems, with the reverse explanation also being plausible.

Female *An. gambiae* mosquitoes clearly retain some sensory capabilities with regards to auditory stimulation however as harmonic convergence is still the primary courtship mechanism in this species (353).

This hypothesis could be tested by repeating the experiments vibrometry experiments conducted in sections 7 and 8 for male and female *An. gambiae* caught in the wild and as such had not had an opportunity to adapt to laboratory conditions. Testing other *Anopheles* species, such as *An. arabiensis*, could also be useful for comparative purposes. In addition to this, more male and female mosquitoes from the insecticidal resistant lines (particularly Ngusso) could be tested to avoid problems associated with drawing conclusions from such a small sample size.

Following pymetrozine injection there was an elimination of CAP response to stimulation and a reduction of the increases in compliance around the resting position seen before exposure in all mosquitoes tested. These changes are in agreement with the results shown in section 8.3.3 for mosquito species that are susceptible to all insecticides. The low sample sizes used for the insecticide resistant lines however prevent concrete conclusions from being drawn – more repeats of the experiment are necessary before firm conclusions can be made.

In addition to this, the method of pymetrozine exposure used in section 8.3.4 was compound injection; as mentioned previously, given that the regular field exposure method of pymetrozine is oral ingestion, experiments investigating the impact of pymetrozine feeding are needed to ensure that both pymetrozine ingestion and injection result in the same effect on the mosquito auditory system. This would also help to guide estimates of suitable concentrations of pymetrozine for use in field situations that would have a significant impact on mosquito populations.

At the very least however, taken together with the earlier results from injecting pymetrozine into insecticide resistant *Drosophila melanogaster* lines, the effectiveness of pymetrozine when used against insects (including both mosquitoes and *Drosophila melanogaster*) that have developed significant levels of insecticide resistance is highly promising and suggests that further experiments be initiated in order to look in more detail at pymetrozine's effectiveness and efficacy against other insecticidal resistant mosquito lines.

8.4.5. The effect of pymetrozine injection on displacement gain as calculated in mosquito auditory systems via pure tone stimulation

The time period over which the potential modulation of pure tone stimuli presented in section 8.3.5 occurs is slightly under 100ms, placing it within the time scale of possible efferent feedback loop control (which can vary extensively according the requirements of the system) (489, 490). As discussed with regards to TTX and TeNT injections in section 7.4.7, efferent modulation has been reported in male *Cx. quinquefasciatus* mosquitoes and it has been suggested that this modulation is linked to identifying flight tones produced by females and is potentially related to mechanical control of sensitivity and the quality factor Q (as described in terms of the free fluctuation velocity fit function in section 6.2.6) (290, 314). This could imply that the modulation of pure tone stimuli is done under the control of an efferent feedback loop.

Female *Cx. quinquefasciatus* have been suggested to also have auditory efferent feedback loops, with the results of TTX/ TeNT injection presented in section 7.3.7 supporting this conclusion (314). In spite of this, flagellar amplitude modulation of pure tone stimuli is at best minor for females from this species, especially when compared to males. This could be indicative once again of the different roles filled by efferent feedback for both sexes.

The elimination of the potential modulation seen in male mosquitoes following pymetrozine exposure is therefore highly interesting, both in terms of the impact that the compound could have on courtship attempts (as males may no longer be able to identify female flight tones) as well as the mechanism by which this modulation is abolished following pymetrozine injection. Pymetrozine's mechanism of action (deterioration of ChO neurons due to a surge in the influx of calcium ions) could directly cause damage to the systems of efferent control for example, or the abolition of ChO mechanosensory function could prevent further mechanisms of control from being initiated (27).

The evidence provided in section 8.4.5 is based on exploratory testing of both this modulation of pure tone stimuli and the impact of pymetrozine on this modulation, and as such would require further testing before statistically valid conclusions could be made. It is also important to test male and female *Ae. aegypti* and *An. gambiae* mosquitoes in order to investigate the potential existence of this modulation in other mosquito species for which efferent feedback loops have not yet been reported (especially in light of the conclusions drawn from section 7.3.7, in which no evidence of efferent feedback could be identified for either female *Ae. aegypti* or *An. gambiae* mosquitoes).

9. Conclusions

9.1. Discussion

9.1.1. Conclusions of behavioural experiments

Pymetrozine has previously been reported to target insect ChOs, with compound exposure leading to an ablation of the organ's mechanosensory function (26, 27). Behavioural assays conducted within this thesis strongly support these previous reports and also explored the effects that a loss of ChO mechanosensory function can have on *Drosophila melanogaster*, with significant reductions in both flight ability and reproductive fitness observed in section 4.

ChOs, and by extension mechanosensation, have been regarded as inessential for the survival of *Drosophila melanogaster* because of the continued viability of ChO mutants (316, 317). This could be seen in the lifespan assay presented in section 4.3.2, with some pymetrozine exposed *Drosophila melanogaster* able to survive as long as control flies. These assumptions should be placed in the context of laboratory conditions however, where there is abundance of available food and no predators. The competitive fitness analysis presented in section 4.3.7 demonstrated that as soon as competition was introduced to the environment then the severe disadvantages associated with a loss of mechanosensation became more apparent.

Such competitive fitness comparisons were not made for mosquito species however, and the experiment described in section 5.3.2 investigating the effect of pymetrozine on *An. gambiae* female fertility and fecundity did not contain sufficiently large group sizes to adequately assess the compound's potential. The issue of small group sizes was also a problem for the larvicidal investigation detailed in section 5.3.1. Experimental repeats are therefore necessary before concrete recommendations regarding pymetrozine can be made.

Drosophila melanogaster are partly able to adapt to a loss of mechanosensation because the mechanism of feeding in this species does not require the insertion of a stylus and as such could be considered not significantly challenging from a mechanosensory control perspective (particularly when compared to mosquito species) (491, 492). Whilst male and female mosquitoes do not need to insert their proboscis into plants in order to feed from sources of glucose, such insertion is necessary for females to acquire a blood meal, which suggests that pymetrozine may have a stronger effect on mosquito species (in terms of female feeding behaviour) than on *Drosophila melanogaster* (493). The decline in flight ability noted for *Drosophila melanogaster* (in sections 4.3.3 and 4.3.4) following pymetrozine exposure is also highly promising with regards to potential compound effects on mosquito species, with flight forming an essential component of host seeking actions for female mosquitoes (336). If these decreases are also found to be present for mosquito species, which seems likely given the results presented in this thesis and the presumed involvement of the mosquito JO in mosquito flight regulation, pymetrozine could have a highly significant impact on host seeking behaviour (76, 333). These assessments could be made for mosquito species using modifications to the experiments conducted with *Drosophila melanogaster* in the Flycube system described in section 4.2.8, or by utilising other published systems of monitoring mosquito flight behaviour (401, 448).

Overall therefore the behavioural experiments presented in section 4 suggested that pymetrozine exposure had a significant impact on various aspects of the *Drosophila melanogaster* lifecycle and could thus also have a strong negative effect on mosquitoes – these experiments need to be conducted in mosquito species however before firm conclusions can be made with regards to the translation of these effects to other species.

9.1.2. Conclusions of electrophysiological experiments

Sexual dimorphisms in auditory function between male and female mosquitoes have been repeatedly reported - most commonly however, such reports have covered the increased sensitivity of the male flagellum as compared to the female, or have focused on male-specific spontaneous oscillations of the flagellum (34, 289, 363). The analyses provided in this thesis (particularly with regards to the calculated displacement and energy gains discussed in section 7.3) suggest that in a quiescent state, female *Ae. aegypti, An. gambiae* and *Cx. quinquefasciatus* mosquitoes can demonstrate equivalent (if not greater) levels of mechanical sensitivity to conspecific males.

In general however the level of energy gain calculated for both male and female mosquitoes in section 7.3.1 was of a relatively low level when compared to *Drosophila melanogaster* (283). This may be the result of laser measurements taken in a single plane and thus not accounting for the energy input of other prongs. Once male mosquitoes began to display spontaneous oscillations of their flagellum following TTX or TeNT injection however the calculated energy gains became far greater than those of *Drosophila* species.

That in the unstimulated state male mosquitoes do not make full use of their amplification machinery is in some ways comparable to *Drosophila melanogaster*, who also do not seem to fully exploit the maximum potential of their amplification machinery; energy gains following DMSO injection have been reported as approximately ten times larger than the equivalent gains calculated prior to injection, although it could be the case that DMSO changes the physiology of the fly such that energy gains that could not be attained otherwise become possible (283). Similarly, for nan^{36a} and iav^1 mutants energy gains have been calculated which are an order of magnitude greater than those of control flies (though still far less than energy gains calculated in this thesis for male *An. gambiae* mosquitoes) (27).

This control and adjustment of energy gain could serve a crucial purpose if its role is to keep the system stable near a critical point of oscillatory instability whilst maximising the antennal sensitivity (494). This critical point (a Hopf bifurcation) has been reported for many different species, including fruit flies, mosquitoes and vertebrates (279, 290, 495). The recent identification of efferent signalling networks in the auditory system of male *Cx. quinquefasciatus* could indicate the potential involvement of efferent control mechanisms with regards to modulation of mechanical sensitivity and energy gain (290, 314).

Maintenance of frequency specificity (in terms of audition) is highly important for both male and female mosquitoes (34, 276). The comparative changes in both K_{PEAK} and K_{STEADY} calculated in section 7.3.3 suggest that for the mosquito species tested in this thesis, the mosquito auditory system is composed of an interlocking system of balancing forces that seek wherever possible to maintain a constant frequency but can also inject a significant amount of energy when required.

A relatively higher degree of similarity (in terms of the measurements of auditory function conducted) was observed between *Ae. aegypti and Cx. quinquefasciatus* females than between either one of these species and *An. gambiae* females; these similarities included the minor (at most) differences in the flagellar best frequency observed between active and sedated or pymetrozine states, results of the single transducer population model fit, displacement gain estimates obtained via pure tone stimulation and comparative sizes of CAP responses to stimulation (shown in sections 7.1 – 7.6).

By comparison *An. gambiae* females seemed distinctly different, underlined by the observed increases in flagellar best frequency observed after pymetrozine injection and the significantly greater extent of nonlinearity calculated. These similarities and/or differences may be the result of *Ae. aegypti and Cx. quinquefasciatus* belonging to the same sub-family (the Culicinae) whilst *An. gambiae* belongs to a different group (the Anophelinae) (56, 76).

However such a level of similarity was not evident for male *Ae. aegypti* and *Cx. quinquefasciatus* with regards to the previously mentioned results – this perhaps suggests that species specific differences in auditory systems have evolved to lie within differences between male mosquitoes rather than females (34).

The uniting factors across both male and female mosquitoes from each of the three species investigated however were the mechanical signatures of transducer gating that were identified in each individual mosquito tested. These signatures included a nonlinear compliance and adaptation to force step stimulation, both of which can be seen 7.3.3 (particularly in the return of the nerve response to baseline following adaptation, rather than engaging in constant neuronal firing or displaying a shift from baseline) (282, 285).

9.2. Critique

9.2.1. Experimental critique

All experiments utilising mosquitoes in sections 5, 7 and 8 used mosquitoes from laboratory colonies. Although the *An. gambiae* Ngusso and Tiassale lines had only been kept in laboratory conditions for several generations, the *Ae. aegypti, An. gambiae* (Kisumu) and *Cx. quinquefasciatus* mosquitoes used had been kept as colonies for far longer (and indeed the length of time necessary for adaptation to laboratory conditions to take place is unclear, meaning that several generations may be sufficient for significant changes to have appeared). Whilst this made mosquito rearing in general easier (as blood feeding could be done via the Hemotek system) and allowed for logistical issues to be dealt with quickly, potential adaptations to laboratory conditions can lead to significant differences forming between field mosquitoes and those used for testing (496, 497).

This is particularly important in this case because adaptation to laboratory conditions could entail changes in flight ability or auditory function, two of the major focal points of this thesis. Should there be significant differences between laboratory and field populations with regards to these two important aspects then the results of the behavioural and electrophysiological investigations into hearing should be treated cautiously. Given the conservation of Nanchung and Inactive across species and the initial testing done using insecticidal resistant mosquito lines it seems unlikely that pymetrozine would be completely ineffective on field mosquitoes - the efficacy, or the effect on mechanosensory behaviours, may be significantly altered however (318).

A number of experiments described in section 4 involving *Drosophila melanogaster* utilised CO₂ sedation in order to select virgin flies following eclosion. Whilst CO₂ sedation makes this process relatively straightforward, it can also create experimental problems because of reported after-effects of this type of sedation on *Drosophila melanogaster* behaviour (422, 498, 499). Utilising ice sedation (as was done for many other experiments throughout this thesis, particularly when such experiments involved mosquito species) should minimise the potential effects of sedation and therefore should be utilised for all future experimental procedures.

Testing tethered, solo-flying mosquitoes provided estimations of the WBF for male and female *Ae. aegypti* and *Cx. quinquefasciatus* that were broadly in line with previous reports (as seen in section 5.3.5) (352, 354). Although the results of the same estimations for male and female *An. gambiae* mosquitoes were lower than those given in the literature, the ratio of male to female WBF remained constant at around 1.5 (353). However, assessing these measurements using individuals does not allow for judgement of a number of variables only assessable when multiple mosquitoes are present; examining WBF modulation and variability for example requires pairs of mosquitoes that interact with one another (34).

In addition to this, the processes involved in preparing tethered mosquitoes (namely the application of glue) could also have led to changes in flight tone characteristics, with experiments using tethered mosquitoes previously being suggested to result in lower WBF estimates than those using free flying mosquitoes (306). This could then have resulted in unreliable estimates of the WBF being made, with corresponding effects on the estimation of distortion products (which were utilised in section 7.3.6).

Interestingly, whilst the frequency of the male flagellar mechanical tuning was consistently at least 40Hz lower across all three species than would have been predicted from the WBF estimates (suggesting that there could indeed have been an issue with this method of measuring the WBF), both the estimated difference tone and the cubic distortion product were broadly in line with the experimental results of the relevant peak mechanical and nerve responses (as discussed in section 7.4.6).

Previous tests exploring harmonic convergence used either the introduction of pairs of tethered mosquitoes to each other or free flying arenas in which the entire courtship procedure could be observed (352, 357, 358). Such courtship interactions provided insight into what fundamental requirements were necessary for copulation to take place, particularly with regards to harmonic convergence, as well as providing information on the heritability of successful copulatory traits (34, 355).

No such experiments were included in this thesis and as such it remains to be seen whether male mosquitoes exposed to pymetrozine are able to harmonically converge with females from the same species (or whether exposed female mosquitoes are able to converge with control males). Without this key information it is difficult to judge exactly how effective pymetrozine could be when targeting mosquito species, though previous reports that deafened mosquitoes were unable to harmonically converge could suggest that pymetrozine exposed mosquitoes would also be unable to harmonically converge (352).

Similarly no trials were conducted to test the ability of pymetrozine exposed female mosquitoes to locate hosts and acquire blood meals; such assays exist and have been published previously (500, 501). These trials would also be essential before any further recommendations could be made regarding pymetrozine's potential as an insecticide targeting mosquito species.

9.2.2. Statistical and modelling critique

A recurring issue noted throughout this thesis was the problem of sample sizes that were insufficiently large enough to guarantee a reasonable level (i.e. at least > 80%) of statistical power. This meant for example that the lack of a significant difference in fertility between pymetrozine exposed *Drosophila melanogaster* and controls discussed in section 4.3.1 may in fact be due to the study being underpowered rather than there being no true difference between the groups (a problem further compounded by the distributions of egg laying habits reported for *Drosophila melanogaster*) (502).

Whilst post-hoc power calculations at least allowed for identification of statistical tests that may be considered underpowered, it would represent a significant improvement if future tests could take advantage of the results presented throughout this thesis and use them to estimate the minimum sample sizes needed to ensure that statistical tests are reasonably powered. The mosquito larvicidal assay (described in section 5.3.1) and the experiment assessing the potential effect of pymetrozine on mosquito fertility and fecundity (in section 5.3.2) for example would both benefit greatly from such an increase in group size, with both of these experiments being crucial for the basis of recommendations regarding pymetrozine's potential involvement in mosquito control programmes.

It is also important to consider the distinctions between statistical and biological significance; for example, female *Ae. aegypti* demonstrated statistically significant changes in flagellar best frequency between active and sedated states, but the absolute frequency difference was less than 10Hz. Distinguishing between the different forms of significance is crucial for accurate sample size calculation and for ensuring that conclusions drawn from the experimental analyses are biologically relevant (484, 485).

The velocity amplitude fit function used to analyse mosquito free fluctuation data in section 7.3 well described female mosquito flagellar free fluctuations (with R² values of over 0.95 for all fits included in this thesis) as well as male *Ae. aegypti* data (283). The function described male *An. gambiae* and *Cx. quinquefasciatus* mosquito flagellar fluctuations less well however because of differences in the best frequency of the active and passive system; the passive system was still observable in the free fluctuation recordings taken whilst a mosquito from one of these two groups was in the active state (as shown in section 7.3.1).

As such, modifications to this fit function may be necessary to account for these differences because these fit functions form the basis of energy gain and effective stiffness calculations and therefore it is essential that they describe the data as accurately as possible. In spite of this however, the fit function was still able to perform relatively well for male *An. gambiae* and *Cx. quinquefasciatus* mosquitoes (with all R² values calculated being greater than 0.85), suggesting that any necessary modifications may not need to be substantial.

The gating spring model of a single transducer population used in section 7.3.3 to analyse mosquito force-displacement data provided some interesting results as well as a basis for future investigations into mosquito auditory mechanotransduction, but is clearly not sufficient to fully describe such a complex auditory system (especially as it is not sufficient to describe the relatively simple *Drosophila melanogaster* system) (394). Initial attempts to use a more complicated model including multiple populations of transducers were made in section 7.3.4 but were only preliminary because of the lack of knowledge about the biological bases for these transducer populations.

The results of the comparisons of fit quality between the two fit types (i.e. fits including a single transducer population as compared to fits involving two independent populations) also did not produce a comprehensive conclusion as to the number of transducer populations, and future AICc analyses would require greater depth to provide clearer suggestions. This means that whilst more detailed molecular studies are necessary to identify the different populations in the JO, improvements to the gating spring model should also be simultaneously made in order to modify its focus to the more complex auditory systems of mosquitoes.

9.3. Future work

Although the experimental analyses presented in this thesis indicate that pymetrozine has potential for use against mosquito species, because of the effect on mosquito auditory function and *Drosophila melanogaster* flight ability and competitive fitness, much more work is required before the compound could be potentially included in any control programmes.

The first such experiment that must be completed is the aforementioned experiment investigating whether male mosquitoes (from any of the species used in this thesis) exposed to pymetrozine and control females from the same species are able to display the phenomenon of harmonic convergence, with the reverse experiment (i.e. pymetrozine exposed female and control males) being equally important (34).

Given the abolition of auditory function reported for pymetrozine exposed mosquitoes (as shown in section 8.3), and the reported difficulties that deafened – and more generally, ChO impaired - mosquitoes have with flight maintenance, it is unlikely that harmonic convergence could occur but the experimental proof of a possible reduction in harmonic convergence events is vital for the future potential of pymetrozine (352).

Further than this, experiments investigating potential reductions in human biting events following pymetrozine exposure at different concentrations could provide information on the effect that pymetrozine could have if incorporated as part of a mosquito control programme. Cone tests and arm-in-cage assays could thus be utilised to judge any possible changes in biting frequency, after which more extensive trials in experimental huts could be proposed if necessary (501, 503). These tests could then be followed by investigations into tracking female mosquitoes attempting to locate sources of blood meals following pymetrozine exposure (448).

The potential impact of pymetrozine as a larvicide also needs to be examined in greater detail. Low sample sizes used in section 5.3.1 meant that the study was not sufficiently powered to detect the existence of true significant differences reliably and thus could be repeated. The auditory function of adult mosquito exposed to pymetrozine only in the larval stage is also interesting when considering that *Drosophila melanogaster* adults exposed only as larvae demonstrated ablated auditory ability (laboratory data, unpublished). If the same is true for pymetrozine then the compound becomes potentially more useful for application in the field.

Further testing on mosquito lines that have demonstrated insecticidal resistance is also important; should pymetrozine be found to be effective against such strains in field situations (as has been suggested in section 8.3) then the compound would become much more promising in terms of targeting these mosquitoes.

The growing importance of control mechanisms which target females from mosquito species that tend to feed outdoors (such as *An. arabiensis*) alongside the previously mentioned increases in insecticidal resistance prevalence mean that compounds with novel mechanisms of action which can be applied outdoors are of potentially great usefulness (9, 504). The mechanism by which pymetrozine should be administered has still yet to be defined but seems likely to be either as part of an attractive toxic sugar bait programme or as a larvicidal/ oviposition trap toxin.

Pymetrozine has been proven to be effective against insects which feed from plants (26). Whilst mosquitoes do not insert their proboscis into sources of sugar provided by plants in the same manner that aphids insert their stylus into the plant phloem, feeding experiments included in section 8.3.1 utilising *Cx. quinquefasciatus* females (which had no documented resistance to insecticides) showed that merely consuming a source of sugar doped with pymetrozine was sufficient to induce the pymetrozine positive phenotype (76). This suggests that pymetrozine could be sprayed onto vegetation in a similar manner to that previously described (160).

Previous trials of outdoor applications of attractive toxic sugar bait have reduced exposure of non-target species such as bees by spraying the compound onto non-flowering vegetation; pymetrozine could be thus applied in a similar manner (160). It would be necessary to first estimate the minimum concentration of pymetrozine that would be required to have a significant effect on the mosquito population; initial estimates could be gained from using the currently recommended concentrations to target aphids, with field trials which estimate the comparative effect on mosquito population size further refining these estimates.

Finally, it is of the greatest importance that before any new method of mosquito control is introduced into a control programme the potential long term impacts of utilising the control measure are fully assessed. These assessments range from potential toxicity to other non-mosquito species (including humans) to long term environmental effects to the impact that the intervention could have on mosquito population dynamics, all of which are essential in order to avoid repeating previous mistakes made with regards to novel interventions (505, 506).

These preliminary assessments should also take into account not only the selection pressure on the mosquito species targeted, in terms of insecticidal resistance for example, but also on pathogens which are transmitted by this species (because of the influence that pathogens can have on their hosts) (102, 103, 507). For example, should an intervention exert such a strong selective pressure that female mosquitoes begin to take human blood meals earlier in the developmental cycle, or more often, then this could cause a significantly negative outcome on the local human population.

All of the tests described above explore the possible use of pymetrozine in field trials; whilst pymetrozine has the potential for such an application, the evidence presented throughout this thesis also provides support for the use of pymetrozine as a tool to explore mechanosensation. Whilst genetic methods currently exist of ablating ChO function, the specificity of pymetrozine in terms of a pharmacological intervention allows for a greater degree of flexibility in assay design (27).

The possible use of mosquitoes as models of mechanotransduction is also highly promising. The mosquito auditory system in both male and females is highly complex and contains a significant number of interesting properties, such as the manipulation of distortion products for mate seeking, that have yet to be fully investigated (276). The existence of efferent feedback loops in male mosquitoes from species other than *Cx. quinquefasciatus* (or the confirmation of such loops in female *Cx. quinquefasciatus*) for example has not yet been confirmed (314). By modifying the already existing gating spring models, mathematical modelling of mosquito auditory transduction should become more refined which would thus help create higher quality models of potentially multiple transducer populations (in line with previous reports in *Drosophila melanogaster*) (285).

The recent innovations in genetic tools with which to manipulate the mosquito genome also allow for the development of similar assays as those previously published for other insect species (508). The development of a deeper understanding of the mosquito auditory system could therefore not only provide information on potential targets for insecticides but also result in a more complete understanding of the role of auditory (and non-auditory) mechanotransduction in mosquito – and more widely, insect - behaviours.

References

1. World Health Organization. Department of communicable disease prevention cae. Malaria fact sheet. 2016.

2. World Health Organization. Department of communicable disease prevention cae. Lymphatic filariasis fact sheet. 2016.

3. Bhatt S, Weiss DJ, Cameron E, Bisanzio D, Mappin B, Dalrymple U, et al. The effect of malaria control on *Plasmodium falciparum* in Africa between 2000 and 2015. Nature. 2015;526(7572):207.

4. Kim D, Fedak K, Kramer R. Reduction of malaria prevalence by indoor residual spraying: a meta-regression analysis. Am J Trop Med Hyg. 2012;87(1):117-24.

5. Lim SS, Fullman N, Stokes A, Ravishankar N, Masiye F, Murray CJL, et al. Net benefits: a multicountry analysis of observational data examining associations between insecticide-treated mosquito nets and health outcomes. Plos Med. 2011;8(9).

6. Steinhardt LC, Yeka A, Nasr S, Wiegand RE, Rubahika D, Sserwanga A, et al. The effect of indoor residual spraying on malaria and anemia in a high-transmission area of northern Uganda. Am J Trop Med Hyg. 2013;88(5):855-61.

 Ashley EA, Dhorda M, Fairhurst RM, Amaratunga C, Lim P, Suon S, et al. Spread of artemisinin resistance in *Plasmodium falciparum* malaria. New Engl J Med. 2014;371(5):411-23.

8. Dondorp AM, Nosten F, Yi P, Das D, Phyo AP, Tarning J, et al. Artemisinin resistance in *Plasmodium falciparum* malaria. New Engl J Med. 2009;361(5):455-67.

9. Ranson H, N'Guessan R, Lines J, Moiroux N, Nkuni Z, Corbel V. Pyrethroid resistance in African anopheline mosquitoes: what are the implications for malaria control? Trends Parasitol. 2011;27(2):91-8.

10. Oduola AO, Idowu ET, Oyebola MK, Adeogun AO, Olojede JB, Otubanjo OA, et al. Evidence of carbamate resistance in urban populations of *Anopheles gambiae s.s.* mosquitoes resistant to DDT and deltamethrin insecticides in Lagos, South-Western Nigeria. Parasite Vector. 2012;5.

11. Wondji CS, De Silva WAPP, Hemingway J, Ranson H, Karunaratne SHPP. Characterization of knockdown resistance in DDT- and pyrethroid-resistant *Culex quinquefasciatus* populations from Sri Lanka. Trop Med Int Health. 2008;13(4):548-55.

12. World Health Organization. Department of communicable disease prevention cae. Indoor residual spraying: An operational manual for IRS for malaria transmission, control and elimination, second edition. 2015.

283

13. Strode C, Donegan S, Garner P, Enayati AA, Hemingway J. The impact of pyrethroid resistance on the efficacy of insecticide-treated bed nets against African Anopheline mosquitoes: systematic review and meta-analysis. Plos Med. 2014;11(3).

14. Read AF, Lynch PA, Thomas MB. How to make evolution-proof insecticides for malaria control. Plos Biol. 2009;7(4).

Kelly-Hope L, Ranson H, Hemingway J. Lessons from the past: managing insecticide resistance in malaria control and eradication programmes. Lancet Infect Dis. 2008;8(6):387-9.

16. Kraemer MUG, Sinka ME, Duda KA, Mylne AQN, Shearer FM, Barker CM, et al. The global distribution of the arbovirus vectors *Aedes aegypti* and *Ae. albopictus*. Elife. 2015;4.

17. Weaver SC, Lecuit M. Chikungunya virus and the global spread of a mosquito-borne disease. New Engl J Med. 2015;372(13):1231-9.

18. Martens P, Kovats RS, Nijhof S, de Vries P, Livermore MTJ, Bradley DJ, et al. Climate change and future populations at risk of malaria. Global Environ Chang. 1999;9:S89-S107.

19. McMichael AJ, Woodruff RE, Hales S. Climate change and human health: present and future risks. Lancet. 2006;367(9513):859-69.

20. Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013;496(7446):504-7.

21. Liu-Helmersson J, Stenlund H, Wilder-Smith A, Rocklov J. Vectorial capacity of *Aedes aegypti*: effects of temperature and implications for global dengue epidemic potential. Plos One. 2014;9(3).

22. Ausborn J, Wolf H, Mader W, Kayser H. The insecticide pymetrozine selectively affects chordotonal mechanoreceptors. J Exp Biol. 2005;208(23):4451-66.

23. Harrewijn P, Kayser H. Pymetrozine, a fast-acting and selective inhibitor of aphid feeding. In-situ studies with electronic monitoring of feeding behaviour. Pestic Sci. 1997;49(2):130-40.

24. Boekhoff-Falk G, Eberl DF. The *Drosophila* auditory system. Wires Dev Biol. 2014;3(2):179-91.

25. Kernan MJ. Mechanotransduction and auditory transduction in *Drosophila*. Pflug Arch Eur J Phy. 2007;454(5):703-20.

26. Maienfisch P. New unknown modes of action. Modern Crop Protection Compounds, Vols 1-3, 2nd Edition. 2012:1327-46.

27. Nesterov A, Spalthoff C, Kandasamy R, Katana R, Rankl NB, Andres M, et al. TRP channels in insect stretch receptors as insecticide targets. Neuron. 2015;86(3):665-71.

28. Kavlie RG, Albert JT. Chordotonal organs. Curr Biol. 2013;23(9):R334-R5.

29. Fuller SB, Straw AD, Peek MY, Murray RM, Dickinson MH. Flying *Drosophila* stabilize their vision-based velocity controller by sensing wind with their antennae. P Natl Acad Sci USA. 2014;111(13):E1182-E91.

30. Ejima A, Griffith LC. Courtship initiation is stimulated by acoustic signals in *Drosophila melanogaster*. Plos One. 2008;3(9).

31. Ferveur JF. *Drosophila* female courtship and mating behaviors: sensory signals, genes, neural structures and evolution. Curr Opin Neurobiol. 2010;20(6):764-9.

32. Bohbot JD, Sparks JT, Dickens JC. The maxillary palp of *Aedes aegypti*, a model of multisensory integration. Insect Biochem Molec. 2014;48:29-39.

33. Simoni A, Wolfgang W, Topping MP, Kavlie RG, Stanewsky R, Albert JT. A mechanosensory pathway to the *Drosophila* circadian clock. Science. 2014;343.

34. Gibson G, Warren B, Russell IJ. Humming in tune: sex and species recognition by mosquitoes on the wing. Jaro-J Assoc Res Oto. 2010;11(4):527-40.

35. Kassebaum NJ, Arora M, Barber RM, Bhutta ZA, Brown J, Carter A, et al. Global, regional, and national disability-adjusted life-years (DALYs) for 315 diseases and injuries and healthy life expectancy (HALE), 1990-2015: a systematic analysis for the Global Burden of Disease Study 2015. The Lancet. 2016;388(10053):1603-58.

36. Murray CJL, Lopez AD. Mortality by cause for eight regions of the world: Global Burden of Disease Study. Lancet. 1997;349(9061):1269-76.

37. Corruccini RS, Kaul SS. The epidemiological transition and anthropology of minor chronic non - infectious diseases. Medical Anthropology. 1983;7(3):36-50.

38. World Health Organization. Global Health Observatory. The 10 leading causes of death by country income group. 2012.

39. Hotez PJ, Alvarado M, Basanez MG, Bolliger I, Bourne R, Boussinesq M, et al. The Global Burden of Disease study 2010: Interpretation and implications for the Neglected Tropical Diseases. Plos Neglect Trop D. 2014;8(7).

40. Vos T, Barber RM, Bell B, Bertozzi-Villa A, Biryukov S, Bolliger I, et al. Global, regional, and national incidence, prevalence, and years lived with disability for 301 acute and chronic diseases and injuries in 188 countries, 1990-2013: a systematic analysis for the Global Burden of Disease Study 2013. Lancet. 2015;386(9995):743-800.

41. Wilkerson RC, Linton YM, Fonseca DM, Schultz TR, Price DC, Strickman DA. Making mosquito taxonomy useful: A stable classification of tribe Aedini that balances utility with current knowledge of evolutionary relationships. Plos One. 2015;10(7).

42. Hay SI, Sinka ME, Okara RM, Kabaria CW, Mbithi PM, Tago CC, et al. Developing Global Maps of the Dominant *Anopheles* Vectors of Human Malaria. Plos Med. 2010;7(2).

43. Greenwood BM, Bojang K, Whitty CJM, Targett GAT. Malaria. Lancet. 2005;365(9469):1487-98.

44. Morrison AC, Zielinski-Gutierrez E, Scott TW, Rosenberg R. Defining challenges and proposing solutions for control of the virus vector *Aedes aegypti*. Plos Med. 2008;5(3):362-6.

45. Arensburger P, Megy K, Waterhouse RM, Abrudan J, Amedeo P, Antelo B, et al. Sequencing of *Culex quinquefasciatus* establishes a platform for mosquito comparative genomics. Science. 2010;330(6000):86-8.

46. World Health Organization. Department of communicable disease prevention cae. Zika virus fact sheet. 2016.

47. Scott TW, Takken W. Feeding strategies of anthropophilic mosquitoes result in increased risk of pathogen transmission. Trends Parasitol. 2012;28(3):114-21.

48. Wilke ABB, Marrelli MT. Paratransgenesis: a promising new strategy for mosquito vector control. Parasite Vector. 2015;8.

49. Vinogradova EB. *Culex pipiens pipiens* mosquitoes: taxonomy, distribution, ecology, physiology, genetics, applied importance and control: Pensoft; 2000.

50. Chan M, Johansson MA. The incubation periods of dengue viruses. Plos One. 2012;7(11).

51. Goel TGG, A.;. Lymphatic filariasis: Springer; 2016.

52. Paaijmans KP, Read AF, Thomas MB. Understanding the link between malaria risk and climate. P Natl Acad Sci USA. 2009;106(33):13844-9.

53. Clements AN. The biology of mosquitoes. Vol 3: Transmission of viruses and interactions with bacteria2012.

54. Charrel RN, Leparc-Goffart I, Gallian P, de Lamballerie X. Globalization of Chikungunya: 10 years to invade the world. Clin Microbiol Infec. 2014;20(7):662-3.

55. Brady OJ, Gething PW, Bhatt S, Messina JP, Brownstein JS, Hoen AG, et al. Refining the global spatial limits of dengue virus transmission by evidence-based consensus. Plos Neglect Trop D. 2012;6(8).

56. Clements AN. The biology of mosquitoes. Vol 1: Development, nutrition, and reproduction: Chapman & Hall; 1992.

57. Vanhandel E, Edman JD, Day JF, Scott TW, Clark GG, Reiter P, et al. Plant sugar, glycogen, and lipid assay of *Aedes aegypti* collected in urban Puerto Rico and rural Florida. J Am Mosquito Contr. 1994;10(2):149-53.

58. Putnam JL, Scott TW. The effect of multiple host contacts on the infectivity of dengue-2 virus-infected *Aedes aegypti*. J Parasitol. 1995;81(2):170-4.

59. De Benedictis J, Chow-Shaffer E, Costero A, Clark GG, Edman JD, Scott TW. Identification of the people from whom engorged *Aedes aegypti* took blood meals in Florida,

Puerto Rico, using polymerase chain reaction-based DNA profiling. Am J Trop Med Hyg. 2003;68(4):437-46.

60. Wong J, Astete H, Morrison AC, Scott TW. Sampling considerations for sesigning *Aedes aegypti* (Diptera: Culicidae) oviposition studies in Iquitos, Peru: substrate preference, diurnal periodicity, and gonotrophic cycle length. J Med Entomol. 2011;48(1):45-52.

61. World Health Organization. Department of communicable disease prevention cae. Dengue control: the mosquito. 2016.

62. Faraji A, Unlu I. The eye of the tiger, the thrill of the fight: effective larval and adult control measures against the Asian tiger mosquito, *Aedes albopictus* (Diptera: Culicidae), in North America. J Med Entomol. 2016;53(5):1029-47.

63. World Health Organization. Department of communicable disease prevention cae. World Malaria Report 2016 fact sheet. 2016.

64. Garrett-Jones C, Boreham PFL, Pant CP. Feeding habits of Anophelines (Diptera, Culicidae) in 1971-78, with reference to the human-blood index - a review. B Entomol Res. 1980;70(2):165-85.

65. Quinones ML, Lines JD, Thomson MC, Jawara M, Morris J, Greenwood BM. *Anopheles gambiae* gonotrophic cycle duration, biting and exiting behaviour unaffected by permethrin-impregnated bednets in The Gambia. Med Vet Entomol. 1997;11(1):71-8.

66. Takken W, Klowden MJ, Chambers GM. Effect of body size on host seeking and blood meal utilization in *Anopheles gambiae* sensu stricto (Diptera : Culicidae): The disadvantage of being small. J Med Entomol. 1998;35(5):639-45.

67. Foster WA, Takken W. Nectar-related vs. human-related volatiles: behavioural response and choice by female and male *Anopheles gambiae* (Diptera : Culicidae) between emergence and first feeding. B Entomol Res. 2004;94(2):145-57.

68. Gary RE, Foster WA. Diel timing and frequency of sugar feeding in the mosquito *Anopheles gambiae*, depending on sex, gonotrophic state and resource availability. Med Vet Entomol. 2006;20(3):308-16.

69. Kabbale FG, Akol AM, Kaddu JB, Onapa AW. Biting patterns and seasonality of *Anopheles gambiae* sensu lato and *Anopheles funestus* mosquitoes in Kamuli District, Uganda. Parasit Vectors. 2013.

70. Alonso PL, Tanner M. Public health challenges and prospects for malaria control and elimination. Nat Med. 2013;19(2):150-5.

71. Artis ML, Huestis DL, Lehmann T. The effects of oviposition-site deprivation on longevity and bloodfeeding rate in *Anopheles gambiae*. Parasite Vector. 2014;7.

72. David MR, Ribeiro GS, de Freitas RM. Bionomics of *Culex quinquefasciatus* within urban areas of Rio de Janeiro, Southeastern Brazil. Rev Saude Publ. 2012;46(5):858-65.

73. Samy AM, Elaagip AH, Kenawy MA, Ayres CFJ, Peterson AT, Soliman DE. Climate change influences on the global potential distribution of the mosquito *Culex quinquefasciatus*, vector of West Nile virus and lymphatic filariasis. Plos One. 2016;11(10).

74. Janssen N, Fernandez-Salas I, Gonzalez EED, Gaytan-Burns A, Medina-de la Garza CE, Sanchez-Casas RM, et al. Mammalophilic feeding behaviour of *Culex quinquefasciatus* mosquitoes collected in the cities of Chetumal and Cancun, Yucatan Peninsula, Mexico. Trop Med Int Health. 2015;20(11):1488-91.

75. Takken W, Verhulst NO. Host preferences of blood feeding mosquitoes. Annual Review of Entomology, Vol 58. 2013;58:433-+.

76. Clements AN. The biology of mosquitoes. Vol. 2: Sensory reception and behaviour. New York: CABI Publishing; 1999.

77. Muturi EJ, Muriu S, Shililu J, Mwangangi JM, Jacob BG, Mbogo C, et al. Bloodfeeding patterns of *Culex quinquefasciatus* and other culicines and implications for disease transmission in Mwea rice scheme, Kenya. Parasitol Res. 2008;102(6):1329-35.

78. Barbazan P, Baldet T, Darriet F, Escaffre H, Djoda DH, Hougard JM. Control of *Culex quinquefasciatus* (Diptera: Culicidae) with *Bacillus sphaericus* in Maroua, Cameroon. J Am Mosquito Contr. 1997;13(3):263-9.

79. Muller GC, Junnila A, Qualls W, Revay EE, Kline DL, Allan S, et al. Control of *Culex quinquefasciatus* in a storm drain system in Florida using attractive toxic sugar baits. Med Vet Entomol. 2010;24(4):346-51.

80. Kirby MJ, West P, Green C, Jasseh M, Lindsay SW. Risk factors for house-entry by culicine mosquitoes in a rural town and satellite villages in The Gambia. Parasite Vector. 2008;1.

81. Klowden MJ, Briegel H. Mosquito gonotrophic cycle and multiple feeding potential - contrasts between *Anopheles* and *Aedes* (Diptera, Culicidae). J Med Entomol. 1994;31(4):618-22.

82. Aly ASI, Vaughan AM, Kappe SHI. Malaria parasite development in the mosquito and infection of the mammalian host. Annu Rev Microbiol. 2009;63:195-221.

83. Beerntsen BT, James AA, Christensen BM. Genetics of mosquito vector competence. Microbiol Mol Biol R. 2000;64(1):115-+.

84. Erickson SM, Xi ZY, Mayhew GF, Ramirez JL, Aliota MT, Christensen BM, et al. Mosquito infection responses to developing filarial worms. Plos Neglect Trop D. 2009;3(10).

85. Ponlawat A, Harrington LC. Blood feeding patterns of *Aedes aegypti* and *Aedes albopictus* in Thailand. J Med Entomol. 2005;42(5):844-9.

86. Scott TW, Amerasinghe PH, Morrison AC, Lorenz LH, Clark GG, Strickman D, et al. Longitudinal studies of *Aedes aegypti* (Diptera : Culicidae) in Thailand and Puerto Rico: blood feeding frequency. J Med Entomol. 2000;37(1):89-101.
87. Yohannes M, Boelee E. Early biting rhythm in the afro-tropical vector of malaria, *Anopheles arabiensis*, and challenges for its control in Ethiopia. Med Vet Entomol. 2012;26(1):103-5.

88. Gunathilaka N, Denipitiya T, Hapugoda M, Abeyewickreme W, Wickremasinghe R. Determination of the foraging behaviour and blood meal source of malaria vector mosquitoes in Trincomalee District of Sri Lanka using a multiplex real time polymerase chain reaction assay. Malaria J. 2016;15.

89. Harrington LC, Edman JD, Scott TW. Why do female *Aedes aegypti* (Diptera : Culicidae) feed preferentially and frequently on human blood? J Med Entomol. 2001;38(3):411-22.

90. McBride CS, Baier F, Omondi AB, Spitzer SA, Lutomiah J, Sang R, et al. Evolution of mosquito preference for humans linked to an odorant receptor. Nature. 2014;515(7526):222-U151.

91. Fernandez-Grandon GM, Gezan SA, Armour JAL, Pickett JA, Logan JG. Heritability of attractiveness to mosquitoes. Plos One. 2015;10(4).

92. Gibson G, Torr SJ. Visual and olfactory responses of haematophagous Diptera to host stimuli. Med Vet Entomol. 1999;13(1):2-23.

93. Corfas RA, Vosshall LB. The cation channel TRPA1 tunes mosquito thermotaxis to host temperatures. Elife. 2015;4.

94. Dekker T, Carde RT. Moment-to-moment flight manoeuvres of the female yellow fever mosquito (*Aedes aegypti* L.) in response to plumes of carbon dioxide and human skin odour. J Exp Biol. 2011;214(20):3480-94.

95. Logan JG, Birkett MA, Clark SJ, Powers S, Seal NJ, Wadhams LJ, et al. Identification of human-derived volatile chemicals that interfere with attraction of *Aedes aegypti* mosquitoes. J Chem Ecol. 2008;34(3):308-22.

96. Turner SL, Li N, Guda T, Githure J, Carde RT, Ray A. Ultra-prolonged activation of CO₂-sensing neurons disorients mosquitoes. Nature. 2011;474(7349):87-U114.

97. McMeniman CJ, Corfas RA, Matthews BJ, Ritchie SA, Vosshall LB. Multimodal integration of carbon dioxide and other sensory cues drives mosquito attraction to humans. Cell. 2014;156(5):1060-71.

98. Zwiebel LJ, Takken W. Olfactory regulation of mosquito-host interactions. Insect Biochem Molec. 2004;34(7):645-52.

99. Hallem EA, Fox AN, Zwiebel LJ, Carlson JR. Olfaction: Mosquito receptor for humansweat odorant. Nature. 2004;427(6971):212-3.

100. Liu C, Pitts RJ, Bohbot JD, Jones PL, Wang GR, Zwiebel LJ. Distinct olfactory signaling mechanisms in the malaria vector mosquito *Anopheles gambiae*. Plos Biol. 2010;8(8).

101. Riabinina O, Task D, Marr E, Lin CC, Alford R, O'Brochta DA, et al. Organization of olfactory centres in the malaria mosquito *Anopheles gambiae*. Nat Commun. 2016;7.

102. Cator LJ, Lynch PA, Read AF, Thomas MB. Do malaria parasites manipulate mosquitoes? Trends Parasitol. 2012;28(11):466-70.

103. Koella JC, Sorensen FL, Anderson RA. The malaria parasite, *Plasmodium falciparum*, increases the frequency of multiple feeding of its mosquito vector, *Anopheles gambiae*. P Roy Soc B-Biol Sci. 1998;265(1398):763-8.

104. Smallegange RC, van Gemert GJ, van de Vegte-Bolmer M, Gezan S, Takken W, Sauerwein RW, et al. Malaria infected mosquitoes express enhanced attraction to human odor. Plos One. 2013;8(5).

105. De Moraes CM, Stanczyk NM, Betz HS, Pulido H, Sim DG, Read AF, et al. Malariainduced changes in host odors enhance mosquito attraction. P Natl Acad Sci USA. 2014;111(30):11079-84.

106. Izumi H, Suzuki M, Aoyagi S, Kanzaki T. Realistic imitation of mosquito's proboscis: electrochemically etched sharp and jagged needles and their cooperative inserting motion. Sensor Actuat a-Phys. 2011;165(1):115-23.

107. Shirai Y, Kamimura K, Seki T, Morohashi M. Proboscis amputation facilitates the study of mosquito (Diptera : Culicidae) attractants, repellents, and host preference. J Med Entomol. 2000;37(4):637-9.

108. Salama HS. Function of mosquito taste receptors. Journal of Insect Physiology. 1966;12(9).

109. Lee SJ, Kim BH, Lee JY. Experimental study on the fluid mechanics of blood sucking in the proboscis of a female mosquito. J Biomech. 2009;42(7):857-64.

110. Lee RMKWC, D. A. Fine structure of the sense organs on the labella and labium of the mosquito *Aedes aegypti* (L.). The Open Entomology Journal. 2009;3:7-17.

111. Pappas LGL, J.R. Labellar chordotonal organs of the mosquito *Culiseta inornata* (Williston) (Diptera: Culicidae). Int J Insect Morphol & Embryol 1976;5(2):145-50.

112. Tirados I, Costantini C, Gibson G, Torr SJ. Blood-feeding behaviour of the malarial mosquito *Anopheles arabiensis*: implications for vector control. Med Vet Entomol. 2006;20(4):425-37.

113. Pates H, Curtis C. Mosquito behavior and vector control. Annu Rev Entomol. 2005;50:53-70.

114. Gubler DJ. Population growth, urbanization, automobiles and airplanes: The dengue connection. In: Greenwood B, De Cock K, editors. New and Resurgent Infections: Prediction, Detection and Management of Tomorrow's Epidemics. London: London School of Hygiene and Tropical Medicine; 1998. p. 117–29.

115. Hay SI, Guerra CA, Tatem AJ, Atkinson PM, Snow RW. Urbanization, malaria transmission and disease burden in Africa. Nat Rev Microbiol. 2005;3:81 - 90.

116. Nutman TB. Lymphatic filariasis: Imperial College Press, London; 2000.

117. World Health Organization. Department of communicable disease prevention cae. World malaria report 2015. 2015.

118. Ramzy RMR, El Setouhy M, Helmy H, Ahmed ES, Abd Elaziz KM, Farid HA, et al. Effect of yearly mass drug administration with diethylcarbamazine and albendazole on bancroftian filariasis in Egypt: a comprehensive assessment. Lancet. 2006;367(9515):992-9. 119. Adjuik M, Agnamey P, Babiker A, Baptista J, Borrmann S, Brasseur P, et al. Artesunate combinations for treatment of malaria: meta-analysis. Lancet. 2004;363(9402):9-17.

120. Kobylinski KC, Deus KM, Butters MP, Hongyu T, Gray M, da Silva IM, et al. The effect of oral anthelmintics on the survivorship and re-feeding frequency of anthropophilic mosquito disease vectors. Acta Tropica. 2010;116(2):119-26.

121. World Health Organization. Department of communicable disease prevention cae. Yellow fever fact sheet. 2016.

122. World Health Organization. Department of communicable disease prevention cae. Questions and answers on dengue vaccines. 2016.

123. World Health Organization. Department of communicable disease prevention cae. Questions and answers on RTS,S/ASO1 malaria vaccine. 2016.

124. Davies TGE, Field LM, Usherwood PNR, Williamson MS. DDT, pyrethrins, pyrethroids and insect sodium channels. lubmb Life. 2007;59(3):151-62.

125. Soderlund DM, Clark JM, Sheets LP, Mullin LS, Piccirillo VJ, Sargent D, et al. Mechanisms of pyrethroid neurotoxicity: implications for cumulative risk assessment. Toxicology. 2002;171(1):3-59.

126. Fukuto TR. Mechanism of action of organophosphorus and carbamate insecticides. Environ Health Persp. 1990;87:245-54.

127. Casida JE. Mode of action of carbamates. Annu Rev Entomol. 1963;8:39-&.

128. World Health Organization. Department of communicable disease prevention cae. Global plan for insecticide resistance management in malaria vectors. 2012.

129. Gamble C, Ekwaru JP, Ter KFO. Insecticide-treated nets for preventing malaria in pregnancy. Cochrane Db Syst Rev. 2006(2).

130. Rowland M, Boko P, Odjo A, Asidi A, Akogbeto M, N'Guessan R. A new long-lasting indoor residual formulation of the organophosphate insecticide pirimiphos methyl for prolonged control of pyrethroid-resistant mosquitoes: an experimental hut trial in Benin. Plos One. 2013;8(7).

131. Wilson AL, Dhiman RC, Kitron U, Scott TW, van den Berg H, Lindsay SW. Benefit of insecticide-treated nets, curtains and screening on vector borne diseases, excluding malaria: a systematic review and meta-analysis. Plos Neglect Trop D. 2014;8(10).

132. Fullman N, Burstein R, Lim SS, Medlin C, Gakidou E. Nets, spray or both? The effectiveness of insecticide-treated nets and indoor residual spraying in reducing malaria morbidity and child mortality in sub-Saharan Africa. Malaria J. 2013;12.

133. Pinder M, Jawara M, Jarju LBS, Salami K, Jeffries D, Adiamoh M, et al. Efficacy of indoor residual spraying with dichlorodiphenyltrichloroethane against malaria in Gambian communities with high usage of long-lasting insecticidal mosquito nets: a cluster-randomised controlled trial. Lancet. 2015;385(9976):1436-46.

134. West PA, Protopopoff N, Wright A, Kivaju Z, Tigererwa R, Mosha FW, et al. Indoor residual spraying in combination with insecticide-treated nets compared to insecticide-treated nets alone for protection against malaria: A cluster randomised trial in Tanzania. Plos Med. 2014;11(4).

135. Killeen GF, Govella NJ, Lwetoijera DW, Okumu FO. Most outdoor malaria transmission by behaviourally-resistant *Anopheles arabiensis* is mediated by mosquitoes that have previously been inside houses. Malaria J. 2016;15.

136. Kitau J, Oxborough RM, Tungu PK, Matowo J, Malima RC, Magesa SM, et al. Species shifts in the *Anopheles gambiae* complex: do LLINs successfully control *Anopheles arabiensis*? Plos One. 2012;7(3).

137. Wilson AL, Chen-Hussey V, Logan JG, Lindsay SW. Are topical insect repellents effective against malaria in endemic populations? A systematic review and meta-analysis. Malaria J. 2014;13.

138. Sluydts V, Durnez L, Heng S, Gryseels C, Canier L, Kim S, et al. Efficacy of topical mosquito repellent (picaridin) plus long-lasting insecticidal nets versus long-lasting insecticidal nets alone for control of malaria: a cluster randomised controlled trial. Lancet Infect Dis. 2016;16(10):1169-77.

139. Gryseels C, Uk S, Sluydts V, Durnez L, Phoeuk P, Suon S, et al. Factors influencing the use of topical repellents: implications for the effectiveness of malaria elimination strategies. Sci Rep-Uk. 2015;5.

140. Orsborne J, Banks SD, Hendy A, Gezan SA, Kaur H, Wilder-Smith A, et al. Personal protection of permethrin-treated clothing against *Aedes aegypti*, the vector of dengue and Zika virus, in the laboratory. Plos One. 2016;11(5).

141. Tusting LS, Thwing J, Sinclair D, Fillinger U, Gimnig J, Bonner KE, et al. Mosquito larval source management for controlling malaria. Cochrane Db Syst Rev. 2013(8).

142. Maheu-Giroux M, Castro MC. Impact of community-based larviciding on the prevalence of malaria infection in Dar es Salaam, Tanzania. Plos One. 2013;8(8).

143. Imbahale SS, Githeko A, Mukabana WR, Takken W. Integrated mosquito larval source management reduces larval numbers in two highland villages in western Kenya. Bmc Public Health. 2012;12.

144. Fillinger U, Lindsay SW. Larval source management for malaria control in Africa: myths and reality. Malaria J. 2011;10.

145. Mbare O, Lindsay SW, Fillinger U. Dose-response tests and semi-field evaluation of lethal and sub-lethal effects of slow release pyriproxyfen granules (Sumilarv (R) 0.5G) for the control of the malaria vectors *Anopheles gambiae sensu lato*. Malaria J. 2013;12.

146. Mbare O, Lindsay SW, Fillinger U. Aquatain (R) Mosquito Formulation (AMF) for the control of immature *Anopheles gambiae sensu stricto* and *Anopheles arabiensis*: dose-responses, persistence and sub-lethal effects. Parasite Vector. 2014;7.

147. Devine GJ, Perea EZ, Killeen GF, Stancil JD, Clark SJ, Morrison AC. Using adult mosquitoes to transfer insecticides to *Aedes aegypti* larval habitats. P Natl Acad Sci USA. 2009;106(28):11530-4.

148. Achee NL, Gould F, Perkins TA, Reiner RC, Morrison AC, Ritchie SA, et al. A critical assessment of vector control for dengue prevention. Plos Neglect Trop D. 2015;9(5).

149. Unlu I, Farajollahi A, Healy SP, Crepeau T, Bartlett-Healy K, Williges E, et al. Areawide management of *Aedes albopictus*: choice of study sites based on geospatial characteristics, socioeconomic factors and mosquito populations. Pest Manag Sci. 2011;67(8):965-74.

150. Fonseca DM, Unlu I, Crepeau T, Farajollahi A, Healy SP, Bartlett-Healy K, et al. Area-wide management of *Aedes albopictus*. Part 2: Gauging the efficacy of traditional integrated pest control measures against urban container mosquitoes. Pest Manag Sci. 2013;69(12):1351-61.

151. Williams GM, Faraji A, Unlu I, Healy SP, Farooq M, Gaugler R, et al. Area-wide ground applications of *Bacillus thuringiensis* var. *israelensis* for the control of *Aedes albopictus* in residential neighborhoods: from optimization to operation. Plos One. 2014;9(10).

152. Suman DS, Farajollahi A, Healy S, Williams GM, Wang Y, Schoeler G, et al. Pointsource and area-wide field studies of pyriproxyfen autodissemination against urban container-inhabiting mosquitoes. Acta Tropica. 2014;135:96-103.

153. Muriu SM, Coulson T, Mbogo CM, Godfray HCJ. Larval density dependence in *Anopheles gambiae* s.s., the major African vector of malaria. J Anim Ecol. 2013;82(1):166-74.

154. Hawley WA. The effect of larval density on adult longevity of a mosquito, *Aedes sierrensis* - epidemiological consequences. J Anim Ecol. 1985;54(3):955-64.

155. Rapley LP, Johnson PH, Williams CR, Silcock RM, Larkman M, Long SA, et al. A lethal ovitrap-based mass trapping scheme for dengue control in Australia: II. Impact on populations of the mosquito *Aedes aegypti*. Med Vet Entomol. 2009;23(4):303-16.

156. Barrera R, Amador M, Acevedo V, Caban B, Felix G, Mackay AJ. Use of the CDC autocidal gravid ovitrap to control and prevent outbreaks of *Aedes aegypti* (Diptera: Culicidae). J Med Entomol. 2014;51(1):145-54.

157. Beier JC, Muller GC, Gu WD, Arheart KL, Schlein Y. Attractive toxic sugar bait (ATSB) methods decimate populations of *Anopheles* malaria vectors in arid environments regardless of the local availability of favoured sugar-source blossoms. Malaria J. 2012;11.

158. Qualls WA, Muller GC, Traore SF, Traore MM, Arheart KL, Doumbia S, et al. Indoor use of attractive toxic sugar bait (ATSB) to effectively control malaria vectors in Mali, West Africa. Malaria J. 2015;14.

159. Enayati A, Hemingway J. Malaria management: past, present, and future. Annu Rev Entomol. 2010;55:569-91.

160. Muller GC, Beier JC, Traore SF, Toure MB, Traore MM, Bah S, et al. Successful field trial of attractive toxic sugar bait (ATSB) plant-spraying methods against malaria vectors in the *Anopheles gambiae* complex in Mali, West Africa. Malaria J. 2010;9.

161. Stewart ZP, Oxborough RM, Tungu PK, Kirby MJ, Rowland MW, Irish SR. Indoor application of attractive toxic sugar bait (ATSB) in combination with mosquito nets for control of pyrethroid resistant mosquitoes. Plos One. 2013.

162. Donnelly B, Berrang-Ford L, Ross NA, Michel P. A systematic, realist review of zooprophylaxis for malaria control. Malaria J. 2015;14.

163. Habtewold T, Prior A, Torr SJ, Gibson G. Could insecticide-treated cattle reduce Afrotropical malaria transmission? Effects of deltamethrin-treated Zebu on *Anopheles arabiensis* behaviour and survival in Ethiopia. Med Vet Entomol. 2004;18(4):408-17.

164. Howard AFV, Koenraadt CJM, Farenhorst M, Knols BGJ, Takken W. Pyrethroid resistance in *Anopheles gambiae* leads to increased susceptibility to the entomopathogenic fungi *Metarhizium anisopliae* and *Beauveria bassiana*. Malaria J. 2010;9.

165. Kikankie CK, Brooke BD, Knols BGJ, Koekemoer LL, Farenhorst M, Hunt RH, et al. The infectivity of the entomopathogenic fungus *Beauveria bassiana* to insecticide-resistant and susceptible *Anopheles arabiensis* mosquitoes at two different temperatures. Malaria J. 2010;9.

166. Ye YXH, Carrasco AM, Frentiu FD, Chenoweth SF, Beebe NW, van den Hurk AF, et al. *Wolbachia* reduces the transmission potential of dengue-infected *Aedes aegypti*. Plos Neglect Trop D. 2015;9(6).

167. Scholte EJ, Ng'habi K, Kihonda J, Takken W, Paaijmans K, Abdulla S, et al. An entomopathogenic fungus for control of adult African malaria mosquitoes. Science. 2005;308(5728):1641-2.

168. Lynch PA, Grimm U, Thomas MB, Read AF. Prospective malaria control using entomopathogenic fungi: comparative evaluation of impact on transmission and selection for resistance. Malaria J. 2012;11.

169. Blanford S, Jenkins NE, Read AF, Thomas MB. Evaluating the lethal and pre-lethal effects of a range of fungi against adult *Anopheles stephensi* mosquitoes. Malaria J. 2012;11.

170. Caragata EP, Dutra HLC, Moreira LA. Exploiting intimate relationships: controlling mosquito-transmitted disease with *Wolbachia*. Trends Parasitol. 2016;32(3):207-18.

171. Walker T, Johnson PH, Moreira LA, Iturbe-Ormaetxe I, Frentiu FD, McMeniman CJ, et al. The wMel *Wolbachia* strain blocks dengue and invades caged *Aedes aegypti* populations. Nature. 2011;476(7361):450-U101.

172. Loreto ELS, Wallau GL. Risks of *Wolbachia* mosquito control. Science. 2016;351(6279):1273-.

173. Marinotti O, Jasinskiene N, Fazekas A, Scaife S, Fu GL, Mattingly ST, et al. Development of a population suppression strain of the human malaria vector mosquito, *Anopheles stephensi*. Malaria J. 2013;12.

174. McGraw EA, O'Neill SL. Beyond insecticides: new thinking on an ancient problem. Nature Reviews Microbiology. 2013;11(3):181-93.

175. Benedict MQ, Robinson AS. The first releases of transgenic mosquitoes: an argument for the sterile insect technique. Trends Parasitol. 2003;19(8):349-55.

176. Alphey N, Coleman PG, Donnelly CA, Alphey L. Managing insecticide resistance by mass release of engineered insects. J Econ Entomol. 2007;100(5):1642-9.

177. Phuc HK, Andreasen MH, Burton RS, Vass C, Epton MJ, Pape G, et al. Late-acting dominant lethal genetic systems and mosquito control. Bmc Biol. 2007;5.

178. Fu GL, Lees RS, Nimmo D, Aw D, Jin L, Gray P, et al. Female-specific flightless phenotype for mosquito control. P Natl Acad Sci USA. 2010;107(10):4550-4.

179. Evans DS, Alphonsus K, Umaru J, Eigege A, Miri E, Mafuyai H, et al. Status of onchocerciasis transmission after more than a decade of mass drug administration for onchocerciasis and lymphatic filariasis elimination in central Nigeria: challenges in coordinating the stop MDA decision. Plos Neglect Trop D. 2014;8(9).

180. Lo NC, Bogoch II, Blackburn BG, Raso G, N'Goran EK, Coulibaly JT, et al. Comparison of community-wide, integrated mass drug administration strategies for schistosomiasis and soil-transmitted helminthiasis: a cost-effectiveness modelling study. Lancet Glob Health. 2015;3(10):E629-E38.

181. World Health Organization. Department of communicable disease prevention cae. Dengue and severe dengue. 2016.

182. World Health Organization. Department of communicable disease prevention cae. Chikungunya fact sheet. 2016.

183. Kuehne A, Tiffany A, Lasry E, Janssens M, Besse C, Okonta C, et al. Impact and lessons learned from mass drug administrations of malaria chemoprevention during the Ebola outbreak in Monrovia, Liberia, 2014. Plos One. 2016;11(8).

184. Radeva-Petrova D, Kayentao K, ter Kuile FO, Sinclair D, Garner P. Drugs for preventing malaria in pregnant women in endemic areas: any drug regimen versus placebo or no treatment. Cochrane Db Syst Rev. 2014(10).

185. Poirot E, Skarbinski J, Sinclair D, Kachur SP, Slutsker L, Hwang J. Mass drug administration for malaria. Cochrane Db Syst Rev. 2013(12).

186. World Health Organization. Department of communicable disease prevention cae. Seasonal malaria chemoprevention (SMC). 2016.

187. Cairns M, Roca-Feltrer A, Garske T, Wilson AL, Diallo D, Milligan PJ, et al. Estimating the potential public health impact of seasonal malaria chemoprevention in African children. Nat Commun. 2012;3.

188. Anderson RM, May RM. Infectious diseases of humans: dynamics and control: Oxford University Press; 1992.

189. World Health Organization. Department of communicable disease prevention cae. Guidelines for the treatment of malaria, third edition. 2015.

190. World Health Organization. Department of communicable disease prevention cae. Q&A on artemisinin resistance. 2016.

191. World Health Organization. Department of communicable disease prevention cae. Dengue guidelines for diagnosis, treatment, prevention and control: new edition. 2009.

192. Andre FE, Booy R, Bock HL, Clemens J, Datta SK, John TJ, et al. Vaccination greatly reduces disease, disability, death and inequity worldwide. B World Health Organ. 2008;86(2):140-6.

193. Moormann AM, Stewart VA. The hunt for protective correlates of immunity to *Plasmodium falciparum* malaria. BMC Med. 2014;12.

194. Marsh K, Kinyanjui S. Immune effector mechanisms in malaria. Parasite Immunol. 2006;28(1-2):51-60.

195. World Health Organization. Department of communicable disease prevention cae. Malaria vaccine development. 2016.

196. Tinto H, D'Alessandro U, Sorgho H, Valea I, Tahita MC, Kabore W, et al. Efficacy and safety of RTS,S/AS01 malaria vaccine with or without a booster dose in infants and

children in Africa: final results of a phase 3, individually randomised, controlled trial. Lancet. 2015;386(9988):31-45.

197. Olotu A, Fegan G, Wambua J, Nyangweso G, Leach A, Lievens M, et al. Seven-year efficacy of RTS, S/AS01 malaria vaccine among young African children. New Engl J Med. 2016;374(26):2519-29.

198. Minsoko PA, Lell B, Fernandes JF, Abossolo BP, Kabwende AL, Adegnika AA, et al. Efficacy and safety of the RTS,S/AS01 malaria vaccine during 18 months after vaccination: a phase 3 randomized, controlled trial in children and young infants at 11 African sites. Plos Med. 2014;11(7).

199. World Health Organization. Department of communicable disease prevention cae. Tables of malaria vaccine projects globally. 2016.

200. Garske T, Van Kerkhove MD, Yactayo S, Ronveaux O, Lewis RF, Staples JE, et al. Yellow fever in Africa: estimating the burden of disease and impact of mass vaccination from outbreak and serological data. Plos Med. 2014;11(5).

201. Ferguson NM, Rodriguez-Barraquer I, Dorigatti I, Mier-y-Teran-Romero L, Laydon DJ, Cummings DAT. Benefits and risks of the Sanofi-Pasteur dengue vaccine: Modeling optimal deployment. Science. 2016;353(6303):1033-6.

202. World Health Organization. Department of communicable disease prevention cae. Zika virus vaccine product development. 2016.

203. Abbink P, Larocca RA, De La Barrera RA, Bricault CA, Moseley ET, Boyd M, et al. Protective efficacy of multiple vaccine platforms against Zika virus challenge in rhesus monkeys. Science. 2016;353(6304):1129-32.

204. World Health Organization. Department of communicable disease prevention cae. Insecticide resistance. 2015.

205. Harris AF, Rajatileka S, Ranson H. Pyrethroid resistance in *Aedes aegypti* from Grand Cayman. Am J Trop Med Hyg. 2010;83(2):277-84.

206. Low VL, Chen CD, Lee HL, Tan TK, Chen CF, Leong CS, et al. Enzymatic characterization of insecticide resistance mechanisms in field populations of Malaysian *Culex quinquefasciatus Say* (Diptera: Culicidae). Plos One. 2013;8(11).

207. Mulamba C, Riveron JM, Ibrahim SS, Irving H, Barnes KG, Mukwaya LG, et al. Widespread pyrethroid and DDT resistance in the major malaria vector *Anopheles funestus* in East Africa is driven by metabolic resistance mechanisms. Plos One. 2014;9(10).

208. Agossa FR, Gnanguenon V, Anagonou R, Azondekon R, Aizoun N, Sovi A, et al. Impact of insecticide resistance on the effectiveness of pyrethroid-based malaria vectors control tools in Benin: decreased toxicity and repellent effect. Plos One. 2015;10(12). 209. Kamgang B, Marcombe S, Chandre F, Nchoutpouen E, Nwane P, Etang J, et al. Insecticide susceptibility of *Aedes aegypti* and *Aedes albopictus* in Central Africa. Parasite Vector. 2011;4.

210. Penilla RP, Rodriguez AD, Hemingway J, Trejo A, Lopez AD, Rodriguez MH. Cytochrome P-450-based resistance mechanism and pyrethroid resistance in the field *Anopheles albimanus* resistance management trial. Pestic Biochem Phys. 2007;89(2):111-7.

211. Protopopoff N, Matowo J, Malima R, Kavishe R, Kaaya R, Wright A, et al. High level of resistance in the mosquito *Anopheles gambiae* to pyrethroid insecticides and reduced susceptibility to bendiocarb in north-western Tanzania. Malaria J. 2013;12.

212. Hougard JM, Duchon S, Darriet F, Zaim M, Rogier C, Guillet P. Comparative performances, under laboratory conditions, of seven pyrethroid insecticides used for impregnation of mosquito nets. B World Health Organ. 2003;81(5):324-33.

213. Ishak IH, Riveron JM, Ibrahim SS, Stott R, Longbottom J, Irving H, et al. The Cytochrome P450 gene CYP6P12 confers pyrethroid resistance in kdr-free Malaysian populations of the dengue vector *Aedes albopictus*. Sci Rep-Uk. 2016;6.

214. John R, Ephraim T, Andrew A. Reduced susceptibility to pyrethroid insecticide treated nets by the malaria vector *Anopheles gambiae s.l.* in western Uganda. Malaria J. 2008;7.

215. Shi LN, Hu HX, Ma K, Zhou D, Yu J, Zhong DB, et al. Development of resistance to pyrethroid in *Culex pipiens pallens* population under different insecticide selection pressures. Plos Neglect Trop D. 2015;9(8).

216. Balabanidou V, Kampouraki A, MacLean M, Blomquist GJ, Tittiger C, Juarez MP, et al. Cytochrome P450 associated with insecticide resistance catalyzes cuticular hydrocarbon production in *Anopheles gambiae*. P Natl Acad Sci USA. 2016;113(33):9268-73.

217. Dusfour I, Thalmensy V, Gaborit P, Issaly J, Carinci R, Girod R. Multiple insecticide resistance in *Aedes aegypti* (Diptera: Culicidae) populations compromises the effectiveness of dengue vector control in French Guiana. Mem I Oswaldo Cruz. 2011;106(3):346-52.

218. Nwane P, Etang J, Chouaibou M, Toto JC, Koffi A, Mimpfoundi R, et al. Multiple insecticide resistance mechanisms in *Anopheles gambiae s.l.* populations from Cameroon, Central Africa. Parasite Vector. 2013;6.

219. Okoye PN, Brooke BD, Koekemoer LL, Hunt RH, Coetzee M. Characterisation of DDT, pyrethroid and carbamate resistance in *Anopheles funestus* from Obuasi, Ghana. T Roy Soc Trop Med H. 2008;102(6):591-8.

220. Awolola TS, Oduola OA, Strode C, Koekemoer LL, Brooke B, Ranson H. Evidence of multiple pyrethroid resistance mechanisms in the malaria vector *Anopheles gambiae sensu stricto* from Nigeria. T Roy Soc Trop Med H. 2009;103(11):1139-45.

221. Djouaka RF, Bakare AA, Coulibaly ON, Akogbeto MC, Ranson H, Hemingway J, et al. Expression of the cytochrome P450s, CYP6P3 and CYP6M2 are significantly elevated in multiple pyrethroid resistant populations of *Anopheles gambiae s.s.* from Southern Benin and Nigeria. Bmc Genomics. 2008;9.

222. Ranson H, Jensen B, Vulule JM, Wang X, Hemingway J, Collins FH. Identification of a point mutation in the voltage-gated sodium channel gene of Kenyan *Anopheles gambiae* associated with resistance to DDT and pyrethroids. Insect Mol Biol. 2000;9(5):491-7.

223. Killeen GF, Chitnis N. Potential causes and consequences of behavioural resilience and resistance in malaria vector populations: a mathematical modelling analysis. Malaria J. 2014;13.

224. Gatton ML, Chitnis N, Churcher T, Donnelly MJ, Ghani AC, Godfray HCJ, et al. The importance of mosquito behavioural adaptations to malaria control in Africa. Evolution. 2013;67(4):1218-30.

225. Russell TL, Beebe NW, Cooper RD, Lobo NF, Burkot TR. Successful malaria elimination strategies require interventions that target changing vector behaviours. Malaria J. 2013;12.

226. Djegbe I, Cornelie S, Rossignol M, Demettre E, Seveno M, Remoue F, et al. Differential expression of salivary proteins between susceptible and insecticide-resistant mosquitoes of *Culex quinquefasciatus*. Plos One. 2011;6(3).

227. Shi MA, Lougarre A, Alies C, Fremaux I, Tang ZH, Stojan J, et al. Acetylcholinesterase alterations reveal the fitness cost of mutations conferring insecticide resistance. Bmc Evol Biol. 2004;4.

228. Alout H, Ndam NT, Sandeu MM, Djegbe I, Chandre F, Dabire RK, et al. Insecticide resistance alleles affect vector competence of *Anopheles gambiae s.s.* for *Plasmodium falciparum* field isolates. Plos One. 2013;8(5).

229. Ndiath MO, Cailleau A, Diedhiou SM, Gaye A, Boudin C, Richard V, et al. Effects of the kdr resistance mutation on the susceptibility of wild *Anopheles gambiae* populations to *Plasmodium falciparum*: a hindrance for vector control. Malaria J. 2014;13.

230. Tokponnon FT, Ogouyemi AH, Sissinto Y, Sovi A, Gnanguenon V, Cornelie S, et al. Impact of long-lasting, insecticidal nets on anaemia and prevalence of *Plasmodium falciparum* among children under five years in areas with highly resistant malaria vectors. Malaria J. 2014;13.

231. Chouaibou MS, Chabi J, Bingham GV, Knox TB, N'Dri L, Kesse NB, et al. Increase in susceptibility to insecticides with aging of wild *Anopheles gambiae* mosquitoes from Cote d'Ivoire. Bmc Infect Dis. 2012;12.

232. Rajatileka S, Burhani J, Ranson H. Mosquito age and susceptibility to insecticides. T Roy Soc Trop Med H. 2011;105(5):247-53.

233. Sibley CH. Observing in real time the evolution of artemisinin resistance in *Plasmodium falciparum*. BMC Med. 2015;13.

234. Osei-Atweneboana MY, Eng JL, Boakye DA, Gyapong JO, Prichard RK. Phenotypic evidence of emerging ivermectin resistance in some population of *Onchocerca volvulus*, the causative agent of onchocercasis. Am J Trop Med Hyg. 2007;77(5):111-.

235. Chinappi M, Via A, Marcatili P, Tramontano A. On the mechanism of chloroquine resistance in *Plasmodium falciparum*. Plos One. 2010;5(11).

236. Hetzel MW, Page-Sharp M, Bala N, Pulford J, Betuela I, Davis TME, et al. Quality of antimalarial drugs and antibiotics in Papua New Guinea: a survey of the health facility supply chain. Plos One. 2014;9(5).

237. Nayyar GML, Breman JG, Newton PN, Herrington J. Poor-quality antimalarial drugs in southeast Asia and sub-Saharan Africa. Lancet Infect Dis. 2012;12(6):488-96.

238. Nilsen A, Miley GP, Forquer IP, Mather MW, Katneni K, Li YX, et al. Discovery, synthesis, and optimization of antimalarial 4(1H)-Quinolone-3-Diarylethers. J Med Chem. 2014;57(9):3818-34.

239. Pedrique B, Strub-Wourgaft N, Some C, Olliaro P, Trouiller P, Ford N, et al. The drug and vaccine landscape for neglected diseases (2000-11): a systematic assessment. Lancet Glob Health. 2013;1(6):E371-E9.

240. Chaccour CJ, Kobylinski KC, Bassat Q, Bousema T, Drakeley C, Alonso P, et al. Ivermectin to reduce malaria transmission: a research agenda for a promising new tool for elimination. Malaria J. 2013;12.

241. Chaccour C. Ivermectin to reduce malaria transmission, prospects and challenges. Trop Med Int Health. 2015;20.

242. Panic G, Duthaler U, Speich B, Keiser J. Repurposing drugs for the treatment and control of helminth infections. Int J Parasitol-Drug. 2014;4(3):185-200.

243. Maciel-de-Freitas R, Avendanho FC, Santos R, Sylvestre G, Araujo SC, Lima JBP, et al. Undesirable consequences of insecticide resistance following *Aedes aegypti* control activities due to a dengue outbreak. Plos One. 2014;9(3).

244. Marcombe S, Mathieu RB, Pocquet N, Riaz MA, Poupardin R, Selior S, et al. Insecticide resistance in the dengue vector *Aedes aegypti* from Martinique: distribution, mechanisms and relations with environmental factors. Plos One. 2012;7(2).

245. Norris LC, Norris DE. Insecticide resistance in *Culex quinquefasciatus* mosquitoes after the introduction of insecticide-treated bed nets in Macha, Zambia. J Vector Ecol. 2011;36(2):411-20.

246. Shiff C. Integrated approach to malaria control. Clin Microbiol Rev. 2002;15(2):278-+.

247. Mnzava AP, Knox TB, Temu EA, Trett A, Fornadel C, Hemingway J, et al. Implementation of the global plan for insecticide resistance management in malaria vectors: progress, challenges and the way forward. Malaria J. 2015;14.

248. Pandey UB, Nichols CD. Human disease models in *Drosophila melanogaster* and the role of the fly in therapeutic drug discovery. Pharmacol Rev. 2011;63(2):411-36.

249. Sokolowski MB. *Drosophila*: Genetics meets behaviour. Nat Rev Genet. 2001;2(11):879-90.

250. Vosshall LB. Into the mind of a fly. Nature. 2007;450(7167):193-7.

251. Gratz SJ, Cummings AM, Nguyen JN, Hamm DC, Donohue LK, Harrison MM, et al. Genome engineering of *Drosophila* with the CRISPR RNA-guided Cas9 nuclease. Genetics. 2013;194(4).

252. Jinek M, Chylinski K, Fonfara I, Hauer M, Doudna JA, Charpentier E. A programmable dual-RNA-guided DNA endonuclease in adaptive bacterial immunity. Science. 2012;337(6096):816-21.

253. Zimmer CT, Garrood WT, Puinean AM, Eckel-Zimmer M, Williamson MS, Davies TGE, et al. A CRISPR/Cas9 mediated point mutation in the alpha 6 subunit of the nicotinic acetylcholine receptor confers resistance to spinosad in *Drosophila melanogaster*. Insect Biochem Molec. 2016;73:62-9.

254. Stenesen D, Moehlman AT, Kramer H. The carcinine transporter CarT is required in *Drosophila* photoreceptor neurons to sustain histamine recycling. Elife. 2015;4.

255. Christie KW, Sivan-Loukianova E, Smith WC, Aldrich BT, Schon MA, Roy M, et al. Physiological, anatomical, and behavioral changes after acoustic trauma in *Drosophila melanogaster*. P Natl Acad Sci USA. 2013;110(38):15449-54.

256. Toivonen JM, O'Dell KMC, Petit N, Irvine SC, Knight GK, Lehtonen M, et al. technical knockout, a *Drosophila* model of mitochondrial deafness. Genetics. 2001;159(1):241-54.

257. Ffrench-Constant RH. The molecular genetics of insecticide resistance. Genetics. 2013;194(4):807-15.

258. Pittendrigh B, Reenan R, FfrenchConstant RH, Ganetzky B. Point mutations in the *Drosophila* sodium channel gene *para* associated with resistance to DDT and pyrethroid insecticides. Mol Gen Genet. 1997;256(6):602-10.

259. Behura SK, Haugen M, Flannery E, Sarro J, Tessier CR, Severson DW, et al. Comparative genomic analysis of *Drosophila melanogaster* and vector mosquito developmental genes. Plos One. 2011;6(7).

260. Ditzen M, Pellegrino M, Vosshall LB. Insect odorant receptors are molecular targets of the insect repellent DEET. Science. 2008;319(5871):1838-42.

261. Brandt SM, Jaramillo-Gutierrez G, Kumar S, Barillas-Mury C, Schneider DS. Use of a *Drosophila* model to identify genes regulating *Plasmodium* growth in the mosquito. Genetics. 2008;180(3):1671-8.

262. Schneider D. Using *Drosophila* as a model insect. Nat Rev Genet. 2000;1(3):218-26.

263. Eberl DF, Hardy RW, Kernan MJ. Genetically similar transduction mechanisms for touch and hearing in *Drosophila*. J Neurosci. 2000;20(16):5981-8.

264. Gibson G, Russell I. Flying in tune: sexual recognition in mosquitoes. Curr Biol. 2006;16(13):1311-6.

265. Villella A, Hall JC. Neurogenetics of courtship and mating in *Drosophila*. Advances in Genetics. 2008;62:67-184.

266. Keil TA. Functional morphology of insect mechanoreceptors. Microsc Res Techniq. 1997;39(6):506-31.

267. Yack JE. The structure and function of auditory chordotonal organs in insects. Microsc Res Techniq. 2004;63(6):315-37.

268. Walker RG, Willingham AT, Zuker CS. A *Drosophila* mechanosensory transduction channel. Science. 2000;287(5461):2229-34.

269. Yan ZQ, Zhang W, He Y, Gorczyca D, Xiang Y, Cheng LE, et al. *Drosophila* NOMPC is a mechanotransduction channel subunit for gentle-touch sensation. Nature. 2013;493(7431):221-5.

270. Lehnert BP, Baker AE, Gaudry Q, Chiang AS, Wilson RI. Distinct roles of TRP channels in auditory transduction and amplification in *Drosophila*. Neuron. 2013;77(1):115-28.

271. Kamikouchi A, Inagaki HK, Effertz T, Hendrich O, Fiala A, Göpfert MC, et al. The neural basis of *Drosophila* gravity-sensing and hearing. Nature. 2009;458(7235):165-U1.

272. Bechstedt S, Albert JT, Kreil DP, Muller-Reichert T, Göpfert MC, Howard J. A doublecortin containing microtubule-associated protein is implicated in mechanotransduction in *Drosophila* sensory cilia. Nat Commun. 2010;1.

273. Park J, Lee J, Shim J, Han W, Lee J, Bae YC, et al. dTULP, the *Drosophila melanogaster* homolog of Tubby, regulates transient receptor potential channel localization in cilia. Plos Genet. 2013;9(9).

274. Albert JT, Göpfert MC. Hearing in *Drosophila*. Curr Opin Neurobiol. 2015;34:79-85.

275. Kim SE, Coste B, Chadha A, Cook B, Patapoutian A. The role of *Drosophila* Piezo in mechanical nociception. Nature. 2012;483(7388):209-12.

276. Albert JT, Kozlov AS. Comparative aspects of hearing in vertebrates and insects with antennal ears. Curr Biol. 2016;26(20):1050-61.

277. Boo KS, Richards AG. Fine structure of the scolopidia in the Johnston's organ of male *Aedes aegypti* (L.) (Diptera: Culicidae). Int J Insect Morphol Embryol. 1975;4:549–66.

278. Boo KS, Richards AG. Fine-structure of scolopidia in Johnston's organ of female *Aedes aegypti* compared with that of male. Journal of Insect Physiology. 1975;21(5):1129-39.

279. Nadrowski B, Effertz T, Senthilan PR, Göpfert MC. Antennal hearing in insects - New findings, new questions. Hearing Res. 2011;273(1-2):7-13.

280. Kamikouchi A, Shimada T, Ito K. Comprehensive classification of the auditory sensory projections in the brain of the fruit fly *Drosophila melanogaster*. J Comp Neurol. 2006;499(3):317-56.

281. Göpfert MC, Robert D. The mechanical basis of *Drosophila* audition. J Exp Biol. 2002;205(9):1199-208.

282. Nadrowski B, Albert JT, Göpfert MC. Transducer-based force generation explains active process in *Drosophila* hearing. Curr Biol. 2008;18(18):1365-72.

283. Göpfert MC, Humphris ADL, Albert JT, Robert D, Hendrich O. Power gain exhibited by motile mechanosensory neurons in *Drosophila* ears. P Natl Acad Sci USA. 2005;102(2):325-30.

284. Riabinina O, Dai MJ, Duke T, Albert JT. Active process mediates species-specific tuning of *Drosophila* ears. Curr Biol. 2011;21(8):658-64.

285. Albert JT, Nadrowski B, Göpfert MC. Mechanical signatures of transducer gating in the *Drosophila* ear. Curr Biol. 2007;17(11):1000-6.

286. Hall JL. Auditory distortion products f_2 - f_1 and $2f_1$ - f_2 . J Acoust Soc Am. 1972;51(6):1863-&.

287. Avitabile D, Homer M, Champneys AR, Jackson JC, Robert D. Mathematical modelling of the active hearing process in mosquitoes. J R Soc Interface. 2010;7(42):105-22.

288. Göpfert MC, Robert D. Nanometre-range acoustic sensitivity in male and female mosquitoes. P Roy Soc B-Biol Sci. 2000;267(1442):453-7.

289. Göpfert MC, Robert D. Active auditory mechanics in mosquitoes. P Roy Soc B-Biol Sci. 2001;268(1465):333-9.

290. Jackson JC, Windmill JFC, Pook VG, Robert D. Synchrony through twice-frequency forcing for sensitive and selective auditory processing. P Natl Acad Sci USA. 2009;106(25):10177-82.

291. Slooten E, Lambert DM. Evolutionary studies of the New Zealand coastal mosquito *Opifex fuscus* (Hutton) 1. Mating behavior. Behaviour. 1983;84:157-72.

292. Provost MW, Haeger JS. Mating and pupal attendance in *Deinocerites cancer* and comparisons with *Opifex fuscus* (Diptera - Culicidae). Ann Entomol Soc Am. 1967;60(3).

293. Risler H. Das Gehörorgan der Männchen von *Anopheles stephensi* Liston (Culicidae). Zool Jahrb (Anat Ontog Tiere). 1953;73(175-186).

294. Risler H. Das Gehörorgan der Männchen von *Culex pipiens* L., *Aedes aegypti* L. und *Anopheles stephensi* Liston (Culicidae), eine vergleichend morphologische Untersuchung. Zool Jahrb (Anat Ontog Tiere). 1955;74(478-490).

295. Belton P. The structure and probable function of the internal cuticular parts of Johnston organ in mosquitos (*Aedes aegypti*). Can J Zool. 1989;67(11):2625-32.

296. Lapshin DN, Vorontsov DD. Frequency tuning of individual auditory receptors in female mosquitoes (Diptera, Culicidae). Journal of Insect Physiology. 2013;59(8):828-39.

297. Boo KS. Fine-structure of the antennal sensory hairs in female *Anopheles stephensi*. Z Parasitenkd. 1980;61(2):161-71.

298. Boo KS. Antennal sensory receptors of the male mosquito, *Anopheles stephensi*. Z Parasitenkd. 1980;61(3):249-64.

299. Roth LM. A Study of mosquito behavior - an experimental laboratory study of the sexual behavior of *Aedes Aegypti* (Linnaeus). Am Midl Nat. 1948;40(2):265-352.

300. Beach R. Physiological changes governing the onset of sexual receptivity in male mosquitos. Journal of Insect Physiology. 1980;26(4):245-&.

301. Nijhout HF. Control of antennal hair erection in male mosquitos. Biol Bull. 1977;153(3):591-603.

302. Charlwood JD, Jones MDR. Mating in the mosquito, *Anopheles gambiae* s.l. 2. Swarming behavior. Physiol Entomol. 1980;5(4):315-20.

303. Charlwood JD, Jones MDR. Mating behavior in the mosquito, *Anopheles gambiae* s.l.1. Close range and contact-behavior. Physiol Entomol. 1979;4(2):111-20.

304. Nijhout HF, Martin SK. Alpha-adrenergic activity of isoproterenol in mosquito antennae. Experientia. 1978;34(6):758-9.

305. Jackson JC, Robert D. Nonlinear auditory mechanism enhances female sounds for male mosquitoes. P Natl Acad Sci USA. 2006;103(45):16734-9.

306. Arthur BJ, Emr KS, Wyttenbach RA, Hoy RR. Mosquito (*Aedes aegypti*) flight tones: Frequency, harmonicity, spherical spreading, and phase relationships. J Acoust Soc Am. 2014;135(2):933-41.

307. Gurtovenko AA, Anwar J. Modulating the structure and properties of cell membranes: The molecular mechanism of action of dimethyl sulfoxide. J Phys Chem B. 2007;111(35):10453-60.

308. Warren B, Lukashkin AN, Russell IJ. The dynein-tubulin motor powers active oscillations and amplification in the hearing organ of the mosquito. P Roy Soc B-Biol Sci. 2010;277(1688):1761-9.

309. Lorentz MN, Stokes AN, Rossler DC, Lotters S. Tetrodotoxin. Curr Biol. 2016;26(19):R870-R2.

310. Kamikouchi A, Albert JT, Göpfert MC. Mechanical feedback amplification in *Drosophila* hearing is independent of synaptic transmission. Eur J Neurosci. 2010;31(4):697-703.

311. Montecucco C, Schiavo G. Mechanism of action of tetanus and Botulinum neurotoxins. Mol Microbiol. 1994;13(1):1-8.

312. Pellizzari R, Rossetto O, Schiavo G, Montecucco C. Tetanus and botulinum neurotoxins: mechanism of action and therapeutic uses. Philos T Roy Soc B. 1999;354(1381):259-68.

313. Field LH, Matheson T. Chordotonal organs of insects. Advances in Insect Physiology, Vol 27. 1998;27:1-228.

314. Andres M, Seifert M, Spalthoff C, Warren B, Weiss L, Giraldo D, et al. Auditory efferent system modulates mosquito hearing. Curr Biol. 2016;26(15):2028-36.

315. Mockel D, Seyfarth EA, Kossl M. Otoacoustic emissions in bushcricket ears: general characteristics and the influence of the neuroactive insecticide pymetrozine. J Comp Physiol A. 2011;197(2):193-202.

316. Gong ZF, Son WS, Chung YD, Kim JW, Shin DW, McClung CA, et al. Two interdependent TRPV channel subunits, Inactive and Nanchung, mediate hearing in *Drosophila*. J Neurosci. 2004;24(41):9059-66.

317. Kim J, Chung YD, Park DY, Choi SK, Shin DW, Soh H, et al. A TRPV family ion channel required for hearing in *Drosophila*. Nature. 2003;424(6944):81-4.

318. Na YE, Jung JW, Kwon HW. Identification and expression patterns of two TRPV channel genes in antennae and Johnston's organ of the dengue and Zika virus vector mosquito, *Aedes aegypti*. J Asia-Pac Entomol. 2016;19(3):563-9.

319. Chang GC, Snyder WE. Pymetrozine causes a nontarget pest, the Colorado potato beetle (Coleoptera : Chrysomelidae), to leave potato plants. J Econ Entomol. 2008;101(1):74-80.

320. Lashkari MR, Sahragard A, Ghadamyari M. Sublethal effects of imidacloprid and pymetrozine on population growth parameters of cabbage aphid, *Brevicoryne brassicae* on rapeseed, *Brassica napus* L. Insect Sci. 2007;14(3):207-12.

321. Ban LF, Zhang S, Huang ZY, He YP, Peng YQ, Gao CF. Resistance monitoring and assessment of resistance risk to pymetrozine in *Laodelphax striatellus* (Hemiptera: Delphacidae). J Econ Entomol. 2012;105(6):2129-35.

322. Ovcarenko I, Lindstrom L, Saikkonen K, Vanninen I. Variation in mortality among populations is higher for pymetrozine than for imidacloprid and spiromesifen in *Trialeurodes vaporariorum* in greenhouses in Finland. Pest Manag Sci. 2014;70(10):1524-30.

323. Yang YX, Huang LX, Wang YC, Zhang YX, Fang SQ, Liu ZW. No cross-resistance between imidacloprid and pymetrozine in the brown planthopper: status and mechanisms. Pestic Biochem Phys. 2016;130:79-83.

324. Taylor GK, Krapp HG. Sensory systems and flight stability: What do insects measure and why? Advances in Insect Physiology: Insect Mechanics and Control. 2007;34:231-316.

325. Mamiya A, Dickinson MH. Antennal mechanosensory neurons mediate wing motor reflexes in flying *Drosophila*. J Neurosci. 2015;35(20):7977-91.

326. Mamiya A, Straw AD, Tomasson E, Dickinson MH. Active and passive antennal movements during visually guided steering in flying *Drosophila*. J Neurosci. 2011;31(18):6900-14.

327. Yorozu S, Wong A, Fischer BJ, Dankert H, Kernan MJ, Kamikouchi A, et al. Distinct sensory representations of wind and near-field sound in the *Drosophila* brain. Nature. 2009;458(7235):201-U4.

328. Rohrseitz N, Fry SN. Behavioural system identification of visual flight speed control in *Drosophila melanogaster*. J R Soc Interface. 2011;8(55):171-85.

329. Ewing AW. Role of feedback during singing and flight in *Drosophila melanogaster*. Physiol Entomol. 1979;4(4):329-37.

330. Muir LE, Kay BH. *Aedes aegypti* survival and dispersal estimated by mark-releaserecapture in northern Australia. Am J Trop Med Hyg. 1998;58(3):277-82.

331. Gillies MT. Studies on the dispersion and survival of *Anopheles gambiae* Giles in East Africa, by means of marking and release experiments. B Entomol Res. 1961;52(1):99-127.

332. Bässler U. Versuche zur Orientierung der Stechmucken - die Schwarmbildung und die Bedeutung des Johnstonschen Organs. Z Vergl Physiol. 1958;41(3):300-30.

333. Nachtigall W. Mechanics and aerodynamics of flight. In: Goldsworthy GJ, Wheeler CH, editors. Insect flight. Florida: CRC Press, Inc.; 1989. p. 1-29.

334. Dickinson MH. Haltere-mediated equilibrium reflexes of the fruit fly, *Drosophila melanogaster*. Philos T Roy Soc B. 1999;354(1385):903-16.

335. Christophers SR. *Aedes aegypti* (L.). The yellow fever mosquito: its life history, binomics and structure: Cambridge University Press; 1960.

336. Cummins B, Cortez R, Foppa IM, Walbeck J, Hyman JM. A spatial model of mosquito host-seeking behavior. Plos Computational Biology. 2012;8(5).

337. Hoffmann EJ, Miller JR. Reassessment of the role and utility of wind in suppression of mosquito (Diptera : Culicidae) host finding: Stimulus dilution supported over flight limitation. J Med Entomol. 2003;40(5):607-14.

338. Hall JC. The mating of a fly. Science. 1994;264(5166):1702-14.

339. Billeter JC, Rideout EJ, Dornan AJ, Goodwin SF. Control of male sexual behavior in *Drosophila* by the sex determination pathway. Curr Biol. 2006;16(17):R766-R76.

340. Tauber E, Eberl DF. Acoustic communication in *Drosophila*. Behav Process. 2003;64(2):197-210.

341. Greenspan RJ, Ferveur JF. Courtship in *Drosophila*. Annu Rev Genet. 2000;34:205-32.

342. Miller PB, Obrik-Uloho OT, Phan MH, Medrano CL, Renier JS, Thayer JL, et al. The song of the old mother: Reproductive senescence in female *Drosophila*. Fly. 2014;8(3):127-39.

343. Rybak F, Aubin T, Moulin B, Jallon JM. Acoustic communication in *Drosophila melanogaster* courtship: Are pulse- and sine-song frequencies important for courtship success? Can J Zool. 2002;80(6):987-96.

344. Schilcher FV. Role of auditory-stimuli in courtship of *Drosophila melanogaster*. Anim Behav. 1976;24(Feb):18-26.

345. Coen P, Clemens J, Weinstein AJ, Pacheco DA, Deng Y, Murthy M. Dynamic sensory cues shape song structure in *Drosophila*. Nature. 2014;507(7491).

346. Bennet-Clark HC, Dow M, Ewing AW, Manning A, Schilcher FV. Courtship stimuli in *Drosophila melanogaster*. Behav Genet. 1976;6(1):93-5.

347. Rubinstein CD, Rivlin PK, Hoy RR. Genetic feminization of the thoracic nervous system disrupts courtship song in male *Drosophila melanogaster*. J Neurogenet. 2010;24(4):234-45.

348. Burnet B, Eastwood L, Connolly K. Courtship song of male *Drosophila* lacking aristae. Anim Behav. 1977;25(May):460-4.

349. Burnet B, Eastwood L, Connolly K. Genetic-analysis of courtship song in *Drosophila melanogaster*. Heredity. 1977;39(Dec):425-6.

350. Manning A. The sexual behaviour of two sibling *Drosophila* species. Behaviour. 1959;15:123–45.

351. LaRue KM, Clemens J, Berman GJ, Murthy M. Acoustic duetting in *Drosophila virilis* relies on the integration of auditory and tactile signals. Elife. 2015;4.

352. Cator LJ, Arthur BJ, Harrington LC, Hoy RR. Harmonic convergence in the love songs of the dengue vector mosquito. Science. 2009;323(5917):1077-9.

353. Pennetier C, Warren B, Dabire KR, Russell IJ, Gibson G. "Singing on the wing" as a mechanism for species recognition in the malarial mosquito *Anopheles gambiae*. Curr Biol. 2010;20(2):131-6.

354. Warren B, Gibson G, Russell IJ. Sex recognition through midflight mating duets in *Culex* mosquitoes is mediated by acoustic distortion. Curr Biol. 2009;19(6):485-91.

355. Cator LJ, Harrington LC. The harmonic convergence of fathers predicts the mating success of sons in *Aedes aegypti*. Anim Behav. 2011;82(4):627-33.

356. Trott AR, Donelson NC, Griffith LC, Ejima A. Song choice is modulated by female movement in *Drosophila* males. Plos One. 2012;7(9).

357. Aldersley A, Champneys A, Homer M, Robert D. Quantitative analysis of harmonic convergence in mosquito auditory interactions. J R Soc Interface. 2016;13(117).

358. Simoes PMV, Ingham RA, Gibson G, Russell IJ. A role for acoustic distortion in novel rapid frequency modulation behaviour in free-flying male mosquitoes. J Exp Biol. 2016;219(13):2039-47.

359. Lapshin DN. The auditory system of blood-sucking mosquito females (Diptera, Culicidae): acoustic perception during flight simulation Entomological Review. 2012;93(2):135-49.

360. Bartlett-Healy K, Crans W, Gaugler R. Phonotaxis to amphibian vocalizations in *Culex territans* (Diptera : Culicidae). Ann Entomol Soc Am. 2008;101(1):95-103.

361. Cator LJ, Arthur BJ, Ponlawat A, Harrington LC. Behavioral observations and sound recordings of free-flight mating swarms of *Ae. aegypti* (Diptera: Culicidae) in Thailand. J Med Entomol. 2011;48(4):941-6.

362. Nijhout HF, Sheffield HG. Antennal hair erection in male mosquitos - new mechanical effector in insects. Science. 1979;206(4418):595-6.

363. Göpfert MC, Briegel H, Robert D. Mosquito hearing: Sound-induced antennal vibrations in male and female *Aedes aegypti*. J Exp Biol. 1999;202(20):2727-38.

364. Hartberg WK. Observations on mating behavior of *Aedes aegypti* in nature. B World Health Organ. 1971;45(6):847-&.

365. Spielman A. The mechanics of copulation in *Aedes aegypti*. Biol Bull. 1964;127(2):324-44.

366. Boyer S, Toty C, Jacquet M, Lemperiere G, Fontenille D. Evidence of multiple inseminations in the field in *Aedes albopictus*. Plos One. 2012;7(8).

367. Richardson JB, Jameson SB, Gloria-Soria A, Wesson DM, Powell J. Evidence of limited polyandry in a natural population of *Aedes aegypti*. Am J Trop Med Hyg. 2015;93(1):189-93.

368. Degner EC, Harrington LC. Polyandry depends on postmating time interval in the dengue vector *Aedes aegypti*. Am J Trop Med Hyg. 2016;94(4):780-5.

369. Golombek DA, Rosenstein RE. Physiology of circadian entrainment. Physiol Rev. 2010;90(3):1063-102.

370. Yan OY, Andersson CR, Kondo T, Golden SS, Johnson CH. Resonating circadian clocks enhance fitness in cyanobacteria. P Natl Acad Sci USA. 1998;95(15):8660-4.

371. Meireles ACA, Kyriacou CP. Circadian rhythms in insect disease vectors. Mem I Oswaldo Cruz. 2013;108:48-58.

372. Panda S, Hogenesch JB, Kay SA. Circadian rhythms from flies to human. Nature. 2002;417(6886):329-35.

373. Chen C, Buhl E, Xu M, Croset V, Rees JS, Lilley KS, et al. *Drosophila* Ionotropic Receptor 25a mediates circadian clock resetting by temperature. Nature. 2015;527(7579):516-U238.

374. Faville R, Kottler B, Goodhill GJ, Shaw PJ, van Swinderen B. How deeply does your mutant sleep? Probing arousal to better understand sleep defects in *Drosophila*. Sci Rep-Uk. 2015;5.

375. Gentile C, Rivas GBd, Lima JBP, Bruno RV, Peixoto AA. Circadian clock of *Aedes aegypti*: effects of blood-feeding, insemination and RNA interference. Mem Inst Oswaldo Cruz. 2013;108.

376. Peterson EL. Phase-resetting a mosquito circadian oscillator 1. Phase-resetting surface. J Comp Physiol. 1980;138(3):201-11.

377. Taylor B, Jones MDR. Circadian rhythm of flight activity in mosquito *Aedes aegypti*(L) - phase-setting effects of light-on and light-off. J Exp Biol. 1969;51(1):59-&.

378. Jones MDR, Gubbins SJ. Changes in circadian flight activity of mosquito *Anopheles gambiae* in relation to insemination, feeding and oviposition. Physiol Entomol. 1978;3(3):213-20.

379. Rund SSC, Bonar NA, Champion MM, Ghazi JP, Houk CM, Leming MT, et al. Daily rhythms in antennal protein and olfactory sensitivity in the malaria mosquito *Anopheles gambiae*. Sci Rep-Uk. 2013;3.

380. Meireles-Filho ACA, Kyriacou CP. Circadian rhythms in insect disease vectors. Mem Inst Oswaldo Cruz. 2013;108.

381. Rund SSC, Lee SJ, Bush BR, Duffield GE. Strain- and sex-specific differences in daily flight activity and the circadian clock of *Anopheles gambiae* mosquitoes. Journal of Insect Physiology. 2012;58(12):1609-19.

382. O'Donnell AJ, Schneider P, McWatters HG, Reece SE. Fitness costs of disrupting circadian rhythms in malaria parasites. P Roy Soc B-Biol Sci. 2011;278(1717):2429-36.

383. O'Donnell AJ, Mideo N, Reece SE. Disrupting rhythms in *Plasmodium chabaudi:* costs accrue quickly and independently of how infections are initiated. Malar J. 2013;12(1).

384. Sack RL, Lewy AJ, Blood ML, Keith LD, Nakagawa H. Circadian rhythm abnormalities in totally blind people - incidence and clinical-significance. J Clin Endocr Metab. 1992;75(1):127-34.

385. Chiba Y, Shinkawa Y, Yoshii M, Matsumoto A, Tomioka K, Takahashi SY. A comparative-study on insemination dependency of circadian activity pattern in mosquitos. Physiol Entomol. 1992;17(3):213-8.

386. Sehadova H, Glaser FT, Gentile C, Simoni A, Giesecke A, Albert JT, et al. Temperature entrainment of *Drosophila*'s circadian clock involves the gene *nocte* and signaling from peripheral sensory tissues to the brain. Neuron. 2009;64(2):251-66.

387. Smith DL, Battle KE, Hay SI, Barker CM, Scott TW, McKenzie FE. Ross, Macdonald, and a theory for the dynamics and control of mosquito transmitted pathogens. Plos Pathog. 2012;8(4).

388. Ross R. Some quantitative studies in epidemiology. Nature. 1911;87:466-7.

389. Villela DAM, Codeco CT, Figueiredo F, Garcia GA, Maciel-de-Freitas R, Struchiner CJ. A Bayesian hierarchical model for estimation of abundance and spatial density of *Aedes aegypti*. Plos One. 2015;10(4).

390. Killeen GF, Smith TA. Exploring the contributions of bed nets, cattle, insecticides, and excitorepellency to malaria control: a deterministic model of mosquito host-seeking behaviour and mortality. T Roy Soc Trop Med H. 2007;101(9):867-80.

391. Le Menach A, Takala S, McKenzie FE, Perisse A, Harris A, Flahault A, et al. An elaborated feeding cycle model for reductions in vectorial capacity of night-biting mosquitoes by insecticide-treated nets. Malaria J. 2007;6.

392. Teboh-Ewungkem MI, Ngwa GA, Ngonghala CN. Models and proposals for malaria: a review. Math Popul Stud. 2013;20(2):57-81.

393. Zhu L, Marshall JM, Qualls WA, Schlein Y, McManus JW, Arheart KL, et al. Modelling optimum use of attractive toxic sugar bait stations for effective malaria vector control in Africa. Malaria J. 2015;14.

394. Effertz T, Nadrowski B, Piepenbrock D, Albert JT, Göpfert MC. Direct gating and mechanical integrity of *Drosophila* auditory transducers require TRPN1. Nat Neurosci. 2012;15(9):1198-U43.

395. Adams MD, Celniker SE, Holt RA, Evans CA, Gocayne JD, Amanatides PG, et al. The genome sequence of *Drosophila melanogaster*. Science. 2000;287(5461):2185-95.

396. Bilen J, Bonini NM. *Drosophila* as a model for human neurodegenerative disease. Annu Rev Genet. 2005;39:153-71.

397. Chintapalli VR, Wang J, Dow JAT. Using FlyAtlas to identify better *Drosophila melanogaster* models of human disease. Nat Genet. 2007;39(6):715-20.

398. Mockett RJ, Matsumoto Y. Effect of prolonged coldness on survival and fertility of *Drosophila melanogaster*. Plos One. 2014;9(3).

399. Metzendorf C, Lind MI. *Drosophila mitoferrin* is essential for male fertility: evidence for a role of mitochondrial iron metabolism during spermatogenesis. Bmc Dev Biol. 2010;10.

400. Benzer S. Genetic dissection of behavior. Sci Am. 1973;229(6):24-37.

401. Straw AD, Branson K, Neumann TR, Dickinson MH. Multi-camera real-time threedimensional tracking of multiple flying animals. J R Soc Interface. 2011;8(56):395-409.

402. Stowers JR, Fuhrmann A, Hofbauer M, Streinzer M, Schmid A, Dickinson MH, et al. Reverse engineering animal vision with virtual reality and genetics. Computer. 2014;47(7):38-45.

403. Borst A. Drosophila's view on insect vision. Curr Biol. 2009;19(1):R36-R47.

404. Rosato E, Tauber E, Kyriacou CP. Molecular genetics of the fruit-fly circadian clock. Eur J Hum Genet. 2006;14(6):729-38.

405. Long CE, Markow TA, Yaeger P. Relative male age, fertility, and competitive mating success in *Drosophila melanogaster*. Behav Genet. 1980;10(2):163-70.

406. Markow TA, Quaid M, Kerr S. Male mating experience and competitive courtship success in *Drosophila melanogaster*. Nature. 1978;276(5690):821-2.

407. Collet JM, Fuentes S, Hesketh J, Hill MS, Innocenti P, Morrow EH, et al. Rapid evolution of the intersexual genetic correlation for fitness in *Drosophila melanogaster*. Evolution. 2016;70(4):781-95.

408. Levine JD, Funes P, Dowse HB, Hall JC. Signal analysis of behavioral and molecular cycles. Bmc Neurosci. 2002;3.

409. Arthur BJ, Sunayama-Morita T, Coen P, Murthy M, Stern DL. Multi-channel acoustic recording and automated analysis of *Drosophila* courtship songs. Bmc Biol. 2013;11.

410. R Core Team. R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria; 2016.

411. Soulavie F, Piepenbrock D, Thomas J, Vieillard J, Duteyrat JL, Cortier E, et al. *hemingway* is required for sperm flagella assembly and ciliary motility in *Drosophila*. Mol Biol Cell. 2014;25(8):1276-86.

412. Kavlie RG, Kernan MJ, Eberl DF. Hearing in *Drosophila* requires *tilB*, a conserved protein associated with ciliary motility. Genetics. 2010;185(1):177-U291.

413. Aruna S, Flores HA, Barbash DA. Reduced fertility of *Drosophila melanogaster hybrid male rescue (Hmr)* mutant females is partially complemented by *Hmr* orthologs from sibling species. Genetics. 2009;181(4):1437-50.

414. Barnes AI, Wigby S, Boone JM, Partridge L, Chapman T. Feeding, fecundity and lifespan in female *Drosophila melanogaster*. P Roy Soc B-Biol Sci. 2008;275(1643):1675-83.

415. Joshi A, Mueller LD. Adult crowding effects on longevity in *Drosophila melanogaster*. Increase in age-independent mortality. Curr Sci India. 1997;72(4):255-60.

416. Joshi A, Wu WP, Mueller LD. Density-dependent natural selection in *Drosophila*: Adaptation to adult crowding. Evol Ecol. 1998;12(3):363-76.

417. Libert S, Zwiener J, Chu XW, VanVoorhies W, Roman G, Pletcher SD. Regulation of *Drosophila* life span by olfaction and food-derived odors. Science. 2007;315(5815):1133-7.

418. Boll W, Noll M. The *Drosophila* Pox neuro gene: control of male courtship behavior and fertility as revealed by a complete dissection of all enhancers. Development. 2002;129(24):5667-81.

419. Shaposhnikov M, Proshkina E, Shilova L, Zhavoronkov A, Moskalev A. Lifespan and stress resistance in *Drosophila* with overexpressed DNA repair genes. Sci Rep-Uk. 2015;5.

420. Slack C, Werz C, Wieser D, Alic N, Foley A, Stocker H, et al. Regulation of lifespan, metabolism, and stress responses by the *Drosophila* SH2B protein, Lnk. Plos Genet. 2010;6(3).

421. Straw AD, Lee S, Dickinson MH. Visual control of altitude in flying *Drosophila*. Curr Biol. 2010;20(17):1550-6.

422. Bartholomew NR, Burdett JM, VandenBrooks JM, Quinlan MC, Call GB. Impaired climbing and flight behaviour in *Drosophila melanogaster* following carbon dioxide anaesthesia. Sci Rep-Uk. 2015;5.

423. Censi A, Straw AD, Sayaman RW, Murray RM, Dickinson MH. Discriminating external and internal causes for heading changes in freely flying *Drosophila*. Plos Computational Biology. 2013;9(2).

424. Muijres FT, Elzinga MJ, Iwasaki NA, Dickinson MH. Body saccades of *Drosophila* consist of stereotyped banked turns. J Exp Biol. 2015;218(6):864-75.

425. Marques MD. Biological rhythms and vector insects. Mem I Oswaldo Cruz. 2013;108:59-62.

426. Peixoto AA, Hall JC. Analysis of temperature-sensitive mutants reveals new genes involved in the courtship song of *Drosophila*. Genetics. 1998;148(2):827-38.

427. Wheeler DA, Fields WL, Hall JC. Spectral-analysis of *Drosophila* courtship songs - *Drosophila melanogaster*, *Drosophila simulans*, and their interspecific hybrid. Behav Genet. 1988;18(6):675-703.

428. Ding Y, Berrocal A, Morita T, Longden KD, Stern DL. Natural courtship song variation caused by an intronic retroelement in an ion channel gene. Nature. 2016;536(7616).

429. von Philipsborn AC, Liu TX, Yu JY, Masser C, Bidaye SS, Dickson BJ. Neuronal control of *Drosophila* courtship song. Neuron. 2011;69(3):509-22.

430. Golding N, Wilson AL, Moyes CL, Cano J, Pigott DM, Velayudhan R, et al. Integrating vector control across diseases. Bmc Medicine. 2015;13.

431. Chanda E, Govere JM, Macdonald MB, Lako RL, Haque U, Baba SP, et al. Integrated vector management: a critical strategy for combating vector-borne diseases in South Sudan. Malaria J. 2013;12. 432. Rahuman AA, Gopalakrishnan G, Venkatesan P, Geetha K. Isolation and identification of mosquito larvicidal compound from *Abutilon indicum* (Linn.) Sweet. Parasitol Res. 2008;102(5):981-8.

433. Benelli G. Plant-borne ovicides in the fight against mosquito vectors of medical and veterinary importance: a systematic review. Parasitol Res. 2015;114(9):3201-12.

434. Hertlein MB, Mavrotas C, Jousseaume C, Lysandrou M, Thompson GD, Jany W, et al. A review of spinosad as a natural product for larval mosquito control. J Am Mosquito Contr. 2010;26(1):67-87.

435. Thailayil J, Magnusson K, Godfray HCJ, Crisanti A, Catteruccia F. Spermless males elicit large-scale female responses to mating in the malaria mosquito *Anopheles gambiae*. P Natl Acad Sci USA. 2011;108(33):13677-81.

436. Pompon J, Levashina EA. A new role of the mosquito complement-like cascade in male fertility in *Anopheles gambiae*. Plos Biol. 2015;13(9).

437. Spitzen J, Spoor CW, Grieco F, ter Braak C, Beeuwkes J, van Brugge SP, et al. A 3D analysis of flight behavior of *Anopheles gambiae* sensu stricto malaria mosquitoes in response to human odor and heat. Plos One. 2013;8(5).

438. Takemura S, Bharioke A, Lu ZY, Nern A, Vitaladevuni S, Rivlin PK, et al. A visual motion detection circuit suggested by *Drosophila* connectomics. Nature. 2013;500(7461):175-+.

439. Sanes JR, Zipursky SL. Design principles of insect and vertebrate visual systems. Neuron. 2010;66(1):15-36.

440. Rocha M, Kimler KJ, Leming MT, Hu XB, Whaley MA, O'Tousa JE. Expression and light-triggered movement of rhodopsins in the larval visual system of mosquitoes. J Exp Biol. 2015;218(9):1386-92.

441. World Health Organization. Department of communicable disease prevention cae, Scheme WPE. Guidelines for laboratory and field testing of mosquito larvicides. 2005.

442. Tripet F, Dolo G, Traore S, Lanzaro GC. The "wingbeat hypothesis" of reproductive isolation between members of the *Anopheles gambiae* complex (Diptera : Culicidae) does not fly. J Med Entomol. 2004;41(3):375-84.

443. Kovendan K, Murugan K, Kumar PM, Thiyagarajan P, William SJ. Ovicidal, repellent, adulticidal and field evaluations of plant extract against dengue, malaria and filarial vectors. Parasitol Res. 2013;112(3):1205-19.

444. Sukumar K, Perich MJ, Boobar LR. Botanical derivatives in mosquito control - a review. J Am Mosquito Contr. 1991;7(2):210-37.

445. McCrae AWR. Oviposition by African malaria vector mosquitos 2. Effects of site tone, water type and conspecific immatures on target selection by fresh-water *Anopheles gambiae* Giles, Sensu-Lato. Ann Trop Med Parasit. 1984;78(3):307-18.

446. Okal MN, Lindh JM, Torr SJ, Masinde E, Orindi B, Lindsay SW, et al. Analysing the oviposition behaviour of malaria mosquitoes: design considerations for improving two-choice egg count experiments. Malaria J. 2015;14.

447. Hamady D, Ruslan NB, Ahmad A, Rawi CSM, Ahmad H, Satho T, et al. Colonized *Aedes albopictus* and its sexual performance in the wild: implications for SIT technology and containment. Parasite Vector. 2013;6.

448. Angarita-Jaimes NC, Parker JEA, Abe M, Mashauri F, Martine J, Towers CE, et al. A novel video-tracking system to quantify the behaviour of nocturnal mosquitoes attacking human hosts in the field. J R Soc Interface. 2016;13(117).

449. van Breugel F, Riffell J, Fairhall A, Dickinson MH. Mosquitoes use vision to associate odor plumes with thermal targets. Curr Biol. 2015;25(16):2123-9.

450. Dell AI, Bender JA, Branson K, Couzin ID, de Polavieja GG, Noldus LPJJ, et al. Automated image-based tracking and its application in ecology. Trends Ecol Evol. 2014;29(7):417-28.

451. Pitts RJ, Zwiebel LJ. Antennal sensilla of two female anopheline sibling species with differing host ranges. Malaria J. 2006;5.

452. Risler H. The construction of the auditory organ in male mosquitoes. Forts Zool. 1977;24(2-3):143-7.

453. Josephson RK, Malamud JG, Stokes DR. Asynchronous muscle: A primer. J Exp Biol. 2000;203(18):2713-22.

454. Sutton GP, Clarke D, Morley EL, Robert D. Mechanosensory hairs in bumblebees (*Bombus terrestris*) detect weak electric fields. P Natl Acad Sci USA. 2016;113(26):7261-5.

455. Mhatre N, Bhattacharya M, Robert D, Balakrishnan R. Matching sender and receiver: poikilothermy and frequency tuning in a tree cricket. J Exp Biol. 2011;214(15):2569-78.

456. Nadrowski B, Göpfert MC. Level-dependent auditory tuning: Transducer-based active processes in hearing and best-frequency shifts. Communicative & Integrative Biology. 2009;2(1):7-10.

457. Pavlidi N, Monastirioti M, Daborn P, Livadaras I, Van Leeuwen T, Vontas J. Transgenic expression of the *Aedes aegypti* CYP9J28 confers pyrethroid resistance in *Drosophila melanogaster*. Pestic Biochem Phys. 2012;104(2):132-5.

458. Ffrench-Constant RH, Rocheleau TA, Steichen JC, Chalmers AE. A point mutation in a *Drosophila* GABA receptor confers insecticide resistance. Nature. 1993;363(6428):449-51.

459. Baines RA, Bate M. Electrophysiological development of central neurons in the *Drosophila* embryo. J Neurosci. 1998;18(12):4673-83.

460. Fournier D, Bride JM, Hoffmann F, Karch F. Acetylcholinesterase - 2 types of modifications confer resistance to insecticide. J Biol Chem. 1992;267(20):14270-4.

461. Merrell DJ, Underhill JC. Selection for DDT resistance in inbred, laboratory, and wild stocks of *Drosophila melanogaster*. J Econ Entomol. 1956;49(3):300-6.

462. Steele LD, Muir WM, Seong KM, Valero MC, Rangesa M, Sun WL, et al. Genomewide sequencing and an open reading frame analysis of Dichlorodiphenyltrichloroethane (DDT) susceptible (91-C) and resistant (91-R) *Drosophila melanogaster* laboratory populations. Plos One. 2014;9(6).

463. Hudspeth AJ, Choe Y, Mehta AD, Martin P. Putting ion channels to work: mechanoelectrical transduction, adaptation, and amplification by hair cells. P Natl Acad Sci USA. 2000;97(22):11765-72.

464. Peng AW, Salles FT, Pan BF, Ricci AJ. Integrating the biophysical and molecular mechanisms of auditory hair cell mechanotransduction. Nat Commun. 2011;2.

465. Somers J, Nguyen J, Lumb C, Batterham P, Perry T. *In vivo* functional analysis of the *Drosophila melanogaster* nicotinic acetylcholine receptor *Dα6* using the insecticide spinosad. Insect Biochem Molec. 2015;64:116-27.

466. Nadrowski B, Albert JT, Göpfert MC. Transducer based active amplification in the hearing organ of *Drosophila melanogaster*. Concepts and Challenges in the Biophysics of Hearing. 2009:431-6.

467. Sweeney ST, Broadie K, Keane J, Niemann H, Okane CJ. Targeted expression of tetanus toxin light-chain in *Drosophila* specifically eliminates synaptic transmission and causes behavioral defects. Neuron. 1995;14(2):341-51.

468. Bates D, Machler M, Bolker BM, Walker SC. Fitting linear mixed-effects models using Ime4. J Stat Softw. 2015;67(1):1-48.

469. Holt JR, Corey DP. Two mechanisms for transducer adaptation in vertebrate hair cells. P Natl Acad Sci USA. 2000;97(22):11730-5.

470. Howard J, Hudspeth AJ. Compliance of the hair bundle associated with gating of mechanoelectrical transduction channels in the bullfrogs saccular hair cell. Neuron. 1988;1(3):189-99.

471. Howard J, Roberts WM, Hudspeth AJ. Mechanoelectrical transduction by hair cells. Annu Rev Biophys Bio. 1988;17:99-124.

472. Nadrowski B, Göpfert MC. Modeling auditory transducer dynamics. Curr Opin Otolaryngo. 2009;17(5):400-6.

473. Nakagawa S, Cuthill IC. Effect size, confidence interval and statistical significance: a practical guide for biologists. Biol Rev. 2007;82(4):591-605.

474. Unlu I, Farajollahi A, Rochlin I, Crepeau TN, Strickman D, Gaugler R. Differences in male-female ratios of *Aedes albopictus* (Diptera: Culicidae) following ultra-low volume adulticide applications. Acta Tropica. 2014;137:201-5.

475. Rydqvist B, Swerup C, Lannergren J. Viscoelastic properties of the slowly adapting stretch-receptor muscle of the crayfish. Acta Physiol Scand. 1990;139(3):519-27.

476. Catton WT, Petoe N. A visco-elastic theory of mechanoreceptor adaptation. J Physiol-London. 1966;187(1):35-&.

477. Nadrowski B, Martin P, Julicher F. Active hair-bundle motility harnesses noise to operate near an optimum of mechanosensitivity. P Natl Acad Sci USA. 2004;101(33):12195-200.

478. Downes JA. The swarming and mating flight of Diptera. Annu Rev Entomol. 1969;14(271-298).

479. Cator LJ, Ng'Habi KR, Hoy RR, Harrington LC. Sizing up a mate: variation in production and response to acoustic signals in *Anopheles gambiae*. Behav Ecol. 2010;21(5):1033-9.

480. Hemingway J, Vontas J, Poupardin R, Raman J, Lines J, Schwabe C, et al. Countrylevel operational implementation of the Global Plan for Insecticide Resistance Management. P Natl Acad Sci USA. 2013;110(23):9397-402.

481. Edi CAV, Koudou BG, Bellai L, Adja AM, Chouaibou M, Bonfoh B, et al. Long-term trends in *Anopheles gambiae* insecticide resistance in Cote d'Ivoire. Parasite Vector. 2014;7.
482. Bourguet D, Guillemaud T, Chevillon C, Raymond M. Fitness costs of insecticide resistance in natural breeding sites of the mosquito *Culex pipiens*. Evolution. 2004;58(1):128-35.

483. Diniz DFA, de Melo-Santos MAV, Santos EMD, Beserra EB, Helvecio E, de Carvalho-Leandro D, et al. Fitness cost in field and laboratory *Aedes aegypti* populations associated with resistance to the insecticide temphos. Parasite Vector. 2015;8.

484. Martinez-Abrain A. Statistical significance and biological relevance: A call for a more cautious interpretation of results in ecology. Acta Oecol. 2008;34(1):9-11.

485. Anderson DR, Burnham KP, Thompson WL. Null hypothesis testing: Problems, prevalence, and an alternative. J Wildlife Manage. 2000;64(4):912-23.

486. Munhenga G, Brooke BD, Chirwa TF, Hunt RH, Coetzee M, Govender D, et al. Evaluating the potential of the sterile insect technique for malaria control: relative fitness and mating compatibility between laboratory colonized and a wild population of *Anopheles arabiensis* from the Kruger National Park, South Africa. Parasite Vector. 2011;4.

487. Lane SJ, Frankino WA, Elekonich MM, Roberts SP. The effects of age and lifetime flight behavior on flight capacity in *Drosophila melanogaster*. J Exp Biol. 2014;217(9):1437-43.

488. Day JF, Vanhandel E. Differences between the nutritional reserves of laboratory maintained and field collected adult mosquitos. J Am Mosquito Contr. 1986;2(2):154-7.

489. Boyle R, Rabbitt RD, Highstein SM. Efferent control of hair cell and afferent responses in the semicircular canals. J Neurophysiol. 2009;102(3):1513-25.

490. Rabbitt RD, Brownell WE. Efferent modulation of hair cell function. Curr Opin Otolaryngo. 2011;19(5):376-81.

491. Pool AH, Scott K. Feeding regulation in *Drosophila*. Curr Opin Neurobiol. 2014;29.

492. Itskov PM, Moreira JM, Vinnik E, Lopes G, Safarik S, Dickinson MH, et al. Automated monitoring and quantitative analysis of feeding behaviour in *Drosophila*. Nat Commun. 2014;5.

493. Vizzi FF. The mouthparts of the male mosquito *Anopheles quadrimaculatus* Say (Diptera: Culicidae). Ann ent Soc Am. 1953;46:496 - 504.

494. Camalet S, Duke T, Julicher F, Prost J. Auditory sensitivity provided by self-tuned critical oscillations of hair cells. P Natl Acad Sci USA. 2000;97(7):3183-8.

495. Martin P, Hudspeth AJ, Julicher F. Comparison of a hair bundle's spontaneous oscillations with its response to mechanical stimulation reveals the underlying active process. P Natl Acad Sci USA. 2001;98(25):14380-5.

496. Huettel MD. Monitoring quality of laboratory reared insects - biological and behavioral perspective. Environ Entomol. 1976;5(5):807-14.

497. Brady OJ, Johansson MA, Guerra CA, Bhatt S, Golding N, Pigott DM, et al. Modelling adult *Aedes aegypti* and *Aedes albopictus* survival at different temperatures in laboratory and field settings. Parasite Vector. 2013;6.

498. Nilson TL, Sinclair BJ, Roberts SP. The effects of carbon dioxide anesthesia and anoxia on rapid cold-hardening and chill coma recovery in *Drosophila melanogaster*. Journal of Insect Physiology. 2006;52(10):1027-33.

499. Colinet H, Renault D. Metabolic effects of CO_2 anaesthesia in *Drosophila melanogaster*. Biol Letters. 2012;8(6):1050-4.

500. Krober T, Kessler S, Frei J, Bourquin M, Guerin PM. An *in vitro* assay for testing mosquito repellents employing a warm body and carbon dioxide as a behavioral activator. J Am Mosquito Contr. 2010;26(4):381-6.

501. Banks SD, Orsborne J, Gezan SA, Kaur H, Wilder-Smith A, Lindsey SW, et al. Permethrin-treated clothing as protection against the dengue vector, *Aedes aegypti*: extent and duration of protection. Trop Med Int Health. 2015;20:399-400.

502. LaFlamme BA, Ram KR, Wolfner MF. The *Drosophila melanogaster* seminal fluid protease "Seminase" regulates proteolytic and post-mating reproductive processes. Plos Genet. 2012;8(1).

503. Okumu FO, Moore J, Mbeyela E, Sherlock M, Sangusangu R, Ligamba G, et al. A modified experimental hut design for studying responses of disease-transmitting mosquitoes to indoor interventions: The Ifakara experimental huts. Plos One. 2012;7(2).

504. Ameneshewa B, Service MW. Resting habits of *Anopheles arabiensis* in the Awash River Valley of Ethiopia. Ann Trop Med Parasit. 1996;90(5):515-21.

505. Knabel A, Scheringer M, Stehle S, Schulz R. Aquatic exposure predictions of insecticide field concentrations using a multimedia mass-balance model. Environ Sci Technol. 2016;50(7):3721-8.

506. Barbosa SH, I. M. The importance of modelling the spread of insecticide resistance in a heterogeneous environment: the example of adding synergists to bed nets. Malaria J. 2012;11(258).

507. Barbosa S, Black WC, Hastings I. Challenges in estimating insecticide selection pressures from mosquito field data. Plos Neglect Trop D. 2011;5(11).

508. Hammond A, Galizi R, Kyrou K, Simoni A, Siniscalchi C, Katsanos D, et al. A CRISPR-Cas9 gene drive system-targeting female reproduction in the malaria mosquito vector *Anopheles gambiae*. Nat Biotechnol. 2016;34(1):78-83.

Appendices

Appendix A. Selected free fluctuations of *Drosophila melanogaster* CantonS females used in the fertility assay (in section 4.3.1) that were exposed or unexposed to pymetrozine



Figure 61. Free, unstimulated fluctuations of female Drosophila melanogaster involved in the fertility assay described in section 4.3.1 for either a pymetrozine unexposed female CantonS (A) or a pymetrozine exposed female (B). The examples shown here are representative of all flies measured from each experimental group.

Appendix B. Selected free fluctuations of *Drosophila melanogaster* CantonS males used in the temperature entrainment assay (in section 4.3.5) that were exposed or unexposed to pymetrozine



Figure 62. Free, unstimulated fluctuations of male Drosophila melanogaster involved in the temperature entrainment assay described in section 4.3.5 for either a pymetrozine unexposed male CantonS (A) or a pymetrozine exposed male (B). The examples shown here are representative of all flies measured from each experimental group.

Appendix C. Selected free fluctuations of *Drosophila melanogaster* CantonS females and males used in the courtship song assay (in section 4.3.6) that were exposed or unexposed to pymetrozine



Figure 63. Free, unstimulated fluctuations of a pymetrozine unexposed female CantonS (A), a pymetrozine exposed female CantonS (B), a pymetrozine unexposed male CantonS (C) and a pymetrozine exposed male CantonS (D). Examples shown here were involved in the courtship song assay described in section 4.3.6 and are representative of all flies measured from each experimental group.

Appendix D. Selected free fluctuations of *Drosophila melanogaster* males used in the competitive reproduction assay (in section 4.3.7) that were exposed or unexposed to pymetrozine



Figure 64. Free, unstimulated fluctuations for a pymetrozine unexposed $LH_B - UCL$ male (A), a pymetrozine exposed $LH_B - UCL$ male (B), a pymetrozine unexposed $LH_M - UCL$ male (C) and a pymetrozine exposed $LH_M - UCL$ male (D). Examples shown here were involved in the competitive reproduction assay described in section 4.3.7 and are representative of all flies measured from each experimental group.

Appendix E. Mosquito flagellomere length measurements

Table 27. Median length values in μ m (with standard errors given in brackets) for each of the 13 antennal flagellomeres, the total length of the flagellum from the point of laser focus to the pedicel and the total length of the flagellum in its entirety.

Flagellomere section	Ae. aegypti female/µm (n = 8)	Ae. aegypti male/µm (n = 10)	An. gambiae female/µm (n = 9)	<i>An.</i> gambiae male/µm (n = 22)	Cx. quinque- fasciatus female/µm (n = 23)	Cx. quinque- fasciatus male/µm (n = 16)
1	202.32	381.36	161.58	202.52	184.88	334.42
	(5.45)	(9.03)	(2.34)	(10.12)	(3.70)	(6.69)
2	159.38	368.10	134.69	332.77	129.36	271.30
	(4.57)	(15.99)	(3.59)	(15.47)	(3.90)	(7.08)
3	151.92	75.84	124.14	114.07	132.00	73.00
	(3.56)	(1.70)	(3.52)	(2.71)	(3.31)	(1.12)
4	152.63	76.12	117.59	112.08	140.75	75.82
	(4.24)	(1.34)	(3.35)	(2.00)	(1.77)	(1.05)
5	143.78	74.93	117.43	108.96	143.32	74.72
	(4.23)	(1.73)	(3.80)	(2.39)	(1.58)	(0.98)
6	141.11	76.12	120.09	111.35	141.77	75.45
	(4.59)	(1.94)	(2.43)	(2.91)	(1.47)	(1.09)
7	135.17	77.23	116.50	112.49	142.74	73.36
	(3.53)	(1.49)	(1.46)	(3.20)	(1.39)	(0.94)
8	130.36	75.69	109.08	112.26	145.61	70.99
	(4.09)	(1.39)	(1.93)	(3.47)	(1.19)	(1.02)
9	123.25	75.46	111.32	113.19	144.60	70.24
	(3.74)	(1.30)	(1.81)	(3.00)	(1.74)	(1.34)
10	113.47	73.06	102.12	114.64	144.62	71.83
	(4.41)	(1.94)	(1.95)	(2.75)	(2.16)	(1.04)
11	108.49	78.07	98.12	112.70	140.17	71.12
	(3.39)	(2.04)	(1.75)	(2.65)	(1.70)	(0.83)
12	104.75	78.07	76.03	107.47	133.51	69.89
	(3.37)	(3.24)	(6.01)	(2.59)	(2.54)	(1.30)

10	129.30	113.34	145.81	216.86	147.45	183.09
13	(15.4)	(4.85)	(8.76)	(11.90)	(9.62)	(14.03)
Length from	1282.30	1244.11	1119.26	1551.68	1447.17	1066.57
focus point to	(18 20)	(17 74)	(11 72)	(18.06)	(14 03)	(7.96)
base (µm)	(10.20)	(1717-1)	()	(10.00)	(1100)	(1.00)
Total length	1795.83	1630.86	1548.03	1736.68	1863.21	1400.99
(µm)	(20.09)	(21.52)	(12.08)	(10.32)	(24.24)	(9.65)
			_	1		_
Appendix F. Comparison of pre- and post-ringer nerve and mechanical responses for *Drosophila melanogaster* and mosquito species

Ringer injection in both *Drosophila melanogaster* and mosquito species was observed to consistently reduce maximum CAP amplitudes produced after stimulation by up to 25% compared to the pre-ringer state. This is almost certainly due to a change in ion concentration following the introduction of the control solution to the insect as well as potential damage caused to the internal system by the injection process itself.

Whilst these reductions were significant, they also seemed stable and CAP amplitude did not appear to be further reduced after the system was given time to recover (after accounting for gradual decreases in CAP amplitude seen if the experiment lasts for a prolonged period of time). The post-ringer maximum CAP amplitudes still remained at such a level as to be statistically greater than in a post-pymetrozine state and allowed for nerve responses to the smallest step stimuli to be identified.

The mechanics of the system remain intact after ringer injection, with both free fluctuations and dynamical stiffness plots suggesting that at worst minimal alterations have occurred. Figure 65 contains demonstrations of typical median reductions in CAP size for *Drosophila melanogaster* and *Ae. aegypti* females as well as the maintenance of the overall changes observable in dynamical stiffness. Table 28 contains comparisons between before and after ringer states for fitting a two state model for either a single transducer population (for *Ae. aegypti* female mosquitoes) or two independent transducer populations (for *Drosophila melanogaster*).

Statistical comparisons were made using paired Wilcoxon signed rank tests with a significance level of 0.05 in Sigmaplot. No statistically significant differences were seen between any of the parameters for either the *Drosophila melanogaster* or *Ae. aegypti* female mosquito model fits, though post-hoc calculations of statistical power gave power estimates below 50% in all cases tested. This suggests that the statistical comparisons may be underpowered and as such had an increase likelihood of type II errors occurring.



Figure 65. Pre- and post-ringer injection comparisons for changes in dynamical stiffness and CAP amplitude in response to changes in displacement for a) Drosophila melanogaster males from the BL 1283 line and B) female Ae. aegypti mosquitoes. Individual data points represent median values calculated at that displacement, with vertical black bars representing standard errors.

Table 28. Comparisons between before and after ringer injection states of the median values for a Drosophila melanogaster line (BL1283) and Ae. aegypti female mosquitoes using the two state model for either a single population of transducers or two independent populations (no significant differences identifiable between any of the model parameters when comparing before and after ringer states).

	BL1283 before ringer (n = 7)	BL1283 after ringer (n = 7)	<i>Ae. aegypti</i> females before ringer (n = 8)	<i>Ae. aegypti</i> females after ringer (n = 8)
<u> </u>				
Number of				
first	206.6	182.9	-	-
population	(26.8)	(38.3)		
ion channels				
Number of				
second	14808.5	14725.8	943.9	1492.4
population	(1781.2)	(1947.1)	(178.2)	(297.5)
ion channels				
First				
population	47	48		
gating force	(2.74)	(4.22)	-	-
(fN)				
Second				
population	5.41	5.34	16	12
gating force	(0.36)	(0.35)	(2.85)	(1.07)
(fN)				
KINFINITY	74	75	74	75
(µN/m)	(1.9)	(2.0)	(1.8)	(1.5)
K _{STEADY}	49	48	64	66
(µN/m)	(1.3)	(1.5)	(1.3)	(1.0)
K _{GS}	27	25	9.0	9.6
(µN/m)	(0.9)	(0.6)	(0.7)	(0.7)

Appendix G. Median values obtained from fitting the two-state model for two independent transducer populations to post-ringer injection data for all *Drosophila melanogaster* lines investigated in section 6

Table 29. Median values obtained by fitting the two-state model for two independent transducer populations to post-ringer injection data for the following Drosophila melanogaster lines (numbers in brackets refer to standard errors): 91-S, 91-R, BL 1283 and BL 1675.

	91 – S	91 – R	BL 1283	BL 1675
	(n = 7)	(n = 7)	(n = 7)	(n = 7)
Number sensitive	228.5	271.2	206.6	214.5
channels	(72.7)	(37.0)	(26.8)	(20.6)
Number insensitive	10544.8	12031.3	14808.5	14459.6
channels	(639.4)	(1138.8)	(1781.2)	(2003.3)
Sensitive channel	30	30	47	41
gating force (fN)	(2.77)	(1.39)	(2.74)	(1.05)
Insensitive channel	5.08	6.44	5.41	5.45
gating force (fN)	(0.19)	(0.51)	(0.36)	(0.28)
KINFINITY	47	62	74	64
(µN/m)	(4.1)	(1.1)	(1.9)	(2.0)
K _{STEADY}	31	37	49	40
(µN/m)	(3.3)	(1.0)	(1.3)	(1.3)
K _{GS}	17	24	27	23
(µN/m)	(0.9)	(0.5)	(0.9)	(0.8)

Table 30. Median values obtained by fitting the two-state model for two independent transducer populations to post-ringer injection data for the following Drosophila melanogaster lines (numbers in brackets refer to standard errors): w¹¹¹⁸xGAL4, J28 7:1xGAL4 and J28 8:1xGAL4.

	w ¹¹¹⁸ xGAL4	J28 7:1xGAL4	J28 8:1xGAL4
	(n = 7)	(n = 7)	(n = 7)
Number sensitive	323.9	317.4	436.1
channels	(25.3)	(22.3)	(59.4)
Number insensitive	15112.7	17994.5	18508.2
channels	(1169.1)	(884.4)	(1150.7)
Sensitive channel	32	35	31
gating force (fN)	(1.32)	(1.65)	(1.97)
Insensitive channel	5.59	5.05	5.13
gating force (fN)	(1.57)	(1.32)	(0.13)
KINFINITY	60	64	61
(µN/m)	(0.8)	(1.1)	(1.7)
K _{STEADY}	38	41	41
(µN/m)	(0.6)	(0.9)	(0.7)
K _{GS}	21	22	21
(µN/m)	(0.4)	(0.4)	(1.1)

Appendix H. Median velocity fit values for all mosquito species and sexes in all states recorded from

Table 31. Median velocity fit parameters and energy gains for male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes before ringer injection (with standard errors provided in brackets).

Species/ sex	Sample size	F₀/m	Best frequency/Hz	Q	Energy gain (kBT)
Ae. aegypti	52	6.11 x10 ⁻⁴	203.06	3.32	3.06
females	52	(1.8 x10 ⁻⁵)	(2.22)	(0.21)	(0.62)
Ae. aegypti	20	6.87 x10⁻⁴	293.83	1.59	2.81
males	39	(3.7 x10⁻⁵)	(11.72)	(0.47)	(1.69)
An. gambiae	40	7.71 x10 ⁻⁴	219.70	1.19	1.93
females	42	(3.2 x10 ⁻⁵)	(3.55)	(0.24)	(0.25)
An. gambiae	25	5.50 x10⁻⁴	336.46	2.87	2.05
males	30	(2.6 x10⁻⁵)	(8.58)	(1.32)	(0.58)
Cx. quinquefasciatus	27	7.12 x10 ⁻⁴	212.96	5.76	6.26
females	57	(3.8 x10⁻⁵)	(2.41)	(2.74)	(2.05)
Cx. quinquefasciatus	40	5.70 x10 ⁻⁴	387.89	7.70	1.85
males	43	(2.9 x10⁻⁵)	(6.60)	(4.34)	(2.40)

BEFORE RINGER INJECTION

Table 32. Median velocity fit parameters and energy gains for male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes after ringer injection (with standard errors provided in brackets).

Species/ sex	Sample size	F₀/m	Best frequency/Hz	Q	Energy gain (kBT)
Ae. aegypti	20	5.58 x10 ⁻⁴	198.5	3.80	1.72
females	29	(2.2 x10 ⁻⁵)	(1.42)	(0.22)	(0.24)
Ae. aegypti	24	6.58 x10⁻⁴	290.0	1.50	3.61
males	24	(3.1 x10⁻⁵)	(11.91)	(3.24)	(1.68)
An. gambiae	26	8.63 x10 ⁻⁴	215.0	1.00	1.84
females	30	(3.8 x10 ⁻⁵)	(3.95)	(0.25)	(0.52)
An. gambiae	20	5.40 x10 ⁻⁴	352.0	3.12	3.78
males	20	(4.1 x10⁻⁵)	(13.01)	(2.79)	(0.83)
Cx. quinquefasciatus	20	7.92 x10 ⁻⁴	221.6	10.6	15.35
females	29	(3.6 x10 ⁻⁵)	(5.16)	(3.96)	(4.88)
Cx. quinquefasciatus	28	5.29 x10⁻⁴	366.0	19.3	1.46
males	20	(4.8 x10⁻⁵)	(6.76)	(5.93)	(1.65)

AFTER RINGER INJECTION

Table 33. Median velocity fit parameters and energy gains for male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes after TTX injection (with standard errors provided in brackets).

Species/ sex	Sample size	F₀/m	Best frequency/Hz	Q	Energy gain (kBT)
Ae. aegypti	10	4.81 x10 ⁻⁴	196.9	2.76	1.56
females	10	(3.7 x10 ⁻⁵)	(3.09)	(0.45)	(0.40)
Ae. aegypti	0	6.39 x10 ⁻⁴	445.3	68.13	45.43
males	9	(1.2 x10 ⁻³)	(29.5)	(58.90)	(14.62)
An. gambiae	10	8.73 x10 ⁻⁴	214.5	1.21	1.70
females	12	(4.1 x10 ⁻⁵)	(7.22)	(0.07)	(0.31)
An. gambiae	7	4.13 x10 ⁻³	339.6	245.8	19280.68
males	1	(1.6 x10 ⁻³)	(6.52)	(945.4)	(14592.2)
Cx. quinquefasciatus	11	7.70 x10 ⁻⁴	210.6	16.01	24.63
females	14	(8.0 x10 ⁻⁵)	(4.54)	(4.89)	(4.39)
Cx. quinquefasciatus	15	1.64 x10 ⁻³	317.1	156.9	722.72
males	15	(2.5 x10 ⁻⁴)	(7.67)	(290.01)	(6004.67)

AFTER TTX INJECTION

Table 34. Median velocity fit parameters and energy gains for male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes after TeNT injection (with standard errors provided in brackets).

Species/ sex	Sample size	F₀/m	Best frequency/Hz	Q	Energy gain (kBT)
Ae. aegypti	11	6.78 x10 ⁻⁴	199.27	2.55	1.56
females		(4.6 x10 ⁻⁵)	(2.05)	(0.24)	(0.25)
Ae. aegypti	Q	6.16 x10 ⁻³	353.04	94.60	83.43
males	0	(7.0 x10 ⁻⁴)	(26.76)	(628.65)	(37.33)
An. gambiae	10	8.51 x10 ⁻⁴	218.70	0.99	1.55
females	12	(6.5 x10⁻⁵)	(3.52)	(0.16)	(0.45)
An. gambiae	Q	4.67 x10 ⁻³	327.01	3034.38	44934.32
males	0	(6.6 x10 ⁻⁴)	(8.90)	(1374.7)	(469401)
Cx. quinquefasciatus	15	7.96 x10 ⁻⁴	222.44	14.07	21.89
females	15	(4.9 x10 ⁻⁵)	(6.39)	(6.98)	(9.20)
Cx. quinquefasciatus	14	1.76 x10 ⁻³	301.21	309.65	1078.95
males	14	(2.3 x10 ⁻⁴)	(7.32)	(162.60)	(2055.62)

AFTER TeNT INJECTION

Table 35. Median velocity fit parameters and energy gains for male Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes whilst their flagella are spontaneously oscillating following either TTX or TeNT injection (with standard errors provided in brackets).

Species/ sex	Sample size	F₀/m	Best frequency/Hz	Q	Energy gain (kBT)
Ae. aegypti	16	6.39 x10 ⁻⁴	384.43	93.05	49.70
males	10	(6.9 x10 ⁻⁴)	(20.45)	(296.39)	(19.87)
An. gambiae	15	4.40 x10 ⁻³	332.04	355.79	40134.86
males	15	(8.4 x10 ⁻⁴)	(5.81)	(895.09)	(254561.08)
Cx. quinquefasciatus	20	1.64 x10 ⁻³	311.23	239.28	1041.56
males	29	(1.7 x10⁻⁴)	(5.52)	(166.47)	(3218.82)

Spontaneously oscillating (male mosquitoes only)

Table 36. Median velocity fit parameters for CO_2 sedated male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes (with standard errors provided in brackets).

Species/ sex	Sample size	F₀/m	Best frequency/Hz	Q
Ae. aegypti	25	5.16 x10 ⁻⁴	207.47	1.04
females	35	(1.9 x10 ⁻⁵)	(3.99)	(0.04)
Ae. aegypti	20	5.78 x10 ⁻⁴	301.43	0.94
males	29	(4.3 x10⁻⁵)	(12.32)	(0.06)
An. gambiae	22	6.62 x10 ⁻⁴	325.00	0.67
females	33	(3.2 x10 ⁻⁵)	(7.44)	(0.03)
An. gambiae	24	5.68 x10 ⁻⁴	283.55	0.91
males	24	(4.2 x10 ⁻⁵)	(5.97)	(0.07)
Cx. quinquefasciatus	20	5.41 x10 ⁻⁴	206.45	1.11
females	20	(2.4 x10 ⁻⁵)	(3.32)	(0.04)
Cx. quinquefasciatus	31	6.11 x10 ⁻⁴	263.20	1.00
males	51	(3.0 x10 ⁻⁵)	(8.70)	(0.05)

PASSIVE STATE BEFORE PYMETROZINE INJECTION

Table 37. Median velocity fit parameters and energy gains for male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes after pymetrozine injection (with standard errors provided in brackets).

Species/ sex	Sample size	F₀/m	Best frequency/Hz	Q	Energy gain (kBT)
Ae. aegypti	36	6.08 x10 ⁻⁴	210.21	1.31	0.15
females	50	(3.3 x10 ⁻⁵)	(3.57)	(0.047)	(0.08)
Ae. aegypti	25	6.35 x10⁻⁴	292.67	1.27	0.15
males	20	(6.4 x10 ⁻⁵)	(11.49)	(0.08)	(0.16)
An. gambiae	26	7.30 x10 ⁻⁴	305.99	0.94	0.12
females	20	(3.0 x10 ⁻⁵)	(7.07)	(0.04)	(0.12)
An. gambiae	18	4.52 x10 ⁻⁴	319.12	1.66	0.15
males	10	(4.4 x10 ⁻⁵)	(10.65)	(0.30)	(0.07)
Cx. quinquefasciatus	27	4.81 x10 ⁻⁴	208.93	1.59	0.17
females	21	(4.6 x10 ⁻⁵)	(1.90)	(0.10)	(0.15)
Cx. quinquefasciatus	35	6.88 x10 ⁻⁴	258.68	1.16	0.19
males		(3.5 x10⁻⁵)	(8.65)	(0.05)	(0.09)

AFTER PYMETROZINE INJECTION

Table 38. Median velocity fit parameters for CO_2 sedated male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes after pymetrozine injection (with standard errors provided in brackets).

Species/ sex	Sample size	F₀/m	Best frequency/Hz	Q
Ae. aegypti	21	6.62 x10 ⁻⁴	218.38	1.03
females	21	(2.9 x10 ⁻⁵)	(4.41)	(0.05)
Ae. aegypti	15	6.11 x10 ⁻⁴	313.92	0.94
males	15	(7.5 x10⁻⁵)	(13.96)	(0.08)
An. gambiae	17	6.04 x10 ⁻⁴	345.99	0.98
females	17	(4.4 x10 ⁻⁵)	(9.97)	(0.04)
An. gambiae	0	5.30 x10 ⁻⁴	317.26	1.04
males	0	(4.8 x10 ⁻⁵)	(12.59)	(0.11)
Cx. quinquefasciatus	20	5.49 x10 ⁻⁴	217.83	1.09
females	20	(3.4 x10⁻⁵)	(2.36)	(0.06)
Cx. quinquefasciatus	21	5.84 x10 ⁻⁴	263.32	0.92
males	21	(3.5 x10⁻⁵)	(6.34)	(0.05)

PASSIVE STATE AFTER PYMETROZINE INJECTION

Appendix I. P-values for comparisons between mosquitoes of two state model of a single transducer population parameters

Table 39. P-values for statistical comparisons between male and female mosquitoes from the same species for parameters obtained from the two state model fits of a single transducer population.

Parameter	P-values for <i>An.</i> <i>gambiae</i> females vs males	P-values for <i>Ae.</i> <i>aegypti</i> females vs males	P-values for <i>Cx.</i> <i>quinquefasciatus</i> females vs males
Number of ion channels	P <0.001	P <0.001	P <0.001
lon channel gating force	P <0.001	P <0.001	P <0.001
KINFINITY	P <0.001	P = 0.023	P = 1.000
K _{STEADY}	P <0.001	P <0.001	P = 0.469
K _{GS}	P <0.001	P = 0.388	P = 0.001
Extent of nonlinearity	P <0.001	P <0.001	P <0.001

Table 40. P-values for statistical comparisons between female mosquitoes from different species for parameters obtained from the two state model fits of a single transducer population.

	P-values for An.	P-values for Cx.	P-values for Cx.	
Parameter	gambiae vs Ae.	quinquefasciatus vs	quinquefasciatus vs An.	
	<i>aegypti</i> females	<i>Ae. aegypti</i> females	gambiae females	
Number of ion	P - 0.436	P = 0.436	P - 0.436	
channels	1 = 0.400	1 = 0.400	1 = 0.430	
Ion channel	P = 0.007	P = 0.738	P <0 001	
gating force	1 - 0.007	1 - 0.700	1 (0.001	
KINFINITY	P <0.001	P = 0.945	P <0.001	
K _{STEADY}	P <0.001	P = 0.141	P <0.001	
K _{GS}	P <0.001	P = 0.974	P <0.001	
Extent of	P <0.001	P = 0.297	P <0.001	
nonlinearity				

Table 41. P-values for statistical comparisons between male mosquitoes from different species for parameters obtained from the two state model fits of a single transducer population.

	P-values for An.	P-values for Cx.	P-values for Cx.	
Parameter	gambiae vs Ae.	quinquefasciatus vs	quinquefasciatus vs An.	
	<i>aegypti</i> males	<i>Ae. aegypti</i> males	<i>gambiae</i> males	
Number of ion	P = 0.230	P <0.001	P - 0.051	
channels	1 = 0.230	1 <0.001	F = 0.051	
Ion channel	P - 0.025	P <0.001	P - 0 530	
gating force	1 = 0.023	1 <0.001	1 = 0.000	
KINFINITY	P = 0.036	P = 0.023	P = 1.000	
K _{STEADY}	P = 0.145	P <0.001	P = 0.469	
K _{GS}	P = 0.001	P = 0.388	P = 0.001	
Extent of	P - 0 552	P - 0.552	P - 0.552	
nonlinearity	1 - 0.002	1 - 0.002	1 - 0.002	



Appendix J. Changes to K_{PEAK} and K_{STEADY} for sedated mosquitoes

Figure 66. Changes in dynamical stiffness in response to changes to displacement, calculated for an Ae. aegypti male using either values estimated at the peak displacement (K_{PEAK}) or at the steady state (K_{STEADY}); A) K_{PEAK} values for both active and sedated states, B) K_{PEAK} and K_{STEADY} comparison for the active state and C) K_{PEAK} and K_{STEADY} for the sedated state.

D) Changes in CAP amplitude responses in response to changes in displacement calculated for an Ae. aegypti female in both active and sedated states.



Figure 67. A and B) Changes in dynamical stiffness in response to changes to displacement, calculated using either values estimated at the peak displacement (K_{PEAK}) or at the steady state (K_{STEADY}), for a Cx. quinquefasciatus female before (A) and during (B) sedation.

C and *D*) Changes in dynamical stiffness in response to changes to displacement, calculated using either values estimated at the peak displacement (K_{PEAK}) or at the steady state (K_{STEADY}), for a Cx. quinquefasciatus male before (A) and during (B) sedation.

Appendix K. Statistical comparisons between single population and two independent population models

Table 42. Number of individuals from each mosquito species and sex which are fit better assuming two independent transducer populations rather than a single population, as determined using the AICc (following previous reports (394)), as well as the AICc for each fit type for female and male mosquitoes from each species.

	Ae.	An.	Cx. quinq-	Ae.	An.	Cx. quinq-
	aegypti	gambiae	uefasciatus	aegypti	gambiae	uefasciatus
	females	females	females	males	males	males
	(n = 21)	(n = 18)	(n = 17)	(n = 18)	(n = 15)	(n = 15)
Sample size	21	17	18	18	15	15
Number of						
individuals						
with better fit						
for two						
population	21	17	18	18	1	2
model than						
for one						
population						
model						
AICc value						
for one						
population	-391	-306	-485	-350	-251	-321
model fit to						
median data						
AICc value						
for two						
population	-398	-320	-490	-354	-250	-315
model fit to						
median data						

Appendix L. CAP amplitude and dynamical stiffness changes following TTX and TeNT injection in mosquitoes

From figures 68 and 69 it is clear that for both male and female mosquitoes TTX injection caused an essentially total loss of CAP production, as has been previously reported (353). Following TeNT injection on the other hand female CAP responses persisted although they were somewhat reduced (as shown in figure 68 part D), with this reduction being even more significant in males – the remaining nerve response was still significantly greater than the nerve response after pymetrozine injection however in both sexes.



Figure 68. Comparisons of dynamical stiffness and CAP amplitude for an Ae. aegypti female following either TTX (top) or TeNT (bottom) injection – post-ringer and post-pymetrozine states are included as references.

TTX injection in the female *Ae. aegypti* led to a decrease in $K_{INFINITY}$, though remained some decrease in dynamic stiffness around the resting position, whilst TeNT injection did not significantly change this stiffness decrease and the steady state stiffness remained constant. Estimation of dynamical stiffness was more difficult for *Ae. aegypti* males after TTX and TeNT injection as the system began to spontaneously oscillate which made measuring flagellar responses to force step stimulation challenging (hence no equivalent model fits are presented in figure 69 for male dynamical stiffness data sets). No significant changes in $K_{INFINITY}$ or K_{STEADY} were identified following TTX/ TeNT injection however, with pymetrozine acting as a comparative measure for potential shifts in these two stiffness values.



Figure 69. Comparisons of dynamical stiffness and CAP amplitude for an Ae. aegypti male following either TTX (top) or TeNT (bottom) injection – post-ringer and post-pymetrozine states are included as references.

Appendix M. Effective stiffness and best frequency shifts between control, sedated and pymetrozine injected states in mosquitoes, as well significance values for all relevant comparisons

Table 43. Median values of best frequency for sedated or pymetrozine exposed male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes (numbers in brackets are standard errors) as well as P-values for all comparisons between the two states.

Mosquito species/ sex	Best frequency (sedated state) (Hz)	Best frequency after pymetrozine (Hz)	P-value
Ae. aegypti females	207.5	210.2	P – 1 0
(n = 36/ 35/ 36)	(3.99)	(3.57)	1 = 1.0
Ae. aegypti males	297.99	292.7	D = 0.706
(n = 25/ 30/ 25)	(11.90)	(11.49)	P = 0.700
An. gambiae	325	306.0	
females	(7.44)	(7.07)	P = 1.0
(n = 29/ 33/ 26)	()	(****)	
An. gambiae males	283.7	319.1	P = 0.06
(n = 20/ 22/ 18)	(5.70)	(10.65)	1 = 0.00
Cx. quinquefasciatus	206.5	208.9	
females	(3.26)	(1.90)	P = 1.0
(n = 29/ 29/ 27)	(0:=0)	(
Cx. quinquefasciatus	274.1	258.7	
males	(10.81)	(8.65)	P = 1.0
(n = 28/ 33/ 35)	(10.01)	(0.00)	

Table 44. Median values of effective stiffness for sedated or pymetrozine exposed male and female Ae. aegypti, An. gambiae and Cx. quinquefasciatus mosquitoes (numbers in brackets are standard errors) as well as P-values for all comparisons between the two states.

Mosquito species/ sex	Effective stiffness (sedated state) (µN/m)	Effective stiffness after pymetrozine (µN/m)	P-value
Ae. aegypti females	74.0	40.5	P < 0.003
(n = 36/ 35/ 36)	(7.34)	(5.04)	1 < 0.003
Ae. aegypti males	148.1	91.7	P = 0.261
(n = 25/ 30/ 25)	(24.4)	(27.1)	F = 0.201
An. gambiae	157.0	125.9	
females	(21.2) (13.5)	(13.5)	P = 0.399
(n = 29/ 33/ 26)	()	(1010)	
An. gambiae males	162.5	182.6	P - 1 0
(n = 20/ 22/ 18)	(12.4)	(22.6)	F = 1.0
Cx. quinquefasciatus	58.7	42.3	
females	(2.35) (2.63) P	P = 0.179	
(n = 29/ 29/ 27)	· · · · · · · · · · · · · · · · · · ·		
Cx. quinquefasciatus	107.9	86.6	
males	(21.6)	(8.36)	P = 0.160
(n = 28/ 33/ 35)	(2)	(0.00)	